





AFRPL-TR-82-032 F04611-80-C-0046



# POST ACCIDENT PROCEDURES

# FOR

# CHEMICALS AND PROPELLANTS

Authors: Deborah K. Shaver

Robert L. Berkowitz

SYSTEMS TECHNOLOGY LABORATORY, INC. 2045 North 15th Street Arlington, Virginia 22201

September 1982

**က** က

S

AD A 1 2

200

Interim Report for the Period April 1981 through January 1982

Approved for Public Release; Distribution Unlimited

The AFRPL Technical Services Office has reviewed this report and it is releasable to the National Technical Information Service, where it is releasable to the general public including foreign nationals.

Prepared for

AIR FORCE ROCKET PROPULSION LABORATORY DIRECTOR OF LABORATORIES ' AIR FORCE SYSTEMS COMMAND EDWARDS AIR FORCE BASE, CALIFORNIA 93523

U.S. DEPARTMENT OF TRANSPORTATION FEDERAL RAILROAD ADMINISTRATION OFFICE OF RESEARCH AND DEVELOPMENT WASHINGTON, D.C. 20590 11 26 184

#### FOREWARD

This report was prepared by Systems Technology Laboratory, Inc., 2045 North 15th Street, Arlington, Virginia 22201 under contract F04611-80-C-0046 with the Air Force Rocket Propulsion Laboratory (AFRPL), Edwards' Air Force Base, California 93523. Mr. John Marshall was COTR for AFRPL. The project was also jointly sponsored through the Federal Railroad Administration (FRA), Office of Rail Safety Research.

This technical report has been reviewed and is approved for publication in accordance with the distribution statement on the cover and on the DD Form 1473.

DAVID DANCER Federal Railroad Administration.

huma chal OHN W. MARSHALL

Project Manager Air Force Rocket Propulsion Laboratory

FOR THE DIRECTOR

EUGENE G. HABERMAN Chief, Liquid Rocket Division Air Force Rocket Propulsion Laboratory FORREST S. FORBES

Chief, Propellant Systems Section Air Force Rocket Propulsion Laboratory

No warranty, guarantee or representation, expressed or implied, is made by or an behalf of the U.S. Government as to the absolute correctness or sufficiency of any representation contained in this document, and the U.S. Government assumes no responsibility in connection therewith, nor can it be assumed that all acceptable safety measures are contained in this or associated documents, or that other or additional measures may not be required under particular or exceptional conditions or circumstances,

REPORT D	DCUMENTATION PAGE	BEFORE COMPLETING FORM
1. REPORT NUMBER	2. GOVT ACCESSIO	N NO. 3. RECIPIENT'S CATALOG NUMBER
AFRPL-TR-82-032	AD-A12/7	23
4. TITLE (and Subtitle)	,	5. TYPE OF REPORT & PERIOD COVER
POST ACCIDENT PROCE	EDURES FOR CHEMICALS AND	Interim
PROPELLANTS	· · ·	April 1981-January 1982
		6. PERFORMING ORG. REPORT NUMBE
		F04611-80-C-0046
Deborgh K Shaver		
Robert L. Berkowitz	2	F04011-80-C-0040
	-	
9. PERFORMING ORGANIZATION	NAME AND ADDRESS	10. PROGRAM ELEMENT, PROJECT, TA
SYSTEMS TECHNOLOGY	LABORATORY, INC.	
2045 North 15th Str	reet	
Arlington, Virginia	22201	
11. CONTROLLING OFFICE NAM	E AND ADDRESS	12. REPORT DATE
U.S. DEPARTMENT OF	TRANSPORTATION	September 1982
rederal Kallroad Ad	Ministration	13. NUMBER OF PAGES
14. MONITORING AGENCY NAME	ADDRESS(if different from Controlling Off	ICO) 15. SECURITY CLASS. (of this report)
Air Force Rocket Pr	opulsion Laboratory/LKCP	INCLASSIFIED
Edwards AFB, Califo	ornia 93523	
		15. DECLASSIFICATION/DOWNGRADIN SCHEDULE
Document is availab Service, Springfiel	RELEASE: DISTRIBUTION UNLI le to the public through th d, Virginia 22151.	MITED Ne National Technical Informati
Document is availab Service, Springfiel 17. DISTRIBUTION STATEMENT ( 18. SUPPLEMENTARY NOTES Refer to AFRPL-TR-82-	RELEASE: DISTRIBUTION UNLI le to the public through th d, Virginia 22151. of the abstract entered in Block 20, if differe 031 which provides database	MITED ne National Technical Informati
<ul> <li>DOCUMENT IS AVAILAB DOCUMENT IS AVAILAB Service, Springfiel</li> <li>17. DISTRIBUTION STATEMENT (</li> <li>18. SUPPLEMENTARY NOTES Refer to AFRPL-TR-82 materials and to AFRPL</li> <li>19. KEY WORDS (Continue on rever hazardous materials production transportation accident assessment emergency response</li> </ul>	RELEASE: DISTRIBUTION UNLI le to the public through th d, Virginia 22151. of the abstract entered in Block 20, 11 differe 031 which provides database L-TR-82-075 which is the tw vapor dispersion hazards assessment hazards mitigation wreckage removal cleanup/disposal	MITED ne National Technical Informati nu from Report) for additional 12 hazardous o volume guidelines manual.
Document is availab Service, Springfiel 17. DISTRIBUTION STATEMENT ( 18. SUPPLEMENTARY NOTES Refer to AFRPL-TR-82 materials and to AFRP. 19. KEY WORDS (Continue on rever hazardous materials production transportation accident assessment emergency response 20. ABSTRACT (Continue on rever This report presents p containers used for his and disposal sites and emergency response gui groups and state and p ment; assessment of me by police, fire chiefs interactive feedback of	RELEASE: DISTRIBUTION UNLI le to the public through th d, Virginia 22151. of the abstract entered in Block 20, 11 differe 031 which provides database L-TR-82-075 which is the tw vapor dispersion hazards assessment hazards mitigation wreckage removal <u>cleanup/disposal</u> or olde 11 necessary and identify by block nur vapor dispersion hazards assessment hazards mitigation wreckage removal <u>cleanup/disposal</u> or olde 11 necessary and identify by block nur production volumes, storage ighway and rail shipments, 1 i capabilities for 16 hazard idelines and resources avail local agencies; key findings wholes for implementing crists, and other disaster respon trisis management system. A	MITED ne National Technical Informati net from Report) for additional 12 hazardous o volume guidelines manual. mbor) capacities, shipment quantitic general commodity flow patterns dous materials; an overview of lable to selected industry s based on STL's accident asset sis management techniques used nse officers; and criteria for
Document is availab Service, Springfiel 17. DISTRIBUTION STATEMENT ( 18. SUPPLEMENTARY NOTES Refer to AFRPL-TR-82 materials and to AFRP. 19. KEY WORDS (Continue on rever hazardous materials production transportation accident assessment emergency response 20. ABSTRACT (Continue on rever This report presents p containers used for his and disposal sites and emergency response gui groups and state and p ment; assessment of me by police, fire chiefs interactive feedback ( D) 1 JAN 73 1473 EDITION (	RELEASE: DISTRIBUTION UNLI le to the public through th d, Virginia 22151. of the abstract entered in Block 20, 11 differe 031 which provides database L-TR-82-075 which is the tw vapor dispersion hazards assessment hazards mitigation wreckage removal <u>cleanup/disposal</u> or olde 11 necessary and identify by block num production volumes, storage ighway and rail shipments, 1 d capabilities for 16 hazard idelines and resources avail local agencies; key findings who for implementing criss, and other disaster respon trisis management system. A	MITED ne National Technical Informati net from Report) for additional 12 hazardous o volume guidelines manual. mbor) capacities, shipment quantitic general commodity flow pattern dous materials; an overview of lable to selected industry s based on STL's accident asset sis management techniques used nse officers; and criteria for

SECURITY GLASSIFICATION OF THIS PAGE (When Date Entered) VILLEN ACCEPTED THE PARTICUL PARTICUL SECURITY CLASSIFICATION O THIS PAGE (When Date Entered)

1

GANTSTIC.

### TABLE OF CONTENTS

.

.

When a real strategy of

LIS	t of a	CRONYMS	· · · · · · · · · · · · · · · · · · ·
1.	EXEC	UTIVE SU	MMARY
2.	BACH	GROUND	
3.	CHEN DAT	MICALS AN	ID PROPELLANTS PRODUCTION AND TRANSPORTATION 3-1
	3.1	Data So	urces
	3.2 3.3	Physical Producti	/Chemical Data
		2 2 1	A return 2.7
		3.3.1	
		3.3.6	
		3.3.3	
	•	225	Fthml Accordista 3-0
		3.3.6	Hydrocvanic Acid
		3.3.7	Technitane
		3.3.8	Methanol
		3.3.9	Methyl Bromide
		3.3.10	Monomethylamine Nitrate
		3.3.11	Pronvlene 3-16
		3.3.12	Sodium Hydrosulfide Solution
		3.3.13	Sodium Hydroxide Solution
		3.3.14	Styrene Monomer
		3.3.15	Toluene
		3.3.16	Vinvl Acetate
	• •		
	3.4	Shipmen	
	3.5	Containe	ers Required for Shipping
		3.5.1	Acetone
		3.5.2	Acetone Cyanohydrin 3-38
		3.5.3	Acrylonitrile
		3.5.4	Aerozine-50
		3.5.5	Ethyl Acrylate
		3.5.6	Hydrocyanic Acid 3-44
		3.5.7	Isobutane
		3.5.8	Methanol
		3.5.9	Methyl Bromide
		3.5.10	Monomethylamine Nitrate
		3.5.11	Propylene
		3.5.12	Sodium Hydrosulfide
		3.5.13	Sodium Hydroxide
		3.5.14	Styrene
		3.5.15	
		3.5.10	V1nyl Acetate

\_

## TABLE OF CONTENTS (cont'd)

	3.6	Commod	ty Flow Patterns.			
4.	ACCI	DENT ASSE	SSMENT	4-1		
	4.1 4.2 4.3 4.4	Notificat Emergene Emergene On-scene	ion/Emergency Responders			
		4.4.1 4.4.2	Highway Accidents	••••••••••••••••••••••••••••••••••••••		
	4.5 4.6 4.7 4.8 4.9 4.10 4.11	Hazardou Release I Firefight Cleanup : Structura Wreckage Key Find	s Material Identification and Location Iandling procedures	4-11         4-11         4-13         4-14         4-15         4-15         4-16		
5.	ASSES TECH DISAS	SSMENT OF NIQUES US TER RESPO	METHODS FOR IMPLEMENTING CRISIS MANAG ED BY POLICE, FIRE CHIEFS, AND OTHER ONSE OFFICERS	EMENT		
	5.1 5.2	Methodol Emergene Citi <b>es</b>	ogy for Obtaining Information	····· 5-1		
		5.2.1 5.2.2 5.2.3 5.2.4 5.2.5 5.2.6 5.2.7 5.2.8 5.2.9 5.2.10	Baltimore, MD Los Angeles, CA Nashville, TN Newark, NJ Pensacola, FL Sacramento, CA Tallahassee, FL Waverly, FL Wilmington, DE Youngstown, FL	5-3 5-7 5-7 5-11 5-12 5-13 5-14 5-14 5-15 5-16 5-16 5-16 5-17		
	5.3	Assessme	Assessment of Methods of Crisis Management			
		5.3.1	Contingency Planning for HM Transportation Accidents	5-24		
,			<ul> <li>5.3.1.1 Overview of Community Emergency Response Planning and Recommendations for Improved Community Plans</li></ul>	s 5-24 s 5-28		
		5.3.2	Training of Emergency Response Personnel			

POPOSICE REAL A PURATURA POWER POWER POWER POWER

「「「「「「」」」」

### TABLE OF CONTENTS (cont'd)

· .		5.3.2.1 5.3.2.2 5.3.2.3 5.3.2.4	Communications Training	
	5.3.3	Availabili Requirem	ity of On-site Resource ents	
		5.3.3.1	Protective Clothing, Gear and Equipment Availability and Uses	
		5.3.3.2	Specialized Treatment Chemicals, Equipment, Resources Availability and Uses	
		5.3.3.3	Treatment Chemicals and Methods 5-37	
		5.3.3.4	Sorbent Methods	
		5.3.3.5	Monitoring Devices	
		5.3.3.6	Heavy Equipment	
		5.3.3.7	Wreckage Removal Contractors	
		5.3.3.8	Product Transfer, Cleanup and	
			Disposal Contractors	
		5.3.3.9	Specialized Response Capabilities	
		· ·	of Each of the Cities Reviewed	
	5.3. <del>4</del> 5.3.5	Identifica by Cities Hazardou Emergence Innovative Hazardou Accidents	tion of Problems Encountered in Actual On-scene Response to s Materials Transportation ties	
5.4	Recomm for Imple	endations fo ementing Cr	or Improved Municipal Methods	
CRITE	RIA FOR	INTERACTI	VE FEEDBACK CRISIS MANAGEMENT SYSTEM	
6.1	Overviev Feedbac	w and Assess k Crisis Man	ment of SOA Interactive agement Systems	
6.2	5.2 Feasibility of Interactive Feedback System as a Training Aid/Accident Simulator			
	6.2.1 6.2.2	Interactiv Feasibility as a Train	re Feedback System Capability	
6.3 6.4	Potentia Criteria Managen	l Users of In for Interactionent System	teractive Feedback System	
	6.4.1	Criteria f Requirem	or Hardware/Software Component ents	

6.

AN AN AN AN AN AN AN AN

#### TABLE OF CONTENTS (cont'd)

Charles and the second of

.

APPENDIX A	DOT HAZARDOUS MATERIALS EMERGENCY RESPONSE GUIDE
APPENDIX B	AAR-EMERGENCY HANDLING OF HAZARDOUS MATERIALS IN SURFACE TRANSPORTATION
APPENDIX C	EPA-MANUAL FOR THE CONTROL OF HAZARDOUS MATERIALS SPILLS: VOLUME I
APPENDIX D	USCG-CHRIS HAZARDOUS CHEMICAL DATA: VOLUME II D-1
APPENDIX E	CITY OF BALTIMORE FIRE DEPARTMENT MANUAL OF PROCEDURES
APPENDIX F	BAY COUNTY (PANAMA CITY/YOUNGSTOWN), FL HAZARDOUS MATERIAL INCIDENT REPORTING FORMS F-1
APPENDIX G	ANNOTATED BIBLIOGRAPHY G-1
APPENDIX H	GLOSSARY



### LIST OF TABLES

Table

Sec. 15

1200

a - I and all and and

3-1	Producers and Production Capacity of Acetone
3-2	Producers and Production Capacity of Acrylonitrile
3-3	Producers and Production Capacity of Hydrocyanic Acid
3-4	Producers and Production Capacity of Methanol
3-5	Propylene Producers and Production Volume
3-6	Sodium Hydrosulfide Solution Producers and Production Volume
3-7	Sodium Hydroxide Producers and Production Volume
3-8	Styrene Monomer Producers and Production Volume
3-9	Toluene Producers and Production Volume
3-10	Vinyl Acetate Producers and Production Volume
3-11	Distribution of Commodities Shipped by Various Modes of Transportation in the U.S
3-12	Regulations Applicable to Container Requirements
3-13	Approved Container Types for All Commodities
3-14	Cylinders, Tank Cars, Cargo Tanks, and Portable Tank Containers Specified for Acetone Service
3-15	Tank Cars, Cargo Tanks, and Portable Tank Containers Specified for Acetone Cyanohydrin Service 3-43
3-16	Cylinders Approved for Shipping Isobutane
3-17	Cylinders, Tank Cars, and Cargo Tanks Specified for Methyl Bromide Service
3-18	Tank Cars, Cargo Tanks, and Portable Tank Containers Specified         for Sodium Hydroxide Service
3-19	Origin-Destination of Sodium Compounds
3-20	Origin-Destination of Miscellaneous Acrylic Organic Chemical Products
3-21	Origin-Destination of Miscellaneous Cyclic Organic Chemical Products
3-22	Origin-Destination of Alcohols 3-55
3-23	Origin-Destination of Organic Acids and Salts
3-24	Origin-Desintation of Industrial Organic Chemicals, nec
3-25	Origin-Destination of Industrial Inorganic Acids
3-26	Origin-Destination of Liquefied Petroleum and Coal Gases
3-27	Origin-Destination of Industrial Gases, nec

### LIST OF TABLES (cont'd)

¢.2

100 million (1997)

.

3-28	Origin-Destination of Sodium Alkalias 3-61
3-29	Origin-Destination of Crude Products from Coal and Petroleum Tar
3-30	Aerosine-50 Trailer Shipment Routes
4-1	Chemical/Propellant Assidents (1971-1988)
4-2	Accidents Studied
5-1	Hazardous Substance Spill Report
5-2	Material Involvement in All Citigs
5-3	Accident Histories of the Chemicals and Propellants in the Cities Examined
5-4	Estimated Losses Due to Incidents Involving the 16 Chemicals and Propellants
5-5	Assessment of Community Contingency Plans Based Upon Crisis Management Criteria
5-6	Sources of Protective Clothing, Breathing Apparatus, Gear, and Equipment
5-7	HM Treatment Methods
58	Analytical Monitoring Equipment
59	Heavy Equipment

÷

۰. .

#### LIST OF ILLUSTRATIONS

### Figure

D CALL LAN

T.L. L.L.

( DATE:

## Page

. .

3-1	Acetone Production
3-2	Acrylonitrile Production 3-8
3-3	Hydrocyanic Acid Production
3-4	Isobutane Production by Region at Natural Gas Processing Plants
3-5	Methanol Production
3-6	Propylene Production 3-20
3-7	Sodium Hydrosulfide Production 3-22
3-8	Sodium Hydroxide Production
3-9	Styrene Monomer Production 3-29
3-10	Toluene Production
3-11	Vinyl Acetate Production 3-35
3-12	Census Regions and Geographic Divisions of the US
3-13	Variation Between Commodity Production and Consumption by Region
3-14	Commodity Flow Pattern of Aerozine-50
5-1	Geographic Location of Cities Selected for Review of Crisis Management Techniques to HM Transportation Accidents

### LIST OF ACRONYMS

A-50	-	Aerozine-50, rocket fuel consisting of 50/50 mixture of hydrazine and unsymmetrical dimethylhydrazine
AAR	-	Association of American Railroads
AE	-	Acoustic Emission
AFB	-	Air Force Base
AFCRL	-	Air Force Cambridge Research Laboratories
BE	-	AAR Bureau of Explosives
BLEVE	-	Boiling Liquid Expanding Vapor Explosion
CCC	-	Communications Coordination Center
CFR	-	Code of Federal Regulations
CHEMTREC	-	Chemical Transportation Emergency Center
CHRIS	-	Chemical Hazards Response Information System
СМА	-	Chemical Manufacturers Association
DDESB	-	Department of Defense Explosives Safety Board
DOD	-	U.S. Department of Defense
DOT	-	Department of Transportation
EMT	-	Emergency Medical Technician
ERG	-	Emergency Response Guide
EPA	-	Environmental Protection Agency
HACS	-	USCG Hazard Assessment Computer System
HM	-	Hazardous Material
HMER	-	Hazardous Materials Emergency Response System developed by Bell Systems for NRC/CHEMTREC
HMRAC	-	Hazardous Material Risk Advisory Committee (Nashville, TN)
IR	-	Infared
JANNAF	-	Joint Army-Navy-NASA-Air Force
LEL	-	Lower Explosive Limit
LN/SCL	-	Louisville & Nashville/Seaboard Coastline Railroads, part of the Family Lines
LPG	<del>.</del>	Liquefied Petroleum Gas
MMH	-	Monomethylhydrazine
МТВ	-	U.S. Department of Transportation Materials Transportation Bureau
NIOSH	-	National Institute of Occupational Safety and Health
NFPA	-	National Fire Protection Association

-i-

# LIST OF ACRONYMS (cont'd)

NOAA	- National Oceanic and Atmospheric Administration
NRC	- National Response Center
NTSB	- National Transportation Safety Board
OMHTADS	- Oil and Hazardous Material Technical Assistance Data System
OSC	- On-Scene Coordinator
OSHA	- Occupational Safety and Health Administration
PEL	- Public Emergency Limit
RCRA	- Resource Conservation and Recovery Act
RAM	- Radioactive Materials
RMA	- Rocky Mount Arsenal
SHELL R&D SPILLS	- Shell Vapor Dispersion Model
SRI	- Stanford Résearch Institute
STCC	- Standard Transportation Commodity Code
тсс	<ul> <li>Transportation Commodity Code</li> </ul>
TLV	- Threshold Limit Value
UDMH	- Unsymmetrical Dimethylhydrazine
UEL	- Upper Explosive Limit
UN	- United Nations
USAF	- United States Air Force
USCG	- United States Coast Guard
UV	- Ultraviolet

PARAMETERS INCOMENTAL INTERACTIONS INTERACTION

.

.

#### 1. EXECUTIVE SUMMARY

The overall objective of this program is to perform a state-of-the-art assessment to develop technology which will minimize hazards and environmental damage from transportation-related accidents or other spills of certain chemicals and propellants. This report has been prepared as an addendum to the Final Interim Report and constitutes the deliverable requirements set forth in CDRL item 11.

An addendum to the Final Interim Report was required because during the performance of the project additional effort on Tasks 1 through 5 was required. In addition, a new set of subtasks was initiated to develop guidelines for managing spills of Titan II propellants. This report represents the additional work completed on Tasks 1 through 3. A separate report will document the work performed during Tasks 4, 5 and 6E of the project. Work accomplished on new Subtasks 6A through 6D has already been delivered in a separate document. The additional chemicals and propellants which were examined are acetone, acetone cyanohydrin, acrylonitrile, aerozine-50, ethyl acrylate, hydrocyanic acid, isobutane, methanol, methyl bromide, monomethylamine nitrate (PRM), propylene, sodium hydrosulfide, sodium hydroxide, styrene monomer, toluene and vinyl acetate. The following paragraphs provide summary data on the additional efforts and new subtasks.

In Task 1A an in-depth literature search was conducted to identify physical/ chemical data, production sites and volumes, commodity flow patterns, container types used for highway and rail transport, and accident histories of the above mentioned chemicals and propellants. Data were obtained from numerous federal, state and local government agencies, trade associations, rail and highway carriers, chemical manufacturers, wreckage removal and cleanup and disposal contractors. These data were obtained to develop the transportation picture for these commodities so that their accident histories can be better understood in the context of all hazardous materials transportation.

In Task 2A, in-depth analyses of the initial response, special emergency equipment and materials, on-scene coordination and communications, hazardous material identification, release handling, firefighting, cleanup and disposal, structural integrity assessment and wreckage removal activities used at 19 selected rail and highway accidents involving the additional 16 chemicals and propellants were conducted. This effort was necessary to identify areas in which emergency response personnel have experienced difficulty in handling the complexity of hazards associated with hazardous

1-1

materials transportation accidents. The assessment of the 10 accidents investigated earlier in this project were combined with the additional 19 to form a larger accident sample. These 29 accidents were then reviewed to identify trends, improvements or changes in accident management. A few examples of SOA accident management techniques identified include:

2

- All of the existing contingency plans that have been activated during HM accidents were designed for generic emergencies and none dealt specifically with response to HM transportation accidents.
- The utilization of special emergency equipment, materials, and techniques at serious transportation accidents has not become standard procedure until the last few years and still appears to depend on the resources, sophistication and pre-planning done in the community in which the accident occurs.
- Response personnel at highway accidents have not significantly changed their communications techniques nor are changes indicated in coordinating response efforts at the scene.
- On-scene communication and coordination techniques at railroad accidents have changed significantly.

Also during this task the appropriate hazards mitigation and cleanup/disposal methods for the chemicals and propellants identified were compiled from the U.S. Coast Guard CHRIS system, the EPA Hazardous Material Spill Control Manual, the AAR Emergency Handling of Hazardous Materials in Surface Transportation and the DOT Emergency Response Guidebooks. These guidelines were documented as a means of providing the first personnel arriving on-scene with personal protection and hazards mitigation actions. It should be noted that several of these materials have disproportionately high accident histories in relation to other hazardous materials.

The emergency response/contingency planning; training requirements and capabililities; and sources of specialized personnel, equipment and gear in ten cities in four states were also assessed. This effort was performed to identify existing response capabilities of representative municipalities in the U.S. as a mechanism for recommending improved use of resources and for developing a "model" hazardous materials crisis management plan for communities. This assessment resulted in several recommendations for improved methods for municipal crisis management at the scene of a hazardous materials transportation accident. Examples of a few of the recommendations made include among:

- Volunteer firefighters should be given the same hazardous materials training that paid personnel receive.
- Emergency response personnel should be required to receive follow-up hazardous materials training after a specified period of time.

1-2

- A designated communications network should be established which will provide for a mechanism to alert the public and to handle communications between the communications command center, the accident site and other off-scene support organizations.
- Contingency plans should identify radio and television stations that will continuously inform the public during the initial phases of the emergency.

Task 3 involved the development of detailed criteria to serve as the basis for the detailed procedures to be developed in Task 4. These procedures will provide for improved hazardous mitigation, wreckage removal and cleanup and disposal techniques at hazardous materials transportation emergencies. In Task 3A the specific criteria for optimum hazards mitigation, wreckage removal and cleanup and disposal methods were expanded. They originally included criteria for on-scene identification of materials; onscene communications; assessing toxic, flammable and explosive vapor hazards; determining meteorological conditions for establishing air dispersion limits; handling leaks; assessing container structural integrity; remote sensing of container temperature and pressure; transfer operations; wreckage removal and cleanup and disposal. A logic sequence for determining optimum operational procedures for the accident scene was also developed along with criteria for developing necessary training aids for emergency response teams. Additionally, criteria for methods for implementing crisis management techniques at hazardous materials transportation accidents were developed. The purpose of the additional work was to develop criteria for an interactive feedback crisis management system. Concerning software requirements, a few of the software criteria cited include:

- The data base for which the software will be developed should include at a minimum the following parameters:
  - color of placard on tank or tank car for use in identifying materials hazard class;
  - commodity involved;
  - STCC and/or UN number of the material(s) involved;
  - specification cylinder, tank car, cargo tank or portable tank container involved in accident;
  - quantity of material being shipped;
  - source strength;
  - time since initial release;
  - leak/no leak conditions;
  - wind direction and speed;
  - precipitation/condensation conditions;
  - ambient temperature;

1-3

- percent cloud cover;
- proximity and location of population centers to accident site;
- population density at or near accident site;
- local topography;
- arrival time and distance of wreckage removal, cleanup and disposal contractor to site (i.e., inventory of these groups is necessary);
- location and type of neutralizing agent which should be used based upon the material released;
- container structural integrity assessment; and
- environmental pollution considerations.
- Once the commodity has been identified the user should be provided with the material's LEL, UEL, TLV, PEL, at the accident scene, boiling point, freezing point, critical temperature, critical pressure, critical density, vapor pressure and autoignition temperature.
- Be developed so that the systems software can be made available to cities through which hazardous commodities are shipped.

#### 2. INTRODUCTION

The overall objective of this program is to perform a state-of-the-art assessment to develop technology which will minimize hazardous and environmental damage from transportation-related accidents or other spills of certain chemicals and propellants.

This report presents the additional work accomplished in Tasks 1, 2 and 3 and is structured into the following sections:

- Section 1 Executive Summary
- Section 2 Introduction
- Section 3 Chemicals and Propellants Production and Transportation Data
- Section 4 Accident Assessment
- Section 5 Methods for Implementing Crisis Management Techniques for Hazardous Materials Transportation Accidents
- Section 6 Criteria for an Interactive Feedback Crisis Management System
- Appendices A H

Sections 1 and 2 present highlights and a structural overview of the report. Section 3 lists data sources and presents analysis of production volumes, containers used for highway and rail shipments, general commodity flow patterns, and transportation accident histories of the chemicals and propellants. Section 4 presents an in-depth assessment of selected accidents covering chronological analysis of initial response; emergency special equipment and materials; on-scene coordination and communications; hazardous material identification and location; release handling, firefighting, cleanup and disposal, structural integrity assessment and wreckage removal activities. The accidents selected were 28 NTSB-investigated accidents and the events at Mississauga in Ontario, Canada. Section 5 presents an investigation of the response and planning capabilities of ten cities as a basis of developing state-of-the-art methods for municipalities to implement crisis management techniques at hazardous materials transportation accidents. Section 6 presents criteria for hardware and specific software for use in the eventual development of an interactive feedback crisis management system. Appendices A, B, C and D have been prepared to present the appropriate hazards mitigation, cleanup and disposal guidelines for the additional 16 chemicals and propellants outlined in the DOT, AAR, EPA and CHRIS response manuals, respectively; Appendix E shows selected response procedures used by the City of Baltimore fire department when responding to a hazardous materials transportation accident; Appendix F provides selected hazardous material incident reporting forms used in Bay County (Panama City/ Youngstown),

Florida; Appendix G is an annotated bibliography which summarizes the current state-ofthe-art for emergency response methods, procedures and systems, hazards mitigation and cleanup activities for rail and highway accidents and user transfer operations involving releases of the additional 18 selected hazardous commodities; and Appendix H is a glossary of terms used in this report.

#### 3. CHEMICALS AND PROPELLANTS PRODUCTION AND TRANSPORTATION DATA

#### 3.1 DATA SOURCES

A literature search has been conducted and a data base compiled for the additional sixteen chemicals and propellants in this project. The data includes production volume, shipping quantities, shipping containers and general commodity flow patterns. Several Federal and State agencies as well as industrial organizations, trade associations and academic institutions listed below were contacted.

#### Federal

- Bureau of Census
- United States Air Force (USAF)
- Department of Transportation (DOT)
  - Federal Railroad Administration (FRA)
  - Materials Transportation Bureau (MTB)
- Interstate Commerce Commission (ICC)
- National Transportation Safety Board (NTSB)

#### State/City

- California
  - Los Angeles
  - Sacramento
- New Jersey
  - Newark
- Maryland
  - Baltimore
- Delaware
  - Wilmington
- Tennessee
  - Nashville
  - Waverly
- Florida
  - Youngstown
  - Tallahassee
  - Pensacola

Crestview

#### **Trade Associations**

• Chemical Manufacturers Association (CMA)

#### Industrial

- Cleanup/Disposal Contractors
  - IT Corporation

#### Academic Institutions

Stanford Research Institute

#### 3.2 PHYSICAL/CHEMICAL DATA

Physical/chemical data were obtained on each of the sixteen commodities and are listed in Appendix G. Other pertinent data such as thermal and chemical reactivity, synergistic/antagonistic effects with other materials, toxicity, exposure and environmental effects were also compiled from the above as well as from the data provided in several of the emergency response systems studied. The hazards mitigation guidelines for each of the commodities found in the response manuals are found in Appendices A, B, C and D.

#### **3.3 PRODUCTION QUANTITIES**

Data on annual production volume for the sixteen chemicals and propellants were compiled from the <u>1981 SRI Directory of Chemical Producers</u> in the U.S. and <u>1980-81</u> <u>OPD Chemical Buyers Directory</u>.

The data are presented in the following sections for each commodity (where available) in terms of location of production sites, annual production on a state-by-state basis, and total U.S. annual production.

#### 3.3.1 Acetone

Table 3-1 lists producers, production sites and capacities for acetone. The total annual U.S. production capacity is 1,588,000 metric tons. Figure 3-1 shows annual production capacity by state. Production is highest in the Middle Atlantic and East North Central Regions. The major end uses of acetone include the chemical manufacturing of methyl isobutyl ketone, methyl isobutyl carbinol, methyl methacrylate

# TABLE 3-1PRODUCERS AND PRODUCTION CAPACITY OF ACETONE

Producer	Production Sites	Annual Capacity (Thousands of <u>Metric Tons)</u>
Allied Chem. Corp Chems Co.	Frankford, PA	163
American Cyanamid Co. Organic Chems. Div.	Willow Island, WV	5
Atlantic Richfield, Co. Oxirane International Subsid. Oxirane Chem. Co.	Bayport, TX	18
Clark Oil & Refining Corp. Clark Chem. Corp. Subsid.	Blue Island, IL	24
Dow Chem. U.S.A	Oyster Creek, TX	127
Eastman Kodak Co. Eastman Chem. Products Inc., subsid. Tennessee Eastman Co.	Kingsport, TN	36
Exxon Corp. Exxon Chem Co., div. Exxon Chem. Americas	Bayway, NJ	63
General Electric Co. Engineered Materials Group Plastics Business Operations	Mount Vernon, IN	109
Georgia-Pacific Corp. Chem. div.	Plaquemine, LA	92
Getty Oil Corp. Getty Refining and Marketing Co., subsid.	El Dorado, KS	25
The Goodyear Tire and Rubber Co., subsid.	Bayport, TX	5
Monsanto Co. Monsanto Chem. Intermediates Co.	Chocolate Bayou, TX	136

3-3

· • •

# TABLE 3-1 (cont'd)

Shell Chem. Co.	Deer Park, TX Wilmington, CA Wood River, IL	317 45 136
Standard Oil Co. of California Chevron Chem. Co. subsid. Petrochems. Div.	Richmond, CA	15
Union Carbide Corp. Chems. and Plastics, div.	Bound Brook, NJ Institute, WV	50 77
United States Steel Corp. USS Chems. div.	Haverhill, OH	145

TOTAL

1,588

•••••

.



and bisphenol-A; as a paint, varnish and lacquer solvent; for use in manufacturing cellulose acetate, especially as a spinning solvent to clean and dry parts of precision equipment; as a solvent for potassium iodide and permanganate; as a delusterant for cellulose acetate and permanganate and cellulose acetate fibers; and in the specification testing of vulcanized rubber products.

#### 3.3.2 Acetone Cyanohydrin

Acetone cyanohydrin is produced by only four manufacturers in the U.S. They are:

Producer		<b>Production Site</b>	
•	Cy/Ro Indust., Inc.	Westwego, LA	
•	E.I. du Pont de Nemours & Co., Inc. Chems. and Pigments Dept.	Memphis, TN	
•	Monsanto Co. Monsanto Chem. Intermediates Co.	Texas City, TX	
•	Rohm and Haas Co. Rohm and Haas Texas Inc., subsid.	Deer Park, TX	

Since the number of producers is limited to four, data regarding annual production volumes is not presently available from the Census Bureau. The major end uses of acetone cyanohydrin include the manufacture of insecticides and as an intermediate for organic synthesis, especially of methyl methacrylate.

#### 3.3.3 Acrylonitrile (inhibited)

Table 3-2 lists producers, production sites and capacities for acrylonitrile. The total annual U.S. production capacity is 950,000 metric tons. Figure 3-2 shows annual production capacity by state, with 73 percent of the total produced in Texas and Louisiana. The major end uses for acrylonitrile include monomer for acrylic and modacrylic fibers and high-strength whiskers; manufacture of alkyl benzene sulfonate and acrylonitrile-styrene copolymers; production of nitrile rubber; in the cyanoethylation of cotton; in the making of synthetic soil blocks; in organic synthesis; as a fumigant grain; as a monomer for a semiconductive polymer that can be used like inorganic oxide catalysts in dehydrogenation of tert-butyl alcohol to isobutylene in water; and in the production of bottles for soft drinks.

### TABLE 3-2

# PRODUCERS AND PRODUCTION CAPACITY OF ACRYLONITRILE (INHIBITED)

Producer	Production Site	Annual Capacity (Thousands of <u>Metric Tons</u>
American Cyanamid Co. Indust. Chems. div.	New Orleans, LA	120
E.I. du Pont de Nemours & Co., Inc. Petrochems. Dept. Freon Products div.	Memphis, TN	122
Polymer Intermediates Dept.	Beaumont, TX	159
Monsanto Co. Monsanto Chem. Intermediates Co.	Chocolate Bayou, TX Texas City, TX	209 204
The Standard Oil Co. (Ohio) Vistron Corp., Subsid. Chems. Dept.	Lima, OH	<u>136</u>
	TOTAL	950



3.3.4 Aerozine-50

Aerozine-50 is primarily used as a propellant and is a 50%/50% mixture of hydrazine and unsymmetrical dimethylhydrazine (UDMH). This propellant is formulated only at Rocky Mountain Arsenal (Denver, Colorado) for exclusive consumption of the USAF. According to the USAF Directorate of Energy Management at Kelly AFB, Texas the 1980 production volume of Aerozine-50 was 246 metric tons (272 tons). Based on USAF projections it can be expected that production will increase considerably in fiscal year 1981 to an estimated 473 metric tons (522 tons).

#### 3.3.5 Ethyl Acrylate (inhibited)

Ethyl acrylate (inhibited) is produced by only four manufacturers in the U.S. They are:

Producer	Production Site
Badische Corp.	Freeport, TX
Celanese Corp. Celanese Chem. Co., Inc.	Clear Lake, TX Pampa, TX
Rohm & Haas Co. Rohm & Haas Texas Inc., subsid.	Deer Park, TX
Union Carbide Corp. Chems, and plastics div.	Taft, LA

Since the number of producers is limited to four, data regarding annual production volumes is not available from the Census Bureau. The major end uses of ethyl acrylate (inhibited) are as polymers; in the manufacture of acrylic paints; and as chemical intermediates.

#### 3.3.6 Hydrocyanic Acid

Table 3-3 shows the producers, production sites and annual production volume for hydrocyanic acid in the U.S. It can be seen that a total of 551,000 metric tons are produced annually. Figure 3-3 shows annual production capacity by state. It can be seen that the majority produced is manufactured in Texas (67%). The major end uses of hydrocyanic acid include manufacturing of acrylonitrile, acrylates, adiponitrile, cyanide salts and dyes and chelating agents. Most of the hydrocyanic acid produced is used captively to produce other chemicals within a plant. It is estimated that 97 percent of

# TABLE 3-3PRODUCERS AND PRODUCTION CAPACITY OF HYDROCYANIC ACID

Producer	Production Site	Annual Capacity (Thousands of <u>Metric Tons)</u>
American Cyanamid Co. Indust. Chems. Div.	New Orleans, LA	15
Ciba-Geigg Corp. Agricultural Div.	St. Gabriel, LA	41
Plastics & Additives Div. Pigments Dept.	Glenn Falls, N.Y.	1
Degussa Corp. Alabama Group	Theodore, AL	24
Dow Chem. USA	Freeport, TX	9
E.I. du Pont de Nemours & Co., Inc.	Memphis, TN	82
Chems. and Figments Dept	Recument TY	22
Petrochems. Dept.	Desumonit, IA	05
Polymer Intermediates	Urange, IA	90
Dept.	Victoria, IX	95
Monsanto Co. Monsanto Chem.	Chocolate Bayou, TX	29
Intermediates Co.	Texas City, TX	29
Rohm & Haas Co. Rohm & Haas Texas Inc., subsid.	Deer Park, TX	91
The Standard Oil Co. (Ohio) Vistron Corp., subsid. Chems. Dept.	Lima, OH	18
TOTAL		551



the HCN is produced as a by-product of other processes, and that 3 percent is sold on the merchant market thereby entering the transportation system.

#### 3.3.7 Isobutane

Isobutane production figures were provided by the National LP-Gas Association on a state regional basis. This data consists of isobutane produced at natural gas processing plants. Total annual U.S. production of isobutane is roughly 288,819 metric tons. Isobutane is produced by only four manufacturers in the U.S.

Producer	<b>Production Site</b>
Air Products & Chemicals, Inc., Specialty Gas Dept. Gardner Cryogenics	Allentown, PA
Matheson	Lyndhurst, NJ
Phillips Chemical Co., Div. Phillips Petroleum Co. Petrochemicals Div. Technical Petroleum Co.	Borger, TX
Union Carbide Corp., Linde Specialty Gases	Linden, NJ

Figure 3-4 shows the annual U.S. production of isobutane by region. The major end uses of isobutane are for organic synthesis; as refrigerants, fuel, aerosal propellants, and highoctane gasoline (aviation fuel); and in the manufacture of synthetic rubber and instrument calibration fluid.

#### 3.3.8 Methanol

Table 3-4 shows producers, production sites and annual production capacity for methanol in the U.S.. It can be seen that the average annual production capacity of methanol is roughly 5,041,000 metric tons. Figure 3-5 shows production capacity by state. The majority of methanol (71%) is produced in Texas. The major end uses of methanol (also known as methyl alcohol) are in the manufacture of formaldehyde and dimethyl terphthalate; in chemical synthesis of methyl amines, methyl chloride, methyl methacrylate, etc.; as an aviation fuel (for water injection); in the manufacture of automotive antifreeze; as a solvent for nitrocellulose, ethylcellulose, polyvinyl butyral, shellac, rosin, manila resin, dyes; as a denaturant for ethyl alcohol; as a dehydrator for



# TABLE 3-4. PRODUCERS AND PRODUCTION CAPACITY OF METHANOL\*

Producer	Production Bitm	Annual Capācity (Thousands of <u>Mātric Tona)</u>
Air Products and Chems., inc. Plastics Div.	Pensacola, PL	163
Allemania Chem. Co	Plaquemiae, LA	<b>Ż90</b>
Borden Inc. Borden Chem. Div. Petrochems Div.	Oelemar, LA	580
Celanese Corp. Celanese Chem. Co., Inc.	Bishop, TX Clear Lake, TX	<del>514</del> 834
E.I. du Pont de Nemours & Co., Inc. Chems., Dyes and Pigments Dept.	Beaumont, TX Deer Park, TX	<del>544</del> 726
Georgia-Pacific Corp. Chem. Div.	Plaquemine, LA	<b>4</b> 35
Monsanto Co. Monsanto Chem. Intermediates Co.	Texas City, TX	363
Tenneco Inc. Tenneco Chems., Inc.	Passdena, TX	290
	TOTAL	5,041

\* Source: SRI International, 1981 Directory of Chemical Producers.


automotive antifreeze; as a solvent for nitrocellulose, ethylcellulose, polyvinyl butyral, shellac, rosin, manila resin, dyes; as a denaturent for ethyl alcohol; as a dehydrator for natural gas; as fuel for utility plants (methyl fuel); and as feedstock for manufacture of synthetic proteins by continuous fermentation.

## 3.3.9 Methyl Bromide

According to SRI International's <u>1961 Directory of Chemical Producers</u>, methyl bromide is manufactured by only two manufacturers in the U.S. They are Dow Chemicals, U.S.A. in Midland, MI and Great Lakes Chemical Corp. in El Dorado, AR. Since the number of producers are small, production volumes are not available from the Bureau of Census. The major end uses of methyl bromide are as a soil and space fumigant; in disinfestation of potatoes, tomatoes and other crops; and in organic synthesis.

## 3.3.10 Monomethylamine Nitrate

According to the NTSB investigation of the Wenatchee, WA transportation accident involving monomethylamine nitrate solution (PRM) (NTSB-RAR-76-1), PRM was originally manufactured by E. I. DuPont in Biwabik, MI and shipped to its operations in Dupont, WA where it was used as a sensitizer in the formulation of an explosive called TOVEX. Prior to August 6, 1974 when an explosion occurred in the Apple Yard at Wenatchee, WA approximately 18 cars per year were shipped to DuPont's plant. Assuming an average of 10,000 gallons of PRM shipped per tank car, this means a total of 180,000 gallons of 86 percent monomethylamine nitrate solution were moved each year (approximately 560 metric tons). However, the DOT special transportation permit for this material was suspended on August 8, 1974, thus suspending shipments.

## 3.3.11 Propylene

Table 3-5 was prepared to show producers, production sites and annual production volume for propylene in the U.S. It can be seen that approximately 9,980,000 metric tons are produced annually. Of these, 64% are produced in Texas. Figure 3-6 shows propylene production by state. The major end uses of propylene are in the manufacture of isopropyl alcohol, polypropylene, synthetic glycol, acrylonitrile, propylene oxide, heptene, cumene, polymer gasoline, acrylic acid, vinyl resins, and oxo-chemicals.

## TABLE 3-5PROPYLENE PRODUCERS AND PRODUCTION VOLUME

Producer	Production Site	Annual Capacity (Thousands of <u>Metric Tons</u> )
Allied Chem. Corp./ BASF Wyandotte Corp/ Borg-Warner Corp.	Geismar, LA	23
American Petrofina Inc. Cosden Oil & Chem. Co., subsid.	Big Spring, TX Groves, TX	59 52
Ashland Oil, Inc. Ashland Chem. Co., div Petrochems Div.	Ashland, KY Louisville, KY	75 14
Atlantic Richfield Co. Arco Chem. Co., div.	Channelview, TX Houston, TX Wilmington, CA	726 91 36
The Charter Co. Charter Oil Co., subsid. Charter International Oil Co., subsid.	Houston, TX	109
Chemplex, Co.	Clinton, IA	79
Cities Service Co. Chems & Minerals Group Petrochems. Div.	Lake Charles, LA	254
Clark Oil & Refining Corp. Clark Chem. Corp. Subsid.	Blue Island, IL Wood River, IL	132 29
The Coastal Corp. Coastals States Marketing, Inc. subsid.	Corpus Christi, TX	25
Conoco Inc. Conoco Chems. Co. Div.	Chocolate Bayou, TX Lake Charles, LA	249 11
Corpus Christi Petrochem Co.	Corpus Christi, TX	254
Dow Chem. U.S.A.	Freeport, TX Plaquemine, LA	249 349

.

.

<u>\_\_\_\_</u>

## TABLE 3-5 (cont'd)

E.I. du Pont de Nemours & Co., Inc. Polymer Products Dept.	Orange, TX	45
Eastman Kodak Co. Eastman Chem. Products, Inc., subsid. Texas Eastman Co.	Longview, TX	204
El Paso Natural Gas Co. El Paso Products Co., subsid.	Odessa, TX	77
Enterprise Products Co. Enterprise Petrochems Co., subsid.	Mont Belvieu, TX	172
Exxon Corp., Exxon Chem. Co., div. Exxon Chem. Americas	Baton Rouge, LA Baytown, TX Bayway, NJ	726 658 181
Getty Oil Co., Getty Refining & Marketing Co., subsid.	Delaware City, DE El Dorado, KS	54 25
The BF Goodrich <b>Co.</b> BF Goodrich <b>Chem. Group</b>	Calvert City, KY	59
Gulf Oil Corp Culf Oil Chems. Co. Petrochems. Div.	Cedar Bayou, TX Philadelphia, PA Port Arthur, TX	356 82 286
InterNorth, Inc. Northern Petrochem, Co. subsid. Petrochems. Div.	Morris, IL	91
Marathon Oil Co.	Detroit, MI Texas City, TX	50 68
Mobil Corp., Mobil Oil Corp, Mobil Chem. Co., div. Petrochems., div.	Beaumont, TX	109
Monsanto Co. Monsanto Chem. Intermediates Co.	Chocolate Bayou, TX	249
Phillips Petroleum Co. Petrochems. Div.	Sweeny, TX	254

3-18

1

TABLE 3-5 (cont'd)

Shell Chem. Co.	Deer Park, TX Norco, LA Wilmington, CA Wood River, IL	862 227 50 113
Standard Oil Co. of California. Chevron Chem. Co., subsid. Petrochems. Div.	El Segundo, CA Richmond, CA	48 86
Standard Oil Co. (Indiana) Amoco Chems. Corp, subsid.	Chocolate Bayou, TX Sugar Creek, MO Texas City, TX	363 16 175
Standard Oil Co. (Indiana) Amoco Oil Co., subsid.	Whiting, IN Wood River, IL Yorktown, VA	136 59 7
The Standard Oil Co. (Ohio)	Lima, OH	122
Sun Co., Inc. Sun Oil Co. of PA subsid. Sun Petroleum Products Co., subsid	Corpus Christi, TX Marcus Hook, PA Toledo, OH	45 179 23
Texaco Inc. Texaco Chem. Co. Div.	Port Arthur, TX Port Neches, TX Westville, NJ	227 66 ^5
Texas City Refining Inc.	Texas City, TX	50
Tosco Corp.	Duncan, OK	34
Union Carbide Corp. Chems and Plastics Div	Seadrift, TX Taft, LA Texas City, TX Torrance, CA	50 191 88 20
Union Oil Co. of CA	Beaumont, TX	29
Union Pacific Corp. Champion Petroleum Co., subsid.	Corpus Christi, TX	73
United States Steel Corp. USS Chems., Div.	Houston, TX	54
	TOTAL	9,980

and the second second



3-20

 ÷.

ļ

Ľ

## SODIUM HYDROSULFIDE SOLUTION PRODUCERS AND PRODUCTION VOLUME

Producer	Production Site	Annual Capacity (Thousands of <u>Metric Tons)</u>
Chem. Products Corp.	Cartersville, GA	2
Dow Chem. USA	Magnolia, AK	n/a
Merichem Co.	Houston, TX	n/a
PPG Indust., Inc. Chems. Group Chem. Division-U.S.	Natrium, WV	29
Stauffer Chem. Cc. Indust. Chem. Div.	Delaware City, DE Dominguez, CA	18 4
Tosco Corp.	El Dorado, AK	n/a
West Chem. Products. Inc. West Agro-Chemical Inc., subsid.	Eighty Four, PA	n/a
Witco Chem. Corp. Argus Chem. Div.	Taft, LA	_2
	TOTAL	55



state. It appears that the major production of this commodity occurs in West Virginia and Delaware. Sodium hydrosulfide is used principally in paper pulping; processing of dye stuffs; in rayon and cellophane desulfurizing; for dehairing hides; and as a bleaching reagent.

## 3.3.13 Sodium Hydroxide Solution

Table 3-7 shows producers, production sites and volumes for sodium hydroxide in the U.S. It can be seen that approximately 13,844,000 metric tons of this commodity are produced annually. Figure 3-8 shows the U.S. distribution of sodium hydroxide manufacture by state. The major end uses of sodium hydroxide are in chemical manufacture; in rayon and cellophane production; in petroleum refining; in pulp and paper; in aluminum; in detergents, soap, and textile processing; in vegetable oil refining; for reclaiming rubber; for regenerating ion exchange resins; in organic fusions; for peeling of fruits and vegetables in the food industry; for laboratory applications; and etching and electroplating.

## 3.3.14 Styrene Monomer (inhibited)

Table 3-8 shows producers, production sites and volumes for styrene monomer (inhibited) in the U.S. This table shows that approximately 4,078,000 metric tons of styrene monomer are produced annually. Figure 3-9 shows styrene monomer production by state. It can be seen that 94% of the total produced is manufactured in Texas and Louisiana by oil, natural gas and chemical companies. Much of the styrene produced is used captively for production of other chemicals and materials. The major end uses of styrene monomer are for manufacture of polystyrene plastics; in the production of ion exchange, alkyl benzene sulfonate and styrene-acrylonitrile polymer resins; as protective coatings (styrene-butadiene latex; alkyds); in styrenated polyesters; in the production of rubber-modified polystyrene and copolymer resins; and as intermediates.

## 3.3.15 Toluene

Table 3-9 shows producers, production sites and annual production volume for toluene in the U.S. Approximately 591,000 metric tons of toluene are produced annually. The production of toluene by state is shown in Figure 3-10. It can be seen that 62% of all toluene is manufactured in Texas. Most of the toluene is manufactured from catalytic reformate from oil and natural gas catalytic cracking operations. The major end uses of toluene are in aviation gasoline and high-octane blending stock; in benzene,

## TABLE 3-7SODIUM HYDROXIDE PRODUCERS AND PRODUCTION VOLUME

|

Producers	Production Sites	Annual Capacity (Thousands of <u>Metric Tons)</u>
Aluminum Co. of America	Point Comfort, TX	165
BASF Wyandotte Corp. Indust. Chems. Group Basic Chems. Div.	Geismar, LA	327
Brunswick Pulp & Paper Inc., Brunswick Chem. Co., div.	Brunswick, GA	31
Champion International	Canton, NC	29
divChems and Associated Products	Pasadena, TX	22
Chemtech Indust. Inc. Eastern Div.	Solvay, NY	N.A.
Convent Chem. Corp.	Calvert City, KY	127
Diamond Shamrock Corp. Indust Chems & Plastics Unit., Electro Chems Div.	Deer Park, TX Delaware City, DE La Porte, TX Mobile, AL Muscle Shores, AL	383 161 533 44 164
Dow Chemical, U.S.A.	Freeport, TX Midland, MI Oyster Creek, TX Pittsburg, CA Plaquemine, LA	2,445 282 349 249 1,148
E.I. du Pont de Nenours & Co., Inc. Petrochems. Dept. Freon Products Div.	Corpus Crhistie, TX	326
FMC Corp. Indust. Chem. Group	South Charleston, WV	290
Fort Howard Paper Co.	Green Bay, WI Muskogee, OK	7 5

TABLE 3-7 (cont'd)

Gen. Electric. Co. Engineered Materials Group Plastics Business Operations	Mount Vernon, IN	54
Georgia-Pacific Corp. Chem. Div.	Bellingham, WA Plaquemine, LA	78 425
Hercules Inc.	Hopewell, VA	32
ICI Americas Inc. Petrochems. Div.	Baton Rouge, LA	154
International Minterals and Chem. Corp. IMC Chem. Group Indust. Chems. Div.	Niagra Falls, NY Orrington, ME	N.A. 79
Kaiser Aluminum and Chem. Corp. Kaiser Indust. Chems. Div.	Gramercy, LA	209
K.A. Steel Chems. Inc. Steelco Chem. Corp. subsid.	Lemont, IL	N.A.
Linden Chems and Chems. Div.	Brunswick, GA Acme, NC Linden, NJ Moundsville, WV Syracuse, NY	104 50 163 86 91
Monsanto Co. Monsanto Chem. Intermediates Co.	Sauget. IL	44
Occidental Petroleum Corp. Hooker Chem. COrp subsid. Indust. Chems Group Operations Div.	Montague, MI Niagara Falls, NY Tacoma, WA Taft, LA	82 279 195 562
Olin Corp. Olin Chems Group	Augusta, GA Charleston, TN McIntosh, AL Niagra Falls, NY	112 253 508 93
Pennwah Corp. Chems Group Inorganic Chem. Div.	Calvert City, KY Portland, OR Tacoma, WA Wynndotte, MI	120 188 90 100

## TABLE 3-7 (cont'd)

PPG Indust. Inc. Chems. Group Chem. Division-U.S.	Barberton, OH Lake Charles, LA Natrium, WV	127 1,138 267
Richardson-Merrell, Inc. J.T. Baker Chem. Co., subsid.	Phillipsburg, NJ	3
Shell Chem. Co.	Deer Park, TX	104
Stauffer Chem. Co. Indust. Chem. Div.	Henderson, NE Le Moyne, AL St, Gabriel, LA	113 77 165
Vulcan Materials Co. Chems. Div.	Denver City, TX Geismar, LA Port Edwards, WI Wichita, KA	13 206 72 182
Weyeerhaeuser, Co.	Longview, WA	139
	TOTAL	13,844

•

× 2



3-27

.

.

## STYRENE MONOMER (INHIBITED) PRODUCERS AND PRODUCTION VOLUME

Producers	Production Sites	Annual Capacity (Thousands of <u>Metric Tons)</u>
American Hoechst Corp Indust. Chems. Div. Petrochems. Div. Plast cs Div.	Baton Rouge, LA Bayport, TX	272 408
Atlantic Richfield Co. Arco/Polymers, Inc., subsid.; Oxirane Internat'l., subsid. Oxirane Chem. Co. (Channelview)	Beaver Valley, PA Channelview, TX	100 454
Cos-Mar. Inc.	Carville, LA	590
Dow-Chem. USA	Freeport,TX Midland, MI	689 136
El Paso Natural Gas Co., El Paso Products Co., subsid.	Odessa, TX	115
Gulf Oil Corp. Gulf Oil Chems. Co. Petrochems. Div.	St. James, LA	272
Monsanto Co. Monsanto Chem. Intermediates Co.	Texas City, TX	680
Standard Oil Co. (Indiana) Amoco Chems. Corp., subsid.	Texas City, TX	272
Sun Co., Inc., Sun Oil Co. of Pennsylvania, subsid. Sun Petroleum Products Co., subsid.	Corpus Christi, TX	36
United State Steel Corp. USS Chems., Div.	Houston, TX	54
	TOTAL	4,078



## TABLE 3-9 TOLUENE PRODUCERS AND PRODUCTION VOLUME

Producer	Production Site	Annual Capacity (Thousands of <u>Metric Tons)</u>
American Petrofina Inc. American Petrofina Co. of Texas, subsid.	Beaumont, TX	17
Cosden Oil & Chem. Co., subsid.	Big Spring, TX	23
Ashland Oil, Inc. Ashland Chem. Co. div. Petrochems. div.	Ashland, KY North Tonawanda, NY	14 11
Atlantic Richfield Co. Arco Chem. Co., div	Channelview, TX Houston, TX Wilmington, CA	15 17 17
Bethlehem Steel Corp.	Sparrows Point, MD	7
CF & I Steel Corp.	Pueblo, CO	N.A.
The Charter Co. Charter Oil Co., subsid. Charter International Oil Co., subsid	Houston, TX	5
The Coastal Corp. Coastal States Marketing Inc., subsid.	Corpus Christie, TX	8
Crown Central Petroleum Corp. Chem.Div.	Pasedena, TX	6
Dow Chem. U.S.A.	Freeport, TX	2
Exxon Corp. Exxon Co., U.S.A.	Baytown, TX	57
Getty Oil Co. Getty Refining and Marketing Co. subsid.	Deleware City, DE El Dorado, KS	15 3
Gulf Oil Corp. Gulf Oil Chems. Co. Petrochems. Div.	Alliance, CA Philadelphia, PA Port Arthur, TX	27 13 24

## TABLE 3-9 (cont'd)

•

オンドー・

Kerr-McGee Corp. Southwestern Refining Co., Inc., subsid.	Corpus Christie, TX	20
LTV Corp. Jones and Laughlin Steel Corp. subsid. Eastern Div.	Aliquippa, PA	1
Marathon Oil Co.	Texas City, TX	10
Mobil Corp. Mobil Oil Corp. MobilChem. Co., div. Petrochems div.	Beaumont, TX	51
Monsanto Co. Monsanto Chem. Intermediates Co.	Chocolate Bayou, TX	23
Nueoes Petrochem Co.	Corpus Christie, TX	8
Phillips Petroleum Co. Petrochems. Div. Phillips Puerto Rico Core Inc., subsid	Sweeney, TX	5
Shell Chem. Co.	Deer Park, TX	27
Sun Co., Inc. Sun Oil Co. of PA, subdiv, Sun Petroleum Products Co. Subsid.	Corpus Christie, TX Marcus Hook, PA Toledo, OH Tulsa, OK	19 21 34 9
Tenneco Inc. Tenneco Oil Co., div.	Chalmette, LA	16
Texaco Inc. Texaco Chem. Co., div.	Port Arthur, TX Westville, NJ	13 18
Union Carbide Corp. Chems. & Plastics, div.	Taft, LA	9
Union Oil Co. of CA	Beaumont, TX Lemont, IL	N.A. 8
Union Pacific Corp. Champion Petroleum Co., subsid.	Corpus Christie, TX	14
United States Steel Corp. USS Chems., div.	Clarion, PA Geneva, UT	4 <u>N.A</u>
	TOTAL	591



phenol and caprolactam; as a solvent for paints and coatings, gums, resins, most oils, rubber, vinyl organosols; as a diluent and thinner in nitrocellulose lacquers; as an adhesive solvent in plastic toys and model airplanes; in chemical processing of benzoic acid, benzl and benzoyl derivatives, saccharin, medicines, dyes and perfumes; as a source of toluene diiscyanates (polyurethane resins); in explosives (TNT) manufacture; in the production of toluene sulfonates (detergents); and as a scintillation counter.

## 3.3.16 Vinyl Acetate

Table 3-10 shows producers, production sites and annual production volume for vinyl acetate in the U.S. This table shows that approximately 1,088,000 metric tons of vinyl acetate are produced annually. Figure 3-11 shows vinyl acetate production volume by state. It can be seen that all production of vinyl acetate occurs in the South Central region, with 1,020,000 metric tons (94% of the total) being manufactured in Texas. The major end uses of vinyl acetate include the manufacture of polyvinyl acetate, polyvinyl alcohol, polyvinyl butyral, and polyvinyl chloride-acetate resins which are used in latex paints; in paper coating; as adhesives; in textile finishing; and in safety glass interlayers.

## **3.4 SHIPMENT QUANTITIES**

Table 3-11 shows the distribution of commodities shipped by various modes of transportation in the U.S. Data was obtained from the <u>Census of Transportation</u> and shipment modes were classified as rail, highway and other. Transport modes classified as "other" include such carrier types as barge, pipeline, and aircraft. Based upon the classification indices used in the census data several commodities were grouped into commodity types. Thus, not all modal distributions are commodity specific, but appear to be indicative of the variation of shipments between different transportation modes. Of the 11 commodity classes it appears that the rail mode resulted in the largest shipment volume in six classes; while both highway and "other" transport classes resulted in the largest shipment frequency in two classes, respectively. Overall, the rail mode resulted in 30 percent and "other" modes 14 percent

## 3.5 CONTAINERS REQUIRED FOR SHIPPING

Types of containers which can be used in transporting the chemicals and propellants are outlined in Title 49 of the Code of Federal Regulations. It should be noted that the types of containers which can be used for highway and rail transport will

## VINYL ACETATE PRODUCERS AND PRODUCTION VOLUME

Producer	Production Site	Annual Capacity (Thousands of <u>Metric Tons)</u>
Borden Inc., Bordon Chem. Div. Petrochems. Div.	Geismar, LA Bay City, TX	68 193
Celanese Chem. Co., Inc.	Clear Lake, TX	193
E.I. du Pont de Nemours and Co., Inc. Polymere Products Dept.	La Porte, TX	181
National Distillers & Chem. Corp. Chems. Div. U.S. Indust Chems. Co., Div.	Deer Park, TX	272
Union Carbide Corp. Chems & Plastics, div.	Texas City, TX	181
	TOTAL	1,088



ł

La a a a a a a a a a

4

Þ

h ŀ

# DISTRIBUTION OF COMMODITIES SHIPPED BY VARIOUS MODES OF TRANSPORTATION IN THE U.S.

Other of (Thousands ( <u>Metric Tons</u>	5	482	1,998	1,084	377	ทล		616
Highway (Thousands <u>Metric Ton</u>	23	. 711	1,060	1,774	214	na	6	3,574
Rail (Thousands of <u>Metric Tons)</u>	27	1,345	1,020	2,183	794	ทล	7	6,079
Thousands of Metric Tons <u>Transported</u>	55	2,538	4,078	5,041	1,088	ла	. 16.5	10,269
Commodity	Sodium Compounds (Sod. Hydrosulfide)	Miscellaneous Acyclic Organic Chemical Products (Acetone, Acrylonitrile, Ethyl Acrylate)	Miscellaneous Cyclic Organic Chemical Products (Styrene)	Alcohols (Methanol)	Organic Acids and Salts (Vinyl Acetate)	Industrial Organic Chemicals, nec. (Acetone Cyanohyrdin)	Industrial Inorganic Acids (Hydrocyanic Acid)	Liqueficd Petroleum and Coal Gases (isobutane, propylene)

-		ther sands of c Tons)	-	554	260
		0 (Thou Metri			
		ıway ands of Tons)	65	81	25
		High (Thouse <u>Metric</u>	1,0	κ. Έ	.1
	ont'd)	l nds of Tons)	2	6	, v
	LE 3-11 (c	Rai (Thousa Metric	55	10,00	20
	TAB	ls of Tons rted			
		Thousand Metric <u>Transpo</u>	-1,629	13,844	591
		ity	Jases, nyl	lies ^oxide)	sts from stroleum ()
		Commod	ndustrial ( .e.c. (metl romide)	odium Alka sodium hydr	rude Produc oal andn Pe ar (toluene

vary based on the type of commodity being shipped. Various shipping containers are available including cylinders, tank cars, cargo tanks, portable tank containers, and overpack containers such as wooden and fiberboard boxes and polystyrene packages. The following discussion identifies the types of cylinders, tank cars, cargo tanks and portable tank containers for shipping each chemical.

Commodity specific requirements for container types and specifications can be found in 49 CFR, Parts 100-179. A list of the applicable container requirements including restrictions on container specifications and volumes for each commodity is given in Table 3-12. An overview of container types approved for the 16 commodities is given in Table 3-13. The container specifications for each chemical are discussed individually.

## 3.5.1 Acetone

Acetone, a flammable liquid with a flash point of  $0^{\circ}$ F, can be shipped in packaging and containers as outlined in Section 173.119 of 49 CFR. These include 5 gallon glass carboys; 17E and 17C metal drums; 5, 5A, 5B, 5C, 5M metal barrels or drums; 10 gallon pails; 42B, 42C or 42H aluminum barrels or drums; cylinders, tank cars, tank motor vehicles, and portable tank containers. The cylinders, tank cars, cargo tanks and portable tanks approved for acetone use with the applicable container specifications are shown in Tables 3-14.

## 3.5.2 Acetone Cyanohydrin

Acetone cyanohydrin, a Poison B, can be shipped in containers specified in Section 173.346 of 49CFR. These include Spec. 5, 5A, 5B, 5C, 17C, 173, 37A, 37B metal drums; 1 quart glass or earthenware and gallon metal inside containers; cylinders, tank cars, cargo tanks and portable tanks. The cylinders specified for acetone cyanohydrin are similar to those for acetone with the exception of specification 4E, 9, 39, 40 or 41 packagings which may not be charged and shipped with a Poison B material. The tank cars, cargo tanks and portable tanks specified for acetone cyanohydrin service and applicable container specifications are shown in Table 3-15.

## 3.5.3 Acrylonitrile

The types of shipping containers and applicable container specifications for acrylonitrile are the same as those for acetone with the addition of some drums; inside and overpack containers (173.119(b)) because acrylonitrile has a higher flash point than acetone.

TABLE 3-12 REGULATIONS APPLICABLE TO CONTAINER REQUIRMENTS<sup>#</sup>

-

Requirements Specific 173.304 173.314 173.315 173.304 173.314 173.315 173.119 173.332 173.353 173.346 173.119 173.119 173.119 Packaging Exceptions 173.118 173.118 173.306 173.306 173.118 None None None None Liquid & Poison Flammable Gas Flammable Liquid Flammable Flammable Flammable Flammable Flammable Required & Poison Labels Poison Liquid Liquid Poison Gas Gas Flammable Liquid Flammable Flammable Flammable Flammable Poison B Flammable Poison A Poison B Liquid Liquid Hazard Class Liquid Gas Gas Ethyl Acrylate Methyl Bromide Acrylonitrile Cyanohydrin Hydrocyanic **Propylene** Isobutane Chemical Methanol Acetone Acetone Acid

Container specifications found in parts 178 and 179 in the commodity specific sections.

\*

TABLE 3-12 (continued)

Specific <u>Requirements</u>	N/A	173.249	173.119	173.119	173.119	N/A
<u>Packaging</u> Exceptions	N/A	173.244	173.118	173.118	173.118	N/A
Labels <u>Required</u>	N/A	Corrosive	Flammable Liquid	Flammable Liquid	Flammable Líquíd	N/A
Hazard Class	N/A	Corrosive	Flammable Liquid	Flammable Liquid	Flammable Liquid	N/A
Chemical	Sodium Hydrosulfide	Sodium Hydrovide	Styrene	Toluene	Vinyl Acetate	Monomethylamine Nitrate

## TABLE 3-13 Approved container types for all commodities

## Drums/Barrels Packaging

	Chea	Acet	Acet Cyan	Acry	Ethy -	Hych	Isot	Meth	Meth	Prop	Sodi Hydr	Sodi	Styr	Tolu	
	<u>ical</u>	one	one ohydrln	lonitrile	l Acrylate	rocyanic Acid	utane	anol	yl Bromide	ylene	u <b>m</b> osulfide	um Hydroxide	ene	ene	
	Tank Motor Ve- hicles	•	٠	٠	•	-		•						٠	
	Tank Cars	•	•	•	•	٠	٠	•	•	•		٠		٠	
	Cargo Tank						•		٠	٠		•			
	Port- able Tank	٠	٠	٠	٠		٠	٠		٠		٠		٠	
	Cy- Linders	•	•	٠	٠	٠		•	•					٠	
Drums/	Steel	•	•	٠	•			•	٠			•		٠	
Barre k	Alum.	•	٠	٠	•			٠						•	
Pa	Wooden Boxes	•	•	٠	٠	٠		٠				٠		•	
ckaging	Fiber- board Boxes	٠	•	•'	•	•		•				٠		vi N	
	Glass Ampules or Viles														
Insid	Cart- ridges														
e Conta	Glass Carboys in <u>Boxes</u>	٠	•	٠	•			•						•	
ners	Glass Bottle	٠		•	•			•						٠	
	Expande Poly- styrene	٠	•	٠	•			•						•	
	Poly Conta														

:

## CYLINDERS, TANK CARS, CARGO TANKS, AND PORTABLE TANK CONTAINERS SPECIFIED FOR ACETONE SERVICE

Cylinders (specifications ` section):	3; 3A (178.36) 3AA (178.37); 3B (178.38); 3BN (178.39); 3D (178.41); 3E (178.42); 4 (178.48); 4A (178.49); 4B (178.50); 4BA (178.51); 4BW (178.61); 4E (178.68); 9; 25; 26; 38; 39 (178.65); 40; 41
Tank Cars (specifications	
section):	103; 103W; 103ALW; 103DW; 104; 104W; 111A60ALWI; 111A60FI; 111A60WI; 111A100W3; 111A100W4; 111A100W6; 115A60WI; 115A60ALW; 115A60W6 (179.200); 105A100; 105A100ALW; 109A100ALW; 109A300W; 112A200W; 112A400F; 114A340W (179.100); 106A500X; 106A800XNC; 106A800NCI; 110A500W (179.300)
Cargo Tanks (specifications section):	MC300; MC301; MC302; MC303; MC304; MC305 (178.340); MC306 (178.341); MC307 (178.342); MC330; MC331 (178.337)

Portable Tanks (specifications section):

DOT 51 (178.245); DOT 57 (158.253)

## TANK CARS, CARGO TANKS AND PORTABLE TANK CONTAINERS SPECIFIED FOR ACETONE CYANOHYDRIN SERVICE

Tank Cars (specifications	
section):	103; 103W; 103A; 103ALW; 103AW; 103BW; 104; 104W; 109A300ALW; 111A60ALW1; 111A60F1; 111A60W1; 111A60WZ; 111A100W4; 115A60W6 (179.200); 105A100; 105A100W (179.100)
Cargo Tanks (specifications section):	MC300; MC301; MC302; MC303; MC305 (178.340); MC306 (178.341); MC310 (178.342); MC312 (178.343)
Portable Tanks (specifications section):	DOT 51 (178.245)

## 3.5.4 Aerozine-50

Aerozine-50 is composed of a 50%/50% mixture of UDMH and hydrazine, and is formulated at Rocky Mountain Arsenal for USAF consumption. It can be shipped only in Air Force R-16 and R-17 trailers or commercially owned and operated tank motor vehicles covered under DOT exemption DOT-E3121. The commercially owned vehicles are operated by WS. Hatch Co., Pacific Intermountain Express, and Lemmon Transport Co., Inc.

## 3.5.5 Ethyl Acrylate

Shipping containers and applicable container specifications for ethyl acrylate are the same as those for acrylonitrile.

## 3.5.6 Hydrocyanic Acid

Hydrocyanic acid can be shipped in metal cans surrounded by absorbent inert material overpacked in wodden or fiberboard boxes with waterproof liners; metal cylinders or tank cars as a Class A poison. The types of cylinders and their specifications approved for hydrocyanic acid service are 3A480 (Section 178.36), 3AA480 (Section 178.37), and 3A480X (Section 178.43). The tank cars in which hydrocyanic acid can be shipped with the sections of 49CFR detailing the specifications are 105A500W (Section 179.100) and 105A600W (Section 179.101). Each tank car in HCN service must be stencilled "Hydrocyanic Acid" as per Section 172.330 of 49CFR.

## 3.5.7 Isobutane

Isobutane can be shipped in cylinders, tank cars, cargo tanks and portable tank containers as a flammable gas. Table 3-16 shows the types of cylinders which can be used for shipping isobutane.

The types of tank cars for shipping isobutane along with the specification sections are 105A100 (179.100), 105A100W (179.100), 111A100W4 (179.200), 105A200W (179.100), 105A200ALW (179.100), 105A300W (179.100), 112A340W (179.100), 114A340W (179.100), 114A400W (179.100), 105A400W (179.100), 112A400F (179.100), 112A400W (179.100), 105A500W (179.100), 106A500X (179.300) and 105A600W (179.100).

The cargo tanks and portable tank containers permitted for isobutane service are the DOT 51 (Section 178.245), MC 330 and MC 331 (Section 178.337).

## CYLINDERS APPROVED FOR SHIPPING ISOBUTANE (LIQUEFIED PETROLEUM GAS)

Cylinders (specifications section):

3; 3A (178.36); 3AA (178.37); 3E (178.42); 4 (178.48); 4A (178.49); 4B (178.50); 4BA (178.51); 4B240ET (178.55); 4BW (178.61); 4P240X; 4B24FLW (178.54); 4E (178.68); 9; 25; 26; 38; 39 (178.65); 41

## 3.5.8 Methanol

Methanol can be shipped in glass carboys, drums and barrels, pails, glass, and earthenware containers, aluminum drums, cylinders, tank cars, cargo tanks and portable tank containers as a flammable liquid with a flash point of  $65^{\circ}$ F. The specified shipping containers and applicable sections of 49 CFR detailing container specifications are the same as those for acrylonitrile.

## 3.5.9 Methyl Bromide

Methyl bromide can be transported in specification 5A metal drums; overpacked 1 pound metal cans; cylinders; tank cars; and cargo tanks. The cylinders, tank cars and cargo tanks specified for methyl bromide service with the applicable container specification sections of 49 CFR are shown in Table 3-17.

## 3.5.10 Monomethylamine Nitrate (PRM)

In 1968 Dupont was authorized by the DOT 's ship this material classed as a flammable solid, further identified as an 85-86% aqueous solution of monomethylamine nitrate crystals in DOT specification 103ALW and 103W tank cars (179.200) and MC306, MC307 and MC312 cargo tanks. However, following the explosion at the Apple Yard in Wenatchee, WA in 1974, the special permit DOT5737 was suspended.

## 3.5.11 Propylene

Propylene, like isobutane, is a liquefied petroleum gas and can be shipped in the same container types as isobutane.

## 3.5.12 Sodium Hydrosulfide

Sodium hydrosulfide is not a DOT regulated hazardous material and has no required container or container specifications in 49 CFR.

## 3.5.13 Sodium Hydroxide

Sodium hydroxide solution can be shipped in Specification 5 metal drums, in glass, earthenware, polyethylene or metal inside containers in certain quantities with overpack fiberboard or wooden boxes, tank cars, cargo tanks and portable tank containers as a corrosive liquid. The tank cars, cargo tanks and portable tank containers specified for

CYLINDERS, TANK CARS AND CARGO TANKS SPECIFIED FOR METHYL BROMIDE SERVICE

MC 330, MC 331 (178.337)

Cargo Tanks (specification section):

sodium hydroxide liquid service and the container specification sections in 49 CFR are given in Table 3-18.

## 3.5.14 Styrene

Styrene, a flammable liquid with a flash point of 93°F can be transported in the same shipping container types as acrylonitrile.

## 3.5.15 Toluene

Toluene, a flammable liquid with a flash point of  $40^{\circ}$ F, can be shipped in the same container types under similar constraints as acrylonitrile.

## 3.5.16 Vinyl Acetate

Vinyl acetate, a flammable liquid with a flash point of 18<sup>0</sup>F, can be shipped in the same container types with their applicable specifications as acetone.

## 3.6 COMMODITY FLOW PATTERNS

A generalized orgin-to-destination pattern was established for the chemicals and propellants between the concensus geographic divisions in the United States. The states included in each of the census geographic divisions are shown in Figure 3-12. The categorization of each commodities shipment origin involved identifying the state in which the chemical has a production plant and then grouping these state specific data into a regional format. Thus, region specific production/origin data was formulated.

To determine the quantity shipped to and consumed in each region the classification indices in the <u>Census of Transportation</u> were once again used. Data was provided as to the quantity of each commodity class being consumed in each region. Thus, a generalized origin-to-destination pattern was completed. The origin-to-destination pattern for each commodity class found in the census data are given in Tables 3-19 through 3-29. Figure 3-13 shows the variation between production volume and consumption volume for each commodity class by geographic region. It can be seen that the majority of the commodities are produced in the West South Central Region while total U.S. consumption exceeds ten percent in each of six different regions.

Origin-to-destination patterns in terms of Aerozine-50 were collected from the USAF Directorate of Energy Management at Kelly AFB, Texas. All Aerozine-50 manufactured in the U.S. is produced from hydrazine and UDMH at Rocky Mountain

TANK CARS, CARGO TANKS AND PORTABLE TANK CONTAINERS SPECIFIED FOR SODIUM HYDROXIDE SERVICE

Tank cars (specification section):	103 (179.200); 103W (179.200); 103A (179.200); 103AW (179.200); 103B (179.200); 103BW (179.200); 104 (179.200); 104W (179.200); 105A100 (179.100); 105A100W (179.100); 111A60F1 (179.200); 111A60W1 (179.200); 111A60W2 (179.200); 111A100F2 (179.200); 111A60W5 (179.200); 111A100W4 (179.200)
Cargo Tanks (specification selection):	MC 303; MC 310; MC 311; MC 312 (178.343)
Portable Tanks (specification section):	DOT 57 (178.253); DOT 60 (178.255)

Arsenal near Denver. This data showed that 92 metric tons (101 tons) are shipped by highway trailer and 155 metric tons (170 tons) are shipped by rail tank car. Table 3-30 and Figure 3-14 shows the trailer routes of Aerozine-50 from Rocky Mountain Arsenal to Little Rock AFB, McConnell AFB, Davis-Monthan AFB and Vandenburg AFB.



3-51

\_ . . .
# ORIGIN-DESTINATION OF SODIUM COMPOUNDS (SODIUM HYDROSULFIDE)

Census Geographic <u>Division</u>	Quantity Produced In Region (Thousands of <u>Metric Tons)</u>	Quantity Consumed In Region (Thousands of <u>Metric Tons)</u>		
Pacific	4	-		
Mountain	-	1		
West North Central	-	5		
West South Central	2	15		
East North Central	-	14		
East South Central	<b>–</b>	6		
South Atlantic	49	6		
Middle Atlantic	-	7		
New England		<u> </u>		
TOTAL (U.S.)	55	55		

## ORIGIN-DESTINATION OF MISCELLANEOUS ACRYLIC ORGANIC CHEMICAL PRODUCTS (ACETONE, ACRYLONITRILE, ETHYL ACRYLATE)

Census Geographic Division	Quantity Produced In Region (Thousands of <u>Metric Tons)</u>	Quantity Consumed In Region (Thousands of <u>Metric Tons)</u>		
Pacific	60	88		
Mountain	-	<b>2</b> 5		
West North Central	25	131		
West South Central	1,387	642		
East North Central	550	336		
East South Central	158	374		
South Atlantic	82	399		
Middle Atlantic	276	498		
New England	<u> </u>	45		
TOTAL (U.S.)	2,538	2,538		

# ORIGIN-DESTINATION OF MISCELLANEOUS CYCLIC ORGANIC CHEMICAL PRODUCTS (STYRENE)

Quantity Produced In Region (Thousands of <u>Metric Tons)</u>	Quantity Consumed In Region (Thousands of <u>Metric Tons)</u>		
-	69		
-	130		
~	6		
3,842	180		
136	107		
-	86		
~	47		
100	2,807		
	646		
4,078	4,078		
	Quantity Produced In Region (Thousands of <u>Metric Tons)</u> - - - 3,842 136 - - 100 - 4,078		

# ORIGIN-DESTINATION OF ALCOHOLS (METHANOL)

Census Geographic <u>Division</u>	Quantity Produced In Region (Thousands of <u>Metric Tons)</u>	Quantity Consumed In Region (Thousands of Metric Tons)		
Pacific	-	2,050		
Mountain	· _	20		
West North Central	-	308		
West South Central	4,878	837		
East North Central	-	610		
East South Central	-	367		
South Atlantic	163	418		
Middle Atlantic	-	323		
New England		108_		
TOTAL (U.S.)	5,041	5,041		

3-55

. . .

.

## ORIGIN-DESTINATION OF ORGANIC ACIDS AND SALTS (VINYL ACETATE)

Censu <b>s</b> Geographic <u>Division</u>	Quantity Produced In Region (Thousands of <u>Metric Tons)</u>	Quantity Consumed In Region (Thousands of <u>Metric Tons)</u>		
Pacific	-	3		
Mountain	-	-		
West North Central	-	153		
West South Central	1,088	804		
East North Central	-	16		
East South Central	-	19		
South Atlantic	-	73		
Middle Atlantic	-	20		
New England				
TOTAL (U.S.)	1,088	1,088		

## ORIGIN-DESTINATION OF INDUSTRIAL ORGANIC CHEMICALS, NEC. (ACETONE CYANOHYDRIN)

ProducedCensusIn RegionGeographic(Thousands of (DivisionMetric Tons)	Consumed In Region Thousands of Metric Tons)
Pacific na	na
Mountain na	na
West North Central na	na
West South Central na	na
East North Central na	na
East South Central na	na
South Atlantic na	na
Middle Atlantic na	na
New England na	na

## ORIGIN-DESTINATION OF INDUSTRIAL INORGANIC ACIDS (HYDROCYANIC ACID)

Census Geographic Division	Quantity Produced In Region (Thousands of <u>Metric Tons)</u>	Quantity Consumed In Region (Thousands of <u>Metric Tons</u> *)		
Pacific	-	1		
Mountain	-	-		
West North Central	-	1.5		
West South Central	444	2		
East North Central	-	5		
East South Central	106	2		
South Atlantic	-	2		
Middle Atlantic	1	2		
New England		1		
TOTAL (U.S.)	551	16.5		

\* Ninety-seven percent of the HCN produced is used as a byproduct of other processes, and that three percent is sold on the merchant market thereby entering the transportation system. This value represents the three percent transported from all HCN production.

# ORIGIN-DESTINATION OF LIQUEFIED PETROLEUM AND COAL GASES (ISOBUTANE, PROPYLENE)

Census Geographic <u>Division</u>	Quantity Produced In Region (Thousands of Metric Tonr	Quantity Consumed In Region (Thousands of <u>Metric Tons)</u>		
Pacific	240	1,206		
Mountain	7	176		
West North Central	162	913		
West South Central	8,425	1,512		
East North Central	755	572		
East South Central	148	2,432		
South Atlantic	65	2,397		
Middle Atlantic	467	874		
New England		187		
TOTAL (U.S.)	10,269	10,269		

# ORIGIN-DESTINATION OF INDUSTRIAL GASES, NEC (METHYL BROMIDE)

Census Geographic <u>Division</u>	Quantity Produced In Region (Thousands of <u>Metric Tons)</u>	Quantity Consumed In Region (Thousands of <u>Metric Tons)</u>			
Pacific	na	35			
Mountain	na	15			
West North Central	na	67			
West South Central	na	53			
East North Central	na	1,014			
East South Central	na	181			
South Atlantic	na	264			
Middle Atlantic	na	-			
New England	na				
TOTAL (U.S.)		<u>&gt;</u> 1,629			

# ORIGIN-DESTINATION OF SODIUM ALKALIES (SODIUM HYDROXIDE)

Census Geographic <u>Division</u>	Quantity Produced In Region (Thousands of <u>Metric Tons)</u>	Quantity Consumed In Region (Thousands of Metric Tons)		
Pacific	939	1,077		
Mountain	-	215		
West North Central	295	479		
West South Central	8,679	2,168		
East North Central	768	2,722		
East South Central	1,293	1,062		
South Atlantic	1,325	2,506		
Middle Atlantic	466	3,204		
New England	79	<u> </u>		
TOTAL (U.S.)	13,844	13,844		

## ORIGIN-DESTINATION OF CRUDE PRODUCTS FROM COAL AND PETROLEUM TAR (TOLUENE)

Census Geographic Division	Quantity Produced In Region (Thousands of <u>Metric Tons)</u>	Quantity Consumed In Region (Thousands of <u>Metric Tons)</u>		
Pacific	44	-		
Mountain	-	-		
West North Central	3	-		
West South Central	398	372		
East North Central	42	-		
East South Central	14	154		
South Atlantic	22	-		
Middle Atlantic	57	65		
New England	<u>_11</u>			
TOTAL (U.S.)	591	591		

## VARIATION BETWEEN COMMODTY PRODUCTION and CONSUMPTION BY REGION

100% LEGEND 90% 80% Preduction - 14 C-Consumption 70% Pa-Pacific 60% M-Mountain WNG-West North Centra 50% WSC-West South Central 40% ENC-East North Central ESC-East South Central SA-South Atlantic 30% MA-Middle Atlantic NE-New England 20% a literation 1 10% 1.916 3. 4 1 9% 30 11 5.0 8% 影 4.5 5 7% 1 13 8% (**1**), 14 -5% 4% . -1. 1 -3 ÷ 3% 54.8 ALC: N ž ٠. ÷., \*\* ÷, 5 2% 1.19 1.6 4 . ~... ÷ Ż -6 Ŀ. . چەنچ P.,  $\tilde{i}$ 4 1 % С Ρ C ¢ Ρ c С С С P С С Ρ P Ρ P Ρ Ρ

Figure 3-13.

ENC

ESC

S A

MA

ΝE

wsc

Pa

.

м

WNC

# TABLE 3-30AEROZINE-50 (TITAN II PROPELLANT)

## TRAILER SHIPPING ROUTES

ORIGIN: ROCKY MOUNTAIN ARSENAL, DENVER CO

ROUTE AND DESTINATION: I-25 SOUTH TO HWY 87 TO I-40 EAST TO LITTLE ROCK AFB, AR

ROUTE AND DESTINATION: I-70 EAST TO I-135 SOUTH TO MCCONNELL AFB, KS

ROUTE AND DESTINATION: I-25 SOUTH TO I-10 WEST TO DAVIS MONTHAN AFB, AZ

ROUTE AND DESTINATION: I-70 WEST TO I-15 TO HWY 58 TO HWY 14 TO HWY 101 TO VANDENBERG AFB, CA



.

•

## 4. ACCIDENT ASSESSMENT

Accident statistics for 1971 through 1980 involving the additional 16 chemicals and propellants are presented in Table 4-1. This table shows accidents involving methanol (methyl alcohol) accounted for the largest percentage (49%) of the total accidents and for 54 percent of the highway and 42 percent of the rail accidents occurring from 1971 to 1980. Sodium hydroxide involvement in accidents was the next highest with 17 percent of the total accidents, 21 percent of highway and 12 percent of rail accidents. The remaining materials with significant accident rates over this time period included the LPGs isobutane and propylene, acetone, toluene, styrene, vinyl acetate, ethyl acrylate and acrylonitrile. On a modal basis, acetone, methyl bromide, methanol, sodium hydrosulfide, sodium hydroxide and toluene were involved in more accidents by highway, while acetone cyanohydrin, acrylonitrile, ethyl acrylate, hydrocyanic acid, LPGs, styrene and vinyl acetate were involved in more accidents by rail. Most of the fatalities and injuries which occurred in this sample were associated with accidents involving LPGs. This is also true of the accident costs. The total costs of the accidents from 1971-1980 involving these materialsis approximately \$25 million which averages \$4,700 per accident.

The following sections present an assessment of 29 NTSB-investigated accidents and the events at Mississauga in Ontario, Canada in terms of initial response, emergency special equipment and materials, on-scene coordination and communications, hazardous materials identification and location, release handling procedures, firefighting procedures, cleanup and disposal procedures, structural integrity assessment and wreckage removal procedures.

A listing of the accidents studied can be found in Table 4-2. These accidents involved both rail an highway mode, spanned twelve years (1968-1979), involved inservice derailments, switching accidents, overspeed impacts in yards, semi-trailer tanker truck collisions, overturns and explosions, caused 96 deaths, more than 1,600 injuries, and cost over \$40 million in property damage, not to mention third party liability damage suits and environmental clean-up, disposal and monitoring costs. These accidents occurred in both urban and rural areas under widely differing climatic conditions and provide a broad representative sample of accident types and conditions, materials involved, hazards encountered, response efforts mounted, emergency response capabilities and the differing perceptions of the effectiveness of accident handling methods. CHEMICAL/PROPELLANT ACCIDENTS (1971-1980)

INT COST 000's)	Average Cost Per <u>Accident</u>	61.1	o	8.22	•	18.1	•	23.15	
		Total Cost of <u>Accidents</u>	318.56	0	682.52	•	152.12	0.01	18,523.27
URIES		Average Numher of Injurles Per Accident	<b>0.0</b> 6	<b>c</b>	0.07	0	0.06	c	0.84
ſNI		Total Number	10	•	÷	0	~	0	668
TIES		Average Number of Fatali- ties Per Accident	0	0	0.02	•	0	0	0.07
FATAL		Total Number	0	0	2	0	•	0	55
N ACCUDENT	IOAD	Per- centage of Total Accidents	71	100	63	0	7	57	70
INVOLVED IN RAILR	RAILF	Total Num <del>ber</del> of <u>Accidents</u>	59	1	52	0	<b>65</b>	•	563
SPORTATION	WAY	Per- centage of Total <u>Accidente</u>	62	0	37	0	ព	43	0£
DE OF TRAN	нон	Total Number of <u>Accidents</u>	222	0	31	0	19	e	245
UM		Average Numher of Accidents <u>Per Year</u>	1.82	0.1	<b>6.8</b>	0	æ	0.7	<b>8</b> .
		Total Number of <u>Accidents</u>	281	-	83	•	2	2	808
		Chemical/ <u>Propellant</u>	Acetone	Acetone Cyanohydrin	Acrylonitrile	Aerozine-50	Eihyl Acrylate	llydrocyanic acid	Isobutane (LPG file)
				4 - 2					

The values for isobutane and propylene are combined in one category for all liquefied petroleum gases and so were only counted once. Thus, the total number of accidents in artificially higher.

**TABLE 4-1** 

í. j

		OW	DE OF TRAN	SPORTATION	I INVOLVED II	N ACCIDENT	FATAL	23 LL	JINI	RIES	ACCIVE (\$10	00'a)
			HOIN	WAY	RAILR	IOAD						
Chemical/ Propellant	Total Number of Accidents	Average Number of Accidenta Per Year	Total Number of <u>Accidenta</u>	Per- centage of Total Accidents	Total Number of Accidenta	Per- centage of Total Accidents	Total Number	Average Number of Fatali- tiea Per <u>Accident</u>	Total Number	Average Number of Injucies Per Accident	Total Cost of <u>Accidents</u>	Average Cost Per Accident
Methanol	2,647	264.7	1,752	66	895	36	•	•	116	0.12	3,903.98	1.47
Methyl Bromide	35	3.5	66	2	~	•	0	0	•	0.17	36.16	1.03
Monomethyla- mine nitrate	G	o	0	o	•	•	0	•	•	•	•	•
Propylene (LPG file)	808 •	2	245	30	563	70	55	0.07	999	0.84	18,523.27	23.15
Sodium Ilydrosulfide	2	ŋ.2	2	100	o	o	¢	0	0	<b>0</b>	1.05	0.53
Sodium Hydroxide	626	93.9	684	13	255	12	2	0	202	0.22	660.36	0.70
Styrene, Monomer	134	r El	16	53	103	"	o	C	12	0.09	298.89	2.23
Toluene	112	L.12	217	78	60	22	0	0	¢	0.03	235.20	58.0
Vinyl Acetate	2	8.4	22	26	29	7	0	0	16	0.19	440.59	5.25
TOTAL	5,382*	<b>538</b>	8,261	19	121,2	39	63•	0.01	1,244	0.23	24,953.84	4.64

121,5

19

8,261

\$38

5,382+

TOTAL

TABLE 4-1 (cont'd)

# TABLE 4-2 ACCIDENTS STUDIED

ï

H. T.

2

Date	Place	Type	Conditions
01/01/68	Dunreith, IN	Dr, train collision	tc puncture, release, fire explosion; vinyl chloride, ethylene oxide explodes
01/25/69	Laurel, MS	Dr	15tc LPG; Army EOD vent & burn 2tc
02/18/69	Crete, NE	Dr	<b>tc NH<sub>3</sub> rupture &amp; release; coupler blow,</b> brittle <b>steel at ambie</b> nt T of 4 <sup>0</sup> F
09/11/69	Glendora, MS	Dr	<pre>8tc vinyl chloride; fire &amp; explosion</pre>
11/29/69	NJ Turnpike Exit 2	Tank truck/ car collision	LPG; leak (12 gph)/ transfer
05/30/70	Brooklyn, NY	Tank truck	Liquefied oxygen, explosion/fire; tank contamination
04/02/71	Berwick, ME	Tank truck	Sodium hydrosulfide; off- load to contaminated storage tank, hydrogen sulfide gas emitted
08/08/71	Near Gretna, FL	Truck	Cylinders methyl bromide; cargo not adequately secured; improper cylinders
10/19/71	Houston, TX	Dr	<b>2tc vinyl</b> chloride <b>punctured, ignition &amp;</b> <b>fire; 45 min later rupture</b> <b>and rocket of another t</b> c
01/22/72	East St. Louis, IL	Rail yard accident	Coupler override, puncture tc propylene - explosion/ fire
03/09/72	Lynchburg, VA	Semitrailer overturn	LPG 9200 gal spill

.

# TABLE 4-2 (cont'd)

Date	<u>Place</u>	Type	Conditions
09/21/72	NJ Turnpike Exit 8	Semitrailer overturn	Propylene 7,200 gal spill; fire impingement & rupture
02/02/74	Oneonta, NY	Dr	7tc LPG; one tc ruptured, fire
07/19/74	Decatur, IL	Railyard accident	Overspeed impact, coupler override, puncture; 5tc isobutane release, explosion
08/16/74	Wenatchee, WA	Railyard accident	Detonation 10,000 gal monomethylamine nitrate (PRM)
09/21/74	Englewood <sup>.</sup> Yard, Houston, TX	Railyard accident	Overspeed impact, coupler override, puncture; butadiene release, fire, explosion
04/29/75	Near Eagle Pass, TX	Semitrailer overturn	8,748 gal LPG release, fire, explosion
05/11/76	Houston, TX	Tank truck overturn	NH <sub>3</sub> release, 7,500 gal
05/16/76	Glen Ellyn, IL	Dr	tc punctured, 20,000 gal NH <sub>3</sub> released
09/24/77	Beattyville, KY	Tank truck	8,000 gal gasoline released, fire
11/09/77	Pensacola, FL	Dr	2tc NH <sub>3</sub> punctured, release
02/22/78	Waverly, TN	Dr	Subsequent rupture, ignition, explosion LPG tc
02/26/78	Youngstown, FL	Dr	2tc Cl <sub>2</sub> punctured, release
03/29/78	Lewisville, AK	Dr.	tc vinyl chloride rupture, explosion, fire

# TABLE 4-2 (cont'd)

Date	Place	Type	Conditions
10/17/78	Claxton, KY (Dawson Springs, KY; Princeton, KY)	Dr	Vinyl chloride, ethyl acrylate; vent and burn tc vinyl chloride
04/08/79	Crest <b>view,</b> FL	Dr	<b>2tc NH<sub>3</sub> rupture, rocket</b> 1 <b>tc Cl<sub>2</sub> leaking</b>
11/08/79	Inwood, IN	Dr	Fire, release acetic anhydride, isobutyl alcohol, ethylacrylate, methacrylic acid, butyl cellosolve
11/10/79	Mississauga Ontario, Canada	Dr	22tc LPG, Cl <sub>2</sub> , styrene, toluene, caustic soda; 7tc LPG ruptured; 3tc LPG ruptured, exploded; tc Cl <sub>2</sub> punctured, leaking
11/11/79	Molino, FL	Dr	<pre>6 LPG tc, tc ethanol, styrene monomer, acetone; 4 LPG tc on fire; vent and burn all 9tc</pre>
NOTE: Dr tc	= derailment = tank car		

LPG = liquefied petroleum gas NH<sub>3</sub> = ammonia

 $Cl_2 = chlorine$ 

The methodology used in the assessment of the mitigation activities in the accidents involved:

- Cataloging the accidents chronologically.
- Identifying and cataloging information relevant to each accident and each accident mitigation activity on a hazardous material transportation accident evaluation data sheet.
- Performing an assessment of each activity for the purpose of identifying chronological changes/improvements in any area of hazardous materials transportation accident mitigation.

The following sections present the results of this in-depth analysis.

#### 4.1 NOTIFICATION/EMERGENCY RESPONDERS

As a result of an in-depth procedural analysis of hazardous materials transportation accidents it was observed that emergency response personnel were, for the most part, notified of the accident by either the railroad conductor, truck driver, carrier's dispatcher or local citizens. Other individuals and organizations were also involved incidentally in the reporting of hazardous materials accidents to local emergency response organizations. These included fire department personnel, passing motorists, local residents, state police department personnel and other railroad employees, but the occurrence of these individuals/organizations notifying emergency response personnel was less frequent than notification by the carrier personnel involved.

For all the accidents reviewed notification time of emergency response personnel ranged between one and thirty (30) minutes. It was found that notification time did not significantly change over the accident time span reviewed. However, notification time should always be minimized so that emergency response personnel can be called and arrive on-scene within a few minutes after an accident occurs. Improved guidelines for notification of emergency response personnel need to be developed to ensure that properly trained and equipped personnel can be dispatched to an accident involving hazardous materials promptly, with as many accurate details as possible. The on-scene arrival times for fire, police, and other emergency response teams were relatively short once notification was effected. The need for accurate information, especially the identity and hazards of the material(s) involved, was repeatedly highlighted as fire fighters were not aware of the extent or intensity of hazards of LPG fires or teams arrived on-scene without sufficient breathing apparatus.

An empirical analysis of the accidents shows that the initial notification time for railroad hazardous materials transportation accidents is approximately 10 and for

highway accidents roughly seven minutes, a difference of three minutes between the two modes of transportation. This difference is probably because rail traffic is in more remote areas, generally. However, time series analyses of railroad accident notification times show that, for the accidents reviewed, initial notification time has increased since 1975. From 1968 through 1974, notification time averaged approximately one minute. However, for the period 1975 through 1980, notification time following railroad accidents increased to an average of 15 minutes. An examination of notification time for highway accidents showed an average of seven minutes from 1968 through 1980. Once response authorities were notified of the occurrence of an accident, it took an average of 14 minutes for response teams to arrive at the scene. This resulted in a minimum of 29 minutes from occurrence of a rail accident to the arrival of police, fire, and/or other response teams on-scene and 21 minutes for a highway accident. It needs to be emphasized that the NFPA considers 3 minutes to be the maximum time necessary to respond to a fire. Therefore, guidelines are needed to decrease notification and response time to hazardous materials transportation accidents. These guidelines should facilitate more accurate information from the scene, more rapid deployment of personnel, equipment and gear to the scene and increase personal safety of responding individuals.

#### 4.2 EMERGENCY RESPONSE/CONTINGENCY PLAN

The second area assessed was the awareness and utilization of the federal/state/ community contingency plan by a locality at the scene of a hazardous material transportation accident. Any unique provisions of contingency plans for special handling of specific hazardous materials, community/shipper response agreements, and mutual aid agreements between nearby communities and between communities and neighboring military installations were identified.

During the period 1968 through 1973, a total of 36 percent of the communities in which serious hazardous materials transportation accidents occurred had existing local or state contingency plans which were implemented/used on-scene. However, for the period 1974 through 1979, this figure increased to 63 percent.

Since the update and expansion of the National Oil and Hazardous Substances Pollution Contingency Plan in 1978, the number of communities having contingency plans has increased. Powever, this analysis also indicates that all of the existing contingency plans that have been activated during HM accidents were designed for generic emergencies such as nuclear attack, natural disaster, warfare, and terrorism and none dealt specifically with response to hazardous materials transportation accidents. In this

regard, it is recommended that specific annexes be developed for integration into each state/community contingency plan which deals specifically with response to hazardous materials transporation accidents. It is believed that hazardous material contingency plans are needed for timely and safe response to hazardous materials transportation accidents. The description of components and mechanisms for developing and utilizing hazardous material transportation accident contingency plans by local communities are discussed in detail in Section 5 of this report.

## 4.3 EMERGENCY SPECIAL EQUIPMENT AND MATERIALS

For 9 (31 percent) of the 29 accidents reviewed in-depth it was observed that no reference was made to specialized equipment and materials available on-scene at the transportation emergency. In fact, at an additional 9 (31 percent) of the transportation accidents reviewed the only emergency special equipment and materials available onscene were normal operational firefighting apparatus and ambulances. Of the remaining 11 (38 percent) of the hazardous material transportation accidents reviewed emergency special equipment and materials included gas masks; self-contained breathing apparatus; communications equipment; material identification information such as "chem cards;" use of a helicopter for air surveillance of vapor clouds and evacuation; mobile hazardous material emergency response van; analytical monitoring equipment; mobile command post; and explosive devices for "vent and burn" operations.

The utilization of special emergency equipment, materials, and techniques at serious transportation accidents has not become standard procedure until the last few years and still appears to depend on the resources, sophistication and pre-planning done in the community in which the accident occurs.

For information concernings the types of personal protective clothing, gear, equipment and treatment chemicals which should be used on-scene refer to Sections 3.1.6 and 3.1.7 of the Task 4 "Draft Guidelines Manual."

#### 4.4 ON-SCENE COORDINATION AND COMMUNICATIONS

#### 4.4.1 Highway Accidents

The local county or state police are typically the first emergency response personnel to arrive on-scene. This may occur because a police patrol vehicle is in the vicinity of the accident and the officer has seen or heard the accident, passing motorists may pass on the information or he may have been notified of the accident via radio

transmission from the police dispatcher, in which case he will proceed directly to the accident.

The local fire department typically arrives after the first police unit(s) and in general has the responsibility for mitigating the potential dangers associated with the accident.

From the NTSB highway accident reports reviewed, it appears that response personnel have not significantly changed their communications techniques nor are changes indicated in coordinating response efforts at the scene. Response to the highway accidents studied was in general rapid, well-coordinated and effective in addressing the situation.

#### 4.4.2 Railroad Accidents

From the NTSB railroad accident reports reviewed, it appears that on-scene communication and coordination techniques have been changing significantly. Prior to 1977, communication and coordination between response personnel was ill-defined, haphazard and subject to change. Communications at large accidents (e.g., Waverly, TN) involving multiple fire, police and rescue unit response have been disastrous as each community had a distinct radio frequency, thus eliminating effective coordination of firefighting and rescue resources. Additional problems have occurred in terms of "who's in charge" at large rail accidents. Both cases of no one in charge (e.g., Oneonta, NY) and too many "chiefs" (e.g., Youngstown, FL) were equally destructive to safety and effective handling of a dangerous situation. There are exceptions to this, but these exceptions are generally in areas where railroads or the community have available radio and communication equipment which can be made available to response personnel. The utilization of communications equipment at an accident site requires more effective coordination between response crews, thus improving communications and providing more complete assessment of the accident site from severa' stategic locations.

During the years following 1977, a trend begins to emerge with each accident site presided over by an on-scene coordinator designated by law or appointed of necessity. This is in large part due to the requirements of National Oil and Hazardous Substances Pollution Contingency Plan, but also because of the increasing awareness and knowledge of response officials cast in the role of handling these emergencies. In the past, this function has often fallen within the domain of the local fire department. More recently, if the accident warrents activating the National or Regional Response Team, the On-Scene Coordinator (OSC) is an official with the Environmental Protection Agency (EPA)

or the U.S. Coast Guard. In other cases, OSCs are typically a member of different response agencies (i.e., fire department, Civil Defense Agency, state police) and are designated in the local/state emergency plan.

Railroads have complained in the past about the sometimes inconsistent direction of the OSC which may conflict with state and local official dictates. Thus it is necessary that effective interface between the federal, state and local authorities be pre-planned to preclude confusion and inconsistency. The management priorities of the accident also ought to be detailed and understood by all parties involved since each has his own "primary" priority and this can lead to teams working against one another or unwittingly undoing previous accomplishments.

## 4.5 HAZARDOUS MATERIAL IDENTIFICATION AND LOCATION

An analysis of the methods used for identifying and locating hazardous materials involved in railroad and highway transportation accidents was performed. It was observed that methods for identifying a hazardous material ranged from the use of tank placards; smell of material; use of waybill, bill-of-lading, consist or "chem card" documentation; specific service stencilling; to contact with CHEMTREC. In one instance DOT placards were burned in a fire requiring the use of other documentation for identifying the commodity hazard class. For both the highway and rail modes it appears that no major improvements were made during 1968 through 1979 for visually identifying the hazardous materials involved. Recent enactment of the DOT regulation which requires the use of UN identification numbers on DOT placards may prove to be a significant improvement in the identification of hazardous materials involved on-scene, but this improvement will require increased distribution of manuals which provide the code key for identifying the hazardous materials by UN numbers. This information can be found in the DOT Emergency Response Handbook. Other methods for identifying HM are needed, especially those which can be used remotely. At this time, however, any such techniques are still in the developmental stages.

### 4.6 RELEASE HANDLING PROCEDURES

The examination of release handling procedures used at rail and highway accidents yielded the following methods: (1) allowing the material to burn out; (2) venting and burning the tanks with explosive charges; (3) applying fog spray to released vapors to dilute their concentration as well as for the purpose of introducing turbulence into the material to allow more rapid dispersal; (4) applying patches or plugs to holes; and 5 in-





field transfer to other containers. Several of the accidents, however, resulted in rupture release, explosion and fire and no release handling procedures could be applied.

The methods used for handling releases of HMs appear to be dependent on the accident conditions, the nature and condition of the HM released and availability of equipment and materials for approaching and patching/plugging leaks and equipment for in-field transfer of lading.

The philosophy of handling releases of compressed, flammable gases involving fire appears to have shifted from allowing the material to burn out at equilibrium to accelerating burning by further venting the container. By allowing material to burn out under ambient conditions exposures to unprotected individuals would be long-term small concentration exposures. The problem with this approach is that it often took several days for the material to burn completely and this stopped commerce on the same rail line or highway; necessitated citizens being away from their homes for long periods of time; and substantially increased the overall costs associated with the accident.

The options for handling releases of hazardous material can be prioritized based upon financial and time constraints, probable exposure to resident population and safety risk to responding personnel. This ranking shown below has been developed based on the aforementioned factors.

Release Handling Approach	Financial and Time Constraints	Probable Exposures to Population	Safety Risk to Responding Personnel
Patch or plug leak	Low cost,	Low	High
	relatively	short-term	short-term
	quick	exposure	exposure
Off-load material	Low cost,	Low	High
	can be time-	short-term	short-term
	consuming	exposure	exposure
Vent and burn	Costly method,	High	High
	lessens time	short-term	short-term
	considerably	exposure	exposure
Allow material to burn out	Low cost method, extremely time-consuming	Low very long-term exposure	Low long-term exposure

It must be noted, however, that vent and burn operations can be performed only under rigorous controls in remote areas of low population density because the risk associated with this method to unprotected individuals is very high.

### 4.7 FIREFIGHTING PROCEDURES

The analysis of firefighting procedures used on-scene reveals that two activities were always performed at the accident site. First, intact tank containers near or impinged by fire were cooled with deluge water spray. This was done to cool and maintain the temperature of the tank, thus minimizing the risk of a tank rupture or a BLEVE. Second, hazardous material fires were extinguished when possible with either the use of foam or water. In some cases, the car was inaccessible, the fire too intense to allow firefighters to approach or ensuing explosions caused firefighters to draw back and reposition. Other spot fires on-scene or nearby were also handled effectively by the firefighting teams.

Firefighters also use fog water spray to knock down hazardous vapors such as ammonia and chlorine at an accident scene. However, care must be taken when directing water streams at leaking containers or problems like the one in Mississauga may occur. The accident in Mississauga saw the puncture of a car of chlorine along with several LPG fires and other HM releases. To reduce  $Cl_2$  vapors, firefighters applied water to the tank car. Water entered the tank car, reacted with  $Cl_2$  to form a solid chlorine hydrate. The car was subsequently patched, but it took 4 days to vacuum pump the  $Cl_2$  to a tank truck because of the solid residue.

Fire hazards were increased at some of the accidents reviewed because the accident occurred in inaccessible areas where no fire hydrants were available on-scene. At one such accident water had to be pumped from a nearby river while, at another incident, hose lines had to be pulled several hundred feet to a rail yard to fight a fire.

These situations may still exist, but fire fighting technology has progressed markedly in the last few years. Another problem historically associated with firefighting at hazardous materials transportation accidents is the lack of adequate training of response personnel. However, many training programs have been developed by the DOT, DOD and NFPA on this subject and fire fighting personnel appear to be better prepared today for handling fires associated with hazardous materials accidents.

## 4.8 CLEANUP AND DISPOSAL

The analysis of cleanup and disposal activities indicated very little information concerning this phase of accident management. Some difficulties identified during the cleanup and disposal operations for the accidents under investigation include: adding water to a concentrated pool of spilled material causing violent splattering and boiling; inability to arrive at a consensus on cleanup and disposal method by all parties involved due to the lack of written guidelines; unnecessary venting of material in a highly populated area; uncontrolled runoff of contaminated firefighting and tank cooling water into water bodies; and contamination of underground water supplies.

Based upon the availability of information, it was observed that not until 1978 did any major efforts in the cleanup and disposal methods area occur. This was probably due to two factors. First, the National Oil and Hazardous Substances Pollution Contingency Plan was updated and expanded in 1978. This measure established a coordinated mechanism for on-site accident management through the National Response Team and the designated regional response team and on-scene coordinator with a specific mandate to mitigate hazards while minimizing environmental pollution. Second, EPA was and is performing extensive research on cleanup and disposal technology and this information is being published as well as being presented and publicized at national meetings and seminars.

Further emphasis has been placed upon the need for adequate disposal of hazardous residues following the passage of the Resource Conservation and Recovery Act (1976) and promulgation of regulations by EPA in 1980 concerning disposal of hazardous wastes. The regulations place stringent requirements on the where, how and in what hazardous wastes may be disposed. Thus, these regulations along with the requirements the Clean Water Act of 1978 and the Superfund bill of 1979, clearly put a considerable additional burden upon the responder to an accident. That responder must now carefully consider his mitigation activities to minimize his cleanup and disposal needs. An entire new industry has been born to fill this need for cleanup and disposal expertise, and their appearance at HM transportation emergencies is becoming common place.

Based upon this review, it can be seen that cleanup and disposal technology has significantly improved in the last five years and will continue to progress. However, technology mus. be continuously updated because of the continual development, manufacture and transportatio... of new hazardous materials. Recommended cleanup and disposal methods should be commodity specific due to the complex properties and hazards associated with each material, but should also take into account possible synergistic or antagonistic effects in the presence of other materials.

## 4.9 STRUCTURAL INTEGRITY

The examination of activities related to the on-site assessment of tank car and highway tank truck structural integrity indicate that very little, if any, effort has historically been made in terms of identifying damage modes and assessing the structural integrity of tanks damaged in accidents. This review showed that tanks have suffered damage in several ways including coupler impact and puncture of tank head; wheel cuts and burns; impact damage from other vehicles; fire impingement; rail punctures; and disorientation and damage to pressure relief valves. It should be noted that several vehicle retrofits have been recommended and implemented to resolve many of these failure modes including the use of thermal coating protection of tank car shells; top and bottom shelf couplers; head shields; and the AAR guidelines related to the visual assessment of tank car structural integrity. The AAR methodology for visually assessing tank car structural integrity was analyzed using fracture mechanics. Based upon this investigation it was found that the AAR approach is practical for on-site application only if the inspector is willing to approach the tank. By so doing, the inspector is significantly increasing personal risk because there is no way to indicate imminent tank failure. The potential application of acoustic emission (AE) technology to the remote identification of tank car structural integrity is being examined as an alternate method. This approach would provide the safe and continuous monitoring of tank car structural integrity, thus allowing rapid evacuations of personnel and equipment if necessary.

Improved guidelines and additional testing of methods for assessing tank structural integrity are needed.

#### 4.10 WRECKAGE REMOVAL

Wreckage removal is one accident mitigation activity about which very little is known by organisations other than those which perform wreckage removal services. Removal of wreckage from a highway accident is typically handled by a local wrecking service in the area where the accident occurred. However, wreckage removal operations at railroad accidents are much more complex than highway accidents due to the multitude of cars and hasardous materials which may be involved at one time. Several commercial organizations have characteristically responded to rail transportation accidents. These organizations include Hulcher Emergency Services and Isringhausen Crane Manufacturing Company. A few railroads, however, have developed their own response teams and have wreckage removal equipment for use. Based on the review of the NTSB-investigated accidents, it appears that the major improvement in wreckclearing operations since 1968 has been the use of analytical monitoring equipment for the purpose of measuring toxic and flammable concentrations in the accident area prior to entering the scene. It was observed at a 1968 accident that wreckage removal was performed during product transfer operations from a damaged tank car into a semitrailer tanker. This could present an extremely hazardous situation to wreckage removal personnel because, if a leak had occurred, the wapor concentrations could have greatly exceeded the TLV or pressurizing a damaged vessel for transfer could stress the vessel beyond its capacity and cause a rupture.

The one NTSB report which commented on wreckage removal practices was Waverly, TN in which a damaged tank car believed stabilized ruptured and exploded just before transfer operations were to begin. The tank car which later ruptured was moved using cable slings around the north end and using the other end as a pivot. Wooden crossties supported the north end with the remainder on the ground. This appeared to be acceptable wrecking practice, with no indication of any problem or mishandling. The issue then becomes one of the stresses exerted, mechanical damage due to the wheel cut and the overall structural integrity of the car rather than one of wreckage removal or transfer techniques.

It appears that the basic philosophy of wreckage removal personnel is that the scene of a hazardous material accident is not approached until all toxic and flammable vapor hazards are dissipated. This is a major improvement in protecting wreckage removal personnel. However, increased knowledge of wreckage removal activities along with better techniques for damage and structural integrity assessment need to be provided to industry so that improvements can be developed.

### 4.11 KEY FINDINGS OF ACCIDENT ASSESSMENT

systemus i mandalara internative . mu

Based upon STL's accident assessment it can be seen that several conclusions can be made concerning SOA accident management techniques from the accidents reviewed. These include:

- 1. Notification time for contacting emergency response personnel did not significantly change between 1968 and 1979.
- 2. The on-scene arrival times for fire, police and other emergency response teams were relatively short once notification was effected.
- 3. All of the existing contingency plans that have been activated during HM accidents were designed for generic emergencies and none dealt specifically with response to HM transportation accidents.

- The utilization of special emergency equipment, materials, and techniques at serious transportation accidents has not become standard procedure until the last few years and still appears to depend on the resources, sophistication and pre-planning done in the community in which the accident occurs.
- 5. Response personnel at highway accidents have not significantly changed their communications techniques nor are changes indicated in coordinating response efforts at the scene.
- 6. On-scene communication and coordination techniques at railroad accidents have changed significantly.
- 7. During 1968 through 1979 no major improvements were made for visually identifying the HM involved on-scene.
- 8. The method used for handling releases of HMs appear to be dependent on the accident conditions, the nature and condition of the HM released and availability of equipment and materials for approaching and patching/plugging leaks and equipment for in-field transfer of lading.
- 9. It was not until 1978 until any major efforts in cleanup and disposal methods occurred.
- 10. Very little, if any, effort has historically been made in terms of identifying damage modes and assessing the structural integrity of tanks damaged in accidents.

Based upon this accident analysis, it can be seen that trends, improvements and changes have been made in accident management activities since 1968. Unfortunately, not all geographic areas and emergency response organizations have experienced a uniform growth in their accident management capabilities. Thus, a SOA of local HM response capabilities was performed to identify in greater detail the specific areas and concerns of municipal emergency response organizations for handling HM transportation accidents as a mechanism for providing improved crisis management techniques. The results of this investigation are presented in Section 5.

## 5. ASSESSMENT OF METHODS FOR IMPLEMENTING CRISIS MANAGEMENT TECHNIQUES USED BY POLICE, FIRE CHIEFS, MAYORS AND OTHER DISASTER RESPONSE OFFICERS

This section presents an assessment of various methods to implement crisis management techniques for hazardous material transportation accidents. The assessment involved the following steps:

- Methodology for the collection of information from each of the ten cities visited;
- Identification of each community's emergency response capabilities for handling hazardous materials transportation accidents; and
- Assessment of each community's emergency training programs and resources. As a result of this assessment, methods for implementing crisis management techniques for handling hazardous materials transportation accidents are being recommended. These methods will provide cities and communities with a mechanism to update their hazardous materials transportation accident contingency plans based upon the latest available technology.

The ten cities selected for analysis were:

- Baltimore, MD
- Los Angeles, CA
- Nashville, TN
- Newark, NJ
- Pensacola, FL
- Sacramento, CA
- Tallahassee, FL
- Waverly, TN
- Wilmington, DE
- Youngstown (Panama City), FL

This selection was made because these localities either: (1) have significant amounts of hazardous materials traffic through them; (2) have been sites of past serious accidents from which past performance and planning improvements could be evaluated; or (3) have already done specific work in contingency planning or crisis management of hazardous materials transportation accidents.

## 5.1 METHODOLOGY FOR OBTAINING INFORMATION

The approach used to identify each city's emergency response capabilities initially involved the identification of key city, county and state decision-makers such as civil defense and emergency preparedness directors, police chiefs, fire chiefs, mayors, and health officials. Contacts were initiated by telephone and letter and were followed up with personal visits, interviews and additional correspondence. Written and verbal procedures, methods and guidelines existing in each city, county or state for handling hazardous materials transportation emergencies were obtained.

7

Electron and

At the beginning of each meeting a brief presentation, entitled "Assessment of Techniques, Plans, Training Aids and Resources for Hazardous Material Accidents," was given to local decision-makers in each city. This briefing addressed STL's approach for performing the assessment; an overview of the "Post-Accident Procedures for Chemicals and Propellants" project; and identified the types of information desired from each city. It was emphasized at each meeting that the project will result in a summary of the best techniques, plans, training aids and resources for management of hazardous material transportation accidents, which will enable local decision-makers to assess their programs and make improvements, when possible.

The type of information requested fell into the following categories: contingency planning; personnel training; specialized personnel, equipment and gear; and problems encounted or foreseen in handling hazardous materials transportation accidents.

Each community was asked if they had developed a specific local contingency plan for handling hazardous materials transportation accidents or if the basic fire plan, state contingency plan or the National Oil and Hazardous Materials Substances Pollution Contingency Plan (1510 Plan) would be used in the event of a hazardous materials transportation accident. Also, organizational responsibilities of all local emergency response service organizations who would be on-scene and at what level they would interact with Federal, State and industrial response groups were identified.

The nature and extent of response personnel training programs especially in handling hazardous materials were also examined. A listing of hazardous material guidebooks and source reference material carried in emergency response vehicles in each city was also compiled.

An inventory of specialized response personnel, equipment and materials was taken for each community. This indicated the level of sophistication in handling an emergency. This inventory, however, could not be used as a basis of assessing "total response capabilities" because it was found that cities having a high population and purchasing power also had an equally high level of sophistication with respect to specialized personnel, equipment and gear. Small population centers in proximity to large cities often relied on the response capabilities of larger nearby cities. For
example, Youngstown, FL relies on Panama City, FL and Waverly, TN relies on Nashville, TN.

Each city was also asked to identify difficulties which had been encountered or could be encountered at a future hazardous materials transportation accident. This information was necessary so that recommendations for improved crisis management techniques could address these issues. These recommendations could be used as a mechanism by local decision-makers for updating their existing techniques, plans, training aids and resources.

## 5.2 EMERGENCY RESPONSE CAPABILITIES IN TEN SELECTED CITIES

Figure 5-1 shows the geographic location of the cities selected for examination. Information on each city's crisis management techniques, plans and training aids for hazardous material transportation accidents is provided in the following sections.

#### 5.2.1 Baltimore, MD

- There is a basic Baltimore City Emergency Operations Plan which deals with all emergencies. This plan has specific guidelines for handling hazardous materials transportation emergencies.
- The Office of Civil Defense will not become involved in a hazardous materials accident unless a "state of emergency" is declared by the Governor.
- The Office of Civil Defense is in-charge of contingency planning and coordination of agencies on-scene as well as procuring required resources for use on-scene such as personnel, equipment and materials (e.g., arrange for living accomodations, obtain cranes and trailers, etc.).
- The Baltimore Fire Department has an advisory committee (i.e., Hazardous Materials Task Force) which has specialized personnel, equipment and gear for responding and coordinating activities on-scene. They also have a CHLOREP patching kit for chlorine emergencies.
- The fire chief serves as the on-scene commander.
- All emergency response vehicles carry the <u>DOT Emergency Response Guidebook</u>. However, if a water pollution threat occurs the City of Baltimore has access to the USCG CHRIS system.
- The fire department is familiar with and has used CHEMTREC (Chemical Transportation Emergency Center) for technical assistance at hazardous materials transportation accidents.
- Concerning hazardous materials training, all fire department personnel take the NFPA hazardous material course entitled "Handling Hazardous Materials Transportation Emergencies."
- The USCG has identified the hazardous materials transported in the Baltimore area and is equipped with a pollution response van with specialized equipment and gear.



- The USCG believes that it is not geared up for immediate response to most hazardous material accidents. They will typically respond with specialized personnel and the CHRIS response system. Their responsibilities on-scene are to support primary hazardous materials control groups.
- The Baltimore police department provides traffic control, escort, crowd control and bomb removal at a hazardous materials transportation accident.

adverseda sundament manchana seasad a adversedar

Charles and the second

- All police officers are required to complete a minimum of eight hours of hazardous materials training. The department utilizes the same NFPA training course as does the fire department.
- Four area hospitals are on 24 hour/day stand-by alert in readiness for an impending accident.
- The Coast Guard has an agreement with the Baltimore City Fire Department that it will respond on an advisory basis at the fire department's request even if the incident is not within its jurisdiction.
- If an evacuation becomes necessary, the Baltimore Office of Disaster Control would activate an order to execute evacuation procedures.
- The following are operating procedures to be used on-scene by the hazardous materials task force (HMTF). These procedures are prescribed in Manual of Procedure (MOP) 625-11.
  - Until proper identification of the product or material has been made, it should be considered toxic and explosive.
  - Members should <u>ANTICIPATE</u> and not delay in calling for assistance as a limited situation can quickly become a major problem if not handled expeditiously. If evacuation is deemed necessary, it should be started immediately, moving those closest to the problem first and working away from the incident.
  - All protective clothing, including breathing apparatus, will be worn in handling these incidents. If initial dispatch indicated a hazardous chemical, air masks will be donned before entering the contaminated area.
  - Upon arrival, the Deputy Chief will assume charge of the incident, assisted by the task force Battalion Chief. The officer in charge will designate the staging area and designate an officer in charge.
  - Points considered in selecting a staging area include wind direction and velocity, topography, and accessibility. First-aid equipment, stand-by manpower, and logistical support will be marshalled here.
  - Subsequent arriving units will, in the absence of specific instructions, report to the staging area.
  - Chief officers will be cognizant of available monitoring equipment (radiological, explosimeter, etc.) and utilize them to best advantage, as well as the available supportive resources; i.e., CHEMTREC, Hazardous Materials Guides, local technical assistance.
  - Chief officers will closely monitor casualties in case the Natural Disaster Plan should be implemented.
  - Use of hose streams for flushing, cooling, or absorption should be considered and stretched where indicated.

#### FIRES INVOLVING FLAMMABLE GASES SHOULD NOT BE EXTINGUISHED UNLESS THE FLOW OF GAS CAN BE STOPPED

Communications will be maintained at all times between operating task force and Fire Communications Bureau.

As wind direction and velocity are extremely important in relation to chemical spills or leaks, the following format will be used by Fire Communications Bureau when giving wind conditions:

"wind from north to south at 12 MPH," or "wind from east to west at 6 MPH," etc.

The above information will be obtained from Weather Bureau, Baltimore Washington International Airport:

> PHONE: 962-2177 (24-hour number) 787-7257

This information will be transmitted:

- 1. At time of dispatch.
- 2. When command post is established.
- 3. When requested by fireground commander.

The Captain Fire Communications will insure that when obtaining weather information, the actual readings at time of request are recorded. The fireground commander will be apprised of any changes that might affect fireground operations.

If type of material is known, all pertinent information; such as fire, explosion, and/or health hazards known about the material or incident will be transmitted:

- 1. At time of dispatch.
- 2. Via radio, after the response of the Battalion Chief has been verified.
- 3. When requested by the fireground commander.

• Specific procedures are also identified for chlorine leaks (MOP 628-1), spill incident-sorbent booms (MOP 600-4), disposable surgical masks (MOP 600-7), prevention of contamination of domestic water supplies (MOP 603-4), harbor protection-fireboats (MOP 605), accidents-derailments railroad right-of-way (MOP 633-1) and oil, chemical and noxious material spill incidents (MOP 644, 644-1, 644-2, 644-2-1, 644-3, 644-4, 644-5, 644-5-1 and 644-5-3).

• In case of an accident-derailment along the railroad right-of-way the following procedures are applicable:

- The Officer in Charge of Unit or Units that respond to an incident along a railroad right-of-way will immediately advise Fire Communications of:
  - o The correct name of the railroad involved.
  - o Location of incident.
  - o Nature and extent of the incident.
- Should the incident be of a serious nature and the emergency operations must le conducted on or across the railroad tracks, the Commanding Officer will notify Fire Communications to have train movement stopped.

5-6

Fire Communications will notify the railroad involved by calling one of the following numbers:

- Baltimore and Ohio 237-3433, Chief Train Dispatcher
- Canton 342-4458, Train Master
- Western Maryland 237-3938, Yardmaster
- Penn Central 685-4827, Power Director
- When the incident involves the Penn Central Railroad, emergency operations must be performed in accordance with (MOP 632-1) "Penn Central Electrified Territory." Until it has been absolutely ascertained that train movement has been stopped, men will be stationed at opposite ends of the incident to alert fire personnel of on-coming trains (using portable radios, if necessary). DANGER: DUE TO THE HIGH SPEED OF SOME TRAINS (UP TO 140 MPH), IT MAY BE NECESSARY TO REQUEST THAT ADDITIONAL UNITS BE DISPATCHED A MINIMUM OF THREE MILES TO EITHER SIDE OF THE INCIDENT IN ORDER TO RADIO AN ADEQUATE WARNING OF ON-COMING TRAINS TO THE UNITS WORKING AT THE INCIDENT.
- The Fire Department is not equipped with apparatus or tools to lift heavy railroad equipment; i.e., freight cars, locomotives, heavy machinery, etc. Therefore, the Officer in Charge shall advise Fire Communications to request proper heavy equipment from railroad involved.
- When the accident or derailment involves freight or tank cars transporting explosives, hazardous chemicals, flammable liquids, or gases and there is spillage of chemicals, oils, or noxious materials, (MOP 644-1) "Spill Procedure" must be initiated.

Hose lines must be placed under rails between ties to prevent hose from being run over by passing trains while performing firefighting operations.

The NFPA hazardous material training course given to emergency response personnel provides information on hazardous materials in-transit; definitions, classes and dangerous properties of hazardous materials incidents; command and control of hazardous materials incidents; and planning for hazardous materials emergencies.

## 5.2.2 Los Angeles, CA

- The Los Angeles Fire Department has experienced problems with hazardous materials identification and container integrity assessment at accidents.
- The California Highway Patrol (CHP) serves as statewide information, assistance and notification coordinator for hazardous spills occurring on all highways throughout the state.
- Los Angeles has only a basic fire plan, with no specific contingency plan devoted to hazardous materials response.
- All fire chief vehicles carry the DOT Emergency Response Guidebook.
- Los Angeles is equipped with one Mobile response van which has analytical testing equipment (GC-MS), technical reference library (EPA, AAR, USCG CHRIS, DOT, etc.), personal protective clothing, breathing apparatus, communications support.

 Response van is operational 24-hours a day. Four men support the van and each assumes command of the van on a weekly rotating basis. These four personnel make up a react team in the area.

Ĺ

- Local cleanup and disposal companies provide basic training to response personnel in the area.
- Los Angeles personnel have taken the NFPA hazardous materials training program and used portions of it as the basis for developing their own.
- Los Angeles personnel have a close working relationship with the chemical manufacturers in the area so they will provide technical assistance on-scene if requested by the incident commander.
- The CHP felt that personal protective clothing, gear and equipment were inadequate for handling an emergency situation but that adequate resources cannot be purchased due to funding restrictions.
- CHP has predesignated highway routes in California where highway vehicles transporting explosive materials may travel as well as stop for fuel and rest. These routes and stops are identified in CHP publication HPH 84.8 entitled "Explosives Routes and Stopping Places."
- CHP has also developed a manual entitled "Hazardous Material Transportation" (HPM 84.2) which was designed to provide guidance and assign responsibilities for enforcement of legal requirements relating to the transportation of hazardous materials and to prevent injury or loss of life resulting from accidents involving hazardous materials. The contents of this manual provide information on policy, laws and regulations, enforcement, commodity identification, explosives, loading, cargo and portable tanks, selection of explosives routes and emergencies.
- From this manual (HPM 84.2) the following information was ascertained in terms of response activities at hazardous materials transportation emergencies:
  - The California Department of Transportation (CALTRANS) is responsible for identifying, containing, removing, or causing to be removed all materials spilled on State highways. This includes reloading or other disposal of hazardous materials cargos in accidentdisabled vehicles and long-term traffic control.
  - CHP will assume the responsibility for immediate notification to local fire departments in event of cargo or vehicle fires.
  - CHP has a hazardous materials emergency operations plan in which response requirements are subdivided into three phases. The three response phases and the activities of organizations under each are:
    - A. <u>Phase I.</u> The officer arriving at the scene takes immediate action to best control the incident, then promptly identifies the commodity and radios its shipping name and hazard class to the dispatch center. Using available reference data, dispatch will advise the field unit of the hazards and precautions to be taken.
    - B. <u>Phase II.</u> The dispatch center promptly notifies appropriate agencies to respond to the scene. Initial and subsequent reports from the officer at the scene, and other inputs from supervisory and industry personnel, determine activity at this point. When necessary, a Departmental supervisor should respond to the scene and establish a command post. For

guidance of dispatch personnel, responding organizations have the following capabilities:

- (1) <u>CALTRANS</u>. CALTRANS personnel can assist in identification, containment, removal and reloading and should routinely be notified as soon as possible of any emergency on a State highway.
- (2) Local Fire Departments/Division of Forestry. In urban areas, local fire departments should be notified of spills of flammable materials or cargo fires involving hazardous materials. In other areas, California Division of Forestry units may be requested to provide fire fighting or standby service. Support from these organizations will not normally include reloading, removal, or decontamination. In many cities fire services may provide the following assistance:
  - (a) Fire protection services.
  - (b) **Protective** clothing and equipment.
  - (c) Ability to enter contaminated area for rescue and/or commodity identification.
  - (d) First aid to the injured.
  - (e) Coordination with the poison control center.
- (3) <u>CHEMTREC</u>. The Chemical Transportation Emergency Center (CHEMTREC) is a voluntary project of industry, sponsored and supported by the Chemical Manufacturers Association, and is located in Washington, D.C. CHEMTREC can provide immediate advice as to precautions to be taken at any emergency scene, can arrange manufacturer or shipper assistance, and operates around the clock: (800) 424-9300 (toll free).
  - (a) CHEMTREC does not directly dispatch assistance, but acts as an information source and coordinating center.
  - (b) Information obtained from CHEMTREC depends upon accurate details provided to CHEMTREC during the initial call for assistance. These details should include type of containers, quantity, name of transporter and shipper or manufacturer, and <u>accurate</u> description (chemical, trade, or shipping name and hazard class, if any). If shipper or manufacturer assistance is required, this should be clearly stated.
  - (c) Assistance in dealing with chlorine incidents under the Chlorine Emergency Plan (CHLOREP) can be obtained through CHEMTREC.
  - (d) Assistance from the National Agricultural Chemicals Association's Pesticide Safety Team Network can be obtained through CHEMTREC. These teams - some of which are located in California - will assume responsibility for the cleanup of agricultural chemicals.

(4) <u>Shippers.</u> Shippers may be able to assist in emergencies, especially if transporting their own products as private carriers. With for-hire carriers, it may be possible to determine shipper from shipping documents.

A STATE I

- (5) Federal Department of Energy (DOE). Assistance in emergencies involving radioactive materials being transported under its auspices may be obtained from the Department of Energy (formerly Energy Research and Development Administration): (415) 273-4237.
- (6) Office of Emergency Services. If transportation of radioactive materials is not under DOE jurisdiction (above), the Office of Emergency Services (OES) should be contacted in emergencies: (916) 421-4990. In addition, OES should be routinely contacted regarding spills of any chemical into or along any surface waterway: (800) 852-7550. OES then notifies the information to concerned Federal and State agencies but does not dispatch assistance.
- (7) The Military. Reloading and other assistance in incidents involving military explosives may be obtained from the Army. The 548th Ordnance Detachment, Presidio of San Francisco, (415) 561-4203 or 561-4312, shall be contacted in all such cases. Dispatch of disposal teams from the nearest base will be coordinated from this central point.
- (8) <u>Division Motor Carrier Safety Units.</u> Motor Carrier Safety Personnel may be able to help identify products and their hasards, especially when products are unknown or marginally dangerous, or when it is necessary to locate the shipper or carrier.
- C. <u>Phase III.</u> The final phase is reloading and cleanup, including decontamination if necessary. When a highway has been restored to safe use, Departmental participation may be terminated with the concurrence of CALTRANS or other public agency personnel concerned with cleanup and disposal.
- CHP provides guidelines for tank structural assessment while impinged with fire.
   These include:
  - <u>Tank No Fire</u>. A tank in any position, even completely overturned, is relatively safe if there is no fire.
  - Upright Tank Moderate Fire. An upright tank with only moderate fire exposure, even if safety valve discharge catches fire, is not extremely hazardous.
  - <u>Upright Tank Intense Fire</u>. A tank exposed to an intense fire, and which contains <u>certain unstable materials</u>, may be hasardous. The Chem card manual contains specific information on these materials. Tank rupture is possible and evacuation to a distance of 1,500 feet is advisable.

- Overturned Tank Intense Fire. A partially or completely overturned tank subject to intense fire is extremely dangerous for two reasons. First, heating of tank heads and shells causes rapid and extreme weakening of the metal (tensile strength of aluminum and steel alloys at several hundred degrees is but a fraction of what it is at normal temperatures). This weakening is pronounced on portions of a tank above the liquid level. Second, relief valves at the top of the tank normally connect with the vapor space above the liquid level. When the tank is overturned, the relief valves no longer connect with the vapor space and the valves cannot function to keep tank pressure from rising to dangerous levels. Under these conditions, tank rupture and the strewing of contents and tank fragments for up to several hundred feet is probable; evacuation to a distance of 1,500 feet is imperative.
- Training is given to all emergency response personnel.
- Emergency response personnel are trained in the areas of personal and personnel protection; incident appraisal; hazardous substance identification methods; rescue techniques; scene control; coordination post establishment and operation; coordination; requesting available resources; containment; cleanup methodology and equipment decontamination; disposal; news media relations; knowledge of liability; and accident reporting.

## 5.2.3 Nashville, TN

- No specific contingency plan has been developed for hazardous materials response. However, a general firefighting plan does exist which designates fire and emergency medical responsibilities on-scene.
- Two problem areas in which Nashville feels that improvements are needed include communications on-scene between organizations and hazardous materials identification. The difficulty in on-scene communications occurs because each of the organizations responding communicate on different radio frequencies which causes confusion when trying to coordinate on-scene activities. This occurred following the derailment, explosion and fire in Waverly, TN. Concerning hazardous materials identification problems have occurred when a material is not marked properly or when limited material hazard information is known and the manufacturer can not be contacted. This significantly increases the time of "uncertainty" for emergency response personnel.
- Nashville has a six man specialized react team with equipment and gear for response on-scene. This team has special training in mitigating hazards associated with releases of hazardous materials. Also, the team utilizes two mobile response vans. These vans are equipped with a CHLOREP kit, burn-off equipment, breathing apparatus, cutting torch, personal protective clothing, disposal drums and lime for neutralization. The Nashville area has established an agreement with the State which provides that the State of Tennessee will provide the emergency response equipment if Nashville will be in charge of responding to hazardous material incidents in central Tennessee. Nashville estimates that they can normally respond to an incident in less than 12 minutes (county is 542 mi.<sup>2</sup>). The Nashville metropolitan area has established a computer-operated emergency response network in which the telephone numbers of CHEMTREC, industrial manufacturers and other emergency response organizations can be activated by pushing only one button for each organization. All emergency vehicles carry the DOT Emergency

<u>Response Guidebook</u>, but the react team has a library of other emergency response source material including a chemical dictionary, USCG CHRIS, EPA and AAR manuals and others. Nashville feels that the use of the UN numbering system on DOT placards will be beneficial. The metropolitan area has established a Hazardous Material Risk Advisory Committee (HMRAC) which identifies policy for handling hasardous materials. HMRAC is comprised of personnel from emergency response, academic and industrial concerns.

- Concerning training of emergency response personnel, all react team members have or are working for an Associates Degree in Fire Science. These members take the NFPA course and an additional 80 hours of hazardous materials training by a chemistry professor. The react team members must then be recertified every two years. All other fire service personnel take the NFPA hazardous materials training course.
- The Metropolitan Nashville-Davidson County Division of Civil Defense and Emergency Preparedness has been designated as the Emergency Coordinating Agency having primary responsibility and authority for planning of disaster preparedness, response and recovery; for coordination and liaison with related agencies of Metropolitan, State and Federal Government and such agencies of other cities, counties and concerned private agencies.

5.2.4 Newark, NJ

- Newark has a general contingency plan for emergencies which does not specifically address hazardous materials emergencies.
- The City of Newark is in the process of developing guidelines for hazardous materials accidents. This is mainly a result of a few significant accidents in the area, one of which involved the release of ethylene oxide in a railroad yard forcing the closing of the Newark International Airport.
- A mutual aid agreement exists between the fire department and local chemical manufacturing companies so that the manufacturers will provide response teams in the event of an emergency. The chemical manufacturers train, provide technical assistance and information to emergency response personnel.
- The City of Newark has their own hazardous materials training program. All response personnel are trained.
- All city emergency response vehicles carry the DOT <u>Emergency Response</u> <u>Guidebook</u>.
- The fire service utilizes positive pressure breathing apparatus rather than Scott air packs. Firefighters are required at all times to wear breathing apparatus even though a fire may not be visible.
- The state of New Jersey has one hazardous materials response van for use onscene. The van, operated by the New Jersey environmental organization, is based in Trenton, approximately one hour from Newark.
- The Newark Office of Environmental Affairs, Department of Engineering, acts as liaison with New Jersey concerning hazardous materials situations.
- A difficulty which Newark felt was a hindrance to hazards mitigation is that technical assistance is usually not available after normal business hours on weekdays and no assistance is available on the weekends.

## 5.2.5 Pensacola, FL (Escambia County)

- Pensacola has an unusually high probability of a hazardous materials transportation accident occurring because an extremely large volume of petroleum products traverse its boundaries and a major interstate highway bisects Pensacola.
- It was estimated that an average 150 car train consist going through Pensacola would include at least 80 cars of HMs.
- Mock-up exercises are held regularly by city and county emergency response personnel.
- Escambia County has an operating 911 emergency communcations network.
- In the city of Pensacola the fire department consists of one station of paid firefighters while the county fire department consists of 15 stations of volunteer firefighters. These two, however, are linked together by a firefighting association.
- In terms of response teams, Civil Defense estimates that the average response time of any firefighting unit in the county is 5 minutes.
- The procedure for notifying a station company of an emergency follows:
  - 1. Dispatcher receives call.
  - 2. Dispatcher notifies service company within one minute after receiving initial notification call.
  - 3. Station notified must then, in then, call back dispatcher within two minutes after notification.
  - 4. If the station company does not return the dispatcher's call within two minutes the dispatcher again notifies the service compnay. This problem occurs only 5-10% of the time.
- Mutual aid agreement exists between the county and city emergency response units.
- When technical expertise is needed on-scene, the county has established a mutual aid agreement with chemical manufacturers in the region. These manufacturers include St. Regis, Monsanto and Air Products. Each of these manufacturers have industrial response teams equipped with acid suits and other personal protective clothing.
- Civil Defense feels that the following organizational structure for on-scene coordination is a logical delegation of responsibilities:
  - The county has been subdivided into 14 response regions, each having a fire chief. The fire chief in each region has been designated the OSC for that region.
- All emergency response vehicles carry the AAR's <u>Emergency Handling of Hazardous</u> Materials in Surface Transportation.
- Pensacola does not recommend the use of the DOT Emergency Response Guidebook (ERG) because they question the validity of a guide book which has a designated guide (Guide No. 11) which addresses no hazardous commodity.
- The county has a mobile van which supplies breathing tanks on-scene and can replenish cylinders in which the air has been depleted. This is known as the "cascade system."
- The county keeps an inventory of individuals and 24-hour access telephone numbers which can be used for search and rescue operations as well as members of industrial response teams.

- Escambia county operates their own hazardous materials training program. This consists of both the NFPA courses in HM and RAM. More extensive training, given at State conferences and seminars, is provided for all fire chiefs, representatives of the firefighting association, civil defense training officer and EMT representative, who is the RAM training individual. All individuals are required to take at a minimum the NFPA courses.
- The county is equipped with a mobile response van, mobile ambulance (which can transport 10 individuals at once carries 100 stretchers), and has a mobile command post.
- The mobile command post (which was used at Molino, FL) is a 55-foot modified bus which is equipped with a meeting room, teletype and 8 monitoring stations. Meteorological data is obtained from NOAA. The post uses microwave communications. Estimated cost of mobile command post is \$30,000.
- County communications consist of a \$1.4 million microwave communications network. Currently, the system provides 100 communications channels, but has the capability to handle 300 channels. It has been found that this system decreases cost while increasing the efficiency of Escambia County's communications network. Prior to installation of this system, a telephone call from one end of the county to the other was long distance (due to three telephone companies servicing the area). Now an individual can call the entire length of the county on a local basis.
- All calls for emergency response personnel are filtered through the county's Office of Civil Defense and Disaster Preparedness. This provides for coordinated response activities.

5.2.6 Sacramento, CA

- Each emergency response vehicle carrys the DOT <u>Emergency Response Guidebook</u>.
- Battalion chief vehicle also carries the AAR <u>Emergency Handling of Hazardous</u> <u>Materials in Surface Transportation</u>.
- Both of the aforementioned publications are also in the dispatchers office.
- Sacramento has no primary hazardous materials analytical testing equipment. They do have an agreement with a chemist in the city engineering dept., several colleges which have laboratory facilities and the I.T. Corp., for testing and identifying materials. The response time that I.T. Corp. requires to arrive at Sacramento is roughly two hours.
- Sacramento has not developed specific guidelines for handling hazardous materials transportation accidents, but relies on the response guidelines identified in the DOT and AAR response manuals for on-scene actions. However, Sacramento has developed guidelines for approaching, locating and securing fire apparatus near the scene of an accident.
- Sacramento has developed a 106 hour hazardous materials training course. This course was developed for members of the two react teams. The 106 hour training program was developed from prepackaged programs such as those developed from NFPA, California Fire Marshall's office, etc. Each captain is also given this training. All other fire personnel are given limited training which consists of familiarizing personnel with the reponse manuals and departmental developed guidelines regarding approaching, locating and securing fire apparatus on-scene.

- The area has two air compressor units which provide a cascade system (capable of replenishing air packs on-scene).
- Two mobile response vans exist. Each van is equipped with 10,000 kw generator, carbon arc search lights, passive cascade system (i.e., cannot produce air but can store enough to replenish 60 tanks), positive pressure breathing apparatus, hydraulic tools, personal protective clothing, MSA breathing apparatus, umbilical cords that air can be forced through (positive pressure - 300 foot maximum length).

## 5.2.7 Tallahassee, FL (Leon County)

- Leon County has a population in excess of 151,000 while the city of Tallahassee comprises 81,000 or 54% of the total county population.
- Tallahassee is bounded on the north by Interstate Highway 10 while LN/SCL have railroad tracks going through the center of town.
- Not one township in Leon County has either volunteers or paid emergency response personnel, exclusive of Tallahassee.
- All emergency response vehicles carry both the 1976 and 1980 version of DOT's Emergency Response Guidebook.
- Tallahassee officials have experienced difficulty in using the 1980 "Guidebook" at night because it was hard to reference the UN number listings in the beginning of the book in the dark or under stressful circumstances.
- There are no chemical manufacturing plants in the immediate areas.
- All fire personnel are given the NFPA Hazardous Materials training course. Fire personnel are required to attend additional hazardous materials training annually.
- Limited hazardous materials training is given to either police or sheriff department personnel, mainly because their on-scene role consists of evacuation and perimeter control.
- Police and sheriff department personnel, however, do get their limited training from full-time training officers of the fire department.
- The fire department has 5 full-time hazardous materials training officers.
- Cooperative agreement has been established between Leon County and LN/SCL about improving track conditions.
- In conjunction with neighboring counties Tallahassee regularly performs mock-up exercises of hazardous materials transportation accidents.
- If foam trucks are required on-scene they are available from the municipal airport.
- Fire department is equipped with a mobile response van and has analytical testing equipment, Scott air packs and acid suits.
- Leon County has an Emergency Operations Plan which considers hazardous material transportation accidents occurring in region.
- Director of Civil Defense was unfamiliar with either the EPA or CHRIS response manuals.
- The 911 emergency telephone system is used in the county.
- The communications command center is located at the Civil Defense's Directors office and is manned 24-hours/day.

- Leon County Emergency Operations Plan identifies designated responsibilities for fire, police, emergency medical and health/rehabilitative services personnel.
- The Tallahassee fire department is equipped with Scott air packs, acid suits and other personal protective clothing, gear and equipment.
- Standard operating procedures for notifying a railroad of an accident involving hazardous materials in the county consists of identifying information on how many cars are involved; nature of contents; possibliity of fire; toxic materials; response agencies on-scene and immediate response needs.

#### 5.2.8 Waverly, TN

- Waverly has no specific guidelines for handling hazardous materials accidents, but relies upon guidelines for responding to fires.
- Civil Defense has the capability to utilize the HMER system in the event of an emergency.
- All emergency response vehicles carry the DOT Emergency Response Guidebook.
- Mock-up exercises have been utilized to train firefighters for possible accidents.
- Waverly employs a training officer who instructs personnel on hazardous materials handling. This training consists of utilizing the NFPA hazardous materials training program, State developed programs, etc.
- Three chemical companies are within the Waverly area, and an agreement has been reached with these manufacturers that they will provide technical assistance and response teams if requested.
- Waverly has meteorological monitoring equipment capabilities and could place the equipment on-scene to identify and monitor climatic conditons and estimate vapor dispersion patterns.
- Waverly has limited specialized personnel, gear and equipment. Sophisticated resources are obtainable from Nashville (response team roughly 1.5 hours), and were made available through a State agreement. The County, however, does have positive pressure breathing appparatus and Scott air packs.
- Waverly had budgeted to purchase personal protective suits, but due to a budget cut of 50% in the fire department's operating revenue these services had to be sacrificed.
- Firefighting personnel are paid volunteers (i.e., paid for time of service).
- The police department has no guidelines; training; or specialized personnel, gear and equipment for handling a hazardous materials release.

#### 5.2.9 Wilmington, DE

- All emergency response vehicles carry the DOT Emergency Response Guidebook.
- Wilmington employs the only paid fire department in the state.
- The State Fire Training Academy handles training of all firefighters. This training includes hazardous materials courses in its curriculum.
- There is a general contingency plan in the city, but this plan does not specifically address response to hazardous materials transportation accidents.

- Neither the fire nor police departments have specific procedures for handling hazardous materials releases except for radioactive materials.
- Two hazardous materials training courses are taught by the Delaware Fire Training Academy. One is geared to all emergency response personnel and instructs personnel in such areas as material characteristics and identification, labeling and placarding. The second course is geared towards individuals which would serve as OSCs at a hazardous materials incident. This course teaches actual tactics and strategies.
- There is no mandatory follow-up training. Firefighters, however, may repeat the course to refresh their memory as they deem necessary.

#### 5.2.10 Youngstown, FL (Panana City-Bay County)

- The Bay County Natural Disaster Operations Plan contains a halardous materials spill emergency response plan (Annex G).
- The Director of Civil Defense is the on-scene coordinator at hazardous material transportation accidents in the county.
- The county maintains a listing of individuals capable of providing technical assistance if an emergency occurrs.
- The plans identifies designated responsibilities for organizations that would be involved in on-scene assistance.
- The plan permits the utilization of heavy equipment resources within the public works agencies for salvage, repair and debris removal operations.
- Guidelines for immediate on-scene actions to be taken by emergency personnel at the scene include the following instructions:
  - 1. Take any feasible steps necessary to protect or save human life and safeguard property.
  - 2. Restrict traffic in and about the scene.
  - 3. Take all necessary actions to contain and/or prevent the spread of the material.
  - 4. If the incident involves fire or material subject to blowing in the wind, conduct operations from an upwind position.
  - 5. Isolate and hold all contaminated persons for further examination by specialists.
  - 6. If there are casualties requiring medical attention, take only necessary life-saving actions prior to the arrival of a qualified hazardous materials specialist and/or physician.
- Attachment No. 1 of Annex G to the Bay County Natural Disaster Operations Plan consists of a Hazardous Substance Spill Report which catalogs on-site accident information including the name and contact number of the reporter, location of incident, type and cause of incident, casualities, personnel and equipment availability. An example of the Hazardous Substance Spill Report is given in Table 5-1.
- Attachment No. 2 of Annex G to the Bay County Natural Disasters Operations Plan contains details of emergency procedures for local authorities when handling hazardous materials. These include:

## TABLE 5-1

## HAZARDOUS SUBSTANCE SPILL REPORT

Å.	****	· (Aganov)	
	can be contacted for further	information at _	
		_	(Phor
	(Location)	·*	
LO	CATION OF INCIDENT		
A.	Structure		
	Building or Company Name		
	Address		
	City	County	
в.	Roadway		
	Highway or Street Name		
	Nearest Intersection		
с.	Off-Shore		
	Nearest identifying landmarks road, street, etc.	(Beach name, pi	er, nea
. <u>TY</u>	Nearest identifying landmarks road, street, etc PE OF INCIDENT	(Beach name, pi	er, nea
. <u>TY</u> A.	Nearest identifying landmarks road, street, etc PE OF INCIDENT Oil Spill	(Beach name, pi	er, nea
. <u>TY</u> A.	Nearest identifying landmarks road, street, etc PE OF INCIDENT Oil Spill Substance	(Beach name, pi	er, nea
. <u>Ty</u> A.	Nearest identifying landmarks road, street, etc PE OF INCIDENT Oil Spill Substance Ship Oilfield	(Beach name, pi Quantity Pipeline	er, nea
. <u>TY</u> A.	Nearest identifying landmarks road, street, etc PE OF INCIDENT Oil Spill Substance Ship Oilfield Name of ship, etc	(Beach name, pi Quantity Pipeline	er, nea
. <u>TY</u> A. B.	Nearest identifying landmarks road, street, etc PE OF INCIDENT Oil Spill Substance ShipOilfield Name of ship, etc Radiological Incident	(Beach name, pi Quantity Pipeline	er, nea
. <u>TY</u> A. B.	Nearest identifying landmarks road, street, etc <u>PE OF INCIDENT</u> <u>Oil Spill</u> Substance ShipOilfield Name of ship, etc <u>Radiological Incident</u> 1. Nature of Incident:	(Beach name, pi	er, nea
. <u>TY</u> A. B.	Nearest identifying landmarks road, street, etc PE OF INCIDENT Oil Spill Substance ShipOilfield Name of ship, etc Radiological Incident 1. Nature of Incident: a. Loss of control	(Beach name, pi	er, nea
. <u>тү</u> А. В.	Nearest identifying landmarks road, street, etc <u>PE OF INCIDENT</u> <u>Oil Spill</u> Substance ShipOilfield Name of ship, etc <u>Radiological Incident</u> 1. Nature of Incident: a. Loss of control b. Lost source	(Beach name, pi	er, nea
. <u>TY</u> A. B.	Nearest identifying landmarks road, street, etc <u>PE OF INCIDENT</u> <u>Oil Spill</u> Substance ShipOilfield Name of ship, etc <u>Radiological Incident</u> 1. Nature of Incident: a. Loss of control b. Lost source c. Radiation producing d	(Beach name, pi	er, nea
. <u>тү</u> А. В.	Nearest identifying landmarks road, street, etc <u>PE OF INCIDENT</u> <u>Oil Spill</u> Substance <u>ShipOilfield</u> Name of ship, etc <u>Radiological Incident</u> 1. Nature of Incident: a. Loss of control b. Lost source c. Radiation producing d d. Exposure	(Beach name, pi	er, nea
. <u>тү</u> А. В.	Nearest identifying landmarks road, street, etc <u>PE OF INCIDENT</u> <u>Oil Spill</u> Substance ShipOilfield Name of ship, etc <u>Radiological Incident</u> 1. Nature of Incident: a. Loss of control b. Lost source c. Radiation producing d d. Exposure e. Transportation accide	(Beach name, pi	er, nea
. <u>TY</u> A. B.	Nearest identifying landmarks road, street, etc <u>Oil Spill</u> Substance ShipOilfield Name of ship, etc Radiological Incident 1. Nature of Incident: a. Loss of control b. Lost source c. Radiation producing d d. Exposure e. Transportation accide f. Nuclear weapon	(Beach name, pi	er, nea

٠..

.

•••	Other Haza	ardous Materia	<u>11</u>	
	Substance		Quantity	
	Generic Na	ames		
		Solid	Liquid	Ga
	ISE OF INCII	DENT		
Des	scribe			
<u> </u>				<u></u>
TN			·····	
INC	IURIES	To which h	conital wana injuned take	52
τn,	(numbe	er)	lospital were injured take	
Are	e injured pe	ersons contami	nated? Yes No Nos.	
Wer	e injured p	persons expose	d? Ies No Nos	•
If	yes, was th	ne hospital an	id ambulance crew so advis	ed?
Des		Wene dead con	taminated?	
DES	SONNEL AND	FOUTPMENT		
Wha	t emergency	v personnel an	d equipment are at the sc	ene?
W 110	to emergency	y personner an	a equipment are at the se	
Tvr	e of additi	lonal assistan	ce requested?	
- , ,				
		COMATTON AND	COMMENTS	
ADI	ITIONAL INF	URMALIUN AND	COMMENTO	

- 1. Take all feasible steps necessary to protect or save human life. Safeguard property insofar as practical.
- 2. Take actions to contain and/or prevent the spread of the material. Spread sand or other collection agents, build dike, etc.
- 3. Keep the public as far from the scene of the incident as reasonably possible. Prevent souvenir hunting and handling of debris. In the case of a nuclear weapons incident, keep the public at least 2,000 feet away.
- 4. Isolate for further examination those persons who may have had contact with the material. Obtain names and address of those involved.
- 5. Remove injured persons from the areas with as little direct personal contact as possible. Hold them at a transfer point for first aid. If serious injury has occurred, demanding more than first aid measures, the patient should be sent at once to the nearest emergency room for medical attention. Advise medical attendants and facilties of possible contamination.

Medical first aid is directed primarily at restoration of breathing, control of hemorrhage, splinting for fractures, prevention of shock and control of pain. These are carried out for an exposed person in the same basic fashion as for a non-exposed individual.

First aid for containinated persons consists of cleansing the skin of obvious dirt (possibly contamination) and if feasible, carefully remove the outer garments and shoes of the patient and wrapping him mummy-fashion in a blanket, sheet, canvas, or large coat. By this measure, any remaining contamination is contained and if the wrapping is carefully done, the victim can be moved with little likelihood of spreading contamination.

- 6. If incidents involve fire or material subject to blowing in the wind, conduct operations from an upwind position. Keep out of smoke, fumes, or dust resulting from the incident. Segregate clothing and tools used at the scene until they can be checked for contamination. Do not handle suspected material until it has been inspected and relased by qualified technical experts.
- 7. In a vehicle accident involving hazardous material, detour all traffic around the accident scene. If this is not possible, move the vehicle or vehicles involved the shortest distance necessary to clear the rightof-way. If the material is spilled, prevent the passage of vehicles and people through the area until it has been surveyed. If right-of-way must be cleared before the assistance team arrives, wash spillage to the shoulders. Do allow wash water to enter drainage system.
- 8. Do not eat, drink, or smoke in the accident area. Do not use food or drinking water that may have been in contact with material from the incident area.
- 9. Take only necessary emergency actions prior to the arrival of a qualified hazardous materials specialist and/or physician.

- The County Plan has a listing of 24-hour/day telephone access numbers to county, State and Federal emergency agencies. Under the listing of Federal Agencies the telephone numbers of Tyndall AFB-DP and Naval Coastal Systems Center is given.
- The Civil Defense has prepared a listing of absorbents, congealing agents, gelling agents, liquids and other equipment which may be required on-scene. This listing has been distributed to each of the service organizations so that they could inventory their resources and have pre-identified materials which may be needed on-scene prior to the occurrence of an accident.
- Fire department equipment consists of breathing apparatus and turn-out gear.
- Civil Defense operates and coordinates specialized equipment utilization on-scene and is equipped with a communications van, breathing apparatus and jump suits.
- The county has only two copies of the DOT Emergency Response Guidebook.
- Emergency response vehicles are not equipped with the DOT guide.
- The County does not train emergency response personnel to handle hazardous materials accidents. Bay County, however, has established a group of decisionmakers from the service organizations in the county (one from each organization) to serve on a hazardous material advisory board. The members of this advisory board are the only response personnel who receive formalized HM training.

## 5.3 ASSESSMENT OF METHODS OF CRISIS MANAGEMENT

This section provides a critical assessment of methods used in each city for implementing crisis management techniques for hazardous materials transportation accidents. Data were collected from each of the ten cities concerning specific guidelines/plans for handling hazardous materials accidents; hazardous materials training programs; and specialized response personnel, equipment and gear for use on-scene. Also, any problems the cities may have encountered or expected to encounter in handling hazardous materials transportation accidents were identified.

Each of the ten cities reviewed has experienced a hazardous material transportation accident involving at least one of the 28 chemicals and propellants being examined by this study. Of the 28 hazardous materials, the following 16 materials were involved in incidents in any of the ten cities: acetone, acrylonitrile, anhydrous ammonia, chlorine, ethylene oxide, liquefied hydrogen, hydrazine, methanol, methyl bromide, propane, propylene, sodium hydroxide, styrene, toluene, UDMH and vinyl acetate. The remaining 12, acetone cyanohydrin, aerozine-50, ethyl acrylate, hydrocyanic acid, isobutane, monomethylamine nitrate, sodium hydrosulfide, monomethylhydrazine, liquefied oxygen, vinyl chloride,  $N_2O_4$  and butadiene were not involved in accidents in these cities during 1971-1980. Table 5-2 has been prepared to show the accident involvement by hazardous materials frequency for the ten cities visited. Table 5-3 has also been prepared to show the distribution of accidents for the chemicals and

## TABLE 5-2

## MATERIAL INVOLVEMENT IN ALL CITIES

Rank	Material	Number of Accidents	Pct. of <u>Sample</u>
1	Sodium hydroxide	26	23.5
2	Methanol	15	13.5
3	Acetone	14	12.6
ų.	Anhydrous Ammonia	13	11.7
5	Toluene	. 9	8.1
	Propane	9	8.1
7	Acrylonitrile	4	3.6
	Chlorine	4	3.6
9	Propylene	3	2.7
	Styrene	3	2.7
	Vinyl Acetate	3	2.7
12	Methyl bromide	2	1.8
	UDMH	2	1.8
	LHo	2	1.8
15	Ethylene oxide	1	0.9
	Hydrazine	_1	0.9
TOTAL		111	100\$

## TABLE 5-3

## ACCIDENT HISTORIES OF THE CHEMICALS AND PROPELLANTS IN THE CITIES EXAMINED

	Baltimore, MD	Los Angeles, CA	Nashville, TN	Newark, NJ	Pensacola, FL	Sacramento, CA	Tallahassee, FL	Waverly, TN	Wilmington, DE	Youngstown, FL	Total
Acetone	6	3		3		1			1		14
Acrylonitrile	1				1				2		4
Anhydrous Ammonia	2	4		1	3	3					13
Chlorine	1	1		1						1	4
Ethylene Oxide			1								1
Hydrazine		1									1
Liquefied Hydrogen		1	1								2
Methanol	3	3	3	2		3			1		15
Methyl Bromide		1				1					2
Propane		2			. 1		1	1			9
Propylene		2	1								3
Sodium Hydroxide	6	8	4	2	1	3	1		1		26
Styrene		3									3
Toluene	2	1		3	1				2		. 9
Unsymmetrical Dimethlhydrazine			1	1 -							2
Vinyl Acetate		2							1		3
TOTAL	21	32	16	13	7	11	2	1	8	1	111

5-23

propellants by city. The DOT Materials Transportation Bureau's data base for the period 1970 through 1980 was used. Using the average hazardous materials accident cost of \$41,900 for 1976 through 1980 and based on the number of accidents in each of the ten cities, the estimated accident cost resulting from the 16 chemicals and propellants was in excess of \$4.6 million.

Table 5-4 shows the estimated loss to each city as a result of accidents involving the 16 chemicals and propellants. Thus it can be seen that each of these communities does have the need for response capabilities for handling possible hazardous materials transportation accidents. The analysis of the response capabilities in each of the cities is presented in the following sections.

Response

Planning

and

## 5.3.1 Contingency Planning for Hazardous Materials Transportation Accidents

5.3.1.1 <u>Overview of Community Emergency R</u> Recommendations for Improved Community Plans

•

STL's "Draft Guidelines Manual" conducted for project F04611-80-C-0046 showed that municipalities having pre-established contingency plans for handling emergency situations were better prepared for handling the hazardous environment surrounding the occurrence of hazardous materials transportation accidents than those areas that had not pre-planned. With the increasing awareness of the potential consequences of an accident involving hazardous materials, many communities (e.g., Nashville, TN; Sacramento, CA; Baltimore, MD) have developed plans specifically aimed at dealing with the hazards posed during an accident involving hazardous materials. However, many localities have not prepared or are not aware of existing hazardous materials transportation accident contingency plans. This fact can be exemplified by such transportation accidents as Beattyville, KY and Youngstown, FL. In fact, communities which have contingency plans usually have a large community-minded chemical manufacturer in the area or have had a major transportation accident which has forced community involvement and awareness.

It is recommended that a viable community contingency plan for hazardous materials transportation accidents be developed for every locality and that this plan contain the following information:

- Hazardous material shipping routes and volumes through community;
- Community transportation network in terms of possible evacuation routes and access by emergency services;
- Location of specialized personnel, materials, and equipment in community or nearest location adequate to handle hazardous materials emergencies;

## TABLE 5-4

## ESTIMATED LOSSES DUE TO INCIDENTS INVOLVING THE 16 CHEMICALS AND PROPELLANTS

City	Accidents	(\$1,000's)				
Baltimore, MD	21	879.9				
Los Angeles, CA	32	1,340.8				
Nashville. ÍN	16	670.4				
Newark, NJ	13	544.7				
Pensacola, FL	Ť	293.3*				
Sacramento, CA	11	460.9				
Tallahassee. FL	2	83.8				
Waverly, TN	1	41.9*				
Wilmington. DE	8	335.2				
Youngstown, FL	1	41.9*				

\* Actual costs were significantly higher due to additional third party liability law suits and environmental cleanup costs.

- Appropriate segments of the emergency response community, with clearly defined individual roles, responsibilities and statutory authorities;
- Methods for accessing relevant technical assistance sources; and

 Designated communications network (radio frequency, network channel siren) to alert the public and to handle communications between/from the Communications Command Center, the accident site and other off-scene support organizations.

These topics are discussed in subsequent paragraphs in this section. At least two states, California and Virginia, have conducted studies to assess the volume of hazardous materials traveling along various segments of the state's transportation network. However, because local emergency responders (i.e., fire, police, medical) are the first groups on-scene, it is recommended that an inventory of hazardous material traffic be conducted at the local community level as well as the state level. The city or regional planning office might be the logical organization to be charged with performing this duty and possibly some other duties associated with municipal contigency pre-planning. Statutes may dictate someone else. The important thing is to have some organization responsible.

It is recommended that an inventory of the community transportation network be conducted. Also, thermophysical/chemical data should be compiled for the hazardous commodities being transported through a community. Based on current emergency response practices, this information should be carried in the cars of each emergency service "chief" (i.e., designated on-scene coordinator or his representative). The types of commodity information which the "chief" should carry include specific gravity, vapor density, explosive limits, toxicity levels and firefighting/first-aid information.

The catalog of appropriate segments of the response community, their responsibilities and authority should also include an inventory of specialized hazardous material response teams in, or available to, the community including local emergency services (fire, police, medical), industrial teams, trade association teams and federal, state and local government personnel. The type of information which should be collected for each specialized responding organization should include the following:

- name and address of key persons/contacts,
- 24-hour emergency phone numbers
- 800 toll-free telephone numbers, if available
- what resources they can provide

Other personnel or organizational information which should be indexed includes:

response speciality (e.g., firefighting, wreck handling, cleanup, disposal)

- specific commodity expertise
- availability of specialized equipment and materials

The inventory of specialized equipment and materials suitable for hazardous materials traveling through a community should identify:

- materials and equipment needed for each hazardous materials being shipped
- location and availability of materials and equipment at public facilities and commercial/industrial facilities

Technical assistance may be obtained in several ways. The Chemical Transportation Emergency Center (CHEMTREC 800/424-9300) which is operated on a 24hour, 7-day-a-week basis by the Chemical Manufacturers Association, can provide some initial response actions for an identified HM and get the shipper in direct contact with the emergency scene. If the HM happens to be one for which a segment of the chemical industry has developed special response teams (e.g., the CHLOREP teams of the chlorine industry through the Chlorine Institute), CHEMTREC alerts such groups. The chemical manufacturers and shippers are the most knowledgeable about the HM(s) they produce and ship and are in the best position to provide technical assistance at the accident scene. CHEMTREC information per se is "cookbook" for specific commodities and no judgments or recommendations are offered.

The National Response Center (NRC), operated by the U.S. Coast Guard (800/424-8802) in conjunction with its joint responsibilities with EPA in handling water spills of hazardous substances, likewise is operated around-the-clock and has a direct tie-in with CHEMTREC through a written agreement.

The Coast Guard and EPA have joint regional response teams, with designated onscene coordinators (OSC), which are dispatched to th scene if either EPA or the Coast Guard deems it necessary. These teams can provide technical advice and/or actually conduct cleanup and disposal operations, if necessary. In addition, the NRC has computer programs for predicting the dispersion of spilled HMs (currently this is essentially confined to water spills, but is being expanded to handle land spills). The computerized data system makes available more detailed technical information than CHEMTREC can provide. EPA and the Coast Guard also have technical experts who may be contacted for technical advice.

There are several cleanup and disposal contractors who specialize in handling and/or advising regarding these aspects of HM spills. Some shippers utilize such contractors when they do not have in-house specialized teams. However, carriers rarely have such capability and would have to rely on a contractor. Normally, the shipper and carrier agree on how the matter will be handled, so it does not become a problem.

5-27

A number of shippers provide a company 800 emergency number on shipping papers and sometimes on the tank cars or cargo tanks.

Without adequate, fully-coordinated communications, it is impossible to handle a HM transportation emergency successfully. The communications network must enable those groups at the scene to communicate within their respective disciplines, and between disciplines and with the OSC. Additionally, the OSC must have direct contact with all the off-scene support activities (e.g., aerial surveillance, weather service, and hospitals) and shipper, carrier, local, State and Federal officials and response centers (e.g., NRC and CHEMTREC) and the news media. All of these communications must be on a non-interfering basis.

Most of the emergency services organizations have communciations systems. The important thing is to tie all communication into a centralized communication center. A system of priorities must be established as to who and what takes precedence. If possible, assignment of specific frequencies to the various groups is recommended. "Ham" operators, particularly, may be a valuable resource. The CB system might be of value under special circumstances.

#### 5.3.1.2 Critical Analysis of Community Plans

Based upon the criteria for effective crisis management techniques to hazardous material transportation accidents given in Section 5.3.1.1, Table 5-5 has been prepared to show the state-of-the-art in community contingency plans for each of the ten cities examined. Based upon a critical assessment of each community's contingency plan for handling hazardous materials transportation accidents it appears that the majority of cities have inadequate on-scene communications capabilities and do not sufficiently provide for the dissemination of accident information to the public. The cities reviewed do have the following components in their contingency plan:

- Knowledge of type of hazardous materials and their transport routes in proximity of the community;
- Specific emergency response guidelines for handling all hazardous materials transportation accidents;
- Inventory of specialized personnel, equipment and materials which could be used on-scene; and
- On-site responsibilities of local emergency response organizations.

The cities which did not have specific guidelines concerning response to hazardous materials would either utilize guidelines for response to fires or the emergency response guidelines identified in Federal and industrial response guidebooks (i.e., AAR,DOT,EPA and USCG).

# TABLE 5-5ASSESSMENT OF COMMUNITY CONTINGENCY PLANSBASED UPON CRISIS MANAGEMENT CRITERIA

1

Criteria	Baltimore, MD	Los Angeles, CA	Nashville, TN	Newark, NJ	Pensacola, FL	Sacramento, CA	Tallahassee, FL	Waverly, TN	Wilmington, DE	Youngstown, FL (Panama City)
Identified hazardous material shipping routes and volumes	٠		•	•	٠		٠			•
Community transportation network in terms of possible evacuation routes and also access to emergency services	•	•	•	٠	•	•	٠		٠	•
Specific hazardous materials response guidelines/contingency plan	•		•	•	•		•		•	•
Location of specialized personnel, materials and equipment in community or nearest location adequate to handle hazardous material emergencies	•	. •	•	, •	•	•	•	•	•	•
Appropriate segments of the emergency response community, with clearly defined individual roles, responsibilities and statutory authorities	٠	•	•	•	•		•			•
Methods for assessing relevant technical assistance sources	٠	•	•		٠		٠			•
Designated communications network to alert the public and to handle communications between the Communications Command Center, the accident site and other off- scene support organizations					• ,					

5-29

•••••

· .

.

TABLE 5-5 (cont'd)

Criteria	Baltimore, MD	Los Angeles, CA	Nashville, TN	Newark, NJ	Pensacola, FL	Sacramento, CA	Tallahassee, FL	Waverly, TN	Wilmington, DE	Youngstown, FL (Panama City)
List of specialized response									_	
- name and address of key	•	٠	٠	•	٠		٠	٠	•	•
persons/contacts - 24-hour emergency phone numbers - resources of group listed	•	•	•	•	•		•		•	•
Contingency plan components:										
<ul> <li>statement of purpose and scope</li> <li>enabling acts and authority</li> </ul>	•		•	•	•		•			•
- identity and functions of	•		•	•	•		•			•
the officials involved - evacuation plan with pre-			•							•
designated shelters and	•		•		•					•
<pre>logistical support - identification of radio and   television stations that will   continuously inform the public   during the initial trauma</pre>										•
- identificaton of the pre- designated local on-scene coordinator (LOSC) by name and/or position	٠		•	•	•		•			•
- a mechanism for updating	٠		٠		٠					
- a document snowing now the local plan interfaces with the federal and state plans	٠		•	•	•		•			
- identification of a liason official to work with the state	٠		•	•	٠		•			•
and federal OSC										
numbers of key officials	•				•					•
- check-in location where key officials can be located or tracked once they have appeared on-scene	•		•	•	•		•			•
<ul> <li>a method for identifying and accounting for individuals who may have authority on-scene</li> </ul>					•					•

## 5.3.2 Training of Emergency Response Personnel

There are a number of training courses available which are taught by government, industry, educational institutes and consulting organizations aimed specifically at responders to hazardous materials incidents. These courses vary from formal academic class sessions to slide-tape presentations, with an insructor's guide and student workbook, to the home-study (correspondence) course. The slide-tape courses run from five to twenty hours of class time. They may or may not be modular (i.e., deal with topical areas such as hazardous materials identification, decision-making and seeking technical assistance).

Hazardous material training courses stress planning but also present some basic information concerning the nature of hazardous materials; how to identify spilled/leaking material; where to find technical help; danger assessment; decision-making; and to a cetain extent some general procedures for on-scene actions such as controlling access to the area, evacuation, surveillance of vapor clouds, firefighting, rescue and In some instances, there is hands-on training such as use of communications. polyurethane foam for sealing holes in drums or diking liquid pools, applying metal patches to tank car holes by means of bolts or stopping leaks with wooden plugs. These are useful techniques, but have limited application to specific situations. Heavy reliance must still be placed on the specialists from the varous disciplines involved. These specialists operate and make decisions based predominately on their own experience and knowledge and, with few exceptions, perform tasks without the benefit of written procedures, particularly with respect to cargo transfer, wreckage removal, cleanup and disposal. Although these courses give some attention to restoring the scene to normal, there is a lack of procedural training in these four activities.

There are many training aids available in addition to the courses themselves. Examples are nomographs, slide rules, pocket manuals, checklists, brochures, guides, films, video tapes, slide-tape combinations, reference books, data bases, resource lists and charts. The U.S. Department of Transportation's Materials Transportation Bureau provides, free-of-charge, quantities of a number of hazardous materials training aids to emergency service organizations. Others may be purchased from the Government or private companies. Rail carriers, in conjunction with the Chemical Manufacturers Association, are putting on hazardous materials transportation emergency training courses in communities where chemical shippers are located or through which rail lines run. The U.S. Department of Transportation, Research and Special Programs Administration, Materials Transportation Bureau has compiled a list of 342 organizations offering training courses on hazardous materials transportation.

Regardless of the type of activity — immediate response, hazard mitigation, cargo transfer, wreckage removal, cleanup and disposal or the specialist discipline involved — training must assure that procedures are understood and utilized to accomplish the following four items:

- 1. Provide adequate on-scene communications
- 2. Evaluate/assess the situation, hazards and actions
- 3. Make decisions

4. Take appropriate actions

These are discussed in the following paragraphs.

## 5.3.2.1 Communications Training

An incident must be recognized and promptly reported to the proper authority. It is very essential that specific information about the accident be provided in this report so that the response network may be activated and those involved can have a reasonable idea of the nature of the accident and hazardous materials involved. Training in how, when, what and to whom to report a hazardous material accident is the first criterion. It involves the ordinary citizen, who by chance may stumble upon an accident scene, as well as those who might become involved as professionals. Communications within a particular response discipline, between groups and with the on-scene coordinator (OSC) are complex but vital. Therefore, training is required in the proper use of communications equipment. Also, the assignment of proper frequencies and responsibility for coordination of communications must be clearly identified in the emergency action plan. The persons responsible for coordinating communications need training:

- To understand the interface between different communication modes, frequencies and equipment;
- To understand, interpret and relay facts and requests being made by or sent to the numerous groups and individual specialists involved in the emergency;
- To recognize and expedite priority communicati s;
- To deal effectively with the news media, by providing appropriate factual information and by utilizing the news media as a means of mitigating hazards to the public such as preventing panic and providing proper instructions or warnings; and

To know how to use communications to coordinate effectively the many activities taking place on-scene and as backup, so that such activities do not interfere with or jeopardize safety of each group and that resources are used most effectively.

The various response groups and individual specialists need communications training in order to learn proper procedures for maintaining constant contact with in their particular groups so that everyone is always accounted for, prompt escape action may be taken if the need arises, and the OSC can be provided with the latest facts on conditions, progress, problems and needs.

The public needs training in such areas as simple self-protection actions (i.e., stuffing cracks in windows or doors) in the event of a hazardous material spill; getting and keeping away from the scene; obeying evacuation orders; and, as previously mentioned, reporting an incident.

The news media can be a real help or can compound the problem. Making the news media aware of and, where possible, a participant in hazardous material spill response training, can make it a strong positive force in a real emergency. Training courses need to contain a portion showing how the news media can assist in the event of an accident. News media representatives should be included in the preparation of the community's HM emergency response plan and in training courses that are given.

All persons who will be concerned with a spill must have further training in evaluation/assessment methods, decision-making, the procedures required in their specific activities and awareness of how their actions impact others.

#### 5.3.2.2 Training for Decision-Makers

It is imperative that training be designed to meet the needs of all decisionmakers. Depending upon the individual responsibilities and the particular types of activities involved, training can range from checkoff lists to computer-aided decisionmaking methods. Essential to all decision-making is consideration of the situation or problem, the alternative courses of action, how the action will be accomplished, when and by whom, and what will be the expected impact or results of each. Evaluation and assessment of the situation are the key factors upon which sound decisions are based. Therefore, a detailed discussion of training requirements for decision-makers in evaluation and assessment is presented.

A chemical, propellant or other hazardous material transportation accident requires initial and continued assessment of the situation and evaluation of the

5-33

requirements and effectiveness of corrective actions. Essentially, these involve obtaining facts and analyzing them. Training is vital to assure that those involved with the emergency know what information is necessary, how it may be obtained and how to analyze it for determining the existing hazards, potential dangers, what damage has been sustained, the magnitude of the spill, who and what are exposed, what resources exist and how they can be used most effectively, what additional resources are required, and the effectiveness of corrective actions. Such training should involve how to:

- identify at a distance any HM's involved or that have been released;
- determine the integrity of the HM containers;
- establish the danger perimeter;
- predict the downwind toxic or flammable vapor concentration versus distance as well as cloud size and travel rate;
- use resources most effectively;
- determine the applicability and effectiveness of corrective actions;
- use remote sensing/detection/analytical equipment;
- interpret data;
- spot changing conditions which pose additional dangers;
- assess risks;
- determine hazards; and
- monitor the scene for toxic or flammable vapor levels and for evidence of personnel exposure.

Such training includes teaching formalized methodologies where appropriate (i.e., risk/hazards analysis).

#### 5.3.2.3 Response Activities Training

Training is required to assure effective and safe performance of all the on-scene and support activities in handling hazardous materials transportation spills. This fourth aspect of training deals with the actual field operations and what type of procedural training deals with the actual field operations and what type of procedural training is appropriate to each of the specialized groups and individual experts involved. This training involves ways to select, use and identify the limitations of equipment and materials (i.e., use only transfer equipment which is compatible with the particular hazardous material or use gravity flow, pressure flow or pumping as cargo transfer means). Training can help train crew members and truck drivers:

- to understand the HM's aboard, their hazards and the precautionary procedures they can use in the event of an accident;
- assist them in seeking response help;

## 5-34

• convey HM information to response personnel; and

• otherwise cooperate with authorities on-scene.

There are continued hazards at chemical and propellant spill scenes and the degree or nature may change. Training for on-scene personnel needs to include procedures for recognizing the actual and potential hazards and the eventuality of a significant change. Their training needs to assure that each person understands not only how to perform his own task efficiently, but to recognize the absolute necessity for safety and accomplishing the task without jeopardizing the safety of others at the scene or creating problems for them, while at the same time protecting the environment and property. Training on the selection and use of proper protective clothing, breathing apparatus, gear, tools, equipment and materials is vital to personnel safety and the successful handling of the spill. Training is needed in the techniques, limitations and safety precautions for cargo transfer, wreckage removal, cleanup and disposal operations. Hazard mitigation involves any means for reducing or eliminating the hazard or threat, so it cuts across the full spectrum of on-scene activities. Training needs to concentrate on the use of common sense coupled with good information and sound technical analysis.

## 5.3.2.4 Training of Emergency Response Personnel in the Cities Reviewed

All of the ten cities reviewed utilized some form of training program for response personnel who deal with hazardous materials under emergency conditions. Eight of the ten cities reviewed stated that they trained personnel in the use of the Department of Transportation's 1980 Emergency Response Guidebook. Of the two cities which did not train emergency response personnel in the use of the DOT Guidebook (i.e., Pensacola, FL and Youngstown (Panama City, FL)), Pensacola does not recommend the use of the DOT Guidebook because they question the Guidebook's utility while Youngstown (Panama City) does not use the guide because they do not have sufficient copies available for all response personnel. It is recommended that if Youngstown (Panama City) chose to utilize the DOT Guidebook that they procure enough manuals to supply each of the emergency response vehicles in the area. The Pensacola area prefers to utilize the Association of American Railraod's Bureau of Explosives response guide entitled Emergency Handling of Hazardous Materials in Surface Transportation. In terms of actual training programs seven of the ten cities utilize at a minimum the NFPA materials training program entitled Handling Hazardous Material hazardous Transportation Emergencies. According to the DOT Material Transportation Bureau the NFPA training course was designed to:

"assist those ...who assume and accept, often at great personal risk, the responsibility of responding to and dealing with transportation accidents involving hazardous materials... provide guidelines for handling hazardous materials during emergency situations and to assist persons with varous emergency services responsibilities in better understanding their roles in the development and implementation of comprehensive and community emergency action plans."

The NFPA hazardous materials training course provides emergency response personnel with information on hazardous materials in-transit; definitions, classes and dangerous properties of hazardous materials; recognizing and identifying hazardous materials incidents; command and control of hazardous materials incidents; and planning for hazardous materials emergencies.

Of the seven cities which utilize the NFPA hazardous materials training course, five use this curriculum in conjunction with specially designed city programs which were developed to meet the specific requirements of each city. Of the ten cities reviewed only Wilmington, DE relies upon the training services of a State Training Academy for the purpose of preparing emergency response personnel in the handling of hazardous materials emergencies.

In all of the cities reviewed except Youngstown (Panama City), FL all emergency response personnel are given some form of hazardous materials training. In Youngstown (Panama City), FL training is provided only to members of a hazardous materials advisory board. It is recommended to the Youngstown (Panama City), FL area that all emergency respnse personnel be provided with at least the NFPA hazardous materials training course.

## 5.3.3 Availability of On-Site Resource Requirements

## 5.3.3.1 Protective Clothing, Gear and Equipment Availability and Uses

When entering a hazardous environment, appropriate protective clothing, gear, breathing apparatus and equipment must be used. This section does not discuss the specific items required for working in specific hazardous environments, but identifies the types which should be available on-scene and their uses.

In terms of personal protection, emergency services should have available protective clothing, acid suits, chemical/gas suits, cooling systems (heat exchangers) and fire entry suits for use by responding personnel. In terms of breathing apparatuses, regulated manifold air supply systems, assorted cannister masks and cartridges should be available. Several commercial manufacturers/suppliers produce this type of equipment. A sample listing of a few manufacturers/suppliers is given in Table 5-6. Table 5-6 is not an endorsement of any of these items or manufacturers and the types of personal protective clothing, equipment and gear which they supply. Chemical manufacturers, the U.S. General Services Administration catalogues, telephone yellow pages, EPA and USCG Regional Offices are other sources of this type of information.

Local emergency services should obtain the obvious personal protective clothing, equipment and gear required for handling the types of hazardous commodities being transported through or consumed in a community. Clothing, equipment and gear should be thoroughly inspected and tested periodically to assure that it will provide the required level of personal safety. Protective clothing, equipment, breathing apparatus or gear which does not meet these strict safety standards should be either discarded or repaired. Once repaired, inspection and testing should be conducted to assure that the repair was adequate. Penalities should be imposed on individuals/organizations which do not abide by these guidelines and who subject personnel to unnecessary hazards due to negligence in enforcing these requirements. These items are expensive and the various segments of the emergency response community have much to gain by pooling resources.

## 5.3.3.2 Specialized Treatment Chemicals, Equipment, Resources Availability and Uses

Specialized treatment chemicals, equipment and resources are needed at a hazardous materials transportation accident. This section identifies treatment chemicals, sorbents and analytical and heavy equipment which can be utilized on-scene.

## 5.3.3.3 Treatment Chemicals and Methods

Several treatment chemicals and methods exist for mitigating the hazards associated with a spill and the decision to use one specific method must be based on technical need, material availability, State and Federal limits for materials in the environment, cost versus effectiveness and residues requiring disposal.

Once the treatment considerations have been analyzed based on the accident conditions and the aforementioned factors, it is then possible to select the appropriate treatment method. Table 5-7 lists some treatment methods for handling spills of hazardous materials.

It needs to be emphasized that only compatible treatment chemicals should be used with the spilled material; mixing incompatible materials may result in worsening the

## TABLE 5-6 SOURCES OF PROTECTIVE CLOTHING, BREATHING APPARATUS, GEAR AND EQUIPMENT

Item	Manufacturer	Location					
Portable resuscitation units	Robert Shaw	Anaheim, CA					
Gas/vapor respirators	3M Company	St. Paul, MN					
Breathing Apparatuses (15-60 minute capacity)	Lab Safety Supply	Janesville, WN					
Gas Mask (30 minute capacity)							
Organic Vapor Respirator							
Acid Gas Respirator							
Gloves							
-neoprene (corrosives)							
-nitrate (aromatic, petroleum							
and solvents)							
-polyethylene	•						
-PVA coated (organic solvents,							
aromatics, ketonics and							
chlorinated solvents)							
Face Shields							
Safety Caps							
Splash Suits							
First Aid Kits							
Breathing Apparatus	Mine Safety Appliances (MSA)	Pittsburgh, PA					

· · · ·
# TABLE 5-7HAZARDOUS MATERIAL TREATMENT METHODS

- Carbon absorption
- PH adjustment
- Air stripping and aeration
- Precipitation
- Neutralization
- Biological
- Destruction
- Solution/Dilution
- Mixing
- Land Spraying

situation. Local contingency plans should identify the local and availability of chemicals for treating any hazardous materials which flow through a community. Some common spill control materials available are acid and caustic neutralizing agents, diatomaceous earth, vermiculite, Portland cement, sawdust, activated carbon and various commercial sorbents.

### 5.3.3.4 Sorbent Materials

Use of sorbent materials to soak up and contain spilled hazardous materials during initial response, product transfer, cleanup and disposal operations at the accident site is common practice. Typical sorbent devices include spill control pillows, which absorb 98% of their capacity in 30 seconds (a number of these can be combined to form a dike); spill squeegee and absorbent paper. It is recommended that a stockpile of such materials be kept by the local community at all times. However, if this is not feasible, the community should identify the location where these supplies can be obtained in a timely fashion, establish cooperative agreements with local chemical manufacturers who would have a supply in-house, or obtain these materials from the product transfer, cleanup and disposal contractor directly.

#### 5.3.3.5 Monitoring Devices

Monitoring devices are a necessity at hazardous materials transportation accidents. Analytical monitoring devices serve such functions as identifying hazardous materials on-site and providing continuous monitoring for toxic, flammable and explosive vapors during all phases of accident response. Table 5-8 is a partial listing of analytical monitoring devices that communities should have available for use in the event of a hazardous materials transportation accident. In the event of a hazardous materials transportation emergency, communities should seek to have most of this equipment available for their use. Analytical monitoring equipment may be purchased exclusively for the use of a city's emergency services, borrowed from a community chemical manufacturer or chemical laboratory, obtained from a region's state emergency preparedness/civil defense office or acquired for use from a nearby military installation. Fire departments have flammable vapor detectors. It is recommended that communities obtain and pool as much of this equipment as possible. However, when a locality cannot fund such purchasing it is recommended that agreements be preestablished with state preparedness/civil defense and military installations for their use and any technical assistance in their operation, as needed.

5-40

#### 5.3.3.6 Heavy Equipment

Similarly, heavy equipment are a necessity at hazardous materials transportation accidents. Table 5-9 is a partial listing of heavy equipment that communities should have available for use in the event of a hazardous materials transportation accident. Heavy equipment can usually be obtained through the state or local Department of Highways and Transportation or through a local construction contractor.

#### 5.3.3.7 Wreckage Removal Contractors

When a transportation accident occurs and the services of a wreckage removal contractor are required on-scene, the choice of a wreckage removal contractor will depend upon the transportation mode involved in the accident, the accident severity in terms of structural damage to equipment, proximity of accident site to contractor's facility, and contractor's available resources. For purposes of this report we are concerned with wreckage removal contractors who respond to rail and highway hazardous materials transportation accidents. Most often, and it is sound safety practice, these contractors will not respond on-scene until all toxic, flammable and explosive vapors related to hazardous materials have been dissipated from the accident site, and the area is considered to be a safe working environment. When wreckage removal contractors are required at rail transportation accidents, special heavy equipment is needed. Two organizations which have historically provided on-site wreckage removal activities to rail transportation accidents are:

### Rail Wreckage Removal Contractors

- Isringhausen Crane Manufacturers, Inc. One Industrial Drive Jerseyville, Illinois 62052 (618) 498-6441
- Hulcher Emergency Services, Inc. Box 191
  Virden, Illinois 62690
  (217) 965-3361
  (800) 252-3371 in Illinois
  (800) 637-5471 outside Illinois

Wreckage removal contractors involved in the response to accidents of highway vehicles are typically local towing or wreckage services. Listing of these organizations can be found in the local telephone directory yellow pages. Unfortunately, their response capabilities are less sophisticated than the groups which respond to railroad accidents.

### TABLE 5-8

### ANALYTICAL MONITORING EQUIPMENT

- Gas detector
- Combustible gas/oxygen detector
- Oxygen deficiency monitor
- Electrical safety hazard analyzer
- Radiation/contamination survey meters
- Infrared radiometer
- Explosimeter

States and states

- Mass spectrometer
- Colorimetric tubes

### TABLE 5-9 HEAVY EQUIPMENT

- Bulldozer
- Crane
- Backhoe
- Highloader
- Dump trucks

Local contingency plans should have these telephone numbers available for use in the event of an accident and only those considered to be qualified and reliable should be listed.

Accident severity influences the level of sophistication required for equipment to be brought on-scene. Proximity of the accident site to the contractor's location influences response time and availability of personnel, equipment and materials. State and local emergency response plans should have pre-identified wreckage removal firms which are capable of responding to transportation accidents that occur in a specific area.

The personnel, equipment and material resources available to the wreckage removal contractor at the time of the accident may influence the choice of a firm to perform the wreckage-removal operations. This can be illustrated by a situation where a wreckage-removal contractor's facility is located one mile from a railroad accident, but due to their inability to provide heavy lifting equipment to the scene, it is concluded that this firm does not have long-term on-scene response capabilities. Since heavy equipment would be needed in this instance, a wreckage-removal contractor having heavy equipment would have to be contacted, even if this firm's facility is located 450 miles from the accident scene. Of course, response time will be increased significantly, but this should not create a big problem because wreckage removal should not begin until the situation is stabilized. The on-scene coordinator should not automatically discount the value of a wreckage removal contractor who does not have the heavy equipment, for this firm may be able to assist in the short-term until the better equipped wreckage removal contractor can arrive on-scene.

Acquiring a wreckage removal contractor is just one of the many decisions which the on-scene coordinator must make regarding the use of outside technical experts.

#### 5.3.3.8 Product Transfer, Cleanup and Disposal Contractors

At most hazardous materials transportation accidents the services of product transfer, cleanup and disposal personnel are required. In many cases, the shipper or association to which the shipper belongs may provide such specialized service. The carrier should involve the shipper in the selection of a product transfer, cleanup and disposal contractor.

The community contingency plan should identify product transfer, cleanup and disposal contractor(s) who are capable of meeting the locality's requirements, based on the hazardous materials being transported in the area, and in the event the shipper and carrier elect not to handle the arrangement or request a recommendation.

In terms of on-site product transfer, cleanup and disposal operations, the following information should be available to the on-scene coordinator and others who must make decisions regarding these activities:

- Waste disposal sites capable of handling the spilled materials as well as state solid waste management agencies;
- Shipping container specifications required for transport of each chemical shipped through the area; and
- Applicable Federal, State and local regulations pertinent to the transport of these commodities.

Local decision makers should familiarize themselves with this information, so that, when local contingency plans are developed, data on the location and capabilities of local waste disposal facilities and availability of shipping containers required for disposal based on the community's hazardous materials transportation needs will have been identified. By so doing it is anticipated that product transfer, cleanup, and disposal activities may be performed at the greatest level of cost-effectiveness and personal safety.

#### 5.3.3.9 Specialized Response Capabilities of Each of the Cities Reviewed

The analysis of the availability of specialized response personnel, equipment and gear in each of the ten communities reviewed concentrated on identifying whether or not each of the cities had the following resources available:

- Hazardous material react team
- Mobile response van
- Specialized equipment and gear (i.e., CHLOREP patching kit, positive pressure breathing apparatus, heavy lifting equipment, foam trucks, analytical testing equipment, hazardous material reference library, personal protective clothing, acid suits, Scott air packs, etc.).

In fifty percent of the cities visited, hazardous materials react teams are available through the local fire service. These cities are Baltimore, MD; Los Angeles, CA; Nashville, TN; Sacramento, CA; and Tallahassee, FL. In Newark, NJ the response team and mobile response van are operated by the state and these resources are located approximately two hours away from Newark. In Pensacola, FL response to hazardous materials transportation accidents are handled by the fire service with the local chemical manufacturers providing technical assistance. Nashville, TN provides response capabilities to the entire central portion of Tennessee under a local/state agreement in which Nashville provides the trained emergency response personnel in exchange for being supplied specialized emergency response equipment at state expense to respond to incidents in the central portion of the state. Under this local/state agreement, the metropolitan Nashville area has the responsibility for responding to hazardous materials incidents at cities like Waverly, TN. This agreement was probably a result of the tragic accident at Waverly in which they did not have the capabilities to adequately handle the assessment required. Waverly, TN also has an agreement with three chemical manufacturers in the area which provides industrial response assistance if an emergency were to occur. In the Youngstown (Panama City), FL area a react team has been developed which consists of one representative from each of the emergency response service organizations in the area.

All of the cities except Newark, NJ; Waverly, TN; and Youngstown (Panama City), FL have direct access to a mobile hazardous materials response van. A response van is available to Newark, NJ but is stationed at Trenton - nearly two hours south.

The majority of cities visited had positive pressure breathing apparatus, personal protective clothing and Scott air packs. Resources which some of the cities had but were more limited included CHLOREP patching kits, analytical testing equipment, extensive hazardous materials reference libraries, acid suits, cascade systems, mobile communications vehicle/command post and a converted bus for use as an ambulance. It is recommended that each city purchase or make a mutual aid agreement so that these aforementioned resources would be available for use in a hazardous materials transportation emergency.

### 5.3.4 Identification of Problems Encountered by Cities in Actual On-Scene Response to Hazardous Materials Transportation Emergencies

During visits to each of the ten cities an attempt was made to identify problems which have been encountered or could be encountered at a hazardous materials transportation accident. A list of the most frequently mentioned problem areas include:

- Identification of hazardous materials involved in the accident;
- Visual assessment of container structural integrity;
- On-scene interagency communications; and
- Unavailability to technical assistance after normal business hours.

### 5.3.5 Innovative Approaches to Handling Hazardous Material Transportation Accidents

During the review of each community's contingency plan, hazardous materials training programs and specialized response personnel, gear and equipment, certain innovative approaches to handling hazardous materials transportation accidents were identified. These approaches are listed here in the hope that they may assist a community in their actual on-scene response activities. This information is intended for cities which may not have been aware of these approaches. In each situation where an innovative approach or equipment is cited a contact and telephone number in the representative city is noted.

1. Development of a procedures manual which contains guidelines for emergency response activities at hazardous materials transportation accidents.

An example of specific guidelines applicable to hazardous materials transportation accident emergency response activities can be found in Appendix E. For additional information on this subject contact Captain James Henry, Baltimore Fire Department (Fire Prevention Bureau) at (301) 396-5752.

2. Deployment of a fire house in the given metropolitan community which has the responsibility of responding exclusively to transportation accidents and other releases involving hazardous materials.

Information on an example of this fire house and its capabilities should be addressed to Chief Cooper, Nashville Fire Department, at (615) 383-8922.

3. Establishment of a cooperative agreement between a metropolitan government and the county (represented by the county fire department) and the State in which the state will provide emergency response equipment if the metropolitan area will be in charge of responding to hazardous materials incidents in a specified geographic region of the state.

This distribution of specialized response personnel, gear and equipment is innovative in that it provides response capabilities on a regional basis rather than upon the individual purchasing power of a specific city. An example of this type of system and further information on this subject can be obtained from Chief Cooper of Nashville.

4. An area's integration into their hazardous materials response plan of the contact and 24-hour/day telephone number of specific types of specialized response equipment such as air compressor, bomb disposal, boom trucks, etc.

An example of an inventory of specialized response equipment in a contingency plan is available from Chief Cooper of Nashville.

5. A community's attempt to identify the location and storage capacity of chemical manufacturers as well as the quantity and type of hazardous material shipped through the area.

An example of a city which is attempting to identify the location and quantity of chemical manufacturers in the area is Newark, NJ. Comments should be addressed to Dr. Leonard Dauerman, Director of the Law and Technology Center of the New Jersey Institute of Technology (NJIT) at (201) 645-5522.

5-46

6. A metropolitan region which has been subdivided into 14 response regions, each having a fire chief who serves as on-scene coordinator if a hazardous materials transportation accident occurs. The area also has available for use on-scene a converted bus which can serve as a 10-passenger ambulance and a mobile command/support post.

For example, such capabilities are available in Pensacola, FL (Escambia County). Details of these capabilities can be obtained through contacting Buck Renfro, Coordinator of Civil Defense for Escambia County at (904) 436-9711.

7. Metropolitan areas having "cascade" systems which can replenish oxygen into breathing apparatus on-scene.

Two such areas are Pensacola, FL and Sacramento, CA. Contact either Buck Renfro of Pensacola, FL at (904) 436-9711 or Captain Jan Dunbar of Sacramento, CA at (916) 449-5267 for further information.

8. An area having reports for hazardous substances spills, response checklists and emergency resources found in their Disaster Operations Plan.

Samples of these reports can be found in Appendix F for Panama City (Youngstown, FL). Further information of this technical documentation can be obtained from Mr. Jim Heisler, Director of Defense Civil Preparedness of Bay County at (904) 769-2181.

## 5.4 RECOMMENDATIONS FOR IMPROVED MUNICIPAL METHODS FOR IMPLEMENTING CRISIS MANAGEMENT TECHNIQUES

Based upon the extensive review and assessment of methods for implementing crisis management techniques for hazardous materials transportation accidents i. the ten selected cities some deficiencies in response capabilities appear to be widespread while others tend to be localized in one or two cities. The recommendations made are general in nature as they apply to all cities and no provision has been made to address any city/area in specific. Based upon this review the following are recommendations which all cities/areas can adopt to provide improved methods for implementing crisis management techniques to address hazardous materials transportation accidents.

- 1. Even though there are special emergency response numbers, in the event of emergency situations the 911 emergency telephone system is of great value for emergency notification and communications.
- 2. Many cities need to develop hazardous materials contingency plans.
- 3. Specific guidelines should be developed for handling the hazards associated with the release of hazardous materials shipped through or stored in the given area.

4. Improved hazardous materials training programs and aids should be provided to municipal police personnel.

- 5. Cities/communities should regularly stage mock-up hazardous materials transportation accidents or other chemical releases as part of their continuous training and readiness programs.
- 6. Guidelines should be developed and refined for visually assessing the structural integrity of tanks involved in hazardous materials accidents.
- 7. Communications capabilities and coordination on-scene should be improved so that the OSC can be aware of the activities of all response organizations involved in hazards mitigation on-scene.
- 8. Fire service personnel should use positive pressure breathing apparatus when approaching the scene of a hazardous materials release.
- 9. Volunteer firefighters should be given the same hazardous materials training that paid personnel receive.
- 10. Emergency response personnel should be required to receive follow-up hazardous materials training after a specified period of time.
- 11. A designated communications network should be established which will provide for a mechanism to alert the public and to handle communications between the communications command center, the accident site and other off-scene support organizations.
- 12. Contingency plans should identify radio and television stations that will continuously inform the public during the initial phases of the emergency.
- 13. Contingency plans should provide for an update mechanism.
- 14. Contingency plans should provide for a method of identifying individuals who have authority cn-scene.
- 15. Contingency plans should give telephone numbers and names of personnel or organizations which would be required to respond on-scene.
- 16. It appears that cities preferred to carry the DOT <u>Emergency</u> <u>Response Guidebook</u>. However, a sufficient supply of these was not available in all the cities reviewed. Thus, every effort should be made to procure these manuals for initial notification and response purposes.
- 17. Cities/communities should provide some form of hazardous materials specific training.
- 18. Municipalities which do not have specialized personnel, equipment and gear for use on-scene may want to pool their resources for the purchase of needed resources.
- 19. Methods for improved access to technical assistance after normal bus sess hours should be obtained.

During the in-depth review of local methods for implementing crisis management techniques for hazardous materials transportation accidents it can be seen that all cities are not adequately prepared for handling HM accidents. One promising method that would provide greater response capabilities and could serve as a training tool is discussed in the next section. 6. CRITERIA FOR INTERACTIVE FEEDBACK CRISIS MANAGEMENT SYSTEM

This section presents criteria for an interactive feedback crisis management system. The eventual development of such a system could serve not only as a training tool for emergency response personnel to simulate the mitigation of hazardous materials transportation accidents, but it could be used by emergency response personnel at actual hazardous material accidents.

The first part of this chapter gives an overview and assessment of state-of-the-art (SOA) interactive feedback crisis management systems in existence; their capabilities and potential utility on-scene. Subsequently, criteria are developed for an interactive feedback crisis management system including hardware/software component requirements.

### 6.1 OVERVIEW AND ASSESSMENT OF SOA INTERACTIVE FEEDBACK CRISIS MANAGEMENT SYSTEMS

This assessment is necessary so that the adequacy or inadequacy of existing interactive systems utilized by emergency response personnel and other individuals can be determined. Interactive feedback systems can vary in complexity. The simplest may be for on-line data retrieval. The more sophisticated may be used as training simulators; some are actually used on-scene for management of the hazardous materials accident.

Information retrieval systems such as the Oil and Hazardous Material Technical Assistance Data Systems (OHMTADS), USCG's Hazardous Assessment Computer System (HACS), Chemical Transportation Emergency Center (CHEMTREC), etc. are not discussed here because as stated by the EPA "information retrieval sources should be considered secondary sources of information because information is from the published literature or past events, and because interaction is limited since the contact usually has no special expertise in spills technology or hazardous chemicals." Thus it is proposed that existing SOA interactive feedback systems, of the information retrieval type, do not at this time warrant additional development efforts because the data base information provided by these systems does appear adequate (i.e., except for predicting the vapor dispersion patterns of certain dense gases including topographic effects), and the sources of information are sufficient in number to meet the needs of Federal, state and local emergency response personnel as well as others involved in hazards mitigation activities on-scene. A detailed overview of SOA information retrieval systems and an assessment of their applicability to hazardous materials accident management is given in Section 3.1.5 of the "Draft Guidelines Manual" for the "Post-Accident Procedures for Chemicals and Propellants" project (F04611-80-C-0046).

6-1

The review of SOA interactive feedback systems demonstrated that the actual number of existing systems which have potential application to on-site hazards mitigation activities is quite limited. One system designed to provide an on-scene coordinator and other emergency response personnel information required on-scene is called the "Emergency Management System," developed by Information Systems, Inc. of Washington, D.C. This system consists of hardware and generic software programs for the following types of emergencies: fire; controlled or uncontrolled radioactive radiation release; airport/aircraft emergencies; military installations emergencies; local emergency operations centers; railroad and surface transportation emergencies; building emergencies (hotel, apartment, commercial); hospital/medical emergencies; and industrial manufacturing plant emergencies. The system does not currently have the capability for use by emergency response personnel at a hazardous materials transportation accident because of software limitations.

## 6.2 FEASIBILITY OF INTERACTIVE FEEDBACK SYSTEM AS A TRAINING AID/ACCIDENT SIMULATOR

#### 6.2.1 Interactive Feedback System Capabilities

Ń,

Ĭ

Interactive feedback systems, as implemented in computer graphics, have been used for manipulation of basic data presentations, i.e. bar chart, pie chart and network modelling. It is a fact that personnel utilizing interactive computer graphics systems can retrieve and segregate large data bases extremely rapidly. The graphics of displaying the data allows the operator to scan and mentally absorb trends, location, relative position, etc. of the displayed data. This information can effectively be used to aid in the development of a logical plan for managing an ongoing accident.

Part of the data to be utilized in a hazardous-material-response interactive graphics system should be locality specific. Examples of this may be the location of emergency response equipment, schools, population centers, unusual topographical features with respect to the designated locality. A locality could very well be a county area or an entire state. One visual display could depict the accident site and then on command one could superimpose the desired locality specific item. Also, the operator will be able to choose specific items that are to be displayed and reformat them in an alternative configuration on the visual graphic terminal.

Another requirement for the system software is the ability to plot vapor cloud dispersion with respect to the local topographical and meteorological conditions. This

cloud should be displayed on the graphics terminal to allow the operator to identify areas that may be encompassed by the anticipated movement of the vapor cloud and need to be evacuated. Population densities should be plotted in color on the visual graphic terminal and overlayed with the vapor cloud to allow emergency response personnel to easily determine where they can be most effective in reducing possible exposure to local residents from the hazardous vapor cloud. This can easily be expanded to include dangerous situations associated with possible explosion caused by the damaged tank car also it would portray impact areas that could be effected by "vent and burn" methods of dissipating hazardous material cargo.

### 6.2.2 <u>Feasibility of Interactive Feedback System as a Training Aid/Accident</u> <u>Simulator</u>

An interactive feedback system may have several applications to hazardous materials transportation accident mitigation activities. These applications include utility in community contingency pre-planning, training of civic and emergency response community leaders as well as a real-time accident emergency response tool.

Concerning the use of an interactive feedback system for community pre-planning purposes a system of this type would be useful in providing data base information to local officials who would be involved in pre-planning for hazardous materials transportation accidents. The types of information which would be beneficial to these individuals includes:

- Hazardous material shipping routes and volumes through community;
- Community transportation networks and designated evacuation routes;
- Location of specialized personnel, materials and equipment in the community adequate to handle hazardous material emergency;
- Listing of appropriate segments of the emergency response community, their responsibilities and authority;
- Sources of technical assistance for specific commodity emergencies;
- Listing of communications organizations in community (i.e., radio and television) which could alert the public of the accident and provide communications support between the communications command post and the accident site;
- Hazard mitigation guidelines for each known substance;
- Guidelines and areas designated for evacuation due to threat of toxic vapor dispersion (i.e., schools, hospitals and large population centers); and
- Recommended evacuation routes as a function of accident location and environmental, topographic, meteorological and demographic considerations.

Regarding the use of an interactive feedback system for training of civic and emergency response community leaders the system could provide great utility in the development and testing of alternative decision/response scenarios based on varying the hazardous material involved and environmental conditions. In the phase where an interactive feedback system is used in the development and testing of alternative decision/response systems it could be used for training emergency response personnel in handling the hazards associated with the accident as well as a simulation technique to represent an accident which has various hazardous materials, envirionmental, topographical, meteorological and demographic conditions.

For use in normative exercises an interactive feedback system would provide the benefit as a practical simulation system that follows a carefully designed logical sequence with a pre-specified "best" solution while at the same time giving interactive guidance to participants during the exercise to the "best" solution.

The normative exercise concept is shown graphically in Figure 6-1. The players can be thought of as the key personnel in charge of managing the emergency; these players would typically be fire chiefs, transportation officials, environmental protection officials, perhaps the local sheriff, and perhaps even the mayor.

These players would be located in the command post which for each state might be in the Office of Emergency Service (OES). In the normative exercise an accident environment is simulated at the command post to train the players how to respond. Each player submits information to the simulation control (SIMCON) on the part of the environment with which he is familiar, each through a series of terminals and graphics interfaces connected to a central processing unit of a computer. Out of this, the computer generates a strategy to guide the players in their response to the emergency, which eventually leads to the "good" solution. Now through the simulation tool, a perturbation is put into the environment; this causes the players to have to redefine their input to the CPU. From here the simulation control provides updated guidance to the players to put them on the path of the "good" solution. The perturbances to the environment might consist of a change in wind patterns (this could affect toxic gas dispersion and fire behavior), onset of heay rain, change from day to night when populations might migrate in or out of the accident zone, etc. It is expected that through this simulation tool the players in charge of the emergency response can be taken through man such scenarios or normative exercises thereby preparing them for proper response during an actual emergency. In the actual accident situation the environmental perturbation comes not from the simulation control but from feedback on

6-4



actual conditions and their fluctuations at the accident scene. The overall flow for the accident/response interaction is shown in Figure 6-2.

As a real-time accident emergency tool an interactive feedback system would provide ready access to existing on-line data bases; serve as an on-site interface between the data bases and communications command post; and provide increased capability and access to information for estimating downwind vapor concentrations; establishing evacuation radii based on population densities downwind from the accident site; etc.

Thus, it can be seen that an interctive feedback system can be a versatile tool for use in community contingency pre-planning, training and real-time emergency response. To implement such a system hardware and software need to be developed. The development of software will require extensive experience in hazardous materials response to transportation accidents, familiarity with local topography, demography and response capabilities available. Interactive computer hardware systems are available but relevant software must be developed.

### 6.3 POTENTIAL USERS OF INTERACTIVE FEEDBACK SYSTEM

This section identifies the individuals which would best be served by using an interactive feedback system in emergency response activities at hazardous materials transportation accidents. The time frame of accident mitigation in which the user would best be served in using such a system has also been identified. To establish both the user groups and the time considerations which would best be served, the NTSB investigated hazardous material accidents as well as the events at Mississauga in Ontario, Canada have been reviewed to identify the time at which specific organizations respond on-scene. The results of this analysis are shown in Figure 6-3.

The analysis showed that fire, police, industrial response teams, civil defense, highway patrol, sheriff, red cross, carriers representative, salvation army and other sources of technical assistance arrive on-scene within one and one-half hours after the accident occurs. It is also observed that it takes anywhere from 11-13 hours for representatives of the shippers and the Bureau of Explosives to arrive on-scene. Effort will be given in the development of guidelines to identify the optimum time period for which a system could most optimally provide utility to on-scene response personnel. However, the system should be available at all times during accident mitigation so that decision-makers can be postel as to changing on-site conditions and make proper restoration decisions based on the latest available data and technology.



100 90 80 70 60 50 40 30 20 MINUTES 10 9 8 7 6 5 4 3 2 1 Fire Sheriff نعا Industrial Response Team Police Fmergency Medical llighway Patrol Red Cross Other Technical Assistance Civil Defense Shippers Rep. Salvation Army Carriers Rep. AAR B of

FIGURE 6-3. RESPONSE TIMES OF VARIOUS EMERGENCY RESPONSE ORGANIZATIONS TO HM TRANSPORTATION ACCIDENTS

ŗ

### 6.4 CRITERIA FOR INTERACTIVE FEEDBACK CRISIS MANAGEMENT SYSTEM

This section identifies criteria for an interactive feedback crisis management system including hardware/software component requirements as well as training aid/simulation modes.

#### 6.4.1 Criteria for Hardware/Software Component Requirements

Previous discussion has shown that an interactive feedback system has many practical applications in hazardous materials accident management including use as an on-line data base information retrieval system for pre-planning, training of civic and emergency response community leaders as well as a real-time accident emergency tool. Hardware criteria for a system which could be used in pre-planning and training would be similar to existing components because these systems could be located in a fixed facility and consist of similar components. Hardware criteria for this system and applications should have the following characteristics:

- 1. Easily maintained;
- 2. Rechargeable power sources;
- 3. Cost-effective;
- 4. Readout should not require technical interpretation or computation;
- 5. Hard copy of the results should be available; and
- 6. Have a user oriented interface for ease of input.

In addition to the above criteria, criteria for an interactive feedback system as a real-time emergency tool must also have the following characteristics:

- 1. Portable (less than 50 lbs.) so that it can be carried over obstacles and irregular terrain easily to obtain accessibility;
- 2. Be able to function over a wide range of temperatures, vibration and environmental conditions;
- 3. Be intrinsically safe; and
- 4. Have constant monitoring capabilities

Concerning software requirements, the software should be developed in such a manner that the following criteria be met:

- 1. Be formatted in a computer language which is easily understood with a minimum of technical assistance and training.
- 2. Be based on actual accident experience and cover all phases of hazards mitigation from initial notification through resumption of normal operations.
- 3. Be interactive in the sense that the computer would prompt the user to make decisions on mitigating hazards.

- 4. Software is dynamic and can be easily updated based on varying data.
- 5. Data base be tailored on a community specific basis.
- 6. Various colors be used for differentiating data elements and quantities.
- 7. Minimal knowledge of computer programming is required to operate the system.
- 8. Video display should be subdivided into two sections. In the middle of the video screen will be a vertical line which will serve as a separator for the interactive question/answer mode displayed on the left of the line and a catalog of accident site conditions displayed on the right of the line.
- 9. The data base for which the software will be developed should include at a minimum the following parameters:
  - color of placard on tank or tank car for use in identifying materials hazard class;
  - commodity involved;
  - STCC and/or UN number of the material(s) involved;
  - specification cylinder, tank car, cargo tank or portable tank container involved in accident;
  - quantity of material being shipped;
  - source strength;
  - time since initial release;
  - leak/no leak conditions;
  - wind direction and speed;
  - precipitation/condensation conditions;
  - ambient temperature;
  - percent cloud cover;
  - proximity and location of population centers to accident site;
  - population density at or near accident site;
  - local topography;
  - arrival time and distance of wreckage removal, cleanup and disposal contractor to site (i.e., inventory of these groups is necessary);
  - location and type of neutralizing agent which should be used based upon the material released;
  - container structural integrity assessment; and
  - environmental pollution considerations.
- 10. Once the commodity has been identified the user should be provided with the material's LEL, UEL, TLV, PEL at the accident scene, boiling point, freezing point, critical temperature, critical pressure, critical density, vapor pressure and autoignition temperature.

11. Be developed so that the systems software can be made available to cities through which hazardous commodities are shipped.

Based upon these criteria, a responsive system could be developed which would be useful as a training tool, as a community pre-planning aid, as a real-time accident response tool and as a way to bring state-of-the-art technology to states and communities who need the information and assistance.

### APPENDIX A

### DOT-HAZARDOUS MATERIALS EMERGENCY RESPONSE GUIDE

### DOT RESPONSE GUIDE 26 ACETONE - UN 1090

### POTENTIAL HAZARDS

### Fire or Explosion

- Will burn. May be ignited by heat, sparks and flames.
- Flammable vapor may spread away from spill.
- Container may explode in heat of fire.
- Vapor explosion hazard indoors, outdoors or in sewers.
- Runoff to sewer may create fire or explosion hazard.

### Health Hazards

- Vapors may cause dizziness or suffocation.
- Contact may irritate or burn skin and eyes.
- Fire may produce irritating or poinsonous gases.
- Runoff from fire control or dilution water may cause pollution.

#### IN CASE OF ACCIDENT

- Keep unnecessary people away.
- Stay upwind; keep out of low areas.
- Isolate hazard area and deny entry.
- Wear self-contained breathing apparatus and full protective clothing.
- Isolate for 1/2 mile in all directions if tank or tank car is involved in fire.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.
- Also, in case of water pollution, call local authorities.

#### Fire

- Small Fires: Dry chemical, CO2, water spray or alcohol foam.
- Large Fires: Water spray, fog or alcohol foam.
- Move container from fire area if you can do it without risk.
- Stay away from ends of tanks.
- Cool containers that are exposed to flames with water from the side until well after fire is out.
- For massive fire in cargo area, use unmanned hose holder or monitor nozzles.
- Withdraw immediately in case of rising sound from venting safety device or discoloration of tank.

### DOT RESPONSE GUIDE 26 (continued) ACETONE - UN 1090

### Spill or Leak

- No flares, smoking or flames in hazard area.
- Stop leak if you can do it without risk.
- Use water spray to reduce vapors.
- Small Spills: Take up with sand, or other noncombustible absorbent material, then flush area with water.
- Large Spills: Dike far ahead of spill for later disposal.

- Move victim to fresh air; call emergency medical care.
- If not breathing, give artificial respiration.
- If breathing is difficult, give oxygen.
- In case of contact with material, immediately flush skin and eyes with running water for at least 15 minutes.
- Remove and isolate contaminated clothing and shoes.

### DOT RESPONSE GUIDE 55 ACETONE CYANOHYDRIN - UN 1541

### POTENTIAL HAZARDS

### Fire or Explosion

- Some of these materials may burn but do not ignite readily.
- Cylinder may explode in heat of fire.

### Health Hazards

- Poison.
- May be fatal if inhaled, swallowed or absorbed through skin.
- Contact may cause burns to skin and eyes.
- Runoff from fire control or dilution water may cause pollution.

### IN CASE OF ACCIDENT

- Keep unnecessary people away.
- Stay upwind; keep out of low areas.
- Isolate hazard area and deny entry.
- Wear positive pressure breathing apparatus and special protective clothing.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.
- Also, in case of water pollution, call local authorities.

### Fire

- Small Fires: Dry chemical, CO2, water spray or foam.
- Large Fires: Water spray, fog or foam.
- Move container from fire area if you can do it without risk.
- Fight fire from maximum distance.

#### Spill or Leak

- Do not touch spilled material.
- Stop leak if you can do it without risk.
- Use water spray to reduce vapors.
- Small Spills: Take up with sand, or other noncombustible absorbent material, then flush area with water.
- Small Dry Spills: Shovel into dry containers and cover; move containers; then flush area with water.
- Large Spills: Dike far ahead of spill for later disposal.

A-3

### DOT RESPONSE GUIDE 55 (continued) ACETONE CYANOHYDRIN - UN 1541

- Move victim to fresh air; call emergency medical care.
- If not breathing, give artificial respiration.
- If breathing is difficult, give oxygen.
- In case of contact with material, immediately flush skin or eyes with running water for at least 15 minutes.
- Speed in removing material from skin is of extreme importance.
- Remove and isolate contaminated clothing and shoes.
- Keep victim quiet and maintain normal body temperature.
- Effects may be delayed, keep victim under observation.

### DOT RESPONSE GUIDE 30 ACRYLONITRILE - UN 1093

### POTENTIAL HAZARDS

### Fire or Explosion

- Will burn. May be ignited by heat, sparks and flames.
- Flammable vapor may spread away from spill.
- Container may explode in heat of fire.
- Vapor explosion and poison hazardous indoors, outdoors or in sewers.
- Runoff to sewer may create fire or explosion hazard.

### Health Hazards

- Poison.
- May be fatal if inhaled, swallowed or absorbed through skin.
- Contact may cause burns to skin and eyes.
- Runoff from fire control or dilution water may cause pollution.

### IN CASE OF ACCIDENT

- Keep unnecessary people away; isolate hazard area and deny entry.
- Stay upwind; keep out of low areas.
- Wear positive pressure breathing apparatus and special protective clothing.
- Isolate for 1/2 mile in all directions if tank or tank car is involved in fire.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.
- Also, in case of water pollution, call local authorities.

### Fire

- Small Fires: Dry chemical, CO2, water spray or foam.
- Large Fires: Water spray, fog or foam.
- Stay away from ends of tanks.
- Do not get water inside container.
- Cool containers that are exposed to flames with water from the side until well after fire is out.
- For massive fire in cargo area, use unmanned hose holder or monitor nozzles.
- If this is impossible, withdraw from area and let fire burn.
- Withdraw immediately in case of rising sound from venting safety device or discoloration of tank.

### DOT RESPONSE GUIDE 30 (continued) ACRYLONITRILE - UN 1093

### Spill or Leak

- No flares, smoking or flames in hazard area.
- Do not touch spilled material.
- Stop leak if you can do it without risk.
- Use water spray to reduce vapors.
- Small Spills: Flush area with flooding amounts of water.
- Do not get water inside containers.
- Large Spills: Dike far ahead of spill for later disposal.

- Move victim to fresh air; call emergency medical care.
- If not breathng, give artificial respiration.
- If breathing is difficult, give oxygen.
- Remove and isolate contaminated clothing and shoes.
- In case of contact with material, immediately flush skin or eyes with running water for at least 15 minutes.
- Keep victim quiet and maintain normal body temperature.
- Effects may be delayed, keep victim under observation.

### DOT RESPONSE GUIDE 27

### ETHYL ACRYLATE - UN 1917 STYRENE - UN 2055 TOLUENE - UN 1294

### POTENTIAL HAZARDS

### Fire or Explosion

- Will burn. May be ignited by heat, sparks and flames.
- Flammable vapor may spread away from spill.
- Container may explode in heat of fire.
- Vapor explosion hazard indoors, outdoors or in sewers.
- Runoff to sewer may create fire or explosion hazard.

### Health Hazard

- Vapors may cause dizziness or suffocation.
- Contact may irritate or burn skin and eyes.
- Fire may produce irritating or poisonous gases.
- Runoff from fire control or dilution water may cause pollution.

#### IN CASE OF ACCIDENT

- Keep unnecessary people away.
- Stay upwind; keep out of low areas.
- Isolate hazard area and deny entry.
- Wear self-contained breathing apparatus and full protective clothing.
- Isolate for 1/2 mile in all directions if tank or tank car is involved in fire.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.
- Also, in case of water pollution, call local authorities.

#### Fire

- Small Fires: Dry chemical, CO2, water spray or foam.
- Large Fires: Water spray, fog or foam.
- Move container from fire area if you can do so without risk.
- Stay away from ends of tanks.
- Cool containers that are exposed to flames with water from the side until well after fire is out.
- For massive fire in cargo area, use unmanned hose holder or monitor nozzles.
- If this is impossible, withdraw from area and let fire burn.
- Withdraw immediately in case of rising sound from venting safety device or discoloration of tank.

### DOT RESPONSE GUIDE 27 (continued)

### ETHYL ACRYLATE - UN 1917 STYRENE - UN 2055 TOLUENE - UN 1294

### Spill or Leak

- No flares, smoking or flames in hazard area.
- Stop leak if you can do it without risk.
- Use water spray to reduce vapors.
- Small Spills: Take up with sand, or other noncombustible absorbent material, then flush area with water.
- Large Spills: Dike far ahead of spill for later disposal.

- Move victim to fresh air; call emergency medical care.
- If not breathng, give artificial respiration.
- If breathing is difficult, give oxygen.
- In case of contact with material, immediately flush skin or eyes with running water for at least 15 minutes.
- Remove and isolate contaminated clothing and shoes.

### DOT RESPONSE GUIDE 13 HYDROCYANIC ACID - UN 1051

### POTENTIAL HAZARDS

### Fire or Explosion

- Some of these materials are extremely flammable.
- May be ignited by heat, sparks and flames.
- Flammable vapor may spread away from spill.
- Container may explode in heat of fire.
- Vapor explosion and poison hazardous indoors, outdoors or in sewers.

### Health Hazards

- Poison; extremely hazardous.
- May be fatal if inhaled or absorbed through skin.
- Vapors non-irritating, deaden sense of smell.
- Runoff from fire control or dilution water may cause pollution.

### IN CASE OF ACCIDENT

- Keep unnecessary people away.
- Stay upwind; keep out of low areas.
- Isolate hazard area and deny entry.
- Wear positive pressure breathing apparatus and special protective clothing.
- Evacuate area endangered by gas (See Isolation and Evacuation Table in back of guidebook; find the material by name).
- Isolate for 1/2 mile in all directions if tank or tank car is involved in fire.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424–9300.

Fire

- Let burn unless leak can be stopped immediately.
- Small Fires: Dry chemical or CO2.
- Large Fires: Water spray, fog or foam.
- Move container from fire area if you can do it without risk.
- Stay away from ends of tanks.
- Cool container with water using unmanned device until well after fire is out.
- Isolate area until gas has dispersed.

### DOT RESPONSE GUIDE 13 (continued) HYDROCYANIC ACID - UN 1051

### Spill or Leak

- Do not touch spilled material.
- No flares, smoking or flames in hazard area.
- Stop leak if you can do it without risk.
- Use water spray to reduce vapors.
- Isolate area until gas has dispersed.

- Move victim to fresh air; call emergency medical care.
- If not breathng, give artificial respiration.
- If breathing is difficult, give oxygen.
- Remove and isolate contaminated clothing and shoes.
- In case of contact with material, immediately flush skin or eyes with running water for at least 15 minutes.
- Keep victim quiet and maintain normal body temperature.
- Effects may be delayed, keep victim under observation.

### DOT RESPONSE GUIDE 22 ISOBUTANE - UN 1969 PROPYLENE - UN 1077

### POTENTIAL HAZARDS

### Fire or Explosion

- Extremely flammable.
- May be ignited by heat, sparks and flames.
- Flammable vapor may spread away from spill.
- Container may explode in heat of fire.
- Vapor explosion hazard indoors, outdoors or in sewers.

### Health Hazards

- Vapors may cause dizziness or suffocation.
- Contact will cause severe frostbite.
- Fire may produce irritating or poisonous gases.

### IN CASE OF ACCIDENT

- Keep unnecessary people away.
- Stay upwind; keep out of low areas.
- Isolate hazard area and deny entry.
- Wear positive pressure breathing apparatus and full protective clothing.
- Isolate for 1/2 mile in all directions if tank or tank car is involved in fire.
  - FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.

### Fire

- Let burn unless leak can be stopped immediately.
- Small Fires: Dry chemical or CO2.
- Large Fires: Water spray, fog or foam.
- Move container from fire area if you can do it without risk.
- Stay away from ends of tanks.
- Cool containers that are exposed to flames with water from the side until well after fire is out.
- For massive fire in cargo area, use unmanned hose holder or monitor nozzles.
- If this is impossible, withdraw from area and let fire burn.
- Withdraw immediately in case of rising sound from venting safety device or discoloration of tank.

### DOT RESPONSE GUIDE 22 (continued)

### ISOBUTANE - UN 1969 PROPYLENE - UN 1077

### Spill or Leak

- No flares, smoking or flames in hazard afea.
- Do not touch spilled material.
- Stop leak if you can do it without risk.
- Use water spray to reduce vapors.
- Isolate area until gas has dispersed.

- Move victim to fresh air; call emergency medical care.
- If not breathng, give artificial respiration.
- If breathing is difficult, give oxygen.
- In case of frostbite, thaw frosted parts with water.
- Keep victim quiet and maintain normal body temperature.

### DOT RESPONSE GUIDE 28 METHANOL - UN 1230

### POTENTIAL HAZARDS

#### Fire or Explosion

- Will burn. May be ignited by heat, sparks and flames.
- Flammable vapor may spread away from spill.
- Container may explode in heat of fire.
- Vapor explosion and poison hazardous indoors, outdoors or in sewers.
- Runoff to sewer may create fire or explosion hazard.

#### Health Hazards

- Poison.
- May be fatal if inhaled, swallowed or absorbed through skin.
- Contact may cause burns to skin and eyes.
- Runoff from fire control or dilution water may cause pollution.

### IN CASE OF ACCIDENT

- Keep unnecessary people away; isolate hazard area and deny entry.
- Stay upwind; keep out of low areas.
- Wear positive pressure breathing apparatus and special protective clothing.
- Isolate for 1/2 mile in all directions if tank or tank car is involved in fire.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.
- Also, in case of water pollution, call local authorities.

Fire

- Small Fires: Dry chemical, CO2, water spray or foam.
- Large Fires: Water spray, fog or foam.
- Move container from fire area if you can do it without risk.
- Stay away from ends of tanks.
- Cool containers that are exposed to flames with water from the side until well after fire is out.
- Withdraw immediately in case of rising sound from venting safety device or discoloration of tank.

#### Spill or Leak

- No flares, smoking or flames in hazard area.
- Do not touch spilled material.
- Stop leak if you can do it without risk.
- Use water spray to reduce vapors.

### DOT RESPONSE GUIDE 28 (continued) METHANOL - UN 1230

- Small Spills: Take up with sand, or other noncombustible absorbent material, then flush area with water.
- Large Spills: Dike far ahead of spill for later disposal.

- Move victim to fresh air; call emergency medical care.
- If not breathng, give artificial respiration.
- If breathing is difficult, give oxygen.
- Remove and isolate contaminated clothing and shoes.
- In case of contact with material, immediately flush skin or eyes with running water for at least 15 minutes.
- Keep victim quiet and maintain normal body temperature.
- Effects may be delayed, keep victim under observation.
# DOT RESPONSE GUIDE 55 METHYL BROMIDE - UN 1062

# **POTENTIAL HAZARDS**

# Health Hazards

- Poison.
- May be fatal if inhaled, swallowed or absorbed through skin.
- Contact may cause burns to skin and eyes.
- Runoff from fire control or dilution water may cause pollution.

# Fire or Explosion

- Some of these materials may burn but do not ignite readily.
- Cylinder may explode in heat of fire.

# IN CASE OF ACCIDENT

- Keep unnecessary people away.
- Stay upwind; keep out of low areas.
- Isolate hazard area and deny entry.
- Wear positive pressure breathing apparatus and full protective clothing.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.
- Also, in case of water pollution, call local authorities.

# Fire

- Small Fires: Dry chemical, CO2, water spray or foam.
- Large Fires: Water spray, fog or foam.
- Move container from fire area if you can do it without risk.
- Fight fire from maximum distance.

# Spill or Leak

- Do not touch spilled material.
- Stop leak if you can do it without risk.
- Use water spray to reduce vapors.
- Small Spills: Take up with sand, or other noncombustible absorbent material; then flush area with water.
- **Small Dry Spills:** Shovel into dry containers and cover; move containers; then flush area with water.
- Large Spills: Dike far ahead of spill for later disposal.

# 

# METHYL BE MILE THE

# First Aid

- Move victim t fres +
- If not breathing goes as
- If breathing is with
- In case of contains, and an analyzing water for at image

. . . . .

- Speed in rem sons or a s
- Remove and is later
- Keep victin que san to s
- Effects may be the symbols

# DOT RESPONSE GUIDE 34 SODIUM HYDROSULFIDE - UN 2318

# POTENTIAL HAZARDS

# Fire or Explosion

- Will burn. May be ignited by heat, sparks and flames.
- May burn rapidly with flare-burning effect.

# Health Hazards

- Poisonous if swallowed.
- Skin contact poisonous.
- Contact may cause burns to skin and eyes.
- Runoff from fire control or dilution water may cause pollution.

# IN CASE OF ACCIDENT

- Keep unnecessary people away.
- Stay upwind; keep out of low areas.
- Isolate hazard area and deny entry.
- Wear positive pressure breathing apparatus and full protective clothing.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.
- Also, in case of water pollution, call local authorities.

# Fire

- Small Fires: Dry chemical, water spray or foam.
- Large Fires: Water spray, fog or foam.
- Move container from fire area if you can do it without risk.
- Cool containers that are exposed to flames with water from the side until well after fire is out.
- For massive fire in cargo area, use unmanned hose holder or monitor nozzles.
- If this is impossible, withdraw from area and let fire burn.

# Spill or Leak

- No flares, smoking or flames in hazard area.
- Do not touch spilled material.
- Small Dry Spills: Shovel into dry containers and cover; move containers; then flush area with water.
- Large Spills: Wet down with water and dike for later disposal.

# DOT RESPONSE GUIDE 34 (continued) SODIUM HYDROSULFIDE - UN 2318

# First Aid

- Move victim to fresh air; call emergency medical care.
- In case of contact with material, immediately flush skin or eyes with running water for at least 15 minutes.
- Remove and isolate ontaminated clothing and shoes.

# **DOT RESPONSE GUIDE 60** SODIUM HYDROXIDE - UN 1823

# POTENTIAL HAZARDS

# Fire or Explosion

- Some of these materials may burn but do not ignite readily.
- Explosive concentrations of gas may accumulate in tanks.
- Some of these materials may ignite combustibles (wood, paper, oil, etc.). Health Hazards

- Contact may cause burns to skin and eyes.
- If inhaled, may be harmful.
- Fire may produce irritating or poisonous gases.
- Runoff from fire control or dilution water may cause pollution.

# IN CASE OF ACCIDENT

- Keep unnecessary people away.
- Stay upwind; keep out of low areas.
- Isolate hazard area and deny entry.
- Wear positive pressure breathing apparatus and full protective clothing.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.
- Also, in case of water pollution, call local authorities.

# Fire

- Some of these materials may react violently with water.
- Small Fires: Dry chemical, CO2, water spray or foam.
- Large Fires: Water spray, fog or foam.
- Move container from fire area if you can do it without risk.
- Cool containers that are exposed to flames with water from the side until well after fire is out.

# Spill or Leak

- Do not touch spilled material.
- Stop leak if you can do it without risk.
- Take up with sand, or other noncombustible absorbent Small Spills: material, then flush area with water.

Small Dry Spills:

- Shovel into dry containers and cover; move containers; then flush area with water.
- Large Spills: Dike far ahead of spill for later disposal.

# DOT RESPONSE GUIDE 60 (continued) SODIUM HYDROXIDE - UN 1823

# First Aid

- Move victim to fresh air; call emergency medical care.
- Remove and isolate contaminated clothing and shoes.
- In case of contact with material, immediately flush skin or eyes with running water for at least 15 minutes.
- Keep victim quiet and maintain normal body temperature.

# DOT RESPONSE GUIDE 26 VINYL ACETATE - UN 1301

# POTENTIAL HAZARDS

# Fire or Explosion

- Will burn. May be ignited by heat, sparks and flames.
- Flammable vapor may spread away from spill.
- Container may explode in heat of fire.
- Vapor explosion and poison hazardous indoors, outdoors or in sewers.
- Runoff to sewer may create fire or explosion hazard.

# Health Hazards

- Vapors may cause dizziness or suffocation.
- Contact may irritate or burn skin and eyes.
- Fire may produce irritating or poisonous gases.
- Runoff from fire control or dilution water may cause pollution.

# IN CASE OF ACCIDENT

- Keep unnecessary people away.
- Stay upwind; keep out of low areas.
- Isolate hazard area and deny entry.
- Wear positive pressure breathing apparatus and full protective clothing.
- Isolate for 1/2 mile in all directions if tank or tank car is involved in fire.
- FOR EMERGENCY ASSISTANCE CALL CHEMTREC (800) 424-9300.
- Also, in case of water pollution, call local authorities.

Fire

- Small Fires: Dry chemical, CO2, water spray or foam.
- Large Fires: Water spray, fog or foam.
- Move container from fire area if you can do it without risk.
- Stay away from ends of tanks.
- Cool containers that are exposed to flames with water from the side until well after fire is out.
- For massive fire in cargo area, use unmanned hose holder or monitor nozzles.
- If this is impossible, withdraw from area and let fire burn.

• Withdraw immediately in case of rising sound from venting safety device or discoloration of tank.

# DOT RESPONSE GUIDE 26 (continued) VINYL ACETATE - UN 1301

# Spill or Leak

- No flares, smoking or flames in hazard area.
- Stop leak if you can do it without risk.
- Use water spray to reduce vapors.
- Small Spills: Take up with sand, or other noncombustible absorbent material, then flush area with water.
- Large Spills: Dike far ahead of spill for later disposal.

# First Aid

- Move victim to fresh air; call emergency medical care.
- If not breathng, give artificial respiration.
- If breathing is difficult, give oxygen.
- In case of contact with material, immediately flush skin or eyes with running water for at least 15 minutes.
- Remove and isolate contaminated clothing and shoes.

# APPENDIX B

# AAR - EMERGENCY HANDLING OF HAZARDOUS MATERIALS IN SURFACE TRANSPORTATION

# ACETONE FLAMMABLE LIQUID STCC 4908105 UN 1090

Acetone is a clear, colorless liquid with a pleasant odor. It is used to make other chemicals, in paint and nail polish removers, as a solvent. It is quite volatile and has a flash point of 0 deg. F. It is lighter than water and soluble in water. Its vapors are heavier than air.

# If Material On Fire or Involved in Fire

- Do not extinguish fire unless flow can be stopped
- Use water in flooding quantities as fog
- Solid streams of water may be ineffective
- Cool all affected containers with flooding quantities of water
- Apply water from as far a distance as possible
- Use 'alcohol' foam, carbon dioxide or dry chemical

# If Material Not on Fire and Not Involved in Fire

- Keep sparks, flames, and other sources of ignition away
- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary
- Attempt to stop leak if without hazard
- Use water spray to disperse vapors and dilute standing pools of liquid

# Personnel Protection

- Avoid breathing vapors
- Keep upwind
- Wear boots, protective gloves, and safety glasses
- Do not handle broken packages without protective equipment
- Wash away any material which may have contacted the body with copious amounts of water or soap and water

- If fire becomes uncontrollable or container is exposed to direct flame evacuate for a radius of 1500 feet
- If material leaking (not on fire), downwind evacuation must be considered

# ACETONE CYANOHYDRIN POISON B, COMBUSTIBLE ENVIRONMENTALLY HAZARDOUS SUBSTANCE (RO-104.54) STCC 4921401 UN 1541

Acetone cyanohydrin is a colorless liquid. It has a flash point of 165 deg. F. It slowly dissociates to acetone, a flammable liquid, and hydrogen cyanide, a flammable poisonous gas, under normal storage and transportation conditions. The rate of dissociation is increased by contact with alkalis and/or heat. It is lethal by inhalation and less readily by skin absorption. It is lighter than water and is soluble in water. Its vapors are heavier than air. Toxic oxides of nitrogen are produced during combustion of this material.

# If Material on Fire or Involved in Fire

- Do not extinguish fire unless flow can be stopped
- Use water in flooding quantities as fog
- Solid streams of water may be ineffective
- Cool all affected containers with flooding quantities of water
- Apply water from as far a distance as possible
- Use 'alcohol' foam, carbon dioxide or dry chemical

# If Material Not on Fire and Not Involved in Fire

- Keep sparks, flames, and other sources of ignition away
- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary
- Attempt to stop leak if without hazard
- Use water spray to disperse vapors and dilute standing pools of liquid

# Personnel Protection

- Avoid breathing vapors
- Keep upwind
- Wear self-contained breathing apparatus
- Avoid bodily contact with the material
- Do not handle broken packages without protective equipment
- Wash away any material which may have contacted the body with copious amounts of water or soap and water

# Evacuation

• If material leaking (not on fire), downwind evacuation must be considered

# ACETONE CYANOHYDRIN (cont'd)

Environmental Considerations - Land Spill

- Dig a pit, pond, lagoon, holding area to contain liquid or solid material
- Dike surface flow using soil, sand bags, foamed polyurethane, or foamed concrete
- Absorb bulk liquid with fly ash or cement powder

Environmental Considerations - Water Spill

- Use natural barriers or oil spill control booms to limit spill motion
- Use surface active agent (e.g., detergent, soaps, alcohols) to compress and thicken spilled material
- If dissolved, apply activated carbon at ten times the spilled amount in region of 10ppm or greater concentration
- Adjust ph to neutral (ph-7)
- Use mechanical dredges or lifts to remove immobilized masses of pollutants and precipitates

# ACRYLONITRILE FLAMMABLE LIQUID, POISONOUS POLYMERIZABLE ENVIRONMENTALLY HAZARDOUS SUBSTANCE (RQ-100/45.4) STCC 4906420 UN 1093

Acrylonitrile is a clear colorless liquid with a strong, pungent odor. It is used in insecticides and to make plastics, fibers and other chemicals. It has a flash point of 32. deg. F. It may polymerize if contaminated with strong bases or if the container is subject to heat, as in fire conditions. Prolonged exposure to the vapors or skin contact may result in death. It is lighter than water and is soluble in water. The vapors are heavier than air. Toxic oxides of nitrogen are produced during combustion of this material.

# If Material on Fire or Involved in Fire

- Do not extinguish fire unless flow can be stopped
- Use water in flooding quantities as fog
- Solid streams of water may be ineffective
- Cool all affected containers with flooding quantities of water
- Apply water from as far a distance as possible
- Use 'alcohol' foam, carbon dioxide or dry chemical

# If Material Not on Fire or Not Involved in Fire

- Keep sparks, flames, and other sources of ignition away
- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary
- Attempt to stop leak if without hazard
- Use water spray to disperse vapors and dilute standing pools of liquid

#### **Personnel** Protection

- Avoid breathing vapors
- Keep upwind
- Wear self-contained breathing apparatus
- Avoid bodily contact with the material
- Wear full protective clothing
- Do not handle broken packages without protective equipment
- Wash away any material which may have contacted the body with copious amounts of water or coap and and water

# ACRYLONITRILE (cont'd)

# Evacuation

- If fire becomes uncontrollable or container is exposed to direct flame evacuate for a radius of 2500 feet
- If material leaking (not on fire), downwind evacuation must be considered

Environmental Considerations - Land Spill

- Dig a pit, pond, lagoon, holding area to contain liquid or solid material
- Dike surface flow using soil, sand bags, foamed polyurethane, or foamed concrete
- Absorb bulk liquid with fly ash, cement powder, sawdust, or commercial sorbents
- Apply "universal" gelling agent to immobilize spill

# Environmental Considerations - Water Spill

- Use natural barriers or oil spill control booms to limit spill motion
- Use surface active agent (e.g., detergent, soaps, alcohols) to compressand thicken spilled material
- Inject "universal" gelling agent to solidify encircled spill and increase effectiveness of booms
- Add calcium hypochlorite
- If dissolved, apply activated carbon at ten times the spilled amount in region of 10ppm or greater concentration
- Use mechanical dredges or lifts to remove immobilized masses of pollutants and precipitates

# Environmental Considerations - Air Spill

- Apply water spray or mist to knock down vapors
- Combustion products include corrosive or toxic vapors

B-5

# ETHYL ACRYLATE, INHIBITED FLAMMABLE LIQUID, POLYMERIZABLE STCC 4907215 UN 1917

Ethyl acrylate is a clear colorless liquid with an acrid odor. It is used to make paints and plastics. It has a flash point of 60 deg. F. If the material is subjected to heat for prolonged periods or becomes contaminated it is subject to polymerization with evolution of heat. If the polymerization takes place inside a container the container may violently rupture. The material is lighter than water and slightly soluble in water. The vapors are heavier than air.

If Material on Fire or Involved in Fire

- Do not extinguish fire unless flow can be stopped
- Use water in flooding quantities as fog
- Solid streams of water may spread fire
- Cool all affected containers with flooding quantities of water
- Apply water from as far a distance as possible
- Use 'alcohol' foam, carbon dioxide or dry chemical

# If Material Not on Fire and Not Involved in Fire

- Keep sparks, flames nd other sources of ignition away
- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary
- Attempt to stop leak if without hazard
- Use water spray to disperse vapors and dilute standing pools of liquid

# Personnel Protection

- Avoid breathing vapors
- Keep upwind
- Wear boots, protective gloves, and safety glasses
- Do not handle broken packages without protective equipment
- Wash away any material which may have contacted the body with copious amounts of water or soap and water

- If fire becomes uncontrollable or container is exposed to direct flame evacuate for a radius fo 2500 feet
- If material leaking (not on fire), downwind evacuation must be considered

# HYDROCYANIC ACID SOLUTION POISON A, FLAMMABLE ENVIRONMENTALLY HAZARDOUS SUBSTANCE (RQ-10/4.54) STCC 4920130 UN 1613

Hydrocyanic acid, solution is hydrocyanic acid, a gas dissolved in water. It is a clear colorless liquid with a faint aromatic odor. It is flammable through the lower concentrations may require some effort to ignite. The vapor is lighter than air, but a flame can flash back to the source of the leak very easily. Lethal amounts may be absorbed through the skin as well as by inhalation.

# If Material on Fire or Involved in Fire

- Do not extinguish fire unless flow can be stopped
- Use water in flooding quantities as fog
- Cool affected containers with flooding quantities of water
- Apply water from as far a distance as possible
- Solid streams of water may be ineffective
- Use 'alcohol' foam, carbon dioxide or dry chemical

# If Material Not on Fire and Not Involved in Fire

- Keep sparks, flames, and other sources of ignition away
- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary
- Attempt to stop leak if without hazard
- Use water spray to knock-down vapors

# Personnel Protection

- Avoid breathing vapors
- Keep upwind
- Wear self-contained breathing apparatus
- Avoid bodily contact with the material
- Wear full protective clothing
- Do not handle broken packages without protective equipment
- Wash away any material which may have contacted the body with copious amounts of water or soap and water

- If fire becomes uncontrollable or container is exposed to direct flame ~ evacuate for a radius of 2500 feet
- If material leaking (not on fire), evacuate for a radius of 2500 feet

# HYDROCYANIC ACID SOLUTION (cont'd)

# Environmental Considerations - Land Spill

- Dig a pit, pond, lagoon, holding area to contain liquid or solid material
- Dike surface flow using soil, sand bags, foamed polyurethane, or foamed concrete
- Absorb bulk liquid with fly ash or cement powder

# Environmental Considerations - Water Spill

• Neutralize with agricultural lime (slaked lime), crushed limestone, or sodium bicarbonate

# Environmental Considerations - Air Spill

ſ

- Apply water spray or mist to knock down vapors
- Vapor knockdown water is corrosive or toxic and should be diked for containment





# ISOBUTANE FLAMMABLE GAS STCC 4905747 UN 1075

Isobutane is a colorless gas with a faint petroleum like odor. It is shipped as a liquefied gas under its vapor pressure. Contact with the liquid can cause frostbite. It is easily ignited. Its vapor is heavier than air and a flame can flash back to the source of leak very easily. The leak can either be a liquid or vapor leak. It can asphyxiate by the displacement of air. Under fire conditions the cylinders or tank car may violently rupture and rocket.

# If Material on Fire of Involved in Fire

- Do not extinguish fire unless flow can be stopped
- Use water in flooding quantities as fog
- Cool all affected containers with flooding quantities of water
- Apply water from as far a distance as possible

# If Material Not on Fire and Not Involved in Fire

- Keep sparks, flames, and other sources of ignition away
- Keep material out of water sources and sewers
- Attempt to stop leak if without hazard
- Use water spray to knock-down vapors

# Personnel Protection

- Avoid breathing vapors
- Keep upwind
- Wear protective gloves and safety glasses
- Do not handle broken packages without protective equipment
- Approach fire with caution

- If fire becomes uncontrollable or container is exposed to direct flame evacuate for a radius of 2500 feet
- If material leaking (not on fire), downwind evacuation must be considered

# SODIUM HYDROSULFIDE SOLUTION CORROSIVE METAL, BASIC ENVIRONMENTALLY HAZARDOUS SUBSTANCE (RQ-5000/2270) STCC 4935268 NA 2922

Sodium hydrosulfide in solution is a colorless to light yellow colored liquid. It is used in paper pulping, manufacturing dyes and dehairing hides. It is soluble in water. It is corrosive to metals and tissue.

#### If Material on Fire of Involved in Fire

- Extinguish fire using agent suitable for type of surrounding fire (Material itself does not burn or burns with difficulty)
- Use water in flooding quantities as fog-
- Apply water from as far a distance as possible

# If Material Not on Fire and Not Involved in Fire

- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary

#### Personnel Protection

- Avoid breathing vapors or dusts
- Avoid bodily contact with the material
- Wear boots, protective gloves, and goggles
- Do not handle broken package, without protective equipment
- Wash away any material which may have contacted the body with copious amounts of water or soap and water
- If contact with the material anticipated, wear full protective clothing

#### **Environmental Considerations - Land Spill**

- Dig a pit, pond, lagoon, holding area to contain liquid or solid material
- Dike surface flow using soil, sand bags, foamed polyurethane, or foamed concrete
- Absorb bulk liquid with fly ash or cement powder

#### Environmental Considerations - Water Spill

- Add soda ash
- Allow to aerate
- Use mechanical dredges or lifts to remove immobilized masses of pollutants and precipitates

# Environmental Considerations - Air Spill

- Apply water spray or mist to knock down vapors
- Evolves flammable hydrogen sulfide gas on contact with acids

# SODIUM HYDROXIDE LIQUID CORROSIVE MATERIAL, BASIC ENVIRONMENTALLY HAZARDOUS SUBSTANCE (RQ-1000/454) STCC 4935240 UN 1824

Sodium hydroxide liquid is the water solution of sodium hydroxide. It is used in chemical manufacturing, petroleum refining, paper making, cleaning compounds, and for many other uses. The concentrated solutions will dissolve in additional water with the evolution of heat. It is corrosive to metals and tissue.

# If Material Involved in Fire

- Extinguish fire using agent suitable for type of surrounding fire (Material itself does not burn or burns with difficulty).
- Use water in flooding quantities as fog
- Apply water from as far a distance as possible

# If Material Not Involved in Fire

- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary

# **Personnel Protection**

- Avoid breathing vapors or dusts
- Avoid bodily contact with the material
- Wear boots, protective gloves and safety glass
- Do not handle broken packages without protective equipment
- Wash away any material which may have contacted the body with copious amounts of water or soap and water
- If contact with the material anticipated, wear full protective clothing

# Environmental Considerations - Land Spill

- Dig a pit, pond, lagoon, holding area to contain liquid or solid material
- Dike surface flow using soil, sand bags, foamed polyurethane, or foamed concrete
- Absorb bulk liquid with fly ash or cement powder
- Neutralize with vinegar or other dilute acid

Environmental Considerations - Water Spill

• Neutralize with dilute acid or removable strong acid

Environmental Considerations - Air Spill

Apply water spray or mist to knock down vapors

# STYRENE MONOMER INHIBITED FLAMMABLE LIQUID, POLYMERIZABLE ENVIRONMENTALLY HAZARDOUS SUBSTANCE (RQ-1000/454) STCC 4907265 UN 2055

Styrene monomer inhibited is a clear colorless liquid with an aromatic odor. It is used to make plastics, paints, and synthetic rubber, and to make other chemicals. It has a flash point of 90 deg. F. Its vapors are irritating to the eyes and mucous membranes. If it becomes contaminated or is sujected to heat, it may polymerize. If the polymerization takes place inside a container, the container is subject to violent rupture. It is lighter than water and insoluble in water. Its vapors are heavier than air. If Material on Fire or Involved in Fire

# • Do not extinguish fire unless flow can be stopped

- Use water in flooding quantities as fog
- Solid streams of water may spread fire
- Cool all affected containers with flooding quantities of water
- Apply water from as far a distance as possible
- Use 'alcohol' foam, carbon dioxide or dry chemical

# If Material Not on Fire and Not Involved in Fire

- Keep sparks, flames, and other sources of ignition away
- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary
- Attempt to stop leak if without hazard
- Use water spray to knock-down vapors

# **Personnel Protection**

- Avoid breathing vapors
- Keep upwind
- Wear boots, protective gloves and safety glasses
- Do not handle broken packages without protective equipment
- Wash away any material whch may have contacted the body with copious amounts of water or soap and water

- If fire becomes uncontrollable or container is exposed to direct flame evacuate for a radius of 2500 feet
- If material leaking (not on fire), downwind evacuation must be considered

# STYRENE MONOMER INHIBITED (cont'd)

# Environmental Considerations - Land Spill

- e Dig a pit, pond, lagoon, holding area to contain liquid or solid material
- Dike surface flow using soil, sand bags, foamed polyurethane, or foamed concrete
- Absorb bulk liquid with fly ash, cement powder, sawdust, or commercial sorbents
- Apply "universal" gelling agent to immobilize spill
- Apply fluorocarbon-water foam to diminish vapor and fire hazard

# Environmental Considerations - Water Spill

- Use natural barriers or oil spill control booms to limit spill motion
- Use surface active agent (e.g., detergent, soaps, alcohols) to compress and thicken spilled material
- Inject "universal" gelling agent to solidify encircled spill and increase effectiveness of booms
- If dissolved, apply activated carbon at ten times the spilled amount in region of 10ppm or greater concentration
- Remove trapped material with suction hoses
- Use mechanical dredges or lifts to remove immobilized masses of pollutants and precipitates

# Environmental Considerations - Air Spill

Apply water spray or mist to knock down vapors

# PROPYLENE FLAMMABLE GAS STCC 4905782 UN 1075

Propylene is a colorless gas with a faint petroleum like odor. It is used to make other chemicals. It is shipped as a liquefied gas under its own vapor pressure. For transportation it may be stenched. Contact with the liquid can cause frostbite. It is easily ignited. Its vapors are heavier than air, and a flame can flash back to the source of leak very easily. This leak can be either a liquid or vapor leak. It can asphyxiate by the displacement of air. Under fire conditions the cylinders or tank cars may violently rupture and rocket.

# If Material on Fire or Involved in Fire

- Do not extinguish fire unless flow can be stopped
- Use water in flooding quantities as fog
- Cool all affected containers with flooding quantities of water
- Apply water from as far a distance as possible

# If Material Not on Fire and Not Involved in Fire

- Keep sparks, flames, and other sources of ignition away
- Keep material out of water sources and sewers
- Attempt to stop leak if without hazard
- Use water spray to knock-down vapors

# **Personnel Protection**

- Avoid breathing vapors
- Keep upwind
- Wear protective gloves and goggles
- Do not handle broken packages without protective equipment
- Approach fire with caution

- If fire becomes uncontrollable or container is exposed to direct flame evacuate for a radius of 2,500 feet
- If material leaking (not on fire), downwind evacuation must be considered

# TOLUENE FLAMMABLE LIQUID ENVIRONMENTALLY HAZARDOUS SUBSTANCE (RQ-1000/454) STCC 4909305 UN 1294

Toluene is a clear colorless liquid with a characteristic aromatic odor. It is used in aviation and automotive fuels, as a solvent for many materials, and to make other chemicals. It has a flash point of 40 deg. F. It is lighter than water and insoluble in water. Its vapors are heavier than air.

If Material on Fire or Involved in Fire

- Do not extinguish fire unless flow can be stopped
- Use water in flooding quantities as fog
- Solid streams of water may spread fire ...
- Cool all affected containers with flooding quantities of water
- Apply water from as far a distance as possible
- Use 'alcohol' foam, carbon dioxide or dry chemical

# If Material Not on Fire and Not Involved in Fire

- Keep sparks, flames, and other sources of ignition away
- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary
- Attempt to stop leak if without hazard
- Use water spray to knock-down vapors

# Personnel Protection

- Avoid breathing vapors
- Keep upwind
- Wear boots, protective gloves, and safety glasses
- Do not handle broken packages without protective equipment
- Wash away any material which may have contacted the body with copius amounts of water or soap and water

#### Environmental Considerations - Land Spill

- Dig a pit, pond, lagoon, holding area to contain liquid or solid material
- Dike surface flow using soil, sand bags, foamed polyurethane, or foamed concrete
- Absorb bulk liquid with fly ash, cement powder, sawdust, or commercial sorbents
- Apply "universal" gelling agent to immobilize spill
- Apply fluorocarbon-water foam to diminish vapor and fire hazard

# TOLUENE (cont'd)

Environmental Considerations - Water Spill

- Use natural barriers or oil spill control booms to limit spill motion
- Use surface active agent (e.g., detergent, soaps, alcohols) to compress and thicken spilled material
- Inject "universal" gelling agent to solidify encircled spill and increase effectiveness of booms
- If dissolved, apply activated carbon at ten times the spilled amount in region of 10ppm or greater concentration
- Remove trapped material with suction hoses
- Use mechanical dredges or lifts to remove immobilized masses of pollutants and precipitates

Environmental Consideration - Air Spill

• Apply water spray or mist to knock down vapors

# VINYL ACETATE FLAMMABLE LIQUID, POLYMERIZABLE ENVIRONMENTALLY HAZARDOUS SUBSTANCE (RQ-1000/454) STCC 4907270 UN 1310

Vinyl acetate is a clear colorless liquid. It is used to make adhesives, paints and plastics. It has a flash point of 18 deg. F. Its vapors are irritating to the eyes and respiratory system. If it is subjected to heat or becomes contaminated it is subject to polymerization. If the polymerization takes place inside a container, the container is subject to violent rupture. It is lighter than water and alightly soluble in water. Its vapors are heavier than air.

#### If Material on Fire or Involved in Fire

- Do not extinguish fire unless flow can be stopped
- Use water in flooding quantities as fog
- Solid streams of water may spread fire
- Cool all affected containers with flooding quantities of water
- Apply water from as far a distance as possible
- Use 'alcohol' foam, carbon dioxide or dry chemical

#### If Material Not on Fire and Not Involved in Fire

- Keep sparks, flames, and other sources of ignition away
- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary
- Attempt to stop leak if without hazard
- Use water spray to disperse vapors and dilute standing pools of liquid

### Personnel Protection

- Avoid breathing vapors
- Keep upwind
- Wear boots, protective gloves, and safety glasses
- Do not handle broken packages without protective equipment
- Wash away any material which may have contacted the body with copious amounts of water or soap and water

- If fire becomes uncontrollable or container is exposed to direct flame evacuate for a radius of 2500 feet
- If material leaking (not on fire), downwind evacuation must be considered

#### Environmental Considerations - Land Spill

- Dig a pit, pend, lagoon, holding area to contain liquid or solid material
- Dike surface flow using soil, sand bags, foamed polyurethane, or foamed concrete
- Absorb bulk liquid with fly ash, cement powder, sawdust, or commercial sorbents
- Apply "universal" gelling agent to immobilize spill
- Apply fluorocarbon-water foam to diminish vapor and fire hazard

# Environmental Considerations - Water Spill

- Use natural barriers or oil spill control booms to limit spill motion
- Use surface active agent (e.g., detergent, soaps, alcohols) to compress and thicken spilled material
- Inject "universal" gelling agent to solidify encircled spill and increase effectiveness of booms
- If dissolved, apply activated carbon at ten times the spilled amount in region of 10ppm or greater concentration
- Remove trapped material with suction hoses
- Use mechanical dredges or lifts to remove immobilized masses of pollutants and precipitates

Environmental Considerations - Air Spill

Apply water spray or mist to knock down vapors

# METHYL BROMIDE, LIQUID (INCLUDING UP TO 2% CHLOROPICRIN) POISON B STCC 4921440 UN 1062

Methyl bromide is colorless liquid with a chloroform-like odor. Under most circumstances it is non-combustible. It is very slightly soluble in water. It is toxic by inhalation. Prolonged or repeated skin contact can cause severe burns and possible absorption of toxic quantities of the material.

# If Material Involved in Fire

- Extinguish fire using agent suitable for type of surrounding fire (Material itself does not burn or burns with difficulty)
- Use water in flooding quantities as fog
- Cool all affected containers with flooding quantities of water
- Use foam, carbon dioxide or dry chemical

# If Material Not Involved in Fire

- Keep material out of water sources and sewers
- Build dikes to contain flow as necessary
- Attempt to stop leak if without hazard
- Use water spray to knock-down vapors

#### Personnel Protection

- Avoid breathing vapors
- Keep upwind
- Wear self-contained breathing apparatus
- Avoid bodily contact with the material
- Wear full protective clothing
- Do not handle broken packages without protective equipment
- Wash away any material which may have contacted the body with copious amounts of water or soap and water

#### Evacuation

If material leaking (not on fire), downwind evacuation must be considered

# EPA-MANUAL FOR THE CONTROL OF HAZARDOUS MATERIALS SPILLS: VOLUME I

TABLE C-1 SPILLS ON LAND

TYPE	APPLICATION OR CONSTRUCTION HETHOD	IISK	ADVANTAGES	2	I SADVANTAGES
Díkes Earthun	Create with buildozer or earth- moving equipment to compact carth (height depends on earth type)	Flat or sloped surface	<ol> <li>Material on si</li> <li>Construct with common equipme</li> <li>Construct quice</li> </ol>	lie l h ent 2 ckly 3	<ul> <li>Natural permeability</li> <li>of soll</li> <li>Seepage through ground</li> <li>Surface composition of</li> <li>soll not suitable in</li> </ul>
Foared Polyurethane	Use trained personnel to construct	Hard, dry surfaces	l. Hold up to sev feet of water	/eral } (1) 2	. Leaks on wet ground . Hard to ohtain disper- sion device
Foatsed Concrete	Use trained personnel to construct	Flat ground Slow moving spill	l. Better adhesio substrates (c) shale/grass)	n co f lay/2	flard to obtain fram and dispersion device . Must set for a time period Will not hold high Will not hold high
Evacual Lon	Buildover or earthmoving equipment - line if possible	Spft ground Matural cavitation	l. Material on al 2. Gonstruct with common equipme	lte 1 1 2 2nt 2 3	. Move large amounts (1) . Move large amounts of material . Natural permeability of soil . Surface of soil not . Surface of soil not
Evacuation & Dikes	Buildozer or earthmoving equipment - line if possible	Soft ground	<ol> <li>Need less spac than separate</li> <li>Alterlat on si Austruct vitipme</li> <li>Construct vitipme</li> </ol>	ce 1 L ce 2 n t 3	Move large amounts of material Natural permeability of soil Surface of soil not suitable in all cases

APPENDIX C

C-1

# TABLE C-2 SPILLS IN WATER-HEAVIER THAN WATER SPILLS

TECTINI QUE	APPLICATION OR CONSTRUCTION METHOD	use -	ADVANTACES	DI SADVANTAGES
Naturai Encavarions and Dikea	None	Mere a natural barrier exists	No construction needed	Can't control the area which comtains the sigli
Construction of Excavation and Difice	Dedges: hydraulic or vacuum pumpa Divers with pumpa then place concrete or sand bags around to form dike if butum material ta not	If bottom can be moved	Material is on sire	<ol> <li>Mard to construct</li> <li>Stirred up bottom may cause dispersion and increased turbidity</li> </ol>

TABLE C-3 SPILLS IN WATER-SOLUBLE OR MISCIBLE SPILLS

ALCONT NUMBER

ŀ

CARLESS IN DURALE AND

PUTCHER DATE

••

NETWOD	AFPLICATION ON CONSTRUCTION MATERIALS	IISE	ADVANTAGES	DISADVANTAGES
Scaled boumes	Buome Device to anchor	Contain depth Limited volumes leaking containers	Contain entire depth of water	<ol> <li>Deployment difficult</li> <li>Not used for large bodies</li> <li>Difficult to get good seal(16)</li> </ol>
blveraton of Uncontaminated Flow	Earthmoving Equipment	Special area where Lopograph is right	l. Can put cleaned water into diverted stream 2. Used for flowing water	1. Difficult to move large amounts of early 2. Clear area needed 3. Impermeability of ground
Diversion of Contaminated Fiu	Block entrance with sandbags, sealed booms or dikes	Special arca where topography is right	l. Can put clean waler back into stream 2. Used for fluuing vater	<ol> <li>Ulfficult to move large amount of earth</li> <li>Clear area needed</li> <li>Impermeability of ground</li> <li>Adverse environmental</li> <li>Impact</li> </ol>
Gelling Agent (40)	Cels, Disperaion Devices, use experienced personnel	If small volumes	l. Stop flowing con- raninenl 2. Stop permeation	1. Nard to ubtain 2. Can't use in large area 3. Must haul to dispose
Containment of Earire Vaterhody	biking Materials Earthmoving Equipment Sandhaga, etc. Lining	For entirely contaminated area	<ol> <li>Can allow containment of a large waterhody</li> <li>Materials on site</li> <li>Easily constructed</li> </ol>	<ol> <li>Rot all waterhoules have containable overflow</li> <li>Permeability</li> <li>May be an unstable condition</li> </ol>

C-3

TABLE C-4 SPILLS IN WATER-FLOATING SPILLS

A STATE OF A

	APPLICATION OF CONSTRUCTION MATERIALS	USE	neference*	AWANTAGES	DISADVANTACES
	Varies, need deployment device	Not roo much current	CC-446-4 (41) P. 6-10 to 6-25	Used on large area. Many varieties	1. Only in vaves < 2-4 feet 2. Current speed < 0.7 hnuts
<b>V</b> etra	Weir à Boat	Calm	CC-446-4 (41) p.6-25	Nut easily clogged: Collecta & contains	Not used in rough water
Pariters Bariters	Alr cumpressor diffuser deployment method	Chily shallow vater	-02-446-4 (41) p-6-25	Do not create a physical barrier to vessels	l. Not in rough water 2. Only shallow water 3. Only thin layers of materials
Sp111 Ner Jing Methods	Chemicals on vater spray or prop. Wesh	To protect shure or other facilities	05-446-4 (41) p. 6-31 10 6-35	Useful in rough valer	1. Not eauly obtaineble 2. Not 100% effective

C-4

4

2

· · · · 1

# TABLE C-5 SPILLS IN AIR

ECIM I QUE	NETINO	ASE	ADVANTAGES	)   SAIN ANT AGES
list Knock Rum	Spray fine mint into air	Water soluble or low lying vepure	Removes hazard from air	Greate water pollution problem and must be contained in solution
ans or Blowers	Disperse air by directing blower Loward it	Very calm and sheltered areas	Can direct air away from populated areas	<ol> <li>Not at all effective if any wind</li> <li>Need large capacity of blowers</li> </ol>

3. Hard to control

C-5

suppress cyanide gas formation but not greater than pH 9 Remove any solids to land fill Be careful not to with NaOH. Option Coments Explosive - Careful to avoid NCN create strong -mell is flam-Raise pH to 2 produces cyanates which are less toxic direct contect volution or reaction moble and Neutralize with NaOH to pH B.5 Adsorb/neutralize to pH 7. C:10-100 1/5 Sol. Matl. Ad NaOH to PH residual react 30 min./ dis-charge to STP Add Ma<sub>2</sub>CO<sub>3</sub> to pH 7--air to 70% max. DO level neutralize to PH 7 with HCI C: 10-100 1/1 Add WaOH to pH 10 then add Treatment Specifications Add acid to pH 7/ EPA-SUGGESTED TREATMENT SCHEMES 8.5/adsorb/ sol. matl. INCI to a di scharge ň Ę 0°-́ щ acetic acid or dilute N2504 or MCI ĩ¥ + To STP 6 4 4 Treatment Scheme then HOCI Backwash Nach 6 Na2CO3 (J) Ξ Solids solids (م ال (<sup>1</sup>) Acrylonitrile When diluted/ 1. 15 . .+ ~ May require acclimatization Amenable To Biological Trmt. at Hunicipal STP may need to be acclimated ۰. When reduced When neutra-lized Acetune Cyanuhydr in Sodium V Hydrosulfide Hazardous Chemical Sodium Hydroxide

.

÷.

:

CALLER & REAL FOR STREET

C-6

.

EPA-SUGGESTED TREATMENT SCHEMES

ALC: NO. N. N. N.

•

. \_|

1

I.

t



C-7

.
~ 5~	0111110	14161110000	<u> </u>		 	
ACT				ACETONE	•	

Same and and

Density i Let watery liquid Cab Plasts and mixes with unter. Platunable, irretating report is gendered.

Stay uppend and use water spray to "Lasesh down" support. Soon of guilton sourcess and call for department. Lase people away, babater and sense discharged anarrach. Aveid reents reach and regar. Needly head backh and genetics conserve generation.

CALL FOR MEDICAL AID.

VAPOR

FLAMMABLE. Plashback along vapor Imili may occiw. Vapor may oxplate if ignited in on the chard oran. Estagouds with dry chromost, alcohol foom. ur tarban dueside Vator may be notificetore on for Coal espared functioner with water.

VADR limitating to syna, naar and threat. Winhaled, may cause difficult breathing or loss of count Noer to freah air. If breathing as suppord, give artificial requiration. If breathing a difficult, gave oxygen.

Desperous to equatic life in high concentrations. May be desperous of it enters water intakas. Notify local health and pollution control efficiels. Notify operature of nearby water intakas.

LIQUID forfacting to eyes. Not inticing to skin. IF IN EYES, hold cyclids open and flugh with plenty of water.

JUL L

Fire

Exposure

Water

Pollution

1. RESPONSE TO DISCHARGE

(Les Berennes Manuel manifester CB salide

Issue warning high flammability

3. CHEMICAL DESIGNATIONS

alibility CI

Deperce and Rush

3.1 Synanyme: Dimethyliketone 2-Propanone

Keton

3.3 Chemical Formula: CH-COCH-

3.4 IMCO/United Nations Numerical Bealgnation: 3 1/1090

5.1 Personal Protective Eaula

52 Symp

5.3 Treeton

3.2 Coast Guard Cam



6.3

...

5.6 6.7

7.3

7.4

12.3



-

÷

5.3 Treatment for Especiets: INHALATION if victim is overcome, remove to fresh air and coll a physician, administer anticical respiration if breathing is irregular or stopped. INGESTION: if victim has smallowed large amounts and is conscious and not having convulsions, induce vemiring and get medical help promptly; no specific antidete haven. SKIN wash will with user EYES. Fush with user iriminated for at least 15 min. Consult a physician 5.4 Toolatly by Inhalation (Threahold Limit Value); 1000 ppm 5.5 Shart-Term Inhelation Limits: 1000 ppm (ar J0 min 6.4 Toolatly by Inhalation Carde 11 (ar J0 have 16 kilden)

leading to dermutitis

at the Support

5.6 Tostally by Ingestion: Grade 1; LDu 3 to 15 g/kg (does

ms Following Exposure:

- 5.7 Loto Tostally: Not pertinent
   5.8 Vapor (Geo) Initiani Characteriolitas: 11 present in high concentrations, support covid moderate
   billion of concentrations and the files of concentration senation of the eyes or respiratory system Effect is temporary

5. HEALTH HAZARDS

ersonal Protoctivo Equipment: Organe vapor conster or oir-supplied mosh; synthetic rubber ploves: chemical safety goggles or face splosh shedd.

INHALATION: vapor writeling to eyes and muctuus membranes: acts as an unachetic in very high concentrations. INGESTION: low order of toxicity but very irritating to muctuus membranes. SKIN: prolonged excessive contact course defatting of the skin, possibly

- 5.9 Limite or Bolid tryteent Characterialiss: No separatelia hazard Productily hermions to the shin sure it is very volatile and evaporates quickly from the skin
- \$ 10 Other Threshold: 100 nom

D-1

2. LARELS

4. OBSERVABLE CHARACTERISTICS

4.1 Physical State (as shipped): Liquid

4.3 Odor: Sumtish: niceson, resembling that

of mon or frust; punpent; shurp, penetrating residual, hotonic, pleasant, non-residual

4.2 Color: Colorina

18 11

:

· . ·



D-2

· .

.

. . . . . .

ACY

# ACETONE CYANOHYDRIN \*

· · · ·

-

			بمواصية كالمتحقق فالقنصي كالمجمع فالشائك		ويستجرب والمستحد والجري تبراك المتحاف فأشاه المحجبة	
Avoin Addades Addades Avoin Vece of Sare of Sare of Sare of Sare of Sare of Sare Sare of Sare of Sare of Sare of Sare of Sare Sare of Sare of	Sonanyon antib - unkayoutable CONTACT WIT minal prostocin minal prostocin minal prostocin minal prostocin minal prostocin prostocin prostocin Martin Possfor Viger m MEAR C BRE Canabus	Watery Equil Plasts and mixes In LIQUID AND VA text with anti-make or uppy 10 "langth- material obstance control age Way OUS GASES ARE I or explored if gains OUS GASES ARE I or explored if gains	Colorium Mild, almond odor. with vote: Posonous reports produced. POR KEDP PEOPLE AWAY. mod broching opparties. down " report. Adda. PRODUCED WHEN HEATED. Is an enclosed area. WITH SELF-CONTAINED US. Is of the account impres.	6.1 6.3 6.4 6.5 6.6 6.7 6.9	E. FIRE MAZARDS     Plank Point: 165°F C C     Planmable Limits in Air:     2.2% – 12%     Pire Estinguishing Agents: Vision disuide     Pire Estinguishing Agents: Not to be Usad:     Not persinent     Special Masards of Combustion Products:     Taxic hydrogen cystude is penerated     when heated     Bahavior in First: Not persinent     Ignition Yamporoture: 1270°F     Electrical Masard: Not persinent     Burning Rate: Data na available	8. WATER POLLUTION     8.1 Aquelle Testelby: Data not available     8.2 Weitertowi Testelby: Not pertinent     8.3 Biological dargen Demand (BOD): Data not available     8.4 Poel Chain Concentration Potential: Data not available     9. SELECTED MANUFACTURERS     1. Alfrech Chemical Co 100 Wist St. Paul Ave
Exposur	Call FO VAPOR POSSON Intenting If breakin If breakin If breakin If breakin If breakin FOSON POSSON Posson Posson If breakin If break	In which water, day ch and containers with R MEDICAL AID. DUS FYS. INHALED, it woh air. IS has taopped, give a g bat stapped, give a DUS IP SWALLOWT shis and syss. DUS IP SWALLOWT shis and syss.	minal, visited from, or arbon davide. webs. webs. untificial responses. system. LD: pand abon. y of webs.	7.1 7.2 7.3 7.4 7.5 7.6	7. CHEMICAL REACTIVITY Reactivity with Water: No reaction Reactivity with Common Materials: No reaction Stability During Transport: Stabile Neutralizing Agents for Acido and Caustics: Not pertinent Polymeritation: Not pertinent Inhibitor of Polymeritation:	Milwaukee, Wii, 53233 2 Essiman Kodak Co Rochewier, New Yuok 13650 3 Rohm and Hass Co Inderendence Mall West Philadelphia, Pa. 10105
Water Pollutio	ARMFI IF SWA CON HARMFI May br d Natify lack Validy opt	In the second se	vision is UNCONSCIOUS OR HAVING vision is UNCONSCIOUS OR HAVING ing except keep vision varia. FE IN VERY LOW CONCENTRATIONS. efficials. efficials.		Noi pariseni	10. SHIPPING INFORMATION     10.1 Grades of Purity: 95–995     10.2 Storage Temperature: Ambient     10.3 Inert Atmosphere: No requirement     10.4 Venting: Pressure-vacuum
1. RESP chro Response issue warning Restrict acce Should be ret	MISE TO DISC Martune revenues prinsan s. marcal- human	MARGE . CO 446-4) IGRICHY	2. LABELS		11. HAZARD ASSESSMENT CODE (Bee many American Industry Contactor A-P-Q	13. PHYSICAL AND CHEMICAL PROPERTIES 13.1 Physical State at 18°C and 1 atm: Linuted 13.2 Mileouter Weight 55.11 13.3 Beiling Point at 1 atm: Decomposes 13.4 Providing Point: -5.8°F = -21°C = 252°A, 13.5 Critical Tampartonics, National States
3. CHEM 3.1 Synanyme: 2 3.2 Coort Guard 3.3 Chamleal Pa 3.4 IMCO-United Designation:	ICAL DESIGNI Ipha-Hydroxyn Metholuctor Compatibilith ynaehydrin whale: (CH)36 Nations Num 6 1/1541	ITIONS Induty romitrile Winde Classification ClOHyCN Initial	4. OBSERVABLE CHARACTERISTICS 4.1 Physical Blate (as shipped): Liquid 4.2 Colorics 4.3 Oddr: Churacterinik: diklinci, strong cyunide	12.1	Codo al Padoral Regulations: Paramous hujed or solid. Class B HAS Hasand Rating for Butk Water Transportation: Cotogory Rating Fire 1 Health Vapor Irritant 1 Legerd or Salid Irritant 2 Pessins 4 Water Pollocios	<ul> <li>13.6 Critical Pressure: Not pertinent</li> <li>13.7 Speaks Gravity: 0.925 at 25*C (liquid)</li> <li>13.8 Liquid Surteen Tendent: Not pertinent</li> <li>13.9 Liquid-Water Interfacial Tendent: Not pertinent</li> <li>13.10 Vegac (Geo) Speaks Gravity: Not pertinent</li> <li>13.11 Antio el Speaks biosts of Vegar (Geo)- (esc.) 1.074</li> <li>13.12 Labort Hood Vegarization: (est.) 28.5 stru/b = (47.5 cst.)</li> </ul>
<ul> <li>\$ 1 Personal Proj n less than victor sum: S.2 Symptoms Per Confesson a increased at S.3 Treatment for to fresh ar vomiting by affected at partice straa     </li> </ul>	octive Equipm 24 concentration volation between Howing Espon and accounting in the beginning in Capoorers Ca (Revour shout horsughly m dr in thoroughly m of water for	5. NEALTH MARE Air-supplied in Mis, rubber or plasti Maret Al low desage nad at later stages b it a physician for at mit atong and water with soop and water stih soop and water this strong solt wate	NAZARDS nask with conversion approved for use with acrylonitrile glows, cover goggliv, or face mask, rubber books; in the context symptoms may be waskness, handsches, with context symptoms may be waskness, handsches, with context symptoms and depth will usuality be ecome slow and gesping (cases of exposure INHALATION: remore victim & INGESTION: if ristim is consensus, induces r SKIN: remove contaminated clothing and wash EYES: hold cyclids apert and wash with continuous,	12.3	Aquatic Taucity	= 6.12 × 10 <sup>9</sup> J/kg 13.13 Hoot of Combustion: (exi 5 - 11.40) His/Ib = -6.400 csi g = -276 × 10 <sup>9</sup> J kg 13.14 Hoot of Decomposition: Not pertinent 13.15 Hoot of Polymonization: Not pertinent 13.16 Hoot of Polymonization: Not pertinent
if breathing administer i in evers min when its stir errives 5.4 Toshelty by tait 5.5 Shark-Tarm in 5.5 Shark-Tarm in 5.5 Cashtiy by tag 5.7 Late Toshelty: 5.8 Vager (Gas) in cuttorn dire	has wapped, gr myl nitrie by a myl nitrie by a ngth is span; o alation (Throw halation Limits sollans. Grade Data nat svoia front Charactio in The effect a	re artificial respirat reshing an ampele rregit artificial resp minime (realistic resp minime (realistic) hatd Limit Value); it Deta not shalab A betan 50 mg/kg ble Plattate: V spers of temporary	ion until physicial arrives. If victim is unconscious, in a class and holding is under his ness for 15 seconds region during this preceders. Regions ampute mill victim's condition improves or physician (mero) relate the even and respiratory system of present in high (Commerd on page 4.	5.0 5.10	S. NEALTH HAZ/ Liquid or Balid Irritant Characteristas: Causa expansis and they down scandary burns on len Oder Threahald: Data ant available	(Cannued as paper 1 and 6 IRDS: (Cont'd.) smaring of the skin and first-degree burn, in chile; g exposure

.



ACN

## ACRYLONITRILE

•

Γ						
	Comment Internet	_			6. FIRE HAZARDS	8. WATER POLLUTION
1		Wetery Squid	Coloriaus to light yellow Arritating odd/	6.1	Flash Point: 30°F C.C.; 31°F O C Flammable Limits in Air:	8.1 Aquatic Testelly: i00 ppm/24 kr/all fish/100% killed/fresh
			income Annothic second is surfaced		3.05%	water 0.05-1 ppm/24 hr/biuegill/icihal/sak
L				J [ "	alcohol fourn, carbon dioxide	8.2 Wateriewi Texicity: Not pertiment
Γ	AVOID CONT.	ACT WITH LIQUID AND VA	NPOR. KEEP PEOPLE AWAY stus and rubher evertlothing (including gloves).	]   **	Fire Extinguishing Agents Not to be Unod: Water or foam may cause frothing	8.3 Biological Oxygen Demand (BOD):
Į	Sing discharge Sing discharge Siny upwind so	in search and call for orpur. If peaking nó we water spray to "knock		6.5	Special Hezords of Combustion Products: When heated or burned, ACN may	8.4 Food Chain Concentration Potential:
	Evarupte area a Isolate and rem	n care of large discharge. Here ducharged material with and ambutum contant are			evolve tozic hydrogen cyanide gas and ozides of nicrogen	None added
┢		FLANMABLE.			Bohavier in Fire: Vapor is heavier than air	
l		POISONOUS GASES MAY & Flashbock along report trail m	E PRODUCED IN FIRE ay essay.		to a source of ignition and flash buck. May notymerize and explode	9. SELECTED MANUFACTURERS
	Fire	Wear goggles, will-contained	I branching apparatus, and rubbes overclothing	67	Ignition Temperature: \$95°F	Electruchemicals Dept Witmmeron, Del 1989a
		Combat fires from a safe data Extinguish with dry chemical	nnce er provincied konsteen. 1, sloekel fusen, er carlinn diexide.		Electrical Hazard: Class I, Group D Burning Role: Data not available	2. Monsanio Co. Monsanio Roberto & Perus hemiscis Co.
		Water may be ineffective on a Coull exposed containers with	fire. Naster	1 I_		800 North Lindbergh Blvd
		CALL FOR MEDICAL AID.			7. CHEMICAL REACTIVITY	3. Vistron Corp
		POISONOUS IF INHALED.		7.1	Reactivity with Water: No reaction Reactivity with Common Meterials:	Cieveland, Unio 44115
	×	Move to frish sir. If breathing has stopped, give	artificial respiration.		Attacks copper and copper alloys; these metals should not be used. Penetrates	ξ
		LIQUID	9 9 <b>9 5 6 1</b> 1		leather, so contaminated leather shoes and gloves should be destroyed. Attacks	
	Exposure	POISONOUS IF SWALLOWE Invitating to skin and cycs.			uluminum in high concentrations	
		Flush affected areas with plet IF IN EYES, hold eyelids ope	ng ann anter. Ity of water. na and fluck with plenty of water.	7.4	Noviralizing Agents for Acids and	10. SHIPPING INFORMATION
		IF SWALLOWED and victim milk and have victim ind	is CONSCIOUS, have victim drink water or here viamitang.	7.5	Counter: Not pertinent Polymorization: May occur spontaneously	10.1 Grades or Purity: Technical 98-1001 10.2 Storage Temperature: Ambient
		CONVULSIONS, do not	a unconscious on his since hing encept heep victim warm.	11	in absence of oxygen or on exposure to visible light or excessive heat violently	10.3 Inort Atmosphere: No requirement
				J I	in the presence of alkali. Pure ACN is subject to self-poly merization	10.4 Venting: Pressure-sacuum
	Water	NARMFUL TO AQUATIC LI Fusing to showing.	FE IN VERY LOW CONCENTRATIONS.		with rapid pressure development. The commercial product is inhibited and not	
	Pollution	May be dangeraug if it others - Notify lucal health and wildli Notify attention of actricy wi	wster intekes. le officie <u>is</u> . stor intekes	7.8	subject to this reaction inhibitor of Polymorization:	
					Methylhydroquinone (35-45 ppm)	
	1. RESPONSE	TO DISCHARGE	Z LABELS		11. HAZARD ASSESSMENT CODE	13. PHYSICAL AND CHEMICAL PROPERTIES
	iter farmen billion Island war tufft St	n Mandhaan, C& aab,ay pisan; hughly		11	(See Heatre Assessment Hendbach, CO 446-31 A-P-O-R-S-Z	13.1 Physical State at 15°C and 1 atm: Liquid 13.2 Molecular Weight: 53.06
	Rememble Reserves access					13.3 Bolking Point of 1 atm: 17/°F = 77 4°C = 350 6°K
	Disperse and flush				·	13.4 Freezing Point:
					19 MATARD CLASSICICATIONS	
				12.1	12. HAZARD CLASSIFICATIONS Code of Poderal Regulations:	13.5 Critical Temperature: 503*F = 263*C = 536*N
3,1	3. CHEMICAL Synanyma: Cyanor	DESIGNATIONS sthylene	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (or shipped): Liquid	12.1 12.2	12. HAZARD CLASSIFICATIONS Cade of Paderal Regulations: Flammable liquid NAS Hazard Rolling for Built Water	12.5 Critical Temperature: S05*F = 263*C = 536*A 13.6 Critical Pressure: Addition at 64 MAx (m)
3,1	3. CHEMICAL Synanyma: Cysnor Fumpi Venton	DESIGNATIONS et hylene rene	4. OBSERVABLE CHARACTERISTICS 4.1 Physical Barb (so phipped): Liquid 4.2 Calor: Colorina 4.3 Other Huld surgery compliantly of	12.1 12.2	12. HAZARD CLASSIFICATIONS Code of Fodoral Regulations: Planmable liquid NA& Hazard Rolling for Built Water Transportation: Cotogory Rating	12.5         Critical Temperature: 503*F = 263*C = 536*h.           13.6         Critical Pressure: 600pus = 45 atm = 4.6 M.N.:m²           13.7         Specific Gravity: 0.8075 a. 20°C (injund)
3.1 3.2	3. CHEMICAL Synanyma: Cyanor Furnigi Venta Vinite Caaat Guard Comp	DESIGNATIONS sthylene roa yanide patibility Classification:	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (so chipped): Liquid 4.2 Color: Colorisus 4.3 Odor: Hild: pergent, recembing that of peach seed kernels	12.1	12. HAZARD CLASSIFICATIONS Code of Federal Regulations: Flammable liquid HAB Haard Reting for Built Woter Transportation: Cotegory Rating Fire	12.5 Gritical Temperature: 50*F = 26*C = 536*K 13.6 Gritical Pressure: 600 psis = 45 atm = 4 6 M N·m <sup>2</sup> 13.7 Specific Growity: 0 2075 a. 20°C (hyud) 13.8 Liquid Suffeet Tension: "sot pertinent 13.9 Liquid Suffeet Interfectal Tension:
3.1 3.2 3.3	3. CHEMICAL Synanyma: Cyanor Fumgr Vento Vinti c Caset Guord Comp Monon Chamical Parimula	DESIGNATIONS thylene yande panieleny Classification: nor c ChjarCHCN	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as phipped): Liquid 4.2 Calor: Coloriesi 4.3 Odor: Mild: pungani, resembling that of pasch und hornels	12.1	12. HAZARD CLASSIFICATIONS Code of Foderal Negulations: Flammable liquid NAS Haared Reting for Built Water Transportation: Category Roting Fire	12.5 Critical Temperature: 503*F = 263*C = 536*A. 13.6 Critical Pressure: 600 pise = 45 stm = 4.6 MN/m <sup>2</sup> 13.7 Specific Gravity: 0.8075 a. 30°C (liquid) 13.8 Liquid Surface Tension: Not pertinen 13.9 Liquid-Wate Interfacial Tension: Not pertinen 13.10 Veper (Gao) Specific Gravity: 1.8
3.1 3.2 3.3 3.4	3. CHEMICAL Synanyma: Cysoo Yonto Vinto Coost Guard Comy Monan Chomical Parmula: IMCO/United Holt Desiangker 31//	DESIGNATIONS Intrine yeande perificities perificities for c CH,=CHCN one Homorreal 09	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (an ohipped): Liquid 4.2 Color: Colorisus 4.3 Odor: Mid: purgent, resombling that of parch send kernels	12.1	12. HAZARD CLASSIFICATIONS Code of Poderal Regulations: Flammable liquid NAS Hazard Roling for Built Water Transportation: Cotogory Rating Fire	12.5 Critical Temperature: 50.75 = 20.17C = 536%. 13.6 Critical Pressure: 600 pite = 45 atm = 4.6 M.N.:m <sup>2</sup> 13.7 Specific Gravity: 0.8075 a. 20°C (liquid) 13.8 Liquid Suffees Temsies: Not pertinent 13.9 Liquid-Water Interfacial Temsies: Not pertinent 13.10 Veper (Geo) Specific Gravity: 1.8 13.11 Retie of Specific Hosts of Veper (GerV- 1.51)
3.1 3.2 3.3 3.4	3. CHEMICAL Synanyma: Cysloo Formy: Vonsic Vonsic Casot Guard Comp Monan Chemical Parthula INSCO/Jahled Notic Designation: 31/1	DESIGNATIONS II bytene rea yeande peribelling Closeffleation: ner c CH,=CHCN pris Numerical 093	<ol> <li>4. OBSERVABLE CHARACTERISTICS</li> <li>4.1 Physical State (so chipped): Liquid</li> <li>4.2 Color: Colories</li> <li>4.3 Odor: Hild: pergent, resembling that of peach seed kernels</li> </ol>	12.1	12. HAZARD CLASSIFICATIONS Code of Pederal Regulations: Plasmable liquid HAB Hazerd Reting for Built Weter Transportation: Cotegory Asting Fire	<ul> <li>12.5 Critical Temperature: 50°F = 263°C = 536°h.</li> <li>13.6 Critical Pressure: 600 pise = 45 atm = 4.6 M N·m²</li> <li>13.7 Specific Gravity: 0.8075 s. 20°C (injuid)</li> <li>13.8 Liquid Suffees Tension: Not pertinent</li> <li>13.9 Liquid-Write Intorfacial Tension: Not pertinent</li> <li>13.10 Vaper (Gas) Specific Gravity: 1.8</li> <li>13.11 Ratio of Specific Heats of Vapor (Gar's 1.151</li> <li>13.12 Latent Heat of Vaportation:</li> </ul>
3.1 3.2 3.3 3.4 5.1	3. CHEMICAL Synanyma: Cysaoc Fumigr Vintic Caset Guard Camp Mone Chemical Parmide INCO/United Nation Designation: 31/1 Persanal Protestive	DESIGNATIONS Ribytene read yende settbility ClosetMestion: ner c CH,=CHCN non Numerical 093 S. MEALTH s Equipment: Ar-uppha	4. OBSERVABLE CHARACTERISTICS     4.1 Physical State (as phipped): Leguid     4.2 Color: Colories     4.3 Oder: Mild: pungani, resembling that of     panch used Lemets     4.4 MAZARDS     mash, industrial chemical type, with approved consister	12.1	12. HAZARD CLASSIFICATIONS Code of Poderal Regulations: Planmable liquid NAS Haanrd Reting for Built Water Transportation: Cotegory Roting Fire	<ul> <li>13.5 Gritical Temperature: S03*F = 263*C = 536*K</li> <li>13.6 Gritical Pressure: 600 pise = 45 sim = 4.6 M/N·m<sup>2</sup></li> <li>13.7 Specific Growity: 0.8075 s. 20°C (liquid)</li> <li>13.8 Liquid Surface Tension: Not pertinent</li> <li>13.9 Liquid-Wate Interfacial Tension: Not pertinent</li> <li>13.10 Vaper (Gao) Specific Growity: 1.8</li> <li>13.11 Rotio of Specific Hoots of Vaper (Gar'- 1.15)</li> <li>13.12 Latent Hoot of Vapertastion: 265 Bic/16 = 147 cst/g = 6.16 X. (0° J. kg</li> <li>13.13 Most of Combustion: (cst.) - 9,900 B.u.16</li> </ul>
3.1 3.2 3.3 3.4 5.1	3. CHEMICAL Synanyma: Cysoo Fynny Venta Venta Caset Guard Comy Monan Chemical Fermids IMCO/Jahad Nelli Designation: J // Personal Protection for acrysontic to foc mach, robber	DESIGNATIONS staylone rea yoande posibility Closellestion: nor s Chi=CHCN one Numerical 093 S. MEALTH s Equipment: Air-supplied new (insi shan 2%) concentr toess, shich wur, slafe yo	OBSERVABLE CHARACTERISTICS     Observation (an onligonal): Liquid     Observation (an onligonal): Liquid     Observation (an onligonal): Liquid     Observation (an onligonal): Constant of     parach send korneth     MAZARDS     mask, industrial chemical type, with approved constant     ations, rindustrial chemical type, with approved constant	12.1	12. HAZARD CLASSIFICATIONS Code of Poderal Regulations: Plan make layed NAS Hazard Robing for Built Water Transportation: Category Rating Fire 3 Heakh Vapor Irritant 3 Luquid or Solid Irriant 1 Porsons 3 Water Potiution Human Tasicaty 4 Aquistic Tasicity 3 Ambetic Effect 2 Rescuraty Other Chemicals 3 Water o	$\label{eq:constraint} \begin{array}{l} \textbf{12.5} & Critical Temperature: \\ & S0^{3}F = 26.3^{2}C = 536^{2}K \\ \hline \textbf{13.6} & Critical Pressure: \\ & 600 pits = 45 stm = 46 M N \cdot m^{2} \\ \hline \textbf{13.7} & \textbf{Specific Gravity: 0 B075 s. CGC (injuid) \\ \hline \textbf{13.8} & Liquid Suffices Tension: Not pertinent \\ \hline \textbf{13.9} & Liquid-Water Interfacial Tension: \\ & Not pertinent \\ \hline \textbf{13.10} Veper (Geo) Specific Gravity: 1 s \\ \hline \textbf{13.11} & Retie of Specific Heats of Veper (Ger V I 1 S) \\ \hline \textbf{13.12} & Labort Heat of Vepertasten: \\ & 205 Bits/Ib = 147 cs1/g = 0 16 X (10^{2} J kg ) \\ \hline \textbf{13.13} & Heat of Combustien: (ect) = -9.003 Bits/Ib \\ & = -5.50 Cs1/g = -200 X (10^{2} J kg ) \\ \hline \textbf{13.14} & Heat of Decomposition: Not pertinent \\ \hline \end{array}$
3.1 3.2 3.3 3.4 5.1 5.2	3. CHEMICAL Synanyma: Cystoc Formgr Vense Vense Cacot Guard Comp Monan Chemical Partuda INCO/Junted Nath Designation: 31/1 Designation: 31/1 Parsanal Protective for scrylandrite u face mask, rubber Symptoma Pallowit vestione. Intide	DESIGNATIONS III bytone read yenide botholity CleaseMeasten: ner c CH.pCHCN botholity CLEAS NEALTH o Equipment: Air supplied to Chart has 2h concern both, such or put; softy be ng Exposure: Sumitar to th c supervise Sumitar to th composition Sumitar to th	4. OBSERVABLE CHARACTERISTICS     4. Observation (an ohipped): Liquid     4.2 Color: Colorisis     4.3 Odor: Hild: proposi, resembling that of     peach seed kernets     HAZARDS     mail. industrial chemical type, with approved consister     times, rubber or plastic gloves, cover gegins or     imal.     out on justice Similar type, milleret on may cause     mail. Industrial chemical type, milleret on the sime of the set	12.1	12. HAZARD CLASSIFICATIONS Code of Poderal Regulations: Flammable liquid NAS Hazard Roting for Built Water Trenepartolise: Cotegory Rating Fire 3 Heakh Vapor Irritan 3 Laquid or Solid Irrnan1 Person 3 Water Pollution Human Teascay 4 Aquasic Teascity 3 Amisbate Effect 2 Rescursty Other Chemicals 3 Water 0 Self-Rescuen 3	<ul> <li>12.5 Critical Temperature: S05°F = 263°C = 536°h.</li> <li>13.6 Critical Pressure: Bollowie = 45 atm = 4.6 M/N·m<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0.8075 a. 20°C (hquid)</li> <li>13.8 Liquid Suffices Temeses: Not pertinent</li> <li>13.10 Veper (Ges) Specific Gravity: 1.8</li> <li>13.11 Retie of Specific Gravity: 1.8</li> <li>13.12 Latent Heat of Veperization: 265 Bar/16 = 147 cat/g = 6.16 × 10° J. kg</li> <li>13.14 Heat of Decomposition: (ex.1 - 9.405 Bar.16 = -5.500 cat/g = -200 × 10° J. kg</li> <li>13.14 Heat of Decomposition: Not pertinent</li> <li>13.15 Heat of Decomposition: Not pertinent</li> <li>13.16 Heat of Decomposition: Not pertinent</li> </ul>
3.1 3.2 3.3 3.4 5.1 5.2	3. CHEMICAL Synanyma: Cysaos Fumigr Vintic Caset Guard Comp Moner Chemical Permide INCO/United Nett Designation: 31/1 Designation: 31/1 Personal Protestive for scrytomitric u foce mask, robbe Symptome Pallowi weakens. hosde smousts of Nett	DESIGNATIONS Rhytene rea pande pande pande pande collection rer c CH_PCHCN men Numericaal 003 S. MEALTH o Equipment, An -supplied n lew (loss laba 2%) concern to bass, sinch or sun; safay he ng Experients, sinch or sun; safay he suns worre crististist. The	4. OBSERVABLE CHARACTERISTICS     4. Obyelant State (so chipped): Liquid     4.2 Color: Colories     4.3 Odor: Mild: pungani, resembling thus of     panch send kernets      MAZARDS     MAZARDS     MAZARDS     MAZARDS     mash, industrial chemical type, with approved constant     ore of hydrogen cyanide. Vapor inhelation may cause     ind, have amount cause stinging and somatimes bioters.	12.1 12.2 12.3	12. HAZARD CLASSIFICATIONS     Coderal Regulations:     Flammable biguid     NAS Haarrd Reting for Built Water     Transportation:     Cotegory Roting     Fire 3     Heath     Vapor Irritant 3     Liquid or Solid Irritant 1     Persons 3     Water Pollution     Human Teaticity 3     Autibatic Effect 2     Rectivity 5     Chemicals 3     Water 0     Self-Reaction 3     Water     Category Cleastification	<ul> <li>12.5 Critical Temperature: 50°F = 263°C = 536°h.</li> <li>13.6 Critical Pressure: 600 pise = 65 stm = 4.6 M N·m<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0.8075 s. 20°C (injuid).</li> <li>13.8 Liquid Suffice Tension: Not pertinent.</li> <li>13.10 Vaper (Gae) Specific Gravity: 1.8</li> <li>13.10 Vaper (Gae) Specific Gravity: 1.8</li> <li>13.11 Ratio of Specific Hoots of Vaper (Gar'- 1.151</li> <li>13.12 Latent Hast of Vaportastian: 265 Blavite = 147 cal/g = 6.16 × 10°J kg.</li> <li>13.14 Hoot of Composition: (nc) J-%gb B.u.1b = -5.500 cal/g = -230 × 10°J kg.</li> <li>13.14 Hoot of Decomposition: Not pertinent.</li> <li>13.15 Hoot of Bolution: Not pertinent.</li> <li>13.16 Hoot of Polymortastion: Not pertinent.</li> <li>13.16 Hoot of Polymortastion: Not pertinent.</li> </ul>
3.1 3.2 3.3 3.4 5.1 5.2 8.3	3. CHEMICAL Synanyma: Cysao Fumgr Vanic Coort Guard Camp Monar Chemical Parmula INCO/United Natin Designation: 31/1 Personal Protostive for scriptontric un for scriptontric un for scriptontric un comas with syna Campacity and syna campact with syn Treatment for Easy Internet of Scriptontric	DESCRIATIONS ethylene read yende betibility Closetfleation: ner c CH,=CHCN one Numerical 093 S. MEALTN o Equipment: Air-upplied n two (trus than 2%) concentry beets, such or sust, sofery he op Exposures: Sundar to the crus absorbed through the til crus absorbed through the til crus absorbed through the til cruss software infraction i read meant in the concent	4. OBSERVABLE CHARACTERISTICS     4.1 Physical State (as phipped): Leguid     4.2 Celer: Colories     4.3 Oder: Mild: pungani, resembling that of     peach used Lemets     4.3 Moder: Mild: pungani, resembling that of     peach used Lemets     4.4 Moder: Mild: pungani, resembling that of     peach used Lemets     4.5 Moder: Mild: pungani, resembling that of     peach used Lemets     mash, industrial chemical type, with approved cansiler     ations, rubber or plastic gloves, cover poggles or     imal, used users and the state of the states     imal, industrial chemical type, with approved cansiler     imal, industrial chemical type, with approved cansiler     imal, industrial chemical figures     imal, industrial chemical figures     imal, industrial chemical figures     imal, industrial chemical figures     imal, industrial of an obtained between     imal, industrial canses, voming and obtainal plain     ment in necessary, call physician for all canses of appointe     where an outpace of residence matintem and when	12.1	12. HAZARD CLASSIFICATIONS         Coderal Regulations:         Flam mable liquid         NAB Hazard Roling for Burk Water         Trensportation:         Category         Rating         Fire         J         Vapor Irritant         J         Liquid or Solid Irritant         I Persons         J         Water Pollution         Human Tasicaty         Anibetic Effect         J         Water         Other Chemicals         J         Water         Self-Rescience         Self-Rescience         Health Histord (Bluec)         4         Assord Classifications:	<ul> <li>12.5 Gritical Temperature: S01*F = 261*C = 536*K.</li> <li>13.6 Gritical Pressure: 60 Doits = 45 stim = 46 M Nrm<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0 2075 a. 20°C (liquid).</li> <li>13.8 Liquid Surface Tension: Not pertinent.</li> <li>13.9 Liquid-Water Interfacial Tension: Not pertinent.</li> <li>13.10 Vaper (Gos) Specific Gravity: 1 a.</li> <li>13.11 Rolls of Specific Hosts of Vaper (Gor V- 1 151.</li> <li>13.12 Lationt Heat of Vaper(zotion: 265 Blu/lb = 147 cal/g = 6 16 X (0' J kg.</li> <li>13.13 Heat of Decemposition: Not pertinent.</li> <li>13.14 Heat of Decemposition: Not pertinent.</li> <li>13.15 Heat of Bolution: Not pertinent.</li> <li>13.16 Heat of Polymortization: Not pertinent.</li> </ul>
3.1 3.2 3.3 3.4 5.1 5.2 8.3	3. CHEMICAL Synanyme: Cysoo Fynny Vento Coort Guard Comy Monar Chemical Fermide IMCO/Johned Neth Designation: J 17 Personal Prototive for encysonative for encysonative uscharte. Neth Symptome Pallowi weaknes. Neth Symptome Pallowi weaknes. Neth Symptome Pallowi weaknes. Neth Somotive of the Symptome Pallowi weaknes. Neth Somotive of the Synanical Article.	DESIGNATIONS Rhytene read yanide by anide by anide by anide control to a second and a second read of the second and a second of the second and a second of the second and a second and the second and a second and the second and a second a second the second and a second a second a second the second a second a second the second a	4. OBSERVABLE CHARACTERISTICS     4.1 Physical State (as phipped): Liquid     4.2 Calor: Colories     4.3 Odor: Mild: purpent, resembling that of     parch cool kernels     4.4 Index (colories)     4.5 Parce (colorie	12.1	12. HAZARD CLASSIFICATIONS         Code of Poderal Regulations:         Flam mable layed         NAB Hazard Robing for Built Water         Trensportation:         Cotogory       Rating         Fire       3         Heakh       3         Vapor Irritant       1         Persons       3         Water Pollution       4         Aquater Tosticty       4         Aquater Effect       2         Rescrivity       0         Self-Rescript       3         HPPA Heazerd Classificationst:       6         Category       Classificationst         Category       Classificationst         Rescrivity       3         NHPA Heazerd Classificationst:       4         Plammobility (Red)       3         Rescrivity (Velow)       2	<ul> <li>12.5 Critical Temperature: S03*F = 263*C = 536*K.</li> <li>13.6 Critical Pressure: 600 pits = 45 stm = 4.6 M Nrm<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0.9075 s. 20°C (liquid)</li> <li>13.8 Liquid Suffices Tension: Not pertinent</li> <li>13.10 Veper (Geo) Specific Gravity: 1.8</li> <li>13.11 Rotie of Specific Gravity: 1.8</li> <li>13.12 Lationt Heat of Veperization: 265 Bis/Ib = 147 cat/g = 6.16 X. (07.1 kg 3.13 Heat of Decemberitien: Not pertinent</li> <li>13.14 Heat of Decemberitien: Not pertinent</li> <li>13.15 Heat of Decemberitien: Not pertinent</li> <li>13.16 Heat of Decemberitien: Not pertinent</li> <li>13.17 Heat of Decemberitien: Not pertinent</li> <li>13.18 Heat of Decemberitien: Not pertinent</li> <li>13.19 Heat of Polymorization: Not pertinent</li> <li>13.16 Heat of Polymorization: Not pertinent</li> </ul>
3.1 3.2 3.3 3.4 5.1 5.2 8.3	3. CHEMICAL Synanyma: Cylato Fumigr Vintic Caset Guard Camp Monen Chemical Period Monen Chemical Period Monen Chemical Period Monen Besignation: 31/1 Designation: 31/1 Design	DESIGNATIONS Ribytene read ) ande ballottiny Closelflastion: ner c CH <sub>2</sub> -CHCN new Numerical 09) S. MEALTH o Equipment: An 2-supplied n low (less the 2-supplied n low (less the 2-supplied n low (less the 2-supplied o to 2-supplied to 2-supplied to 2-supplied to 2	4. OBSERVABLE CHARACTERISTICS     4. Observation (an ohipped): Liquid     4.2 Color: Colories     4.3 Odor: Mild: pungoni, resembling that of     panch seed kernets     MAZARDS     maik. Industrial chemical type, with approved constern     there or plastic glores, cover geggins or     find, organ cyanide. Yapor inhelation may cause     in, lower amount cause stinging and sometime blotteric     vicin produces nauses, romting and sometimes blotteries     find an onyon or forth-inr-suppled mask when     nature noming by administering strang solution of     Y remove commanded cicking and subtaines     You anyon of (resh-inr-suppled mask when     nature comtamanated cicking and subtaines of     Y remove commanisted cicking and wash affacted     head cyatich again and wesh with continuous pantif	12.1	12. HAZARD CLASSIFICATIONS Code of Poderal Regulations: Flammable layoud MAB Hazard Roting for Built Water Trenepartolise: Cotogory Rating Fire 3 Heakh Vapor Irritant 3 Laquid or Solid Irrnani 1 Person 3 Water Pollution Human Teatcaty 4 Aquasic Teatcaty 3 Amisbatic Effect 2 Rescurity Cher Chemicals 3 Water 0 Self-Reaction 3 HFPA Heaard CloueNeators: Cotogory CloueNeatorsiton Heakh Hizard (Blue) 4 Plammability (Red) 3 Rescuring (Velow) 2	<ul> <li>12.5 Critical Temperature: 50°F = 28°FC = 536°N.</li> <li>13.6 Critical Pressure: 600 pile = 45 stm = 4.6 M/N·m<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0.8075 s. 20°C (liquid)</li> <li>13.8 Liquid Suffice Tension: Not pertinent.</li> <li>13.10 Veper (Ges) Specific Gravity: 1.8</li> <li>13.10 Veper (Ges) Specific Gravity: 1.8</li> <li>13.11 Ratie of Specific Hosti of Veper (GerV 1.15).</li> <li>13.12 Latent Host of Veperization: 265 Bir/16 = 147 cat/g = 6.16 × 10° J kg</li> <li>13.14 Host of Decomposition: Not pertinent.</li> <li>13.15 Host of Composition: Not pertinent.</li> <li>13.16 Host of Decomposition: Not pertinent.</li> <li>13.16 Host of Polymortastion: Not pertinent.</li> </ul>
3.1 3.2 3.3 3.4 5.1 5.2 8.3	3. CHEMICAL Synanyma: Cysao Fumgr Vintic Caset Guard Camp Moner Chemical Particle Chemical Particle Chemical Particle Designation: J // Designation: J // Designation: J // Designation: J // Partanal Protestive for acrylandric un for acrylandric un catact un of the area therapythy to areas any fait	DESCRIATIONS Ribytene read y ande bothbility Closelflastion: ner c Chy=CHCN nons Numerical 093 S. MEALTH a Equipment: An-supplied n low (less than 2%) concentr boots, sincker suit; safety be of are absorbed through the si chw, sneeting, abdomined pur- tives, sneeting, abdomined pur- station, sneeting, sneeting, abdomined pur- station, sneeting, abdomined pur- station	4. OBSERVABLE CHARACTERISTICS     4. Obyeleast State (as shipped): Liquid     4.2 Calor: Colories     4.3 Oder: Mild: pungani, rasembing thus of     panch used kerteck     4.3 Oder: Mild: pungani, rasembing thus of     panch used kerteck     4.4 Mild: pungani, rasembing thus of     panch used kerteck     and users of the second seconds.	12.1	12. HAZARD CLASSIFICATIONS         Coderal Regulations:         Flam make layout         NAB Hazard Rolling for Built Water         Trensportation:         Cotogery       Rating         Fire       3         Heakth       3         Uspor Irritant       1         Persons       3         Water Pollution       4         Human Tosicity       4         Aquatic Tosicity       3         Matter Collution       3         Water Collution       3         Water Collution       3         Water Collinicalis       3         Matter Chanicalis       3         Matter Chanicalis       3         Matter Chanicalis       3         Rescurry       Classification         Planmobility (Rad)       3         Reactivity (Velow)       2         S. NEALTH MAZ       5. NEALTH MAZ	<ul> <li>12.5 Critical Temperature: 50°F = 283°C = 536°h.</li> <li>13.6 Critical Pressure: 600 pise = 45 stm = 4.6 M N·m<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0.8075 s. 10°C (injuid)</li> <li>13.8 Liquid Surface Tension: Not pertinent.</li> <li>13.9 Liquid-Write Interfacial Tension: Not pertinent.</li> <li>13.10 Vaper (Gas) Specific Gravity: 1.8</li> <li>13.11 Ratio of Specific Hoots of Vaper (Gar'- 1.151</li> <li>13.12 Latent Hoot of Vaportastion: 265 Blavite = 147 cal/g = 0.16 x (07.1 kg</li> <li>13.14 Hoot of Depempention: Not pertinent.</li> <li>13.15 Hoot of Combustion: (nc1 - m, ND B.u.16 = -5,500 cal/g = -230 x (07.1 kg</li> <li>13.14 Hoot of Beindian: Not pertinent.</li> <li>13.15 Hoot of Beindian: Not pertinent.</li> <li>13.16 Hoot of Polymortastion: Not pertinent.</li> <li>14.6 Combustion: Not pertinent.</li> <li>15.7 Not pertinent.</li> <li>15.8 Cortical and the state of the</li></ul>
3.1 3.2 3.3 3.4 5.1 5.2 8.3	3. CHEMICAL Synanyma: Cysao Fumgr Vintic Cooket Guard Comy Monan Chemical Parthds IMCO/United Notic Designation: 31/1 Parsanal Prototility for scriptantitie un face mask, rubber Symptoms Pollowi weaknes, heste songuts of Heur contact with type Transmant Byr Elaps INIALATION: entering cattamin talt water, but on are a throughly un urasm of water for fir vature a set for cruch an anti i att Do na macrodi	DESCRIATIONS ethylane read yande setibility Closetfleation: ner c CH,=CHCN mon Numerical 093 S. MEALTH o Equipment: Air-uppide now (kess that 2%) concestr bats, sicker aut; sofety be op Exposures: Similar to the cruss severe irritation (ing or absorbed through the st cruss severe irritation (ing nerve the still or metacul treat remover serving for the st is of statism is developed treat remover serving and the st is of statism is developed treat rest anset, 15 mm cathong, pre-striftical resper- rite ampules in a cleah and he cruffictal respiration while du- tionation in the developed treat	4. OBSERVABLE CHARACTERISTICS     4. Obyeleast State (as oblepped): Liquid     4.2 Ceder: Coloriesi     4.3 Oder: Mild: pungent, resembling that of     peach used Lemets     4.3 Oder: Mild: pungent, resembling that of     peach used Lemets     4.4 Market and Lemets     4.5 Oder: Mild: State and Lemets     4.5 Oder: Mild: State and Lemets     4.5 Oder: Mild: State and Lemets     4.6 Market and Lemets     4.6 Market and Lemets     4.7 Oder: Mild: State and Lemets     4.8 Market and Lemets     4.8 Market and Lemets     4.8 Market and Lemets     4.9 Market     4.9 Market and Lemets     4.9 Market     4.9 Market and Lemets     4.9 Market     4.9 Market and Lem	12.3	12. HAZARD CLASSIFICATIONS     Coderal Regulations:     Planmable liquid     NAS Hazard Robing for Built Water     Trensportation:     Cotegory Rating     Fire 3     Heakh 3     Vapor Irritant 3     Liquid or Solid Irriant 1     Porsons 3     Water Pollution     Human Tasicay 4     Aquistic Tasicity 3     Aubietic Effect 2     Rescurity     Other Chamicals 3     Water Columnities 3     NFPA Heazerd (Blue) 4     Planmability (Rad) 3     Rescurity (Vellow) 2     S. NEALTH HAZ Vapor (Gae) Irritant Chargestantions	<ul> <li>12.5 Critical Temperature: S03*F = 263*C = 536*K.</li> <li>13.6 Critical Pressure: 600 pits = 45 stm = 4.6 M Nrm<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0.8075 s. 20°C (liquid)</li> <li>13.8 Liquid Suffices Tension: Not pertinent</li> <li>13.10 Veper (Geo) Specific Gravity: 1.8</li> <li>13.11 Rotie of Specific Gravity: 1.8</li> <li>13.12 Lationt Heat of Vepertasten: 205 Bic//b = 147 cat/g = 0.16 x (07 J. kg</li> <li>13.14 Heat of Desemberities: (nc) = -9.403 Bic//b kg</li> <li>13.15 Heat of Desemberities: (nc) = -9.403 Bic//b kg</li> <li>13.14 Heat of Desemberities: (nc) = -9.403 Bic//b kg</li> <li>13.15 Heat of Desemberities: Not pertinent</li> <li>13.16 Heat of Desemberities: Not pertinent</li> <li>13.18 Heat of Desemberities: Not pertinent</li> <li>13.19 Heat of Desemberities: Not pertinent</li> <li>13.10 Heat of Polymortization: Not pertinent</li> <li>13.16 Heat of Polymortization: Not pertinent</li> <li>13.16 Heat of Polymortization: Not pertinent</li> <li>13.17 Heat of Desemberities: Not pertinent</li> <li>13.18 Heat of Solution: Not pertinent</li> <li>13.19 Heat of Solution: Not pertinent</li> <li>13.10 Heat of Polymortization: Not pertinent</li> <li>13.11 Heat of Solution: Not pertinent</li> <li>13.12 Heat of Solution: Not pertinent</li> </ul>
3.1 3.2 3.3 3.4 5.1 5.2 8.3	3. CHEMICAL Synanyme: Cysaoc Fynny Venta Caoet Guard Comy Monar Chemical Fermide IMCO/Mohed Nelli Designation: J 17 Personal Prototilie foc missi, robbe Symptome Pallowi weaknes, heste Symptome Pallowi weaknes, heste Symptome Pallowi weaknes, heste Symptome Pallowi weaknes, heste Symptome Pallowi weaknes, heste Symptome Pallowi weaknes, heste Symptome Pallowi weaknes, heste Synanic and Statistic Synanic and Statistic S	DESIGNATIONS Rhytene rana yande panbelliny Closelflastion: ner c CH,=CHCN mas Numarisat 093 5. MEALTH o Equipment: Air-upplied now (issi has 2%) concerts to botts, sicker sun; safety he ng Engenure: Similar to th che, sinterne, addeminal by d are absorbed through the si remove victim to fresh ar ( next) success addeminal by d are absorbed through the si remove victim to fresh ar ( next) success addeminal by a set absorbed through the si remove victim to fresh ar ( next) success addeminal by a set absorbed through the si remove victim to fresh ar ( next) fresh and water & VES. to si levis 15 mes cathing, pre artificial respon- rite amoute in a clock and he attificial respiration which do timent until condition impre- on (Throught Lindt Values loss Lindt Lindt Values	A. OBSERVABLE CHARACTERISTICS     A. Developments Starte (an oblepped): Liquid     A.2 Cader: Coloriess     A.3 Oder: Mild: purgent, resembling that of     parch used hornels     MAZARDS     mash, industrial chemical type, with approved constart     atom, robber or plantic gloves, cover peggins or     inn, inver amount cover strange and automatime block     ore of hydrogen cyanide. Vapor inhelation may covie     need vorming Similar symptoms shown if large     into most its previous for all addeminal plan     most in necessary, call physician for all comes of apport     ward an appart of roh-in-trangelad mash when     nature commission of the sub-facient     hered most means taking in a sub-facient     hered most means taken in a creary minute     ing thy, Represe anyore when its aircengts in agent     may the approxement of the sub-facient     hered hypercamment of the sub-facient     hered hypercamment of the sub-facient     hered hypercampled mash in a facient     hered hypercampled mash in a second in     the sub-facient     and with physician and with centineous panils     atom unit physician arrives. If he is unconstroors,     did is under his news for 13 seconds in a creary minute     may thy, Represe anyone when its aircength is agent     more or physician arrives.     if 20 pem	12.1 12.2 12.3	12. MAZARD CLASSIFICATIONS     Coderal Regulations:     Flammabile layed     MAB Hazard Risting for Built Water     Trensportation:     Cotegory Rating     Fire 3     Heakth     Vapor Irritant 3     Liquid or Solid Irriant 1     Persons 3     Water Pollution     Human Teascay 4     Aquatic Teaticity 3     Amibatic Effect 2     Reactivity     Cher Chemissis 3     Water 0     Self-Reaction     Self-Reaction     Self-Reaction     Self-Reaction     Self-Reaction     Self-Reaction     Self-Reaction     S. NEALTH MAZ Vapor (Gao) Irritant Characteritation:	<ul> <li>12.5 Critical Temperature: 50°F = 28°FC = 536°h.</li> <li>13.6 Critical Pressure: 6000 µs = 45 stm = 4.6 M N·m<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0.8075 s. 20°C (liquid).</li> <li>13.8 Liquid Suffices Tension: Not pertinent.</li> <li>13.10 Vaper (Gas) Specific Gravity: 1.8</li> <li>13.11 Ratio of Specific Gravity: 1.8</li> <li>13.12 Latent Hash of Vaporization: 205 Bir/Ib = 147 cat/g = 0.16 × 10° J. kg.</li> <li>13.13 Heat of Department: Not pertinent.</li> <li>13.14 Heat of Department: Not pertinent.</li> <li>13.15 Heat of Department: Not pertinent.</li> <li>13.16 Heat of Department: Not pertinent.</li> <li>13.17 Heat of Department: Not pertinent.</li> <li>13.18 Heat of Departmention: Not pertinent.</li> <li>13.19 Heat of Departmentation: Not pertinent.</li> <li>13.16 Heat of Departmentation: Not pertinent.</li> <li>13.16 Heat of Polymorization: Not pertinent.</li> <li>13.16 Heat of Polymorization: Not pertinent.</li> <li>13.16 Heat of Departmentation: Not pertinent.</li> <li>13.16 Heat of Baintion: Not pertinent.</li> <li>13.16 Heat of Baintion: Not pertinent.</li> <li>13.16 Heat of Baintion: Not pertinent.</li> <li>13.16 Heat of Departmentation: Not pertinent.</li> <li>13.16 Heat of Departmentation: Not pertinent.</li> <li>13.16 Heat of Departmentation: Not pertinent.</li> <li>14 Heat of Departmentation: Not pertinent.</li> <li>15 Heat of Departmentation: Not pertinent.</li> <li>14 Heat of Departmentation: Not pertinent.</li> <li>15 Heat of Departmentation: Not pertinent.</li> </ul>
3.1 3.2 3.3 3.4 5.1 5.2 8.3 8.3 8.4 8.8 8.9	3. CHEMICAL Synanyma: Cysao Fumigr Vintic Cacet Guard Comy Moner Chemical Formula: INCO/United Nation Designation: 31/1 Designation: 31/1 Designation: 31/1 Designation: 31/1 Designation: 31/1 Designation: 31/1 Designation: 31/1 Designation: 31/1 Tractment Internet Contact with cym Contact with cym Cymbal with cym Cymbal with cym Cymbal with cym Cymbal with cym Cymbal with cym Cymbal with cym Cym Cym Cym Cym Cym Cym Cym Cym Cym C	DESCRIATIONS Ribytane read yande batibility Closelification: ner C CH_I=CHCN mon Numerical (00) S. MEALTH a Geutypment. Air-supplied in low (test than 2%) concern to the second state of the interpolation of the second state of the second state of the second state of the second state remove neum to fresh air ( neucle second state of the second state of the second state remove neum to fresh air ( neucle second state of the second state of the second state remove neum to fresh air ( neucle second state of the second state of the	4. OBSERVABLE CHARACTERISTICS     4. Obyelest State (so shipped): Liquid     4.2 Color: Colories     4.3 Oder: Mild: pungent, resembling that of     panch seed hereats     4.3 Oder: Mild: pungent, resembling that of     panch seed hereats     4.4 Market of the seed hereats     4.5 Oder: Mild: pungent, resembling that of     panch seed hereats     4.6 Market of the seed hereats     4.7 Market of the seed hereats     4.8 Market of the seed hereats     4.9 Oder: Mild: pungent, resembling that of     panch seed hereats     4.1 Market of the seed hereats     4.1 Market of the seed hereats     4.2 Market of the seed hereats     4.2 Market of the seed hereats     4.3 Oder: Mild: Yapor shelation may calle     ind, the seed of the seed to see of the seed of the seed to see of the seed of the seed to see of the seed of the seed of the seed to see of the seed to see of market of the seed to seed of the seed means     10 market of the seed of the seed means     10 market of the seed of the seed means     10 market of the seed to seed of the seed means     10 market of the seed to seed of the seed means     10 market of the seed to seed of the seed means     10 market of the seed to seed of the seed means     10 market of the seed to seed of the seed means     10 markets     10 permitted the seed of the seed means     10 market of the seed to seed of the seed means     10 market of the seed to seed of the seed means     10 market of the seed to seed of the seed means     10 markets     10 permitted the seed to seed of the seed means     10 market of the seed to seed to seed of the seed means     10 markets     10 permitted the seed to seed to seed of the seed means     10 markets     10 permitted the seed to seed to seed to seed to seed to the seed     10 mark	12.1 12.2 12.3	12. MAZARD CLASSIFICATIONS Code of Poderal Regulations: Flammabile layoud MAB Hazard Roting for Built Water Transpartation: Cotogory Rating Fire 3 Heakh Vapor Irritant 3 Laquid or Solid Irriant 1 Poison 3 Water Poliviton Human Teacity 4 Aquasic Teacity 3 Aastbatic Effect 2 Rescurity Cher Chemicals 3 Water 0 Self-Reaction 3 MRPA Heaard Clevelloadens: Cotogory Clevelloadens: Cotogory Clevelloadens: Cotogory Clevelloadens: Cotogory Clevelloadens: Cotogory Clevelloadens: Cotogory Clevelloadens: S. MEALTH MAZ Vager (Gao) Irritant Characteristics: 1/ point Laydor Solid Irritant Characteristics: 1/ point Laydor Solid Irritant Characteristics: 1/ point Category Clevelloadens: 1/ point Category Clevelloadens: Cotogory Clevell	<ul> <li>12.5 Critical Temperature: 50°F = 283°C = 536°h.</li> <li>13.6 Critical Pressure: 600 pile = 45 stm = 4.6 M N·m<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0.8075 s. 20°C (liquid)</li> <li>13.8 Liquid Suffice Tension: Not pertinent.</li> <li>13.9 Liquid-Water intofacial Tension: Not pertinent.</li> <li>13.10 Veper (Ges) Specific Gravity: 1.8</li> <li>13.11 Ratio of Specific Gravity: 1.8</li> <li>13.12 Latent Haat of Vepertastion: 265 Bis/the = 147 cat/g = 6.16 × 10° J kg</li> <li>13.13 Host of Combinest: Not pertinent.</li> <li>13.14 Heat of Competitions: Not pertinent.</li> <li>13.15 Heat of Combinest: Not pertinent.</li> <li>13.16 Heat of Solution: Not pertinent.</li> <li>13.16 Heat of Bolistion: Not pertinent.</li> <li>14.16 Heat of Bolistion: Not pertinent.</li> <li>15.17 Heat of Bolistion: Not pertinent.</li> <li>13.18 Heat of Bolistion: Not pertinent.</li> <li>14.18 Heat of Bolistion: Not pertinent.</li> <li>15.19 Heat of Bolistion: Not pertinent.</li> <li>16.10 Cont/d.)</li> <li>moderatels triticating such that persumber with not assatted on closing and Showed to remain, may cause minimum, bit aborted through the sun and cause</li> </ul>
3.1 3.2 3.3 3.4 5.1 5.2 8.3 8.3 8.4 8.6 8.6 8.7	3. CHEMICAL Synanyma: Cysao Fumgr Visio Coost Guard Camp Monor Chemical Partula: INCO/United Natin Designation: 31/1 Personal Protostive for scriptontrite inte O/United Natin Designation: 31/1 Personal Protostive for scriptontrite inte Columbus Synapteme Pederd weakness, hesta Synapteme Pederd weakness, hesta Synapteme Synapteme Pederd weakness, hesta Synapteme Sy	DESCRIATIONS Ribylane rana yande botibility Closefflastion: ner C Chy=CHCN one Numerical 003 S. MEALTH o Equipment: An-yuppide n low (less than 2%) concent where the second second second n low (less than 2%) concent where the second second second n low (less than 2%) concent remove such as to the char absorbed through the si cause severe stratistic Tage remove scient to fresh ar of need area if NGESTION - by of visitim is concensor. Ski i or at less 15 men cething, give artificial reagion resting, give artificial reagion restingent respectives while du- strificial respectives while du- strificial respectives while du- strificial respectives while du- strificial respectives to the du- tinem unit conditions impre- on (Throutabet Limble Values here Limble: 4% pain for 30 o m: Grade 3. L Do 30 o 300 ( not available	4. OBSERVABLE CHARACTERISTICS     4. Obyelest State (so chipped): Liquid     4.2 Color: Colories     4.3 Oder: Mild: pungent, resembling that of     peach used Lemets     4.3 Oder: Mild: pungent, resembling that of     peach used Lemets     4.4 MAZARDS     mash, industrial chemical type, with approved constent     mash, industrial chemical type, with approved constent     mash, industrial chemical type, with approved constent     ind.     original chemical type, with approved constent     ind.     industrial physician for applied mesh when     induce constances have any minute and with continuous pantial     ation until physician for the succonstoues,     id a under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind is under this new fort 15 seconds in every minute     ind physecond arrives.     if: 20 ppm	12.1 12.2 12.3 12.3 12.3	12. HAZARD CLASSIFICATIONS         Coder of Poderal Regulations:         Flam make layout         NAB Hazard Roling for Built Water         Trensportation:         Cotogory       Rating         Fire       3         Heakh       3         Vapor Irritant       1         Poisons       3         Water Pollution       4         Music Taicity       4         Aquisic Taicity       3         Asthetic Effect       2         Rescurity       3         Other Chamicals       3         Water       0         Self-Rescue       4         Planmebidity (Red)       3         Reactivity (Voliow)       2         S. HEALTH HAZ         Vapor (Gae) Invitant Characterstition:       Vapor (Characterstition:         Stef-Rescue on high vapor concentration         Livitic maderate characterstition:       Vapor (Characterstation:         Vapor (Gae) Invitant Characterstition:       Vapor (Characterstation:         Update of Solid Unitions Characterstation:       Vapor (Characterstation:         State maderate and high vapor concentration:       Update: a Solid Unition Characterstation:         Divisitic maderate: 21 4 open (Sense of smeth fatinge	<ul> <li>12.5 Critical Temperature: 50°F = 283°C = 536°h.</li> <li>13.6 Critical Pressure: 600 pise = 45 stm = 4.6 M N·m<sup>2</sup>.</li> <li>13.7 Specific Gravity: 0.8075 s. 20°C (injuid)</li> <li>13.8 Liquid Surface Tension: Not perinent.</li> <li>13.9 Liquid-Write intorfacial Tension: Not perinent.</li> <li>13.10 Vaper (Gas) Specific Gravity: 1.8</li> <li>13.11 Ratio of Specific Gravity: 1.8</li> <li>13.12 Latent Hash of Vaportastion: 265 Bis/16 = 147 cal/g = 0.16 × 10°J kg.</li> <li>13.13 Host of Cambustion: (ect.) = 9.00 Bis/16 = 13° cal/g = -3.500 cal/g = -230 × 10°J kg.</li> <li>13.14 Host of Beisdian: Not perinent.</li> <li>13.15 Host of Beisdian: Not perinent.</li> <li>13.16 Host of Polymortastion: Not perinent.</li> <li>14.6 Host of Polymortastion: Not perinent.</li> <li>15.7 Contract on page 3 and 8.</li> </ul>



EAC

## ETHYL ACRYLATE

Gammas Smar Anyliz and, olipi o Ethyl Syrayman Arond cont Wars pogda Your Sing Sing ducha Sing ogram Sing ducha Sing ogram Judate sad	ners ner Liquid Ploots an watar. Ploots an watar. Net work based and vapor. Keep they ployeet. Net Constructed breathing app heig ployeet. Sal our water sport to "Loos remove ducharped material. Neath and antihum remover.	Coluctum Pruity adar Pannable, invitaing vapor is produced. p propie avery. anona, and robber overclocking rement. et down <sup>1</sup> vapor. et down <sup>1</sup> vapor.	6.1 6.2 6.3 6.4 6.5	6. FIRE HAZARDS Flash Point: 44°F O.C Flashmable Limits in Air: 1.8°9.5° (calc.) Fire Estinguishing Agents: Dry chemical, fuum or carbon dioxide Fire Estinguishing Agents Not is be Used: Not pertinent Special Hazards of Combustion Products: Toxic and irritating vapors generated when beated	B. WATER POLLUTION     B.1 Aquatic Toslethy:     Data not available     B.2 Waterfood Toslethy: Data not available     B Biological Oxygen Demand (BOO):     Data not available     4 Pood Chain Concentration Potential:     None
Fire	PLANNABLE Consumts any explode in Pashback along roper 2nd Vapor any explode it games including glorest. Conhest firm from side da Extragarith with dry chema have may be neffective o Coal expand containers w	fire. may actur any actur d is an enclosed area. d brothing apparatus, and rubber overclothing lance are protocted location. col. forum. or carbon dooxide. n fine. ith water.	6.6 6.7 6.8 6.9	Bohavier in First: Vapor is heaver then air and may travel considerable distance to a source of ignition and flash back. May puls merice and cause sontainer to explode lighthen Temperature: 721°F Electrical Hazard: Data not available Burning Rets: 4.3 mm/min	9. SELECTED MANUFACTURERS     Celanese Corp     Celanese Chemical Co. Division     245 Park Ave     New York, N. 10017     Duw Radinate Co.     Williamsburg Va. 23185
Exposure	CALL FOR YEUCAL AI VAPOR Initiating to erver, near and If subated, will cover beach More to fore have been if browships has useped, pi If browships has useped, pi If browships has and erver. HOUDD Will how shin and erver. HOUDD Plush efficiend arrew with po IF IN ALLOWED and with or wells.	D. cho or nonan. we artificial respiration. e stygen. hing and abora. hart of vaster. part and flush with plently of water. m is CONSCIOUS, have victim drink water	7.1 7.2 7.3 7.4 7.5 7.6	7. CHEMICAL REACTIVITY Reactivity with Water: No reaction Reactivity with Common Materials: No reaction Stability During Transport: Stable Neutralizing Agents for Acids and Counties: Not persinent Pelymerization: May occur exclude moss- ture, light, avoid exposure to high tes- perstures, store in presence of air Inhibiter of Pelymerization: 13-17 ppm monomichyl ether of hydroquinone	3 Rohm and Mass Co Independence Mall West Philadelphia, Pa 19105 10. SHI2P2ING INFORMATION 10.1 Grades or Purity: 98.5 - 49.5% 10.2 Storage Temperature: Ambient 10.3 Inert Atmosphere: No requirement 10.4 Venting: Pressure-sacuum
Water Pollution 1. RESPORS iden Measures Harr Iviace Warning Restrict access Evaluate areas Disperse and fit	Effect of low concentration Fouring to above him. May be despress if it enter Notify local bealth and with Notify operators of nearby C TO DISCMARCE and numbers CO 444-11 -largh flammability with	s on squatic life a unknown. To water intubal. Info officual water intakes. 2. LABELS		11. HAZARD ASSESSMENT CODE ISSE HAMP & AMERICAN HERRESSE CG 448-3; A-P-Q-T-U-Z	13. PHYSICAL AND CHEMICAL PROPERTIES 13.1 Physical State et 15°C and 1 atm. Liquid 13.2 Molecular Weight: 110-12 13.3 Beiling Point et 1 atm: 211.3°F = YU h°C = 3725°K 13.4 Prevening Point:
3. CHENICA 3.1 Synonyme: Acr Eth) 3.2 Caset Guerd Ce Mor 3.3 Chemisel Perm 3.4 IMCO/United N Designation: 3.2	L DESIGNATIONS his acid, ethyl ester i 2-propensate mpatibility Cleanification: nomer ale: CH <sub>2</sub> =CHCOOCH <sub>2</sub> CH <sub>1</sub> ptione Numerical //1917	A. OBSERVABLE CHARACTERISTICS     A. OBSERVABLE CHARACTERISTICS     A. Descent and the second s	12.1	12. HAZARD CLASSIFICATIONS Code of Federal Regulations: Flammable liquid NAS Hazard Rating for Bulk Water Transportation: Category Rating Fire Category Rating Fire Liquid or Solid Irritan Liquid or Solid Irritan S Water Pollution	- 94*5 - 72*C = 201*N 13.5 Critical Temperature: 54*5 - 27*C = 552*N 13.6 Critical Pressure: 544 pire = 37 atm = 3 * M.S.m. 13.7 Specific Gravity: 0.923 at 20*C (locula) 13.8 Liquid Surface Tension: 25 divises ther interfacial Tension: (est ) 40 divises (m = 0.04 N m at 20*C 13.10 Vapor (Gais) Specific Gravity: Not percinent
S 1 Personal Protect Impervious planet S 2 Symptome Patte escessive vapu or estreme triff Tradition if a respiration if a Call a physicia S Shart-Torm Inha	5. HEALTI Iliva Equipment: Organic can vest wing Exposure: May cause is concentrations can also cause tation of the reperatory tract personer: INHALATION re- personer: INHALATION re- personer: SKIN AND EYES in tion (Threahold Limit) Yaku lation Limita: 50 ppm for 13	A HAZARDS Inster or air-supplied mask, acid poggles; rritation and burns of eyes and skin. Exposure to dromsiness accompanied by nauka, headache, move victim to fresh air and administer artificial wash for 13 min. with copious quantities of water b): 23 ppm min.	12.3	Human Tosicity     2       Aqualix Toxicity     2       Aesiheric Effect     2       Reactivity     2       Other Chemicals     2       Water     0       Self-Reaction     3       NPPA Hazard Collections:     2       Cologery     Closelfication       Health Hazard (Blue)     2       Flammability (Red)     3       Reactivity (Yellow)     2	<ul> <li>13 11 Ratio of Specific Heats of Vepor (Ges): 1 0k0</li> <li>13 12 Latent Heat of Veporization: 149 Biu ib = 52 vicit, g = 34° × 10° J kg</li> <li>13 13 Heat of Combustion: - 11 k0° H a kh = - 5KR caling = - 276° X 10° J kg</li> <li>13 14 Heat of Decomposition: Not perturent</li> <li>13 15 Heat of Solution: Not perturent</li> <li>13 16 Heat of Polymorization: - 11° H; to = - 156 caling = -7.79 X 10° J kg</li> </ul>
56 Testchy by Inger 57 Late Testally: R 58 Vaper (Geo) inti usually lutrat 59 Liquid or Solid h short stypurd 510 Oder Thrasheld:	Nion: Grade 2. LDu 0 5 to 5 g ereated exposure may develop ant Characteriolitics: Vapor t e muderate or high vapor color mitant Characteriolitics: Cau and may cause secondary buri 0 00024 ppm	/kg (rat) sensities), is moderedly writating such that personnel will not ensense and the skin and first-degree burns on is on long exposure.		NÖ	·Cuminand an pages 1 and 6.

D-7



D-8

•

HCN

ĩ

## HYDROGEN CYANIDE

Common Synon Hydrocysnic seld Pramic seld	Watery liquid, or gas	Colorian Bitter simond adar	6.1 6.2	6. FIRE HAZARDS Flash Point: 0°F C C Plammable Limits in Alr: 5 6% - 40 0%	8. WATER POLLUTION 8.1. Aquetic Teachty: 0.16.ppm/72.hr/soung.boxs.TLm.fresh water
	Sinks and misce with is produced and	water, Poisonous, fismmable vaper rives. Boiling point is 78° F.	63	Fire Extinguishing Agents: Stap flaw of gas Fire Extensionian Agents Mat to be the st	0.069 ppm/24 hr, pin perch. TL <sub>mi</sub> sale water 8.2. Waterlewi Tealetty: Data nini available
AVOID CON WEAR CHEN BREAT Stop discharg Shut off ignit Stay uppeind Ivolate and re Notify local I	TACT WITH LIQUID AND VARO AICAL PROTECTIVE SUIT WITH NING APPARATUS pr if possible. Call fair department. ion sources. and use water spray to "hanch doo more discharged material. bealth and pollution control agenci	R. EVACUATE AREA. SELF-CONTAINED 78" 19907.	6.6	None Special Hazardo of Combustion Products: Extremely toxic vapurs are penerated even at ordinary temperatures Bohavior In Fire: Container's may explode with spinon of concents	Bological Oxygen Demand (BOD): Date not available     B 4 Food Chain Concentration Petential.     Nose
Fire	FLAMMABLE. Playback along upper trail may occur. Vapor may explode if ignited in an enclosed area. WEAR CHEMICAL PROTECTIVE SUIT WITH SELF-CONTAINED BREATHING APPARATUS. IFO Stop discharge it possible Cool exposed containers and protect men effecting shutoff with water. Let fire burn.		6.7 68 6.9	Ignition Tempurature: 1001°F Electrical Hezard: Data not svoilable Burning Rote: 15 mm/min	9. SELECTED MANUFACTURERS     1. L 1 daPani de Nemuercácio Inc Electrotemmatil Depl Wilmingun, Det 19898     2. Monsulo Polymers & Perushemicals Co 800 Scoth Lindhergh Bird St. Louis, Mo 63160
	CALL FOR MEDICAL AID. VADOR POISONOUS IF INHALED OR Imsting to sea. Nore to frush air. If breathing has stopped, give arri- that ROT mouth to mouth If breathing a difficult, give oxy LIOUID POISONOUS IF SWALLOWED Instring the SWALLOWED	IF SKIN IS EXPOSED. Inclair respiration ref. OR IF SKIN IS EXPOSED.	7.1 7.2 7.3	7. CHEMICAL REACTIVITY Reactivity with Water: Dissuires with a moderate reaction Reactivity with Common Materials: None Stability During Transport: May become unitable and subject to explicit on if stored for extended time or exposed to high temp and pressure	3 Rohm and HaarCo Indexendence Mall West Phriadelphra: Pa 19105
	Intracting to syst. Remore concernisated clothing a Fluch offected areas with planty (IF IN EYES, hold cyclids open a IF SWALLOWED and vectors a C or mills and harv victus mild IF SWALLOWED and victus in a VULSIONS, do nothing exc	nd shoes. 57 water. 14 fhigh-with plenty of water. 00%CrOUS, haw victim driak water new combing. NCONSCIOUS OR HAVING CON- ept keep victim warm.	7.4 7.5 7.6	Neutralizing Agents for Acids and Caustica: The weak acidity can be neutralized by staked lime, but this dues not destroy the privationous property. Polymerization: Not pertinent Inhibitor of Polymerization: Not pertinent	10. SHIPPING INFORMATION 10.1 Grades or Purthy: 94%, sometimes shipped as a water solution, or a horhed or, an inert solid. All forms are estremets toak 10.2 Storage Temperature: Data not availa 10.3 Inert Atmosphere: May he padded 10.4 Venting: Data not available
Water Pollution	HARMFUL TO AQUATIC LIF! May be dangerous if is entern wa Notify local health and wildlife o Notify operators of nearby water	: IN VERY LOW CONCENTRATIONS. ler inteles Miciels inteles.			
1. RESPC riter Annuaria Issue war water c Restrict a Evacuate	DNSE TO DISCHARGE Warnes reveauer 68 each mange - high Rommability. contaminant cortaminant screa area	2 LABELS PRESENT PRESENT		11. HAZARD ASSESSMENT CODE (Bee Hagerd Assessment Handback CG 44-3) A-B-C-K-L-M-N	23. FMYSICAL AND GHEMICAL PROPERTIES     13.1 Physical State at 15°C and 1 atm: Liuurd     13.2 Molecular Weight: 27.03     13.3 Beiling Point at 1 atm: 15.3 *F = 25.1°C = 298.9°K     13.4 Freezing Point: 3.1°E = -11.1°C = 10.0°C
3. CHEM 3.1 Synonyms: 1 3.2 Coest Guard 3.3 Chemical Fe 3.4 IMCO: Unite Designation:	ICAL DESIGNATIONS Hydrocyanic acid Prusic acid Compatibility Classification: Noi applicable rmula: HCN d Nations Numerical 2 0/1051	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (an physical): Liquid 4.2 Color: Coloriess to bluish white 4.3 Odor: Characteristic sweetish, like almond	12.1	LE. RRAIND ULASSIFICATIONS Code of Federal Regulations: Poisonous par or laudi. Class A NA\$ Hazard Reting for Bulk Weter Transportation: Celegory Reting Fire	<ul> <li>13.5 Critical Temperature: 362 3*F = 183 5*C = 456 7*K</li> <li>13.6 Critical Pressure: 725 pila = 50 atm = 5.01 MN m</li> <li>13.7 Specific Gravity: 0.64 at 20*C (liquid)</li> <li>13.8 Liquid Surface Tempion: Not pertinent</li> <li>13.9 Liquid-Water Interfacial Tempion: Not pertinent</li> <li>13.10 Vapor (Gas) Specific Gravity: 0.5</li> <li>13.11 Ratio of Specific Heats of Vapor (Gas): 13.03</li> </ul>
C.SUTION Cless / ar report invoige s 5 1 Personal Prat cylinder W Clear-rise / piores, cles and consume Fe eyes, value and consume 5 3 Treatment las ductor arm contaminas s y prosent	5. HEALTN A powor, szphysierio car be caus kariporticularle ever microu man tective Equipment: Excaps purpo fork purpores - vapor-proof emerg- iork purpores - vapor-proof emerg- mical vafety gogglie, quick-oprening oliowing Ezponure: Irritation of t sion, nuvca, headache, weak heas o uon Ezponure: Call a doctor If breas with INHALATION remose paus del ching and wash kin thorough TACT hold match.	HAZARDS In div ingestion, inholation, or obscription of liquid becames, and feets. New only — air excape mask with 5-minute air new will develop the set of the air mask with and regulator, and 30-minute air cylinder. Rubber wifety shower hroas, pulpitation, difficult breathing, reddening of farms and legs, giddines, — followed by collapse hing has stopped, give artificial respiration entil in to fresh air S. Nix CONTACT: remove ly with coprous quantities of water and supp	12.3	Human Toxici) 4 Aqualic Toxici) 4 Actiletic Effect I Reactivits J Water 0 Self-Reaction J NFPA Hozard Closelfications: Closelfication Meshit Hazard (Bive) 4 Flammability (Red) 4 Reactivity (Yellon) 2	13.12 Latent Heat of Vaporization: 446 bru/le = 247 cal.g = 10.3 × (49) kg 13.13 Heat of Combustion: = -0.100 Btu (46) = - 58rd cal/g = -245.3 × (47) kg 13.14 Heat of Decomposition: Not pertinent 13.15 Heat of Solution: Not pertinent 13.16 Heat of Polymorization: Not pertinent 
reas 15 mil If parient is Inding this respiration patient is to \$ 4 Tostally by Ini \$ 5 Short-Tarm to \$ 6 Tostally by Ini	n n uncensional approximate part and data n uncensional approximate and for Replace ampli netrate part is second netration improves or dector arrives helation (Threshold Limit Value) helation: Limita: 20 ppm for 30 m spotters: (rade 4 LDa less has 5	ege ann aonmaiour penas scream or weige og al re by crushing a pear i fampules in a closh and in every minute. Du not i nierrups artificial alfangli hin spent. Continue i realment until c. 10 ppm in Dimg/kg	5.7 5.0 5.9 5.10	5. HEALTH HAI Late Testelly: Data not available Vaper (Gas) IrrRent Characteristics: Vapor in Liquid or Bolid IrrRent Characteristics: Liquid absorbed through skin or cyn Odor Threshold: Data not available	CARDS (Cont'd.) not serv irritating but is extrumely poisonous d is not irritating but is extremely poisonous if
		(Consider page 1			



.

IBT

# ISOBUTANE

Commun Simon 3 Min-Bylginguae Stop dincha Shot off on 5 Solve oglima Shot off on Natify Ingel	Processing and particular and partic	ned gas Colorian ()dorian volter. Plananable vijäble vopae cloud mensi, down" vopae. encia.		6. FIRE NAZARDS     Plash Paint: -113°F C.C     Plasmabile Limits in Air:     1.8° - 3.4%     Pre Extinguishing Agents:     Nop flue of gas     Pre Extinguishing Agents Net to be Used:     Not perinent     Special Nazards of Combustion Products:     Nut perinent     Behavior in Pire: Not perinent	8. WATER POLLUTION 8.1 Aqualis Tasichy: Name 8.2 Waterfoot Tasichy: Nunc 8.3 Biological Osygen Domand (BOD): Name 8.4 Feed Chain Concentration Potential. Name
Fire	FLANN ABLE Flashback along report trail m Vapor may explode if ipniced Stop flow of gas if possible. Cost exposed containers and Let fire born.	ay occur, is as enclosed area, men effecting shutoff with water.	6.1 6.1	F Ignklon Temporature: 390°F 5 Electrical Hazard: Nut periment 9 Burning Role: 93mm/min	9. SELECTED MANUFACTURERS 1. Atlantik Richfield Co ARCO Chemical Co. Division 200 South Broad St Philadelphia, Pa. 19101 2. Citize Service Co. Inc
Exposure	CALL FOR MEDICAL AID. VAPOR Imitating to eyes. If inhaled, will cause distringent er loss of consciournes. More to fresh air. If breathing has stopped give i breathing i difficult, give I breathing i difficult, give I F IN EYES, hold syelids opt	, difficult breathing artificial reapiration, rygen, in and Rugh with plenty of water.	7.1 7.2 7.3 7.6	7. CHEMICAL REACTIVITY Reactivity with Water: Nu reaction Reactivity with Common Materials: Nu reaction Stability During Transport: Nuble Neutralizing Agents for Acids and Caustics: Nut pertinent Polymorization: Nut pertinent Inhibities of Polymorization: Not pertinent	Perindhemicals Darivian (r) Wall Sa New York N Y 10005 3 Philips Periodeum Ca Barilesville, Okla 74004 10 SHIPPING INFORMATION 10 1 Grades or Party: Parts Inchinali
Water Pollution	Not harmful to aquasic life.				10.2 Siorage Temperature: Ambien 10.3 Inert Almosphere: No requirement 10.4 Venting: Safety relief
L. RESPON IBeo Amaoneo M Issúe Warning Restrict occes Evocuate area	SE TO DISCHARGE Hindeau Ca 44-1; high Nammability d	2. LIBELS		11. HAZARO ASSESSMENT CODE (HT HALVI ADMINISTRATIONALLI) A-D-C-D-E-F-G 12. HAZARO CLASSIFICATIONS	13. PHYSICAL AND CHEMICAL PROPERTIES           13.1 Physical Biologian SPC and 1 atm. co.           13.2 Molecula: Weight: Sr 12           13.3 Boiling Point at tatm.           10.875 = -11.975 = 26.37K           13.4 Processing Point:           -22.575 = -255.35K = 17.97K
3. CHEMIC 3.1 Symmyrms: 2-h 3.2 Coast Guard Ci 5.3 3.3 Chemical Form 3.4 IMCO;United N Designation: 2	AL DESIGNATIONS Activity propanc ampatibility Classification: unsted hydrocarbon unse: CH <sub>2</sub> CH1CH37 rations Numerical U/1960	4. OBSERVABLE CHARACTERISTICS 4.1 Physical Bisse (so shipped): Liquid under pressure 4.2 Color: Coloriess 4.3 Odor: Like gasoline	12.1 12.2 12.3	Code of Foderal Regulations: Flammable compressed gas NAS Hazard Rating for Built Water Transportation: Not Insted NFPA Hozard Classifications: Not Insted	<ul> <li>13.5 Critical Temperature: 27478 = 1344 = 40646</li> <li>13.6 Critical Pressure: 279764 = 1604466 = 160400 pr.</li> <li>13.7 Specific Gravity - 945742 (2000 compute 13.8 Liquid Surface Tension, 3.4 dates com = 0.014 N (a. a. 5.6%) 13.9 Liquid-Water interfacial Tension, edu Audites com = 0.017 N (a. a. 5.6%) 13.10 Vapor (Ges) Specific Gravity - 2.0 13.11 Rate of Specific Gravity - 2.0 13.11 Rate of Specific Gravity - 2.0</li> </ul>
S 1 Personal Protect     S 2 Symptome Policy     incoordination for     advato for     confused, or a     of openprime     puvinoing of     ASPIRATIO     S 5 Short-Torm Inhu	5. HEALTH the Equipment: Self-containe owing Exposure: Central nervo no anexhesia and respiratory a equilar hearbeat it sare but a posure: INHALATION prot neitherized, apply antificial respi- to or other sympathomimes and head, give symptomusic and sup N no treatment required lation (Threathed Limit Volue) interion Limits: Data not availab	MAZARDS d breathing appuratus, safety goggles as system depression ranging from diazaness and rest, depending on concentration and eatent of langerous complication at anotheric levels of victum against setf-injury sifte is supportun. ration if breathing has slopped, avoid administration mis, preven asystation of community by proper portive treatment. INGESTION OR : Data nut available te			. GNN 13.12 Letter Heat of Vaporization: 13.13 Heat of Combustion:
56 Tosicity by Ingo 57 Late Testetty: N 58 Vaper (Gas) Inti 59 Liquid or Betid I Necase it is to 5 10 Odor Threatest	ollon: Noi perineni ione ione ione Charaetoriolles: Nune iritane Charaetoriolies: No ap irit volatile and examiration quickt i Data nui available	prociable hazard. Praicelly harmless to skin s. Some frouble possible		not	(5



**D-12** 

- . .

•

MAL

## METHYL ALCOHOL

Gamman Senarem Nachard Nacid eistel Nacid eistel Nacid eiste Prosector gen Ster off gentin Ster off gentin Armed contextr o Armed senarer Nacid hera ber	Weatery benefit Places and minute which we dependent theory propherents, a warrow and rail for dependences, do not and rail for dependences, do not and rail for dependences, and benefities and more and and production content agenetics.	Colorium Abaghel edur war Planmakke, umanag napar is produced. 		6. FIR, IM2ABDS     1. Posth Petets: S4°FC.C.; 61°FO.C.     2. Planmagha Limits in Air: 6 AS.—36 S4     3. Pire Estinguishing Agonto Hot to be Usadt:     Wiser apy to ineffective.     5. Special Meaords of Combustion Products:     Not perimeti     Defaults in Place: Containers may explade	B. WATER POLLUTION     Aquestic Fooldly:     200 ppm/11 hr/goldfich/diad/fresh-upter     2:3 Waterbest Testably: Data net seviciote     3 Biological Daygon Domand (800):     (Theor 100%, 54 days, 0.1 – 1.1 k/k,     3 days, 0.7% titter, 1.20 days     8.4 Peod Chain Concentration Potentiat:     None
Fire	PLANMABLE Vapor nay orphode of ignized in an Example of the second second second Example of the second second second Vapor and the second second second Call FOR MEDICAL AID Call FOR MEDICAL AID VAPOR If subside, will court distinct. Non or has of continuence.	oostaad araa. Ma ka kal faan, or ayrkan dioxide. r. hete, difficult breeding, rial rayerbian.	7.7	CHEMICAL BEACTIVITY     Resetivity with German Refer.	SELECTED MANUFACTURERS     Borden Inc     Borden Chemical Division     Genmar, La 19734     Celanae Corp     Celanae Corp     Celanae Corp     Celanae Chemical Co Division     245 Part Ane     New York, N Y, 10017     E 1 du Pont de Nemeurs & Co., Inc     Indurreil and Bischemical Dept     Widmington, Del 19998
Exposure Water Pollution	LIQUID ROSCAQUS IF SWALLOWED. Insuling to due and eyes. Remore consummated clushing and Path officers are with planty of IF IN EVES, hold cyclud agent and IF IN EVES, hold cyclud agent and have been state in the WILSIGNS. do nothing every WILSIGNS. do nothing every Angeress to aquate March may be hear to capacity March with the Name of the capacity of the sin high to the hear every of the sin high to hear to approve the sin high to hear to approve the single to approve hear to approve hear to approve the single to approve the single to approve hear to approve the single to approve the single to approve hear to approve the single to approve to approve hear to approve the single to approve the single to approve to approve hear to approve to approve to ap	I shoes. Toto, Rush with pleasy of votor. SCOUTS. how victure drink water Promising. CONSCIOUS OR HAVING CON- 1 Larp victure warm. A starp victure warm. Mathem.	7. 7. 7. 7.	Stability During Transport: Stable     Mentralizing Agents for Acids and     Counties: Naj pertinent     Polymorlastics: Net pertinent     Institute of Polymorlastics: Net pertinent	10. SHIPPING INFORMATION 10.1 Grades or Partig: CP, Crude, ACS all 99.9% 10.2 Blarage Temperature: Ambient 10.3 Grad Almosphere: No requirement 10.4 Veniting: Open (flome prepare) or pressure-recover
1. RESPON data Programs in Nature Warting Restrict access Evacuate and Driperse and	NSE TO DISCHARGE Innen Hunstein, ca 46-9 1- hegh Rammsbildy 9 Resh	2. LABELS Pressbury		11. NAZARD ASSESSMENT CODE data matrix antimitations Cd 44-8 A-P-Q-R-5	13. PHYSICAL AND CHEMICAL PROPERTIES 13.1 Physical Base at 18°C and 1 atm; i must 13.2 Methodar Weight: 32.04 13.3 Basting Public of 1 atm; 148.1°F = 44.3°C = 337.7°K 13.4 Provide Pathot
3. CHEMIC 3.1 Synonymus Co Ce Mil 3.2 Coord Buard C Plu 3.3 Chemisal Porn 3.4 MRC0/United I Peelgnotien: J	CAL DESIGNATIONS bienus sprit. Wood aleohaf bienus sprit. Wood aphtha inhanet. Wood sprit. Bienparthill canhol media: CH(OH Notions Nuunarbad D.2/1230	4. OBSERVABLE CHARACTERISTICS 4.1 Physical Butto (as chipped): Louid 4.2 Color: Colorias 6.3 Odur: Foint slochol; like athyl slochol, faintly sweet; characteristic pungem	12.1	12. HAZARD CLASSIFICATIONS Code of Poderal Regulations: Flammable liquid HAS Heave Parting for Bulk Water Transportation: Category Roting Fire	144 0°F = -97 8°C = 175 4°K 13.6 Critical Temperature: 444°F = 240°C = 313°K 13.6 Critical Pressure: 142.0 peas = 77.7 stm = 7.87 MN/m <sup>2</sup> 13.7 Speakle Garles 7 Poz at 20°C (liquid) 13.6 Liquid-Water Interfact Tension: Not periment 13.10 Vaper (Gas) Speakle Growity: 1.1 13.11 Data of Sanatha North of Vaper (Gas):
S.1 Personal Protoc safety gaggies ache. facigue depression an be absorbed i B-3 Treatment the C brasthing has in plots of so 5.4 Tostelly by Inde 5.6 Moeth-Term Juh 5.6 Tostelly by Inde 5.6 Tostelly by Inde	5. NEALTH stive Equipment: Approved co is rubber gaves. leaving Exposures. High concentra and draw under. High concentra and star under the source of the source of the source of the leaving the source of the source of the source of a physician. Skill OR E storten (Thrushaid Limit Values instant Limits: Outs are availe sources: S to 13 g/bg fra Name	NAZARDS Instar mask for high visper concentrations, inconsive vapor causer are inritation, head- work area produce control inervers system in will probably cause death in 1 to 3 hrs. Can use death or eye domage. Isspenurs and spely artificial respiration if immling, then give 2 casspoors of baking sede VES flood with easer for 15 min. (r 200 ppm) Ms	12.3	Water Pollucion         Human Toxicity       1         Aquatic Toxicity       1         Aastheire Erlest       1         Rescience       2         Other Chemicsle       2         Water       0         Self-Reaction       0         HPPA Hexard Clustifications:       Cologory         Cologory       Classifications         Horth Hazerd (Bure)       1         Pammabelity (Ref)       3         Ractional (Yellow)       9	1.254 13.12 Lationi Hoot of Vaporlasiton: 473 0 Biu-ib = 262 8 cs/g = 11 00 x 107 J/kg 13.13 Hoot of Conduction: = 6410 Biu/ib = = 6677 cs/g = = -141 B x 107 J/kg 13.14 Hoot of Documposition: Not pertinent 13.15 Hoot of Bubymerization: Not pertinent 13.16 Hoot of Pubymerization: Not pertinent 13.16 Hoot of Pubymerization: Not pertinent
5.8 Vaper (Bas) tri system if pro 5.9 Liquid ar Balld to romain, an 5.10 Odar Threahad	Mant Charagterletter, Vapers ( nem in high generativisions. The a livritent Charasterletter. Minu oy cause simersing and reddening & 140 ppm	zene a night amarting of the syna or respiratory flort is somperary num heared. If spitled on clothing ond allowed of the skim.		NO	13

. . . . . .

: :·

,



MTB

## METHYL BROMIDE

			ר ו		
Brownstethers Enteriores	Legerfird ga	Colories Oduries to sweet oder		5. FIRE MAZARDS Flooh Point: Practically not flammable	8. WATER POLLUTION 8.1 Aquetic Toolety: None
	Sinks and built in Bailing print	nyser Polyanuus vapor cloud is formed. 19 997 P.	6.2	Planmable Limite in Airt 10% 13% The Enterthables Assets I has an annual	8 2 Waterburt Taxially: None 8.3 Biological Oxygen Damand (BOD):
AVOID CON	TACT WITH LIQUID AND VA	POR. Keep propie sway.		Fire Estinguishing Agents Hel to be Used:	§ 4 Food Chain Concentration Patentiat
include Stup decharg	ng planas. ng planas. ng si punsible. Call fire departme	al		hiel perlinent Special Hazarda al Combustion Producto:	None
Francist and	a so care of large lasks and one water spray to "knock	40-00" *1967		Tease and writing gates are generated	
footate and e Natury locat	braith and pullutum cuttral ap	NUT		Bahavlar in First Containers may explode	
	Cumbustible POISONOUS AND HIRITAT	ING GASES ARE PRODUCED IN FIRE	6.7	Ignition Temperature: 999°F	
Fire	West pupples officialised (including plown) EXTINGUISH WITH WATE Cool exposed containers well	Heiming appendix, and Heiming Streaming 3, FOAM, OR CARBON DIOXIDE 1 water:		Encourses researce. You pertinent	9. SELECTED MANUFACTURERS 1. Dow Chemical Co Midland Mich 48640 2. Great Lakin Chemical Corp Weyl Lakington, July 4 Millio
	CALL FOR MEDICAL AID		{ }		3 Northwest Industries, Inc.
	VAPOR				Michigan Chemical Curp 351 E. Ohio Si
<b>ु</b> ष्ट ।	POISONOUS IF INHALED.		,	7. CHEMICAL REACTIVITY Reactivity with Webst. Do reactive	Chicago III 80611
×	If breathing has stupped, pro	arteficial regardation ath i	7.2	Asactivity with Common Materials:	
	If breathing a difficult, give a	n <b>y gen</b>		No reaction Stability During Transport: Stable	
Exposure	Will burn shin and even. Harroful of excitomed		1 7.4	Neutralizing Agents for Acids and	
	Remove cuntaminated clothin Flush affected areas with plet	ng and shuta. Hy of water.	,,	Counter: Not pertinent Polymorization: Not pertinent	
	DO NOT RUB AFFECTED / IF IN EYES, hold sychich op IF SWALLOWED and victum or mill. DO NOT INDUCE VOMITIN	REAS. na and Nucle with plenty of water. na CONSCIOUS, have victim drink water G.	7.6	Inhibitor of Polymortastion: Not pertinent	10. SHIPPING INFORMATION 10.1 Grades or Purity: Commercial not less than 99-39
					10.3 Inert Atmosphere: No requirement
Water Pollution	Not harmful to aquatic life May be dangerous if it entern Notify lacal health and wildle Notify operators of dearby w	waser intakes. Fe officials aver intakes			10 4 Vonting: Safety relief
1. RESPONS (See Augenes Main Issue warning Restrict pages	E TO DISCHARGE	2 LABELS		11. HAZARD ASSESSMENT CODE (Ben Haard American Increases CO 445-3) A-B-C-I-J	13. PHYSICAL AND CHEMICAL PROPERTIES 13.1 Physical Bioto at 18°C and 1 alm: Ga- 13.2 Malacular Walgher 44.95 13.3 Bailing Point at 1 alm: 34.5°F = 3.6°C = 276.4°N
				12. HAZARD CLASSIFICATIONS	-ijs*f = -93*C = 180*K
			12.1	Code of Fodural Regulations:	13 S Critical Temperature: 176*F = 191*C = 4rd*A
J. GREWIGH 3.1 Synonyma: Bron	nomaliane	4. UBSCRYADLE CHARACTERISTICS 4.1 Physical State (as shipped):	1 1122	Poisonous inquid or solid, Class B NAS Histord Rating for Bulk Water	13.6 Critical Prosoure: Nut pertinent
Emb	infume	Liquefind gas		Transportation:	13.7 Specific Gravity: 1 of at 20°C (liquid)
Mon	obromomethese	4.2 Gener: Coloriess 4.3 Odor: Relatively adarless; sweet,		Cologory Rating	13.8 Eliquid Surface Tension: 24.5 dyney/cm = 0.0245 N/m at 15°C
).2 Coast Guard Cor Haio	mpetibility Closeffection:	chieroform-lijke		Health	13.9 Liquid-Water Interfacial Tension:
	de: CHjBr			Vapor Tristant	13 10 Vapar (Gao) Specific Gravity: 3 3
3.4 HiCO/United No Designation: 2.0	None Numerical )/1062			Passons	13 11 Ratio of Specific Hoots of Veper (Gas):
	· -			Human Tonicity	13.12 Latent Heat of Vaporization:
	S. HEALTH	HAZARDS		Aquatic Toxicity	106 Blu/Ib = 59 7 cal/g = 2 50 × 10º 3/kg
1.1 Personal Protect	Ivo Equipment: Self-contain	ed breathing apparatus; goggles. I vance causes lung connection, and missionany.		Resolivity Chamarala	$=77t \operatorname{cal/g} = -74.15 \times 10^9 \operatorname{J/kg}$
edema Higher	concentrations cause rapid na	resus and death. Contact with hquid stritates		Weter	13 14 Next of Decemposition: Not pertinent
cy es and burns 5.3 Tructment for Ex	SAM <b>PROVING INHALATION</b> res	nove victim to fresh air; give artificial respira-	12.3	Self-Reaction	13 16 Heat of Polymortastion: Not pertinent
tion if needed. 1	SKIN OR EYES: Auch with -	oter for at least 15 mill.		Cotogory Classification	
.4 Tecleity by Inhole .5 Sheri-Tarm Ishal	enen (Thresheid Limit Velu: Ietlen Limity: 20 ann Jer 5 m	рк I>рфил 16.		Health Hazard (Blue)	
6 Testelly by ingest	tion: Deta not available			Reactivity (Yellow) 0	
7 Late Teslelly: Da	ILE ROL SVEILABLE				- Constant on pages 3 and 6 -
i. <b>5 Voper (Geo) I</b> rrite net usually tole	ent Cheresteristics: Vapor n rate moderate or high vapor ci	a moderately irritating such that personnel will incentrations.			la <u>,</u>
5 9 Liquid or Solid in	Mant Characteristics: Fairl	y savere skin irritant, may cause pain and		NO	163
\$.16 Oder Throsheld:	Data net evailable				



D-16

.•

PPL

### PROPYLENE

r	<u>-</u>				1
Earnen Jonen August Maintaither Maintaither	Plasts and boils on Plasts and boils on a product.	nd gas Calurings Mild oder unter. Planmabit, välikir vapar eland unte	6.1 4.2 6.3 4.4	6. FIRE HAZARDS Pash Point: - 162°FC C Panmobio Limits in Aur: 20% - 11% Pro Estinguishing Agents: Stop form of past Piro Estinguishing Agents Hol to be Used: hat pertinent	B. WATER POLLUTION     Aquatic Taskety: Nune     Waterland Taskety: Nune     Balagical Oxygen Domand (BOD) Nune     Faad Chain Cancentration Palantial     None
Sary uponed i Everyout onto Avoid contact Nacify local b	and use water spray to "Anoch i on oue of large dealarge. I with laged with and pollation control age	ione" ruppe. Intea.	0.5	Special Heards of Combustion Products: Not purinder Bohaviar in Pire: Containers may explode Vaper in heavier than sir and may transf conviderable divigince to a source	
Fire	PLANMARLE. Consumer any explore rull an Photoback along rupper rull an May explore of generation Scop flow of gen of permittion. Coul exposed consumers and Let for basis.	, y aastur. actuard anta. protoci mea effecing abutoff with wotter	6.7 6.8 0.9	o ignilion tea testa back Ignilion Temperatura: 927°F Elestricai Hazard: Clavi I, Group D Burning Réte: Ramn/min (Inguid)	9. SELECTED MANUFACTURERS 1. Dum Chemical Cu Midiand Mich 4and0 2. Excon Chemical Co Housion Tex 77001 3. Linum Carbule Curp Chemic us and Photor University
Exposure	CALL FOR MEDICAL AND VAROR If subside, will couse detaining there to fresh as If benching as different, give a LQUID Mill couse freethore. Plash affected areas with pire DO NOT RUB AFFECTED A	er las of conscientings. entificial reginisses. aygen. aygen. REAS.	7.1 7.2 7.3 7.4 7.5	7. CHEMICAL REACTIVITY Reactivity with Weler: No reaction Reactivity with Common Molerials: No reaction Blability During Transport: Stable Neutralizing Agents for Acids and Coustics: Not pertinent Polymorization: Not pertinent	200 Park Ave New York, N. Y. HID17
Water	Not harmful to aquetic life.		7.6	Inhibitor of Polymorization: Not pertinent	<ol> <li>SHIPPING INFORMATION</li> <li>Grades or Purify: Chemical 92+9 polymerization 94+9; recarch 49+3 proprine concentrate 80+3</li> <li>Biorego Temperature: Amhicht 103 Ineri Atmosphere: No requirement</li> <li>Hond Atmosphere: No requirement</li> <li>Venting: Salety relief</li> </ol>
1. RESPONS Then Property Married Insula Warring - Evacuate area	E TO DISCHARGE unis Instatuus, Cit 446–4) hugh flammability	2. LABELS		1). HAZARD ASSESSMENT CODE (See naure Assessment nonsear CG 46-3) A-B-C-D-E-F-G	13. PHYSICAL AND CHEMICAL PROPERTIES 13.1 Physical Blate at 19°C and 1 atm: Ga- 13.2 Molecular Weight: 42.00 13.3 Boiling Point at 1 atm: -53.4°F = -47.1°C = 225.5°K 13.4 Prover ng Point -10.4°F = -153.2°C = 88°K
3. CHEMICAI 3.1 Symanyma: Mett Prop 3.2 Coast Guard Car Olefi 3.3 Chamical Permu 3.4 IMCO/United Na Designation: 2.0	L DESIGNATIONS hylet byleac mean montability Classification: in da: CH/CH=CH; disso Numericad lylo77	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as shipped): Liquefied gas 4.2 Color: Coloriess 4.3 Odor: Wesh gassy	12.1	IZ: RAZARD CLASSIFICATIONS     Code at Federal Regulations:     Fiamable compressed gas     NAS Mazard Rating for Buth Water     Transportation:     Category Rating     File     Category Rating     File     Vapor Irritant     O     Liquid or Solid Irritant     O     Liquid or Solid Irritant     I     Water Pollution	$\label{eq:constraints} \begin{array}{c} 13.5  Critical Temperature: \\ (97.2^{\circ}F = 9) d^{\circ}C = 365.0^{\circ}K \\ 13.6  Critical Pressure: \\ \hline 570.011d = 43.6 atm = 4.62.51 N m^2 \\ 13.7  Specific Gravity: 0 n(N at = 4^{\circ}C (fiquid)) \\ 13.6  Liquid Surface Tension: \\ \hline 16.7 d h hor/cm = 0.0167 N/m at = 4^{\circ}C \\ 13.9  Liquid-Water Interfacial Tension: \\ \hline Nat pc/(nent) \\ 13.10  Vaper (Gas) Specific Gravity: : 4 \\ 13.11  Rate of Specific Hysts of Vapor (Gas); \\ \end{array}$
S.1 Personal Protect face shield (for S.2 Symptoma Follor and unconscion S.3 Treatment for Em or has stopped. S Short-Term label S Short-Term label S Short-Term label S Voper (Goo) Irrite S.4 Voper (Goo) Irrite	5. HEALTI Ivo Equipments: Organic vap using Exposure: Moderate co senes. Contact with injuefied y pesares: INHALATION: rei start resusciation, give oxyge start resusciation, give oxyge then (Threaded Limit Value one and Characteriotics: No a traporates quickly	4 NAZARDS or caniter or air-supplied mask, goggles or riquid). wopylene will cause "freezing burn " nove vision from exposure, if breathing is irregular move vision from exposure, if breathing is irregular move vision from exposure, if breathing is irregular move vision form exposure, if breathing is irregular move vision form exposure, if breathing is irregular move and the special state of the special state of the are nonivisiating to the eyes and throat percendic hazard. Practically harmless to the	12.3	Human Toxicity 0 Aquatic Toxicity 1 Aesthetic Effect 0 Reactivity Other Chemicals 1 Water 0 Self-Reaction 1 NPPA Hezard Classifications: Catogory Classification Health Hazard (Bluet) 1 Flammability (Red) 4 Reactivity (Yellow) 1	1   15 13 12 Letent Heel of Vaporization: 14 78 July 10 College 432 # 10 U kg 13 13 Heel of Combuston: -19.692 Hu In +-10.940 coll.g = -458 G4 X 10 U kg 13 14 Heel of Decomposition: Not pertinent 13 15 Heel of Solution: Not pertinent 13 16 Heel of Polymorization: Not pertinent (Comment on angel 3 and 6) (Comment on angel 3 and 6)
5 10 Oder Thresheld:	Data not available				

۰.

۰.



D-18

SHS

## SODIUM HYDROSULFIDE SOLUTION

			_		
Laborate Spinlar Sudates Brailfolg Sudates auffordates Sudates In design a		Light vettiner Rutten eggindur för And		6. FIRE HAZARDS 1. Flooh Point: Nuc Ilammanic 2. Floommobile Limite in Air: Nuc Ilaminanic	8 WATER POLLUTION 81 Aquatic Taxicity: 200 mg Linn an introduct of its (E.g. incom
Nersja dan, harr Kuntata and re Noresta fin af f	ge if prouble. Keep peripte ye enning dow harged Material health and publicition control t	ιν χετι, κει		Fire Extinguishing Agents, Nit Schlindent     Pire Extinguishing Agents Not to be Ubadi: Nit Schlindent     Special Hazards of Cambustion Products Nit Schlindent     Antonio Schlindent	<ul> <li>eater</li> <li>8.2 Welerleet Tesscript state totate store</li> <li>8.3 Biological Garges (Demand (BOD) Data tot arausing</li> <li>8.4 Fees Chain Concentration Potential Note</li> </ul>
Fire	sut flammable			Ignition Temperature. No. perfinent Electrical Hazard: No. perfinent Burning Rate, Not pertinent	9 SELECTED MANUFACTURERS 1 Stautter Chemical Cumpans Industrial Chemical Dission Westpurt Cump (misso)
Exposure	Call fee medical and Liquid Innahing to thin and eyes. If sustained and it cause nois Remove confaminated clear Plugh alfected areas onth git (F 18 Net 20 Mill workda to (F 20 Net 20 Mill workda to	eza, vomiting, ur losa uf comecourgena. Ing and vhive prity of varer per and fluch with plents of varer in a COPSTIDUS have rectom dank eater indust number indust 1000 kl 1000 kl 4 v I/G CON- g except keyr on los varm		7. CHEMICAL REACTIVITY Reactivity with Water: Nu reaction Reactivity with Common Materials: Curroles mout metals but reaction is nut harardisus Stability During Transport: Stable Neutralizing Agents for Acids and Causties: Fluid with water	PPro Industries Incomposited     Industries Incomposited     Industries Remusel Network     One Gratemas Center     Provident Products Corporation     Sone Philop Ra     East Providence (R. 1. 02414
			75	Polymonization: Not pertinent Inhibitor of Polymorization: Not pertinent	10. SHIPPING INFORMATION 10. Grades or Purity: 30 "0" substaints in ester 10.2 Storage Temperature: Ph3"F 10.3 Inert Atmosphere, No require test 10.4 Vanting: Pressure sectors
Water Pollution	Dangerious to aquaric life in May he dangerious if it enter Notify limal health and wild Notify operations of nearby	high concentrations t water intakes lefe officials agree intakes			
1. RESPONSI Sea Anterna Com Issue Haffing Official Access Restrict Access Disperse and flu	E 10 DISCHARGE maximum regions regions an anter umfarmer rat	2. LABELS Nor lanet requiring hy Code of Heneral Regulations	] [_	11. HAZARD ASSESSMENT CODE -See Hard Assessment Code 3	<ol> <li>PHYSICAL AND CHEMICAL PROPERTIES</li> <li>Physical State of 19°C and 1 atm. Experies</li> <li>Molecular Weight: Sur perimetri</li> <li>Beining Point at 14m. Suppris 2</li> <li>212°F = 100°C = 371°A.</li> <li>Freezing Point: appris 1</li> </ol>
3. CHEMICAL Statum Andre Statum Andre Statum Andre Statum Andre Statum S	L DESIGNATIONS with misurfi 3e sen suttide misurgist mpatibility Classification. U Ia Numeri HETS strong Numerical I Interd	OBSERVABLE CHARACTERISTICS     Physical State (as shipped): Liquid     traiter solution:     Color: Light lemon pule velum     imber : clars red     Odor: Rotten egy	12 1	12. HAZARD CLASSIFICATIONS Code of Poderal Regulations: Not Noted NAS Hazard Rating for Bulk Water Transportation. Category Rating Fire d Health Vapor Irritant d Linguid or build Irritani d Posume d Water Pollyumen	<ul> <li>6) P. + 171 C 2001k</li> <li>13.5 Critical Temperature: Nik perking m.</li> <li>13.6 Critical Pressure: Nik perking m.</li> <li>13.7 Specific Gravity: J. 41.1510 (huma)</li> <li>13.8 Liquid Surface Tennen Data nut assurable</li> <li>13.9 Liquid Water interfacial Tension Nut permetsia</li> <li>13.10 Veper (Gae) Specific Gravity Nut permet.</li> <li>13.10 Veper (Gae) Specific Gravity Nut permet.</li> <li>13.11 Reflect Specific vision of Veper (Gas) Nut permet.</li> </ul>
S.1. Personal Protects printf gragies, a sprintfession Pollow public scatters questing scatters transition for a constant of house constant of house constant of house constant of house constant for a present of the source source scatters to mere scatters present of the source to mere scatters present of the source to mere scatters present of the source to mere scatters to mere	5. HEALTH the Equipment: Runner pro- sitions and set of parespiration on groupment - Inflation of the provide set of the set of a failure and beath haracid ext. Fail of sching, in groupe the sub- sition and set and set of the of the set and set of the provide anti-the set of set of the set of the set of the provide anti-the set of the set the set of the set of the set of the provide anti-the set of t	NALAROS NALAROS Interequipment success agron touts selection we selecularized threathing apparates must puese instancing in respirators trace and gas anish may the given off anter applies prevent, untrinned exposure can be able to the selection contributed exposure can be able to the selection dispersions in a size can be able to the selection of garded exposed can be able to the selection of garded exposed in the interval be detribed in tauval and is one of a contributed by detribed interval and is one of a contributed by detribed associations in the selection of the selection association in the selection of the selection of the selection of the selection of the previous contributed as the selection of the contributed exposed on the bless emined the selection of the selection of the selection of the selection of the contributed exposed on the bless emined the selection of the contributed of the selection of the selection of the selection of the contributed exposed on the bless emined the selection of the contributed of the selection of the selection of the selection of the contributed the selection of the selection of the selection of the contributed the selection of the selection of the selection of the contributed the selection of th		Human Tususity I Aquatis Extent I Aquatis Extent I Reactivity Other Chemicals I Water I Water I Well Reaction I BIPPA Hazard Classifications Nin twice S. MEALTH HA It patient is uncunicitius do but give anishing this tungue should be kept forward and faile to commute their placed in a fast down position for tungue should be kept forward and faile to commute their placed in a fast down position for the placed in a fast down position for the placed in a fast down position	13-12 Latent Hear of Vegunzetion Nul per inclu 13-13 Meator Combustion - Nul per inclu 13-14 Heator Decomposition - Nul per inclu 13-16 Heator Decomposition - Nul per inclu 13-16 Heator Polystanzation - Nul per inclu 13-16 Heator Polystanzation - Nul per inclu 224RDS (Con 1) Heat Inclusion - Nul per inclusion eth remuned - Hearing heats back of autorate heator - Data non-inclusion
ay unun an property rigitm amà na tra argan an Rin ar a argan an Rin ar an argana an Rin ar an argana an Rin ar an argana an Rin ar an argana an Rin ar an	nie, while awaring in station imprevens applied to the sites a station medical attention of national medical attention of national posters is used multiple accession and applications intering maxing maxing induced hy rus raged ant sche sometaus is clea	citizm phosecon porcest may be replin a data be grandbar Skills, im nedicers they all the office of an operate Skills States and the second state into State of the Skill Skill Skill and the states table with the port of a state, in the measure s long may be parent other at onthe linger. Symiling	5 4 5 7 5 8 5 9 5 9	Testerly by Ingestion: Grace 2: LD-0.4 Stork g Late Testerly: Data and as manife Vesor (Gas) Instant Characteristics: Mapurou Stall high sumestrations anotes and the refer Liquid or Solid Instant Characteristics: A sur- degree Surisy atter a tem minutes contast Odor Threehold; 1: 5547 apm	kg avidem allerik el Malu un sugen i Aurge Lichtonik Fortengung is obteresik in mant Max Lavie Balik (habing) i s

824

.

. . . .

### SHD

:

۰.

÷ .

## SODIUM HYDROXIDE

	فالمطابأ فالجام والمطاطرة ومستحدياتها والمتحاطية الأمر ويحمره		
Connect branning Contra and Lyn Solid Robes or pa Solid Robes or pa Solid Robes or pa Solid Robes or pa Solid Robes or participation A cond contact with and and data. Every p Hear robes contractions a including advant	Ente Where OderWes rich weiter regis swey	6. FIRE HAZARDS 6.1 Plash Point: Not Rammable 6.2 Planmable Limits in Air: Not Rammable 6.3 Fire Estinguishing Agents: Not pertinent 6.4 Pire Estinguishing Agents Not to be Used: Not pertinent 6.5 Special Hazards of Combustion Products:	8. WATER POLLUTION     8.1 Aquastic Taslethy:     125 ppm //b hr //musquite fish. TLgs. Free     150 ppm //23 hr //by vers //ichel-sali wate/     8.2 Waterfood Taslethy: Data not a-zulable     8.3 Biological Dargen Domand (BOD):     None
Soup decharge of possible brokers and remove decharged material. Source break health and pathones control as	Man.	Not perintent 6.6. Bohavior in Piro: Nut perintent 6.7. Ignition Tomporature: Nat flammable 6.8. Electrical Hazard: Nut perintent	6.4 Food Chain Concentration Polontial: None
Nor Remaster May cause fire on consists o Planneshie gas may be prod Near orbiter unrecitableg (u Fire Fire Could decharge serve with o Could exponent constances was	ide combasilitas, anné an opniscu vidé metals, cladéag divens, éne h aoster	6.9 Burning Role: Not flammable	9. SELECTED MANUFACTURERS 1. Diamond Shamiouk Curp Electric Chemicals Diricium Deer Park, Tex 775th 2. Dem Chemical Co Midland Mich Rhindu
CALL FOR MEDICAL AID DUST Initialing to expert, near and Nove to freigh air If beaching is a support, gar If Dearling is a support, gar SOLID Wall been skin and gree. Named of garsliewed Remove contaminated clich IF is EYES, hold cyrcleh ag IF SWALLOWED and victor or walk. Do NOT INDUCE VONTTI:	hreas: argen argen en and fluch with plansty of water ng and shown. ng of water en and fluch with plansty of water is CONSCIOUS, have viction drink water KG.	<ol> <li>CHEMICAL REACTIVITY</li> <li>Reactivity with Water: Dissuisce with literation of much heat, may steam and optimiter</li> <li>Reactivity with Common Materials: When wet, attacks metals such as atummum tin, feed, and zinc to produce flammable hydrogen gas.</li> <li>Stability During Transport: Studie</li> <li>Houristizing Agonts for Actide and Counties: Flesh with water rime with divise acets: acid.</li> <li>Polymorization: Not pertinent</li> <li>Inhibitor of Polymorization: Not pertinent</li> </ol>	PPC Industries Inc Industries Inc Industries Incursion Barberion I Ner-64201      10 SHIPPING INFORMATION      10 I Grades or Purity: Technical Bakes USP reliefs      10.2 Storage Temperature: Ambient      10.3 Inert Atmosphere: No requirement      10.4 Meeting: Com
Water Pollution Series of marks of marks	hgð concentrations. «sner mukka. á officialu. sner inteka.		104 Venting: Opin
1. RESPONSE TO DISCHARGE the Reserve Normal Normalian CO and a live a safety correnae Revinci acces Disperse and flush	2. LABELS No hazard label required by Cude of Føderal Regulations	1). HAZARD ASSESSMENT CODE Iter manuf Australian Roveran CC +40-35 SS	13 PHYSICAL AND CHEMICAL PROPERTIES 13 Physical State of 15°C and 1 atm: Noted 13 Molecular Weight: 40.00 13 Botting Point of Yatm: Vers high 13 & Freezing Point: rest? = 510°C = 101°h.
3. CHEMICAL DESIGNATIONS 3.1. Synonyme: Causin, unda 3.2. Ceasel Guard Competibility Clessification: Not applicable 3.3. Chemical Formula: NaCH, 3.4. IMCO United Nationa Numerical Designation: 8.0/1821	4. OBSERVABLE CHARACTERISTICS 4.1 Physical State (as ahipped): Solid 4.2 Color: White 4.3 Odor: Odories	12. HAZARD CLASSIFICATIONS         12.1 Code of Federal Regulations: Not Inted         12.2 MAB Hazerd Reting for Burk Water Transportation: Not Inted         12.3 NFPA Hazerd Classifications: Category Classifications Health Hazerd (Bluet)         Health Hazerd (Bluet)         Flarm Ables (Red)         Reactivity (Yellow)	<ol> <li>Critical Temperature: Not pertinent</li> <li>Gritical Pressure: Not pertinent</li> <li>Specific Gravity: 213 at 2010 could</li> <li>Liquid Surface Tension: Not pertinent</li> <li>Liquid Water Interfacial Tension: Not certinent</li> <li>To the Cash Specific Gravity Not pertinent</li> <li>To the Cash Specific Gravity Not pertinent</li> <li>The Specific Hosts of Vapor (Gas). Not pertinent</li> <li>Latent Host of Vapor solution: Not pertinent</li> <li>Latent Host of Vapor solution:</li> </ol>
S. HEAL     S. Neal     S. Heal     S. Personal Protective Equipment: Chemical     respirator: rubber book, rubber     book, rubber     S. Symptome Following Exposure: Nirony cor     dust may cause damage to unper repirator     writiation in pneumonits. INGESTION, sig-     formation or perforation may occur. EYE 6     Treatment for Exposure: INHALATION to     physician INGESTION give water or mi-     induce romming. SNIN, such immediately	IN HAZARDS afers pupples. Face shield, filter or dust-type rosive action on contacted tosues. INHAL ATTON itract and tung isself, producing from mild nos- rere damage to macuas membranes, severe scar IONTACT produces service damage emose from exposite, support respiration, call followed to divide singer or true juce do NOT with large quantities of water under emergency		13 13 Heat of Combustion: Nut pertinent 13 14 Heat of Decomposition: Nut pertinent 13 15 Heat of Solution: Nut pertinent 13 16 Heat of Polymerization: Not pertinent
<ul> <li>Burger Carlos State States and States and</li></ul>	icopious amounts of nater for at least 15 min , we): "Sut pertinent olatile ere skin terman. Cause second: and third-degree	NO	Constituent as pages 1 and 8 -
burns on share contract and is very injurious 5-18: Odor Throshold: Not pertinent	io the even		

**D-**20

....



:

.....

D-21

STY

.

.....

· · · ·

- has a sin and

· .

### STYRENE

B WATER POLLUTION
 Aquatic Toxicity:
 22 ppm We be /Nucpill/TLm (fresh eater
 B2 Weterfoor Testelty: Data non-available
 Biological Organ Demand (BOD):
 If % ((Dear) 412 days

8.4 Food Chain Concentration Petential.

9. SELECTED MANUFACTURERS

Minsanio Co Minsanio Polsmers & Perrochemicals Cu VIII Norch I, indheren Blod Ni Luuis Ato A3rbh

10, SHIPPING INFORMATION

510 psia = 39 46 atm = 4.00 MIN m

Concinued an pages 3 and 8 1

13.7 Specific Gravity: U Vikiai 21°C (liquid)
13.8 Liquid Surface Tension: 32.14 di teckimi e 0.0214 N. m. at. 911
13.9 Liquid-Water Interfacial Tension: 35.4 di knowler in 0.0254 N. m. at. 1911

13.10 Vapor (Gae) Specific Gravity: Not periment 13.11 Ratio of Specific Heats of Vapor (Gae)

1 074 13 12 Latent Heat of Vaporization

NOTES

10.1 Grades or Purity: 99.5+7 10.2 Storage Temperature: Ambien; 10.3 Inert Atmosphere: No requirement 10.4 Venting: Open (flame arrester)

None

Amixo Chimicals Curp

910 South Michigan Ar-Chicago, III: Nim05 Dow Chemical Co Midland, M. ch. 45640

MonsahioCo

1

2

٦

			ويراد المتحدين والمتحد فللمحد والمتحد والمحدين والمحدود والمحدوق والمحد			
	Commen Senas Stored Visothensette Physiothylogy	urma Vesiary bquud	Colorina to light yellow - Sweet pirsont adar	6 1 6 2	6. FIRE HAZARDS Flesh Point: 93°F O.C 64° Flammable Limits in Airt. 1	FCC 13 613
		Flasis on we	e Planmable, initiating rapor is produced	63	Fire Extinguishing Agents: fog-loam-carbon-dioxide. Fire Extinguishing Agents F	Water , ur drs chemical fot to be Used:
	Avuid Ciinta Near chemic Stup dischar Call fire dea	ct with liquid and rapor th cal protective suit with self ye if possible actionst	op propie a ve) ontained breathing apparatus	65	Water mas be ineffective Special Hazards of Combut Not pertinent	ilion Products:
	- Instate and r Notify local	where discharged material health and pullution contr	i agencias		Behavior in Fire: A apur is hi air and may travel conside distance to a source of igni	eavier Lhan rable Llion and Nash
		Cumbustible CONTAINERS MAY 8: Flashback along vapor 1 Vapor may explude if y Weak chemical organization	PLODE IN FIRE. sid may occur. wind us an enclowed arts - will with welf-unklaned breathing apparatus		hack. At elevated tempera as in fire conditions, polyn may take place which may container explosion	tures such nerstation lead to
	Fire	Comhat fires frum wie	instance or protected lucation.	6.7	ignition Temperature: 914*	ł
		Water may be ineffective Curil exposed container	un fire with water	6.8 6.9	Electrical Hazard: Class E C Burning Rate: 52 mm/min	Group D
		CALL FOR MEDICAL	ND	1	7 CHEMICAI REACTIN	
		Imiating to eyes, nove a	d throat.		Reactivity with Water: Non	cuction
		Move to fresh air		72	Reactivity with Common Ma	iterials:
		If breathing is difficult.	pre azygen		No reaction	
		LIQUID Will burn skin and even		7.3	Stability During Transport: Neutralizing Agents for Acid	Studie In and
	Exposure	Harmful if swallowed. Remove contaminated	athing and shores		Caustics: Not periment	
	·	Flush affected areas with IF IN EYES, hold evelo	pleniv of water upen and flush with plenty of water	7.5	Polymerization: May occur i 150°E C in cluse runtures	f heated above
		IF SWALLOWED and v	tim is CONSCIOUS have victum drink water		Metalisalis, perovides, and	verong acids
		DO NOT INDUCE VON	ITING		may also cause polymerizat	100
					bulyicatechol, 10-15 ppm	Termary -
				4 1		
	Water	Fauling to shoreline				
	Pollution	Notify local health and	nidiuf officials			
	1 RESPONSE	TO DISCHARGE	2. LABELS		11. HAZARD ASSESSMENT	CODE
	See Response Marine	46 Mandbook CG 446-41			See Hagard Assessment Handbook	CG +++ 31
	Issue warning	air contaminant	No hazard label required by Cude of Ender it Regulations		A-1-U-Z	
	Should be temp	ed				
	Chemical and pl	homeat treatment	·			
				I I	12. HAZARD CLASSIFICAT	TIONS
	3. CHEMICAL	DESIGNATIONS	4. OBSERVABLE CHARACTERISTICS		Not listed	•
31	Synonyms: Phene	thylene Phenviethylene.	4 1 Physical State (as shipped): Liquid	12.2	NAS Hezard Rating for Bulk	Water
	Styrol, Styroler	ne: Vinylbenzene	4.2 Color: Coloriess		Transportation:	Relies
32	Monumer	spenomy creamcanor	4.3 Odor: Sweet at low concentrations characteristic pungent, sharp,		Fire	3
33	Chemical Formul	III: C.H.CHICH;	disigreeable		Health	
34	INCO United Nat	lione Numerical			Vapor Irritant Lound or Solid Irritant	2
	oregnation. 3 3	2034			Poisons	2
				1 1	Water Pollution	L.
					Aquatic Toxicity	3
	Baraanal Brate-H	5. HE	LIN NALARUS		Aesthetic Effect	2
	ploves book p	uggles or face shield	nee nee of approved entries request as presses		Reactivity Other Chemicals	2
52	Symptoms Follow	ring Exposure: Modera	circitation of eyes and skin. High suppr concentrations		Water	0
4.1	cause diffiness.	drunkenness, and anesthe mesure: (Schial ATION	remove to fees the keep a semi-and superture	123	Sell-Keaciion NFPA Nazard Classification	, 
	artificial respira	ition if needed INGESTI	IN do NOT is suce vomiting, call physician, no		Calegory	Classification
	known antidute attention	SKIN OR EVECONT	CT. Rush with plenty of water, for every get medical		Health Hazard (Blue)	2
54	Tezicity by Inhala	tion (Threshold Limit V	lue): (00 ppm		Flammability (Red) Reactivity (Yellow)	3
55	Shert-Term Inhel	ation Limits: 106 ppm fo	i jū min	4 I	·····	-
56	Tericity by Ingest	ion: Grade 2 UDa 031a	5 øj kø			·
57	Lefe Texicity: Da	ta nut available	an a sua muda su su su su su su kukuku manana a M			
	find high concer	nerations appleasant. The	ns sause moderate critation such that personnes will flact is temporary			
59	short expusive.	illant Characteristics: ( mus cause secondars bur	auter smarting of the skill and first degree burns on s on long exposure			
5 10	Oder Thresheld:	0 144 ppm				
				1 1		

D-22



D-23

ø .

TOL

.

## TOLUENE

Commun Syrney Talasi Markiy Hongana athy Hongani	irma Watery kquad Ploats on water Pl	Coluters Pleasant odor - ammable, unitating reports preduced.	6.1 6.2 6.3	6. FIRE HAZARDS Plack Point: 40°F C C , 35°F C C Flammable Limite in Air: 1.276 - 75 Fire Extinguishing Agents: Carbon dounde or dry chemical for small fires, ardinary fount for large fires	8. WATER POLLUTION 8.1. Aquate Tasicity: 1.150 mg/1.456 hr-sunfish.7 Lm. froih-ait 8.2. Waterland Yasicity: Data not assiste 8.3. Biological Dargen Damand (BOD): 0.45. Supy. 343 (Herer 1 & dass
Strip discha Shut off ign Stas upwin Asind cimit fundate and Nutify linial	pe if punchle. Keep peuple away utum sources and call for depart § and our macro spray in "Source co each logue and vapur remove docharped material health and pullutum cuntrul age	ngai diwa" vapar	6 4 6 5 6 6	Fire Estinguishing Agents Not Is be Used: Water may be ineffective Special Hazards of Combustion Products: Not pertinent Bohavior in Flost: V agen is heavier than air and may travel a considerable	8 6 Food Chain Concentration Polonital. None
Fire	FLANMABLE Flachteck survey require trail may accret: Vapor may caplude if spinted in an enclosed area. Very program and wife contained breaking apparetus Eximption with dry chemical foam us carbon drowide Water may be ineffective on file Configuration of the spined containers with water		67 68 69	diviance (ria y survice of ign (rin and Bash hack Ignition Temperature: 1977) Electrical Hazerd: Class 1 Group D Burning Rate: 5.7 mm (mix	9. SELECTED MANUFACTURERS 1 Exion Chemical Cu Housion Tex 72001 2 Sheli Chemical Cu Petrichemical Division Housion Tex 77011 2 Sheli Ce
Exposure	CALL FOR WEDICAL AID VADD Imitating to eyes, nove and th If inhabrd, will cauve nauses, difficult breathing, or la Wove to (rev) ar I breathing has stopped, give I breathing bit firstill, give os LIQUID Imitating to skin and eyes. If iselflowed, will cause nause Remute contamunated Clathu Plush affected areas with plus If IN EVES. hold eyethin op IF SWALLOWED and victim or multiplication of the stopped and DO NOT INDUCE VOMITIN	rust. comming, leadsche, dizzinem, u of cunsciouwnem. artificial empirebiun teen s. vomiting or loss of convciouenem. ng and shoes. ity of water in and flush with plenty of water. is CONSCIOUS: have victim drink water G.	7 1 7 2 7 3 7 4 7,5 7,6	7. CHEMICAL REACTIVITY Reactivity with Water: No reaction Reactivity with Common Materials: No reaction Stability During Transport: Stable Neutrolizing Agents for Acide and Caustics: Not pertinent Polymerizables: Not pertinent Inhibitor of Polymerization: Not pertinent	10 SHIPPING INFORMATION 10 SHIPPING INFORMATION 10 Grades or Purity: Research respect actation att 99 ke %, industrial curvates 44-% with VE science and servic amounts of Resizence and neuralistic budriserbuilts, 40 (20) less pure that industrial
Water Pollution	Dangerous to aquatic lafe in hi Fouling to shorebane May be dangerous if it enters s Nonfy incal nealth and wildli Nutify operators of nearby wi	gh cuncentrations. •ster intakes. fe officials. tier intakes			102-UTAL 102 Storage Temperature: Amhieni 103 Inori Atmosphere: No requirement 104 Venting: Open (fame arrever) or prevere-secular
3. RESPONS ISee Assessed for Issue warning Esacuate area	<b>IS TO DISCHARGE</b> See readance CG 446-44 High Rammability	2. LABELS		1). WAZARD ASSESSMENT CODE IS IN MARK ALLEGATION MONTHAL CO NO -31 A-T-U 12. HAZARD CLASSIFICATIONS	13. PHYSICAL AND CHEMICAL PROPERTIES           13.1 Physical State at 15°C and 1 atm:           13.2 Material State at 15°C and 1 atm:           13.3 Beiling Point at 1 atm:           23.1 °F = 110 *C = 33 8°A           13.4 Freesting Point:           -19°F = -93 0°C = 17h 2°A
3. CHEMICA 3.1 Synanyma: Mai Mai 3.2 Coast Guard Ca Ara 3.3 Chemicst Farm 3.4 JMCO/United N Designation: 3	IL DESIGNATIONS hy:Ibenzene hy:Ibenzene hy:Ibenzel woi wmpetibility Classification: maic hydrocarbon ute: CaH-CH- etions Numerical 2: 1294	<ol> <li>OBSERVABLE CHARACTERISTICS</li> <li>Physical State (as shipped): Liquid</li> <li>Color: Colories</li> <li>Odor: Pungeni, aromatic, benzene-like, distinct, pleasant</li> </ol>	12.1	Code of Federal Regulations: Fiammable liquid NAS Neazard Rating for Built Water Transportation: Category Rating Fire 3 Health Vapor firmant 1 Liquid or Solid firmani 1 Poisons 2 Water Pollucion	13.5         Critical Temperature:           NO.476 = 318 AFC = 501 8FN           13.6         Critical Pressure:           59.1         pus = 40.55 stm = 4.108 MN/m <sup>2</sup> 13.7         Specific Gravity: 0.86 <sup>3</sup> at 20°C (riquid)           13.6         Liquid Surface Tension:           29.0         dyneycm = 0.0200 N m at 20°C           13.9         Liquid-Water Interfacial Tension:           30.1         dyneycm = 0.040 N m at 20°C           13.10         Vapor (Gas) Specific Gravity:           No. pertilient         No.1040 N m at 20°C
S 1 Personal Protect     S 2 Symptoms Fold     headache, an     aspirated, cau     ingevied cause     ovygen if need     fush with wat     S 3 Staactity by Inho     S 5 Shart-Term inho	5. HEALTI tive Equipments : Vapors irriti sitiesia, respiratory arrest Liqs we coupling, gagging, distress voming, griping, distress voming, griping, distress appeare: INHALATION re- led, call a dector INGESTION er for at least 15 min SNIN w lation (Threaheld Limit Valu- bation Limits: 600 ppm for XV	4 HAZARDS [mask, goggles or face shield, plastic gloves are eyes and upper respirators tract, cause diziness and rapidly developing pulmonars edems. If epresed respiration move to fresh arr, give artificial respiration and s do NOT induce somiting, call a distor. EYES ipp off, wash with soap and water b): 100 ppm min	12 3	Human Tosicity i Aquatic Tosicity 3 Aesthetic Effect 2 Reactivity Other Chemicals I Water 0 Setif Reaction 0 NFPA Hazard Classifications: Catogory Classification Health Hazard (Blue) 2 Fiammability (Red) 3 Reactivity (Yellow) 0	<ul> <li>13 11 Natio of Specific Peers of Vapor (Cas): 1312 Latent Neel of Vaporization: 112 Bluello = 86 (cal); = 1.61 × 107 J sp 1313 Neet of Combustion: -17.4 N Bluello = -058 cal; = -055 × 107 J sp 1314 Neet of Decomposition: Nuclearlinet 1315 Neet of Solution: Not perform 1316 Neet of Polymorization: Nuclearlinet</li> </ul>
<ol> <li>Se Testicity by Ingel</li> <li>Lete Testicity: A</li> <li>Vaper (Ges) triticity: Ingel</li> <li>Sector (Sector (Secto</li></ol>	allon: Grade 2, LDo 0 5 to 5 g Johes and liver damage mail to lant Choracterfolice: S apur- lant night consentrations. The renant Charactoriettes: Mini- s cause smarting and reddening c 0 17 ppm	ing How ingestion Cause a slight smarting of the eves of tespirators effect is temporary mum harard. It splited on clushing and allowed pof the skin		NO	Command un parri 1 and A

D-24

:

.



D-25

VAM

## VINYL ACETATE

Comment Stream, Yydd Ywyd a Siog daethargo o' grwn Shod off upper a samu Area Control by Shod off upper a samu Area Control and uro Shod Stream, and Shod S	Wassery liquid Colo Plasts on water. Planmable, union de Long prophe sway net deal for department. Inter party to "lanck dows" repar. harred na printed. publisher control agencian publisher control agencian MABLE	rien Plennni fruity odor ang rapar a produced.	6.1 6.2 6.3 6.4 6.5 6.6	6. FIRE NAZARDS Plack Paint: 18*6 C . 23*6 O C Planmable Limits in Air: 245 – 1345 Pire Extinguishing Agents: Carbon draside or dry chemical far unall fires, ordinary fare in for large fires Pire Extinguishing Agents Not to be Used: Water may be ineffective Bapecial Maaarde of Combustion Products: Not permant Bohevier in Pire: Vapor in heaver than air and may travel considerable distance to a supress of grantman and Bach back May poinmerze when heated in a fire at heavier means	B WATER POLLUTION     Aquatic Teasing:     Id Aquatic Teasing:     Id ppm, 76 hr, Discpill TL,m, Irch 1 are     Plu ppm, 76 hr, Discpill TL,m, Irch 1 are;     2 Waterbard Teasing: Date his are;     Balogical Daypon Domond (BOD):     27% (there is 10 days     Food Chain Concentration Patential:     None     Statistic Automation Patential:     None
Fire Fire Cont	Forbacts along upper real may access. Forbacts along upper real may access. Vaper may explode if general man an enclosed area Were paging and all's consisted because appoints. Fire Combast firms from wire detakene or presected location. Et ingrued with dry chemical, from, carbon desude Cool explosed containers with water		67 64 63	er Stupfall Conjune: Ignition Temperature: 300°F Electrical Hazard: Clavit Group D Burning Role: 33mm-min	3 SEECIED MANUFACTURERS     Creance Corp     Celonce Chemical (or Tossium     245 Park Ave     New York, N. V. 10017     National Distiller-& Chemical Corp     U. Schwarder Chemic, Corp
CALL VAR Imuti If under If bere If bere If bere If bere If bere Fluch IF IN IF Sh	TOR MEDICAL AID is to spee, now and threat. net will count datamens or difficult are here has suppord, and antificult report here has suppord, and antificult report is to this and area. Is realized or it spilled on skin. reasonnaired cirching and shoes. reasonnaired cirching and shoes. Test and area with plents of or opter YES, hold ayaluds open and fleah with LL OWED and vectors is CONSCIOUS. multi-	athing ion plenty of water have viction densk water	7 1 7 2 7 3 7 4 7 5 7 6	<ol> <li>CHEMICAL REACTIVITY</li> <li>Reactivity with Water: No reaction</li> <li>Reactivity with Common Meterials: No reaction</li> <li>Stability During Transport: Nichle</li> <li>Noutralizing Agents for Acids and</li> <li>Coustica: No pertinent</li> <li>Polymerization: Can occur when in contact with perovides and strong such but on - under estieme conditions</li> <li>Inhibitor of Polymerization: 3 - 5 ppm or 1417 ppm bydroquinone: Shipment- usuals also comain 200 ppm of diphenylamine</li> </ol>	Housion, Tex 1786 J. Unich Carthole Curp Cremision and Plasics, Urinion 270 Park Ave New York, N. Y. 10017 10. SHIPPING INFORMATION 10.1. Grades ar Purity: Grades A (Diphenstamme inhibited) 90 kg Grade M (Midniginne inhibited) 10.2. Storage Tempersture: Amilient
Water HARM Foulur Pollution Solid Solid	UL TO AQUATIC LIFE IN VERY L to shorehee dengrous of stenters water intakes, each heath and widdle officials, sporaturs of searby water intakes	OW CONCENTRATIONS.			103 Inert Atmosphere, No reduitenten 104 Venting: Pressure valuem
RESPONSE TO DI- tean Augurea statutes names Issue warning — high Ram air contaminant Es acuate area	CHARGE n. co 44-41 mabile).	2. LABELS DURETCE ITA PRESENT		11. HAZARD ASSESSMENT CODE	13. PhYSICAL AND CHEMICAL PROPERTIES 13.1. Physical State of 15°C and 1 atm: Liver 13.2. Molecular Weight: In ON 13.3. Boiling Point of 1 atm: 10.12°F = 72.4°C = 140.1°A 13.4. Pressing Point: -13.0°F = -22.4°C = 161.4°A
3. CHEMICAL DESIG 3.1 Synonyms: VAM Vinji A mono Vy Ac 3.2 Ceast Guard Compatibul Polymerizable 3.3 Chemical Pormula: CHA 3.4 INCO-United Maximum Nu Designation: 3.2,1301	ATIONS 4. 00 167 4.2 Color: 4.2 Color: 4.3 Oder: 5107 5107 510 SUB 5107 5107 510 SUB 5107 5107 5107 5107 SUB 5107 SUB 5107 SUB 5107 SUB 5107 SUB 5107 SUB 5107 SUB 5107 SUB 51	SERVABLE CHARACTERISTICS of Blote (so ohipped): Liquid Coloriess Not unpleasant, sweet smell in all quantum: pleasant fruits, rectermine	12 1	Code of Federal Regulations: Final make liquid Nammake liquid Transportation: Cotegory Roting Fire 3 Heelth Yapor Irritani 1 Liquid of Solid Irritani 1 Poisono 2 Water Politistion	13.5         Critical Temperature: 48(*F) = 252*C = 325*K           13.6         Critical Pressure: 517 atm = 42 pus + 4.25 M/K m²           13.7         Specific Gravity: 0.44 at 25*C (riguid)           13.8         Linding Surface Tempion; 20.95 dimes/cm = 0.021% N m/at 20*C           13.8         Linding Surface Tempion; 20.95 dimes/cm = 0.021% N m/at 20*C           13.8         Linding Kenster Hootback Tempion; (CS) 1.00 dimes/cm = 0.01% N m/at 20*C           13.0         Value (Geo) Specific Gravity; Not periment
S. HEALTH HAZARDS     S. HEALTH HAZARDS     S. HEALTH HAZARDS     S. HEALTH HAZARDS     S. How the second sec		Aqualic Toxicity II Accileree Effect 2 Reactivity Other Chemicals 2 Water 0 Self-Reaction 3 2.3 NFPA Hazard Classifications: Category Classific Health Hazard (Bjuer 2 Flammability (Red) 3 Reactivity (Yellow) 2		$ \begin{array}{l} (3-1)  \text{relation of Specific Hosts of Vapor (Gas).} \\ (-10) \\ \hline 13-12  \text{Latent Host of Vapor (astion - 1n) thus the } \\ = & \forall 0 \text{ trailing = } \forall 9 \times 10^{12} \text{ stars}^{-12} \\ = & \forall 0 \text{ trailing = } \forall 9 \times 10^{12} \text{ stars}^{-12} \\ = & \neg 3 \text{ stars}^{-12} \\ \end{bmatrix} \\ \begin{array}{l} \text{Host of Combustion:} & \neg 4^{14} \text{ stars}^{-12} \\ = & \neg 3 \text{ stars}^{-12} \\ \text{Host of Occomposition:} \\ \hline 13-14  \text{Host of Occomposition:} \\ \hline 13-16  \text{Host of Solution:} \\ \hline 13-16  \text{Host of Polymort starshine:} \\ \hline 13-16  \text{Host of Polymort starshine:} \\ \hline -244 \ \text{ trailing = } = 10^{12} \text{ stars}^{-12} \text{ stars}^{-12} \\ \hline \end{array}$	
<ol> <li>Vapor (Gos) irritant Characteria: sa: Vapors cau le a slight smarting of the even or respiratory system if presen in high concentrations. The effect is temporary.</li> <li>Uquid as Solid Invitant Characteristica: Minimum hazard. If spilled on clothing and allowed to remain, may cause smarting and reddening of the skin.</li> <li>Oder Thresholdt: 0.12 ppm</li> </ol>				NOT	5 and 201 and 201 and 2



#### DOT ISOLATION AND EVACUATION DISTANCES

		First Isolate in	Then Evacuate in a Downwind Direction	
Material	Initial Isolation	All Directions (feet)	width <u>miles</u>	length <u>miles</u>
Acrylonitrile	30	60	0.1	0.2
Hydrocyanic acid	90	190	0.5	0.7
Methyl bromide	50	90	0.2	0.3

		MOP 600-4
FD	Manual of	SECTION EMERGENCY SERVICES
	Procedure BROADCAST	SUBJECT SPILL INCIDENTS - SORBENT BOOMS
TAKE TH	E FOLLOWING ACTION	ONS IMMEDIATELY TO BRING YOUR
MANUAL	UP-TO-DATE:	
PLACE THIS SECTION 600	DOCUMENT IN NUMERICAL O - EMERGENCY SERVICES.	ORDER AFTER THE DIVIDER FOR
-	9 99 198 99 99 94 96 99 110 110 110	
In Section	644-3 across the top of	page 1, mark SEE BROADCAST 600-4.
This boom w sorption by alone by to spills of v are those m boomed, but <u>PURPOSE</u> :	ill not absorb water an linking together in ov wing through spill area irtually any thickness. ost effectively sorbed. sorption is not effici	nd can be used for booming as well as verlap configuration, or for sorption a linked end-to-end. Will work on Distillates and crudes above 40°F.(4°C.) Heavier oils can be equally well ent.
The primary are as foll	reasons for supplying ows:	sorbent booms to Fire Department units
. To proby boom storm openin for a	vide immediate, complet ning the gutter, stream drain inlet or discharg gs, or linking together catchment area.	e control of small oil spills us, runs, stuffing boom into e rivulets, sanitary sewer for forming levee or dike
. To pro larger up for Govern	vide immediate initial oil spills in accordan ces of the responsible ment can be mobilized.	control measures to contain nce with MOP 644-1 until clean- party, the State, or Federal
DISTRIBUTION:		
l. One so a plas	rbent boom will be carr tic trash bag.	ied on each engine and truck with
6/16/78		7 ** 2*** 2*2** Page 1 of
	-	

ŀ

			MOP 600-4
CTION		SUBJECT	· • ·
EMERG	ENCY SERVICES	SPILL :	INCIDENTS - SORBENT BOOMS
·			
2.	Three replacements an where Battalion Chief	nd/or spares will be re fs are quartered.	tained in stations
3.	Replacements will be (see Replacement Supp	made from Maryland Por ply - page 3 of 3).	t Administration
OPERATI	ONS:		
1.	Upon arrival at the s to Fire Communication	scene, the Officer in C ns as per item 1 [MOP 64	harge will report 44-1].
2.	When a single unit re sorbent booms are rea Communications office and specifically requ will also determine a should be EMERGENCY of	esponds to a spill incid quired, the Unit Offices e to dispatch additional uest that units respond and notify Communication or NON-EMERGENCY.	dent and additional r will notify Fire l units <u>as needed</u> with booms. He ns whether response
3.	Whenever three (3) or incident and a box al office will dispatch assume command at the	r more units are at the larm has <u>not</u> been sound the nearest Battalion ( e scene.	scene of a spill ed, Communications Chief who will
4.	If additional quantit in Charge will arrans Charge of Field Opera	ties of booms are still ge for delivery through ations in accordance wi	needed, the Officer the Officer in th [MOP 644-1].
5.	Units that are not no sorbent booms shall b	eeded after they have de be placed in service.	elivered their
6.	Clean-up and Disposal	1	
	a. Oil-contaminated plastic bag or in can.	sorbent booms shall be n a suitable container s	placed in the such as a metal
	b. Booms can be disp regular station to Utility Operation	posed of in the same mar trash is collected by Bu ns.	nner as the Greau of

ß

			MOP 600-4
TION		SUBJECT	<u></u>
EMERGENCY SE	RVICES	SPIL	L INCIDENTS - SORBENT BOOMS
Manufi da da ante de la compañía de	لأوالمتار ويسياه المتعالمين ويسير والأراب والمت		
c. S	orbent booms that	contain volatile	flammable liquids
a b 0	nd large quantiti e disposed of by perations crew th overnment clean-u	es of oil-contamin supervisor of the at responded or by p crews or contrac	ated booms shall Bureau of Utility State or Federal tors.
d. W 1 f	hen completing Sp t indicates "abso ootage of booms u	ill Report (210-04 rbed by sorbent ma sed.	l), in area where terial," show total
REPLACEMENT SU	PPLY OF SORBENT B	OOMS:	
	hen sorbent boom attalion Chief by	is used, Unit Offi phone and request	cer will notify the a replacement boom.
. W V a	hen the Battalion ill notify Marylan s follows:	Chief's supply is nd Port Administra	depleted, the Chief tion for replacements
	Marine 1 383-5764 237-833	Foreman, Robert F. 4 or 383-5104 or 9 - Voice Beeper	O'Conner
	Chief o: 237-858	f Debris and Oil, 1 2 - Voice Beeper	Marty Wallace
	391-1193	2 - Home (24 hours)	)
		By order,	
		Think	· Aprike_

. . .

.

×.\*

	MOP 600-7
Manual of	SECTION EMERGENCY SERVICES
BROADCAST	SUBJECT DISPOSABLE SURGICAL MASKS
TAKE THE FOLLOWING ACTI MANUAL UP-TO-DATE:	ONS IMMEDIATELY TO BRING YOUR
PLACE THIS DOCUMENT IN NUMER FOR SECTION 600 - EMERGENCY	ICAL ORDER AFTER THE DIVIDER SERVICES.
In Section 644-3 across the BROADCAST 600-7.	top of the page, mark SEE
Large amounts of dust are promaterial at spill incidents. the hazards of inhaling these surgical masks will be provide	oduced while spreading absorbent To safeguard members against e dust particles, disposable ded for respiratory protection.
Each first-line Engine and fist. six (6) masks, and Unit Offic masks when the need arises fi Chief.	ruck Company WIII be issued cers may secure replacement rom their respective Battalion
A supply of masks will be dia Chief's station from the Fire	stributed to each Battalion e Department Storeroom.
Disposable surgical masks are operations involving fire, su instances, self-contained bro	e not to be worn during emergency moke, or toxic gases. In these eathing apparatus must be used.
	By order,
	THOMAS J. BURKE, Chief
DATE 3/14/77	,-4, Page 1 of 1

			MOP 603-4
A PARA	Manual of	SECTION EMERGENCY SERVICE	
Jal Te	Procedure DETAIL PROCEDURE	SUBJECT	PREVENTION OF CONTAMINATION OF DOMESTIC WATER SUPPLY

Water from the Baltimore Harbor; fresh water streams in and around the City; static water supplies such as tanks, swimming pools, and other untreated water supplies, must be considered as polluted and under no circumstances shall water from these sources be interconnected with any piping system or any hose system that is connected to the domestic water supply.

Fireboats and pumpers at draft shall not be used to supply automatic sprinkler and standpipe systems.

Relays shall not be established where one source of water is the City water supply and the other is a fireboat or pumper at draft.

Hose lines being supplied by a fireboat or pumper at draft shall not be connected to any apparatus or appliance that is also connected to the City water supply.

Whenever a pumper has been used at draft, it shall be connected to a City water supply hydrant and shall be thoroughly flushed as soon as possible after such use.

Whenever it becomes necessary to provide emergency water supply for human consumption in buildings or ships, only new fire hose connected to a City water supply hydrant shall be used.

Page 1 of 1

			MOP 605	
A FED	Manual of	SECTION EMERGENCY SERVICE		
DALTY	Procedure POLICY	SUBJECT	HARBOR PROTECTION, FIREBOATS	

Fireboat No. l	Mayor Thomas D'Alesandro, Jr.	Dundalk Marine Terminal 2700 Broening Highway
Fireboat No. 2	Mayor J. Harold Grady	2609 Leany Street Locust Point
Reserve Fireboat No. 1	P. W. Wilkinson	Dundalk Marine Terminal 2700 Broening Hignway
Reserve Fireboat No. 2	August Emrich	Maintenance Bureau 1407 Key Highway

#### Assignment

All fireboats are under the jurisdiction of the Second Battalion who shall also be responsible for maintaining reserve fireboats in operating condition at all times.

#### Response

- A. Fire Communications will dispatch Fireboats No. 1 or No. 2 to box alarms as per card of assignment, each filling in for the other in accordance with standard fill-in policy.
- B. For fires reported aboard vessels in the stream; vessels in inaccessible areas to land units, lire Communications will dispatch both fireboats as per the following:

The nearest fireboat, immediately, to the scene of the incident and the nearest battalion chief, engine and truck to the station of the remaining fireboat, to be transported immediately to the scene of the incident.

- . The remaining fireboat will prepare to respond and stand by until the arrival of the land units.
- . The land units will take with them all equipment from their apparatus necessary for safe and efficient shipboard o erations, such as:

DATE

racourt attractor Page 1 of 4

MOP 605

SUBJECT

SECTION

EMERGENCY SERVICE

HARBOR PROTECTION, FIREBOATS

a. Air masks and spare air bottles

b. Manual and powered forcible entry tools

- c. Any equipment necessary to supplement the equipment carried on the fireboat
- . The pump operator will remain at the fireboat station and care for the apparatus and vehicles of the land units.

. After initial evaluation of the situation at the scene, the first-arriving unit will immediately notify Fire Communications of conditions and: (a) request any special equipment, (b) additional assistance necessary, (c) return unnecessary units.

- . When additional assistance is needed, Fire Communications will notify the Officer in Charge of Field Operations who will coordinate operations. Any available fireboat will be utilized to transport the additional personnel and equipment of the land units to the scene.
- . Fireboat commanders shall make provisions for having sufficient life safety devices aboard to accommodate the fireboat crew and all passengers.

#### Response When a Fireboat is Away From Station

When a fireboat is away from its station and an alarm of fire is received, Fire Communications will determine the fireboat's position and dispatch the nearest fireboat to the incident.

When a fireboat is away from station for other than fire or emergency service and a second alarm is received for an incident to which the other fireboat has responded, the fireboat shall return to station.

#### Transfers

Fireboat No. 1 or No. 2 will not be transferred; however, an activated reserve fireboat may be transferred by the Officer in Charge of Field Operations as conditions necessitate.

SECTION

SUBJECT

EMERGENCY SERVICE

HARBOR PROTECTION, FIREBOATS

MOP 605

#### To Place a Reserve Fireboat in Service

- A. A reserve fireboat will be placed in service at any time to take the place of either Fireboat No. 1 or No. 2 when the latter must be placed out of service for drydock, overhaul or routine maintenance.
- B. On other occasions, the Officer in Charge of Field Operations will arrange for and coordinate activation of a reserve fireboat in accordance with MOP 613 - Activating Reserve Apparatus and MOP 613-1 - Recall of Off-Duty Members to Duty.
- C. Under emergency conditions when Fireboats No. 1 and No. 2 are committed to an incident or incidents, the Officer in Charge of Field Operations will activate one or both reserve fireboats by utilizing the on-duty crews of land units, provided sufficient licensed fireboat personnel are available to properly operate the vessel(s).

#### Training

- A. Two hour harbor training cruises beginning at 0800 hours will be conducted as follows, weather conditions permitting:
  - 1. Fireboat No. 1 The first day of each day shift.

Fireboat No. 2 - The third day of each day shift.

- 2. The Captains of fireboats shall maintain an accurate record of training cruises.
- 3. Fireboats in reserve status will be utilized on alternate training cruises.
- 4. All fireboat personnel shall participate in training exercises with their assigned units.

#### B. Land Unit Participation

 From time to time as required, land units will participate in training cruises and instructions. This shall be devermined by Battalion Chief Fire Academy with the approval of the Chief of Fire Department and the scheduling of the participation will be issued on a general order.

12/3/79

**ž-**3

Page 3 of 4
				MOP 605	
ECTION		SUBJECT			
E	MERGENCY SERVICE		HARB	OR PROTECTION, FIR	EBOATS
	2. The schedulin	e of land units to	participate :	in fireboat	
	training crui training and	ses shall not confl inspections schedu	ict with the ed for the la	existing and units.	
	3. Land units wi to the firebo at 0745 hours	11 be placed out of at station assigned	service and with their a	report Ipparatus	
		•			
			_		

	MOP 605-1
Manual of	SECTION EMERGENCY SERVICE
Procedure DETAIL PROCEDURE	SUBJECT FIREBOAT OPERATIONS

- 1. Fireboats shall be operated in accordance with the U.S. Coast Guard Regulations insofar as they are applicable.
- 2. When marine units respond to alarms of fire or other emergencies where land units have likewise responded, the officer in charge of the marine units shall receive his orders from the officer in charge of the land units who will be in charge of operations. The officer in charge of marine units shall advise the commander of the land units the extent of the fire from the water side and the necessity of summoning additional fireboats should the situation warrant.
- 3. Monitor pipes on fireboats shall not be placed in service at fires unless so ordered by officer in charge of fire.
- 4. Handlines from fireboats must be shut down when monitor pipe is placed in service.
- 5. Upon leaving station for any purpose except alarms of fire, the officer in charge of the fireboat must notify the Fire Communications Office and respective Battalion Chief of his destination. Radio communication must be maintained.
- 6. In case of abnormal weather conditions or other emergency that may endanger a fireboat, the commanding officer of the fireboat shall notify the Battalion Chief of the existing conditions and request remission to move the boat to a more protected or safer location. The Battalion Chief or officer acting as such shall be responsible for the decision as to whether the boat is to be moved to prevent damage. In the event the Battalion Chief cannot be contacted, the officer in charge of the boat shall use his best judgement in the matter and be held responsible accordingly.
- 7. The relocating of fireboats while operating at the scene of a fire must nave the approval of the officer in charge, unless under extreme emergency conditions which are endangering the safety of the boat.
- 8. Officers in charge of hose lines being supplied by fireboats will notify the boat as to length of line and type of nozzle or a pliance being supplied.

12/3/79

DATE

Page 1 of 1

·	MOP 628-1
Manual of	SECTION EMERGENCY SERVICES
Procedure DETAIL PROCEDURE	SUBJECT CHLORINE LEAKS

#### INFORMATION:

There are three (3) Chlorine Safety Kits available for service at the Steadman Fire Station. These kits are assigned to Rescue Company No. 1 and are stored in the air bottle storage closet on the Lombard Street side of the apparatus floor.

The kits are marked as follows:

A - for use on 100 or 150 pound cylinders

- B for use on one ton containers
- C for use on tank cars of 16, 30 and 55 tons

Note: A carrying case containing paint scrapers, cloths, ammonia, and vaseline is also provided for use with the above kits to aid in protecting members and locating the source of chlorine leaks. This carrying case must accompany any Chlorine Safety Kit(s) to an incident.

#### **GENERAL INSTRUCTIONS:**

In the event of an emergency involving a chlorine leak, all three (3) of the above kits (unless only one type is specified) must be obtained from the Steadman Station and delivered to the location of the incident.

- 1. Fire Communications will immediately dispatch Rescue No. 1 and a first-alarm assignment to the scene and notify the Officer in Charge of Field Operations of the incident.
- 2. Rescue 1 will transport the appropriate kit(s) on the apparatus and respond to the incident.

If Rescue No. 1 is out of service, the Officer in Charge of Field Operations will:

a. Authorize the use of another unit in the Steadman Station (Airflex, Scuba Wagon, Pumper) to deliver the kit(s).

			SUBJECT	
EMER	RGENCY SERVI	CES	CHLOR.	INE LEAKS
		<u> </u>		
		h Vaka arrangeme	mes to have the kitle	al delivered to the
		incident in th unavailable.	a event the above-me	ationed units are
	3.	Fire Communication	s will notify the Of	ficer in Charge of
	51	the incident when the scene.	Rescue No. 1 is avai	lable to respond to
	4.	Keys to the storag	e closet and Scuba W	agon will be maintained
		in High Pressure F	umping Station.	
	5.	Members will wear tained breathing a	complete protective (	clothing and self-con-
		to cover delicate	and exposed areas of	the body when making
		emergency repairs	on chlorine containe:	rs.
	6.	The Officer in Cha munications by rad	rge of the incident which are a set of the incident which are a set of the incident of the set of t	will notify Fire Com-
		cerning the leak,	e.g., size of contain	ner, magnitude of leak,
		involved.	: Teak has deen conta.	ined, and size of area
	7.	Fire Communication	s will inform Chemtro	ec (telephone number
		800-424-9300, 24 h	ours a day) of any clurted.	hlorine emergency
		Char has been repu		• . • • • •
	8.	chemtrec will cont gas, and this comp	act the closest manu: any will send represe	facturer of chlorine entatives to the scene
		if necessary.		
	9.	Fire Communication	s will notify the Fin	re Department Safety
		incident.	all pertinent inform	nation concerning the

-

2.1.2

- <del></del>	MOP 625-11
Manual	SECTION
of	EMERGENCY SERVICES
Procedure	SUBJECT
DETAIL	HAZARDOUS MATERIALS TASK FORCE
Salis PROCEDURE	(HMTF)

#### Purpose:

To provide an attack force with specific guidelines for operations at incidents involving hazardous chemicals and/or materials.

### Definition:

A Hazardous Materials Task Force (HMTF) consists of 2 Engines, 1 Truck, Chemical Unit and a Battalion Chief, manned with specially trained personnel, designed to respond as a single entity on all incidents involving hazardous chemicals and/or materials.

# Scope:

Designated HMTF's are:

Task Force 1	Task Force 2
Engine 41	Engine 10
Engine 50	Engine 57
Truck 20	Truck 28
Chemical 1	Chemical 2
Battalion Chief l	<b>Battalion</b> Chief 6

A Deputy Chief, Rescue 1, Airflex 1, and nearest in-service ambulance will respond with whatever task force is dispatched. During normal working hours, the Safety Officer shall be notified and at all other times if it has been established a serious incident. In those situations when all required units are not available from the same task force, Captain, Fire Communications will consider the most expedient dispatch from 1st or 6th Battalion, based on circumstances, to insure that the four required components necessary to complete HMTF are dispatched.

Typical responses would include, but not limited to:

- . Chlorine or ammonia leaks
- . Bulk petroleum storage facilities incidents
- . Radiological incidents
- . Chemical spills and/or explosions
- . Petroleum products transportation fires

4/23/79

DATE

\*\*\* >\*\*\* 2 3-43\*44 Page 1 of 4

	MOP 625-11
TION	SUBJECT
EMERGENCY SERVICES	HAZARDOUS MATERIALS TASK FORCE (HMTF)

## Responsibility:

It will be the responsibility of each member to exercise the appropriate control dictated by his rank in the implementation of this operation procedure.

# Procedure:

A. General

Pre-Planning

Hazardous materials locations will be pre-planned by the local unit. (MOP 610)

Required forms will be updated on an annual basis and theoretical fire-fighting training periods will be conducted to familiarize the members with conditions and to discuss specific fire-fighting operations that may be encountered. Where necessary, due to the size or complexity of the facility, pre-planning tours will be coordinated through respective Battalion Chief. Deputy Chief to be notified of all pending tours. Copies of the pre-plans will be forwarded to all HMTF units who will maintain a separate book containing this information.

B. Dispatch, response, and arrival procedures on box or silent alarms Upon receipt of a report of an incident involving hazardous materials, Fire Communications will transmit the closest fire alarm box to the incident. If the report of the incident is from an employee handling the hazardous materials and the information indicates the incident to be of a serious nature, Fire Communications will also dispatch the nearest HMTF.

If the information received on a reported incident is of a minor nature, Fire Communications may use their discretion and dispatch on a silent alarm those units deemed necessary.

If the task force is included in this assignment, no other units will be required besides the Deputy Chief, Rescue 1, Airflex 1, and the nearest in-service ambulance. During normal working hours, the Safety Officer shall be notified and at all other times if it has been established a serious incident. If the task force is not included in the assignment, they will also be dispatched.

1. Where the cause of the alarm is not known, the first-arriving units will size up the situation and give a complete report to the FCB. If it is determined that a hazardous chemical or material is involved, the officer in charge will determine if the nearest available HMTF should be dispatched.

-1417-7113			
<i>•</i>		MOP 625-11	
CTION	SUBJEC		
EMERGENCY SERVICES		HAZARDOUS MATERIALS TASK FORCE (HMTF)	
z. when mate to t hold port eng: on t pos: or ( meas	erials, the first-assigned of the scene of the incident will positions a minimum of 6 l the incident of the incident the company upon arrival will the assignment should contin tions. If the services of the HMTF are required, he should respond:	engine company will respond directly ith the remaining units going into blocks (2,000 feet) from the re- t. The officer of the first-assigned ll determine if the remaining units nue to the scene from their nold the units in the hold positions hall transmit to them any precautionar ing.	
C. Operatin	g Procedure		
. Unti made	l proper identification of , it should be considered t	the product or material has been toxic and explosive.	
. Memb as a hand shou firs	ers should <u>ANTICIPATE</u> and r limited situation can quic led expeditiously. If evac ld be started immediately, t and working away from the	not delay in calling for assistance tkly become a major problem if not cuation is deemed necessary, it moving those closest to the problem a incident.	
. All worn a ha the	protective clothing, includ in handling these incident zardous chemical, air masks contaminated area.	iing breathing apparatus, will be ts. If initial dispatch indicated s will be donned before entering	
. Upon assi will	arrival, the Deputy Chief sted by the task force Batt designate the staging area	will assume charge of the incident, talion Chief. The officer in charge a and designate an officer in charge.	
. Poin wind aid mars	ts to be considered in sele direction and velocity, to equipment, stand-by manpowe halled here.	ecting a staging area would include pography, and accessibility. First- er, and logistical support will be	
. Subs	equent arriving units will, s, report to the staging ar	, in the absence of specific instruc- ea.	
. Chie ment best i.e.	f officers will be cognizan (radiological, explosive m advantage, as well as the , CHEMTREC, Hazardous Mater	It of available monitoring equip- seter, etc.) and utilize them to available supportive resources; rials Guides, local technical	

•

Page 3 of 4

 		MOP 625-11
	SUBJECT	
EMERGENCY SERVICES	HAZARDOUS	MATERIALS TASK FORCE (HMTF)
Natural Disaster Plan	should be implemented.	les in case the
. Use of hose streams fo be considered and stre	r flushing, cooling, c tched where indicated.	or absorption should
FIRES IN FLAMMABLE GAS. FLOW OF GAS CAN BE STO	ES SHOULD NOT BE EXTIN PPED.	GUISHED UNLESS THE
Communications will be maintaine force and Fire Communications Bu	d at all times between reau.	operating task
As wind direction and velocity a chemical spills or leaks, the fo Communications Bureau when givin	re extremely important llowing format will be g wind conditions:	: in relation to : used by Fire
"wind from north to so "wind from east to wes	uth at 12 MPH," or t at 6 MPH," etc.	
The above information will be ob Washington International Airport	tained from Weather Bu	r <b>eau, Bal</b> timore
PHONE: 962-2177 (24-h 787-7257	our number)	
This information will be transmi	tted:	
<ol> <li>At time of dispatch</li> <li>When command post</li> <li>When requested by</li> </ol>	h. is established. fireground commander.	
The Captain Fire Communications information, the actual readings fireground commander will be apprint fireground operations.	will insure that when at time of request ar rised of any changes t	obtaining weather e recorded. The hat might affect
If type of material is known, all explosion, and/or health hazards will be transmitted:	l pertinent informatio known about the mater	n; such as, fire, ial or incident
<ol> <li>At time of dispatch</li> <li>Via radio, after the Chief has been veri:</li> </ol>	e response of the Batt fied.	alion

4/23/79

	MOP 632 - 1 /
Manual of Procedure DETAIL PROCEDURE	SECTION EMERGENCY SERVICES
	SUBJECT PENN CENTRAL - ELECTRIFIED TERRITORY

In the event of a fire or other incident on or adjacent to the Electrified Territory of the Penn Central Railroad, no major fire operations shall be conducted within two hundred (200) feet of this territory when there is a possibility of a stream or spray contacting charged electrical wires.

Personnel will not approach charged electrical system closer than;

1. Eight (8) feet of 132,000 volt Transmission lines.

2. Three (3) feet of 11,000 volt Catenary lines.

3. Three (3) feet of 6,600 volt Signal power lines.

If absolutely necessary, members will operate fog streams only near charged electric wires or equipment but in no case shall members approach such charged electrical equipment nearer than twenty (20) feet when operating such streams. Hose lines will be operated so that water from the lines does not run, drip, or splash on charged electrical equipment.

If the officer in charge deems it necessary to have the electric power shut off in a respective zone, he will advise Fire Communications Office.

Fire Communications will notify:

Power Director - - Penn Central Railroad - - - 685-4827

Although a call to the Power Director will result in the de-energizing of a portion of the system, stopping electric locomotives, it will not stop diesel locomotives traveling on the system. Therefore notify the Power Director to notify the Movement Director to stop all movement.

When deemed necessary to de-energize the system, operations shall not take place unless the system has been properly grounded by a qualified electrician (Class A Employee) of the railroad. The officer in charge shall personally see that the ground stick is properly applied. No member of the Fire Department will attempt to make any ground or other connections.

All electric equipment shall be considered energized and shall be treated as such until a qualified employee of the company is on the scene and made the proper grounding connections.

DATE

HOP 632 - 1

SUBJECT

EMERGENCY SERVICES

SECTION

PENN CENTRAL - ELECTRIFIED TERRITORY

3

Members must not touch any dangling wires or foreign objects hanging from damaged wires, attachments, or supports nor attempt to move them by any means.

Caution shall be taken not to come in contact with or use water directly on tracks, guy lines, or catenary poles, serious injury could occur.

When the emergency no longer exists, the officer in charge at the scene must make every effort to authorize restoration of the electrical system and/or movement of all trains as quickly as possible.

8/23/73

1 Page 2 of 1

	MOP 633-1
Manual of Procedure DETAIL PROCEDURE	SECTION EMERGENCY SERVICES
	SUBJECT ACCIDENTS - DERAILMENTS RAILROAD RIGHT-OF-WAY

# INCIDENTS - RAILROAD PROPERTY RIGHT-OF-WAY:

The Officer in Charge of Unit or Units that respond to an incident along a railroad right-of-way will immediately advise Fire Communications of:

- . The correct name of the railroad involved.
- . Location of incident.
- . Nature and extent of the incident.

Should the incident be of a serious nature and the emergency operations must be conducted on or across the railroad tracks, the Commanding Officer will notify Fire Communications to have train movement stopped.

Fire Communications will notify the railroad involved by calling one of the following numbers:

- . Baltimore and Ohio 237-3433, Chief Train Dispatcher
- . Canton 342-4458, Train Master
- . Western Maryland 237-3938, Yardmaster
- . Penn Central 685-4827, Power Director

When the incident involves the Penn Central Railroad, emergency operations must be performed in accordance with [MOP 632-1] "Penn Central Electrified Territory." Until it has been absolutely ascertained that train movement has been stopped, men will be stationed at opposite ends of the incident to alert fire personnel of on-coming trains (using portable radios, if necessary). DANGER: DUE TO THE HIGH SPEED OF SOME TRAINS, (UP TO 140 M.P.H.), IT MAY BE NECESSARY TO REQUEST THAT ADDITIONAL UNITS BE DISPATCHED A MINIMUM OF THREE MILES TO EITHER SIDE OF THE INCIDENT IN ORDER TO RADIO AN ADEQUATE WARNING OF ON-COMING TRAINS TO THE UNITS WORKING AT THE INCIDENT.

Page 1 of 2

DATE

SECTION

MOP 633-1

ACCIDENTS - DERAILMENTS RAILROAD RIGHT-OF-WAY

The Fire Department is not equipped with apparatus or tools to lift heavy railroad equipment; i.e., freight cars, locomotives, heavy machinery, etc. Therefore, the Officer in Charge shall advise Fire Communications to request proper heavy equipment from railroad involved.

When the accident or derailment involves freight and/or tank cars transporting explosives, hazardous chemicals, flammable liquids, or gases and there is spillage of chemicals, oils, or noxious materials, [MOP 644-1] "Spill Procedure" must be initiated.

Hose lines must be placed under rails between ties to prevent hose from being run over by passing trains while performing fire-fighting operations.

# INCIDENTS - ACCIDENTS INSIDE OF TUNNELS:

The Officer in Charge of Unit or Units that respond to an incident, accident or other emergency inside a railroad tunnel will follow the same procedure for reporting the incident as stated in "Incidents Along a Railroad Right-of-Way."

Should the incident be a fire in a railroad car, the Officer in Charge will notify railroad that the car is in the tunnel. The railroad will bring the car out of the tunnel, if possible, for fire-fighting operations. However, when fire-fighting and/or rescue operations must be performed inside the tunnel, the Officer in Charge will ensure that all train traffic has been stopped and all safety precautions have been taken to protect fire and civilian personnel.

5/10/77

page 2 of 2

		MOP 644
A BARA	Manual of	SECTION EMERGENCY SERVICE
	Procedure POLICY	SUBJECT OIL, CHEMICAL AND NOXIOUS MATERIAL SPILL INCIDENTS

When a spill of oil, chemical, or noxious material results in pollution of a waterway, a violation of Federal and State Law occurs. A spill, which occurs in any section of the City of Baltimore and enters either the sanitary or storm sewer system, will eventually flow to a stream or tidal waters. Under the terms of the 1972 amendment of the Federal Water Pollution Control Act, practically any waterway is under Federal jurisdiction.

Failure to report a spill is a violation of the law. Responsibility for reporting a discharge or spill rests primarily with the individual who caused it, or who owns or controls the property, equipment, or vehicle on which or from which the emmission of oil, chemical or noxious material occurs. The Maryland Water Resoures Law, Article 96A, Sec. 29a-1 requires, " Any person discharging or permitting the discharge of oil, or any person, actively or passively participating in the discharge or spilling of oil into the waters of the State whether from a land-based installation, including vehicles in transit, or from any vessel, ship or boat of any kind, shall not knowingly fail to report the incident immediately to the appropriate Federal authority and to the Administration and shall not knowingly fail to remain available until clearance to leave is given by appropriate officials ".

Responsibility for the prompt control, containment and removal of any spilled oil rests with the person or persons responsible for the illegal discharge. In case of emergency, actions, which judgement dictates will best control or rectify the conditions constituting the emergency, are permitted.

In response to an incident involving a spill or discharge of oil, chemical or noxious material, the Baltimore City Fire Department will make every effort to prevent the spillage from flowing into storm water drains, sanitary sewers, streams, rivers, harbors or any waterway. When a spill or discharge endangers life or property, the Fire Department shall take whatever action is deemed necessary, to control and eliminate the life, fire and explosion hazard. The Commanding Officer at the scene shall use all available means to restore the hazardous condition to normal safety.

E-21

		MOP 644 - 1
M	anual of	EMERGENCY SERVICE
Pro DI BALT <sup>2</sup> PRO	CEDURE	OIL, CHEMICAL AND NOXIOUS MATERIAL SPILL INCIDENTS

- 1. The Commanding Officer of unit or units that respond, shall size-up the incident and immediately report to Fire Communications Office the following:
  - a. The location of the incident.
  - b. The general conditions at the scene.
  - c. The type and quantity of material spilled.
  - d. Additional Fire Department units, equipment and manpower needed.
  - e. Material, equipment and/or manpower needed from agencies outside the Fire Department.
- 2. The Commanding Officer at the scene of spill incident will make every effort to prevent discharge from flowing into storm water drains, sanitary sewers, streams, rivers, and harbors, by diverting flow to a safe catchment area, by diking, or by utilizing sorbant material to contain the liquid for later disposal, unless there is imminent life, fire or explosion threat.

The Highway Maintenance Division of the Department of Public Works has available for emergency response a vehicle with sand and special chemicals.

The Water Resources Administration of the Maryland State Department of Natural Resources has available for emergency response, a vehicle and crew to assist in containment and clean-up of spillage.

Regardless of the quantity of material involved in the spill, Fire Communications Office must notify the following:

Maryland Water Resources Administration Annapolis -1-267-5551 After office hours, holidays, weekends -Annapolis -1-267-5181

Page 1 of 4

6/18/73

DATE

L-22

CTION		····	6110 IECT	nur 044 - L
ie liųn			SUBJECT	
EM	IERGENCY S	ERVICE	OIL, CH MATERIA	EMICAL AND NOXIOUS L SPILL INCIDENTS
	lf th sanit Commu	ere is any spillage, ary sewer, stream, r nications Office will	that can enter any st iver, harbor or waterw i immediately notify:	orm water drain, ay, Fire
	The O	fficer in charge of 1	Field Operations (day	or night)63086 or 63088
	The O Depar notif	fficer in charge of ] tment Operations and led:	Field Operations will insure that the follo	coordinate Fire wing are
	8.	The Fire Prevention	Bureau, 0800 to 1600 1600 to 0800 and Holidays as per posted	Weekdays65753 Nights, Weekends - Duty Officer list.
	ъ.	Department of Public	: Works, Highway Maint	enance Division63179
	с.	Department of Housin	ng and Community Devel	opment63184
	d.	The United States Co	oast Guard (Group Offi	cer of the Day)789-8050
	e.	Maryland Water Reson	urces Administration -	Office hours- Annapolis -1-267-5551
	£	After	r office hours, holida	ys, weekends - Annapolis -1-267-5181
	1.	ABSISTANT CITY SOLI	Day After	1630 hours
	g.	The Baltimore Police	e Department	Direct line
	When imped Trans	spill is on the public the normal flow of it and Traffic	ic highway or operatio traffic, notify the D	ns in any respect epartment of Direct line
	In ca	se of a chemical spin	ll notify:	
	The	e Environmental Prote i the Water Resources	sction Agency, Philade s Administration (as l	lphia, Pa215-597-9898 isted in e. above)
	If hai inform Offic inform the p	zard of spilled produ mation from er in charge of incid mation that might be roduct.	ict is unknown, ascert ient will attempt to s of assistance to Chem	ain up to date CHEMTREC800-424-9300 ecure any and all trec in identifying
6/1	8/73		0.0	Page 2 of 4
			ピーイン	

ļ

.

			MOP 644 - 1
ECTION		SUBJECT	
DE	RGENCY SERVICE		OIL, CREMICAL AND NOXIOUS MATERIAL SPILL INCIDENTS
<b>#</b>			
	3. If there is the possibili of Baltimore Harbor, the	ty that a large following actio	e spill will enter the waters on shall be implemented:
	<ol> <li>The Officer in char information from Of agencies, will disp the harbor area whe</li> </ol>	ge of Field Ope ficer in charge atch a Battalic re the spill wi	erations upon obtaining at scene or from other on Chief to investigate ill most likely appear.
	2. The Battalion Chief	will:	
	a. Notify Fire C	ommunications of	of conditions in the harbor.
	b. Call for unit control the s	s and equipment pill.	t n <del>ee</del> ded to contain and
	c. Coordinate Fi	re Department (	operations at the harbor site.
	3. The Fire Communicat	ions Office wil	11:
	a. Notify the Of information r	ficer in Charge ecaived.	a of Field Operations of all
	b. Comply with B Junits.	ettalion Chief	's orders for Fire Department
	c. Notify the Un the Day) of t	ited States Coa he situation in	ast Guard (Group,Officer of n the harbor
	d. Notify the Ma list.	ryland Port Adm	ministration as per posted
	4. When volatile flammable l be taken to prevent ignit	iquids are invo ion:	olved, every precaution must
	a. Eliminate all	possible source	ces of ignition.
	b. Utilize explo concentration	simeter to moni s of vapors.	itor build-up of dangerous
	c. Cover exposed Mechanical, C disposal car	surface of fla hemical or High be accomplished	ammable liquid with n Expansion foam until proper i.
	d. Care should b upwind side.	e taken to appr	roach the scene from the
	e. Set up safety	zone and evacu	ate unnecessary personnel

6/18/73

.

£-24

1

		MOP 644 - 1
SECTION		SUBJECT
	EMERGENCY SERVICE	OIL, CHEMICAL AND NOXIOUS MATERIAL SPILL INCIDENTS
	5. If leakage is from owner of tank or of the product pumped Commanding Officer notify company inv	an above ground or underground tank, ascertain ompany responsible for filling the tank and have out to minimize amount of discharge. The at the scene will advise Fire Communications to olved.
	<ol> <li>If spill involves military response will call Fort Geo</li> </ol>	military chemicals or biological material and is required, notify Fire Communications, who rge G. Meade at any of the following numbers:
		0800 to 1630 hours
		1630 to 0800 hours
	<ol> <li>Fire Department pe incident to anyone unless a legal adv</li> </ol>	rsonnel shall make no statements concerning the who is not a member of the City Government, isor from the Law Department is present.
	8. Reports shall be f normal channels.	orwarded as per current procedure through

.

۰.

	MOP 644-2
Manual	SECTION
of	EMERGENCY SERVICE
Procedure	SUBJECT
DETAIL	REPORT OF SPILL INCIDENT
JALTS PROCEDURE	FORM 28-2100-0041

- Report of Spill Incident (form 28-2100-0041) [MOP 644-2-1] will be completed in triplicate and forwarded in lieu of Fire Report (form 28-2100-0042) when incident involves a spill only, without fire or combustion of any kind.
- Fire Report will be forwarded when fire or combustion results from a spill; and if there is a residue of oil remaining after the fire (re. MOP 644-1, Oil, Chemical and Noxious Material Spill Incidents), Report of Spill Incident will be forwarded with the Fire Report.
- If more than two (2) persons are involved in a spill incident, check appropriate box and use the reverse side of (form 28-2100-0041) for the names and information concerning the additional persons.
- 4. Report of Spill Incident will be completed by Officer in Charge of incident in the same manner that Fire Reports are completed and forwarded through normal channels.
- 5. Distribution:
  - a. Forward original and one copy to Headquarters.
    - Original copy retained in Headquarters in the same manner as Fire Reports.
    - 2. One copy will be forwarded to the Law Department through the office of the Safety Officer.
  - b. One copy to be retained in unit file.

Page 1 of 1

DATE , s / : / ...

**E-**26

SECTION

SUBJECT

MOP 644 - 2 - 1

# EMERGENCY SERVICES

# REPORT OF SPILL INCIDENT FORM 210 - 041

PAPE CITES	JRE CITY FIRE D	EPARTMENT	REPO	ORT OF SPILL INCI	DENT ·
Date.	ua: jTime:	,2%d.c	Jrd.: jein		6sn.:
	·	<u></u> 0/3.	nrs. j	Dr§.	hrs. nis.
jue, 31 c245					Mecall:
Zwner			Owner's Address		Zip Coee
		Person(s) Involve	id in Spill incident		
hdøress:		Zie Codei	Adaress		Zip Coder
Company or Firm:			Company or Firm:		
		Zia Cose:	A00/011:		Zip Code:
		<u></u>	Manufa Lucasta Mumor		
	·····			·	
	if Additional	Persons are involved in Spill L	ncident, check here 🗆 and	a use reverse side.	
Type of Substance (Cr	10CH)	How seil occurred (Check		Extent of Senil (Check)	
Gasoline	U No. 2 Fuel Oil	Vehicular Accident		Contained on Land	
- Kerosene	An. 4 Fuel Oil	Personal Error		Entered Storm Sewer	
Jat Fuel	No. 5 Fuel Oil	Mechanical Fellure		Li Entered Sanitary Sewer	
Cheer Ol	No. 6 Fuel Oli	0 Other		Entered Surface Water	-
				(Name of body of water)	
				Amount of spill	
		Fire Departm	tent Response		
[neine					
/rucks	·····	_ Chemical	HI-Ex	Fire Bost	į
Sther Units		BC DC		F.F.B F.J.B	
-		Other Agen			
United States Coase	I Guard Include Annory		Department of Public Wo	Prist	Senitation
Anyland Water Re	nources Administration	•			
Maryland Port Adr	ninistration		Department of Housing a	ine Community Development	
Opportment of Law	,		Building Insee	ctor U Mechanical Inspector	
_		fire Oeparim	ent Operations		
Absorbed in surface	e — No action taken		Foom blanket maintai	ned on spill	
Evaporated - No a	clion laken		Type used:	Ame	ibeeu tra
				<b>.</b>	
Surface skimming	a or barrier			nanical	<u> </u>
Contained by Battle					
Surface skimming Contained by baffit Divertes to safe loc	ation			mical	
Surface skimming Contained by beffit Diverted to safe loc Gispersed by flush- Denser to life and p	iation ng — Extreme Hazard — Inneerty		Asserted by sortent	meterial	
Surface skimming     Contained by baffit     Divertee to tafe loc     Clasersed by flushi     Danger to life and p	iation ng — Extreme Hazard — property		Che Absorbed by sorbent Explosimeter used:	mital	
Surface skimming  Contained by Baffi  Diverted to safe loc  Oliverted to safe loc  Oliverted by Fluchi  Denger to life and p	lation ng — Extreme Mazard — property		Che     Asserbed by sorbent     Explosimeter used:     Initial reading	mital	
Surface skimming     Contained by baffk     Divertee to tafe loc     Origerise by fluch-     Denger to life and p	Lation ng — Extrome Mazard — property		Cho     Cho     Asserbed by sorbent     Explosimeter used:     Initial reading     Maximum read     Maximum read	mical material 	
Surface skimming     Contained by outfit     Divertes to tare lac     Clasersed by fluch     Donger to life and s     Temerkst	cation ng — Extrome Hazard — property		Cho	mical material 	
Surface skimming     Contained by outfit     Orivertee to safe loc     Cleaersed by flush     Denger to life and i	cation ng — Extrome Mazard — property		Absorbed by sorbent Explosimator use: Initial reading Maximum read Final reading	mical	
Surface skimming     Contained by deffi     Diverted to safe let     Clabersed by flushi     Denger to life and j	estion ng — Extrome Masard — property		Abserbed by sorbent Explosimeter used: Initial reading Maximum read Final reading	mical material	
Surface skimming Concerned by deffi Divertee to safe loc Claperise by flushi Denger to life and j Remarks:	estion ng — Extrome Masard — property		Absorbed by sorbent Explosimeter use: Initible reading Maximum read Final reading	mical	
Surface skimming Concerned by beffi Divertee to safe loc Clabersed by flush Canger to life and p Remarks:	cation ng — Extrome Masard — property		Abserbed by sorbent Explosimeter use: Initible reading Maximum read Pinel reading	mical material 	
Surface skimming Concerned by deffi Diverted to safe loc Clabersed by flushi Denger to life and j Remarks:	Estion ng — Extrome Hazard — property		Abserbed by sorbent Explosimeter use: Initial reading Maximum read Final reading	mical material 	
Surface skimming Contained by deffi Overted to safe los Clobersed by flushi Denger to life and j Remarksi	Estion ng — Extrome Hazard — property		Che Abserbed by sorbent Explosimeter used: Initibil reading Maximum read Final reading	mical	
Surface skimming Contained by deffi Diverted to safe loc Clabersed by flushi Denger to life and i Remarks:	Estion ng — Extrome Hazard — property	Perwerded:	Che Abserbed by sorbent Explosimeter use: Initible reading Maximum read Final reading	mical	
Surface skimming Contained by deffi Diverted to safe los Clabersed by flush Clabersed by flush Clabersed by flush Remarks:	Estion ng — Extromo Masard — property	Perwsrdeq:	Che Abserbed by sorbent Explosimeter use: Initible reading Maximum read Final reading	mical	
Surface skimming Contained by deffi Diverted to safe los Clabersed by fluchi Denger to life and p Remarks:	Estion ng — Extromo Masard — property	Perwsrded:	Che Abserbed by sorbent Explosimeter use: Initibil reading Maximum read Final reading	mical	
Surface skimming Contained by deffi Diverted to safe loc Claberand by fluchi Denger to life and p Remarks:	Estion ng — Extromo Masard — property 	Perwerdeg:	Che Abserbed by sorbent Explosimeter use: Initial reading Maximum read Final reading	mical	

Original form is 8 1/2" x 11", white in color.

11/ 5/73

Page 1 of 1

	j
	MOP 644 - 3
Manual of	SECTION EMERGENCY SERVICES
Procedure DETAIL PROCEDURE	SUBJECT SPILL INCIDENTS - ABSORBENT MATERIAL

Absorbent Material for use at oil spill incidents is supplied to the Fire Department by the Maryland Water Resources Administration.

This material will not absorb water, and while it will absorb all petroleum products, it works best with the heavier oils.

# Purpose

The primary reasons for supplying absorbent material to Fire Department Units is as follows:

- 1. To provide immediate complete control of small oil spills.
- 2. To provide immediate initial control measures to contain large oil spills in accordance with [MOP 644 - 1], until clean up forces of the responsibile party, the State or Federal Governments can be mobilized.

# Distribution

- 1. A total of five bags of absorbent material will be maintained at each fire station under the jurisdiction of the captain in charge of quarters.
- 2. From this total, two (2) bags may be carried on each engine and truck if sufficient storage space is available.
- 3. A reserve supply will be maintained at the Supply Bureau.

## Operations

- 1. The unit officer will have all available absorbent material placed on the apparatus before responding from quarters on either a silent or box alarm to a spill incident, automobile accident, flushing operation, or when the dispatcher's response instructions indicate it may be needed.
- 2. Upon arrival at the scene, the officer in charge will report to Fire Communication Office as per item 1 [MOP 644-1].
- 3. When a single unit responds to a spill incident and additional absorbent material is needed, the unit officer will notify Fire Communications Office to dispatch additional units <u>as needed</u> and specifically request that units respond with sorbent materials.

8/22/73

DATE

F-RE DEPT 210-43744

Page 1 of 2

	L. L	_
	NOP 644 - 3	
ECTION	SUBJECT	/
EMERGENCY SERVICES	SPILL INCIDENTS - ABSORBENT MATERIA	L.
4. Whenever three (3) or more Find of a spill incident and a box Communications Office will distance who will assume command at the second second as the second second as the second	re Department Units are at the scene alarm has <u>not</u> been sounded, Fire spatch the nearest Battalion Chief, e scene.	(
5. If additional quantites of ab- the officer in charge will ar Officer in Charge of Field Op	sorbent material are still needed, range for delivery through the erations in accordance with [MOP 644-1].	١
6. Units that are not needed after absorbent material, shall be present material.	er they have delivered their supply of placed in service.	
7. Clean up and Disposal:		
<ul> <li>a. Small quantities of oil conup and placed in a suitable originally contained the mathematic solution of a same manner as the regular Bureau of Utility Operation</li> <li>c. Absorbent material that con and large quantities of oil disposed of by supervisor of Operations crew that respondent and crews or contracted</li> </ul>	ntaminated absorbent shall be cleaned e container such as the empty bags which aterial. ent material can be disposed in the station trash is collected by the ns. ntains volatile flammable liquids l contaminated material shall be of the Bureau of Utilities nded or by State or Federal Government ors.	1
Maintenance of Supply of Absorbent Ma	aterial	
<ol> <li>When absorbent material is use supply officer by phone and re the five (5) bag supply at the</li> </ol>	ed, unit officer will notify the equest sufficient quantity to maintain e station.	
2. The Supply Officer will schede soon as practicable after requ officer.	ule delivery of needed quantities as uest is received from the unit	
3. The Supply Officer will notify form 210-033 thru normal chan storeroom is depleted to fifty	y the Deputy Chief Staff Services on nels when the reserve supply at the y bags.	

いたので、「「いい」のないない。「「いい」のないがないです。

• • • •

			MOP	644-4
T.D.	Manual of	SECTION	EMERGENCY SERVICE	
	Procedure DETAIL PROCEDURE	SUBJECT	SPILL INCIDE.TS - SEALING COMPOUND	

One (1) pound packs of Mac-Seal sealing compound is issued by the Fire Department and is to be carried on the apparatus of all Engine and Truck Companies.

#### **PURPOSE:**

C. C. C. C.

This sealing compound is to be used to seal <u>small</u> leaks or ruptures in gasoline and/or fuel oil tarks to control the flow from a leak and to minimize the amount of product spilled.

#### LIMITATIONS:

This product is intended as a temporary measure only and is not to be used for permanent repair under any circumstances. Product will not retain large leaks and may not control leaks under pressure.

#### INSTRUCTIONS:

Officer in charge will insure that all provisions of MOP 644-1 are complied with and that necessary reports, including Fire Prevention Code violations are forwarded.

The unit's supply of Mac-Seal sealing compound will be replenished by requisition from the Supply Bureau.

	MOP 644-5
Manual of	SECTION EMERGENCY SERVICES
Procedure DETAIL PROCEDURE	SUBJECT POLYCHLORINATED BIPHENYLS (PCB'S)

PCB's or askarels are fire-resistant, chlorinated hydrocarbon insulating fluids which are used in capacitors and transformers. PCB's are insoluble in water. They are also non-biodegradable and cumulative.

PCB's have been used in three broad types of applications for the past 40 years as follows:

- "Open-ended" applications; for example, in paints, specialty inks, paper coatings, plastics, etc.
- (2) "Nominally closed" applications; for example, as the working fluid in hydraulic or heat-transfer systems.
- (3) "Closed electrical system" applications, specifically as the insulating fluid in transformers and capacitors.

Askarel-filled transformers and capacitors present less fire hazard than oil-filled transformers. As a result, askarels have replaced mineral oils in more than 90% of the power and industrial capacitors today and are used in transformers located in buildings without enclosed electrical vaults.

The environmental effects of askarels are under in-depth study by governmental and other agencies. Traces of askarels are being found in the environment and in fish and bird life. The long term genetic and ecological effects are not yet completely understood. For these reasons, care should be taken to contain askarels and minimize their entry into the environment.

Askarel wastes should never be disposed of down drains or sewers.

#### Emergency Handling of PCB's:

Most equipment is under the control of the Baltimore Gas and Electric Company, but some equipment is privately and city owned. Unit commanders during fire prevention and/or pre-fire planning inspections will

DATE 8/27/79

Page 1 of L.

28-1417-7113

SECTION

MOP 644-5

SUBJECT

EMERGENCY SERVICES

POLYCHLORINATED BIPHENYLS (PCB'S)

check transformers and capacitors that are privately and city owned to see that they are properly marked. This information is on the equipment tag located on the unit itself.

In response to an incident involving a PCB spill, the Baltimore City Fire Department will make every effort to prevent the spillage from flowing into storm water drains, sanitary sewers, streams, rivers, narbors or any waterway. When a spill or discharge endangers life or property, the Fire Department shall take whatever action is deemed necessary to control and eliminate the life, fire and explosion hazard. The commanding officer at the scene shall use all available means to protect the public, pending the arrival of a representative of Maryland Water Resources Administration who will take charge of removal of spilled PCB's.

Operating procedures should be to avoid physical contact with any askarels. The use of porous gloves (such as leather) that can absorb and retain askarels should be avoided. Resistant gloves of rubber (neoprene, polyethylene) should be used if contact is unavoidable. Self-contained breathing apparatus must be worn to avoid askarel vapors. The gases produced when askarel is decomposed by high temperatures (such as that of an electric arc) in the presence of air or organic insulating materials contain a high percentage of hydrogen chloride and small percentages of other gases.

In case of spillage on clothing or contamination of equipment, the items should be removed and bagged for proper cleaning. If the appliance is owned by the Baltimore Gas and Electric Company, they will make arrangements to pick up and clean the items.

If the appliance is privately or city owned, the contaminated items should be removed and bagged and forwarded to the contractor who will clean up the PCB spill. This information can be obtained from the Water Resources Representative on the scene. A listing of all items contaminated will be forwarded to the Chief of Fire Department.

A member with an open cut or abrasion that has been exposed to askarel vapors should report to the Department Infirmary for treatment immediately. Eyes that have been exposed to liquid askarel should be irrigated immediately with large quantities of running water for 15 minutes and then examined by a physician.

3/27/79

**E-32** 

Page 2 of 2





SEC.	TION
------	------

MOP 644-5-3

EMERGENCY SERVICES

WARNING IDENTIFICATION



(Polychlorinated Biphenyls)

A toxic environmental contaminant requiring special handling and disposal in accordance with U.S. Environmental Protection Agency Regulations 40 CFR 761 For Disposal Information contact the nearest U.S. E.P.A. Office

In case of accident or spill call toll free the U.S. Coast Guard National Response Center. 800-424-8802 Also Contact B.G.&.E. Trouble Desk Tel. No. 234-5551

8/27/79

		HAZARDOUS SUBSTANCE SPILL REPORT
ī.	RFF	PORTER
	Δ	
	~	(Agency) can be contacted for further information at (Phone)
		(Location)
Ι.		ATION OF INCIDENT
	Α.	Structure
		Building or Company Name
		Address
		City County
	з.	Roadway
		Highway or Street Name
		Nearest Intersection
	С.	Off-Shore
		Nearest identifying landmarks (Beach name, pier, nearby road, street, etc.)
[.	ŢΥΡ	E OF INCIDENT
	Α.	Oil Spill
		Substance Quantity
		ShipOlfieldPipelineRailroad
		Name of ship, etc.
	в.	Radiological Incident

F-1

	<ul> <li>c. Radiation producing device</li> <li>d. Exposure</li> <li>e. Transportation accident</li> <li>f. Nuclear weapon</li> </ul>
	2. Radioactive material involved amount
	C. Other Hazardous Material
	Substance Quantity
	Generic Numes
	Solid Liquid Gas
IV.	CAUSE OF INCIDENT
	Describe
ν.	INJURIES
	Injured To which hospital were injured taken?
	(number)
	Are injured persons contaminated? Yes <u>No Nos.</u>
	Were injured persons exposed? Yes No Nos
	yes, was the hospital and ambulance crew so advised?
	Dead Were dead contaminated?
VI.	PERSONNEL AND EQUIPMENT
	What emergency personnel and equipment are at the scene?
	Type of additional assistance requested?
VII.	ADDITE AND INFORMATION AND COMMENTS

F-2

# RESPONSE CHECKLIST

		YES	NO	NA
1.	Have the proper authorities been notified?			
	CHEMIREC (800) 424-9300			
	COUNTY/CITY			
	Sheriff (or Police Dept.)			
	County Health Department			
	Fire Department		<u> </u>	<del></del>
	Hospitals		<u> </u>	
	Public Works			
	Water Department			
	water Resources			
	Fish and Game			
	Air Pollution District			
	STATE			
	Bureau of Disaster Preparedness	<u> </u>		
	Department of Health	<u> </u>		
2.	Have adequate safety precautions been taken in the polluted area?			
3.	Have you identified the pollutant?	<u> </u>		
4.	If unidentifiable, do you know who to contact for proper identification?			
5.	Are adequate communications available?			

# EMERGENCY RESOURCES

1. Collection Agents

Location

Telephone No.

A. <u>Absorbents</u> Absorbent felt paper Bark Copolymer PVC/PVA Cotton waste Peat Polypropylene fiber Polyurethane foam Rayon floss Rock wool sheets Rope Sawdust

Sawdust Sisal string Straw Waste paper

B. Congealing Agents

Nylon Agglutinants Plastic film Plastic foam

C. Gelling Agents

Molten wax Soap

D. Liquids ie

Chlorine repair kit

# 2. Supplies

A. Signs

B. Rose

APPENDIX G

### ANNOTATED BIBLIOGRAPHY

- 1. Association of American Railroads, Bureau of Explosives. <u>Emergency Handling</u> of Hazardous Materials in Surface Transportation. 1981.
- 2. Association of American Railroads. <u>Standard Transportation Commodity Code</u>. Alphabetical, Numerical, January 1978.
- 3. Association of American Railroads. <u>Standard Transportation Commodity Code</u> Tariff, No. 1-G.
- 4. <u>Bay County Natural Disaster Operations Plan</u>. Panama City/Youngstown, FL. 1981.
- 5. California Department of Transportation, Maintenance Department. <u>Hazardous</u> <u>Substances Spill Procedure Manual.</u> September, 1979.
- 6. California Highway Patrol. <u>Explosives Routes and Stopping Places</u>. HPH 84.3, August 1973.
- 7. California Highway Patrol. <u>Hazardous Materials Transportation</u>. November, 1979.
- 8. Chemical Marketing Reporter. <u>1980-81 OPD Chemical Buyers Directory</u>. Schnell Publishing Company. 1980.
- 9. <u>City of Baltimore Emergency and Major Disaster Assistance Plan</u>. Office of the Mayor.
- Department of California Highway Patrol. <u>Hazardous Substance Highways</u> <u>Spills Study</u>. A study submitted to the California legislature as directed by Senate resolution 52 and House resolution 62. 1978-79.
- 11. Department of Transportation, Materials Transportation Bureau. Student Performance Manual To Accompany <u>Handling Hazardous Materials</u> <u>Transportation Emergencies</u>.
- Department of Transportation, Research and Special Programs Administration, Materials Transportation Bureau. <u>1980 Emergency Response Guidebook</u>. Report No. DOT-P-5800.2. Washington, D.C.: Government Printing Office.
- Department of Transportation, United States Coast Guard, Chemical Hazards Response Information System, <u>A Condensed Guide to Chemical Hazards</u>, January 1974.

This manual is Volume I of four volumes that provide timely information essential for proper decision-making by responsible Coast Guard personnel and others during emergencies involving the water transport of hazardous chemicals. It also provides certain basic non-emergency related information to support Coast Guard efforts to achieve improved levels of safety in the bulk shipment of hazardous chemicals.

- 14. Department of Transportation, United States Coast Guard. <u>CHRIS Hazard</u> <u>Assessment Handbook Volume 3</u>. Report No. CG-446-3. January 1974.
- 15. Department of Transportation, United States Coast Guard. <u>CHRIS Hazardous</u> <u>Chemical Data, Volume 2</u>. Report No. CG-446-2. January 1974.
- Department of Transportation, United States Coast Guard, Chemical Hazards Response Information System, <u>Response Methods Handbook</u>. Volume 4, December 1978.

This manual is Volume IV of our volumes that provide timely decisionmaking information for use by Coast Guard personnel and others involved in responding to and mitigating the effects of hazardous material discharges. It can be properly used only in conjunction with the other manuals of the CHRIS system.

17. Department of Transportation, United States Coast Guard, Office of Research and Development, <u>Chemical Hazards Response Information System for</u> <u>Multimodal Accidents (CHRISMA)</u>, Final Report. April, 1975.

> This report examines the need for improved technical and other information for meeting emergencies connected with the transportation of hazardous materials, particularly actual or potential discharges of mode. The Chemical Hazards Response Information System (CHRIS), under development by the U.S. Coast Guard to furnish in-depth guidance during emergencies involving waterborne transport, was seen as a likely prototype for other modes as well. A reevaluation of CHRIS has been conducted to determine the desirability of enlarging its scope to include all modes of transportation. It is agreed that an expanded system would be beneficial in reducing losses to life, property and environment. This information system would be a decentralized organization providing response guidance on request to local emergency services personnel, a computerized hazard assessment system operated at Headquarters, and three reference manuals furnished to all response organizations.

- 18. Emergency Plan No. 1. Nashville, TN.
- 19. <u>Encyclopedia of Explosives and Related Items</u>, Volume 1. Picatinny Arsenal. Dover, N.J. 1960.

- 20. <u>Escambia County Hazardous Materials Accident Plan and Fire Suppression</u> Guide. Pensacola, FL.
- 21. Enviromental Protection Agency. Illinois Central Gulf Train Derailment: Claxton, Kentucky. August 1980. EPA Report No. 430/9-80-012.
- 22. Environmental Protection Agency, Industrial Environmental Research Laboratory, Office of Research and Development. <u>Manual for the Control of</u> <u>Hazardous Materials Spills: Volume 1: Spill Assessment and Water</u> <u>Treatment Techniques.</u> Report No. EPA-600/2-77-227. Washington, D.C.: Government Printing Office, November 1977.

This report contains suggested procedures for control of hazardous material spills using improvised treatment processes. This manual includes: notification procedures on inventory of information sources; methods for spill identification and assessment; a thought guide for determining the best method of handling a spill; suggested treatment schemes for 303 designated hazardous chemicals; a limiting factor system design approach, and design construction, and operation steps for each of the five treatment processes applicable to improvised systems.

- 23. Enviromental Protection Agency. <u>L&N Train Derailment. Molino, Florida</u>. Report No. 904/9-80-063. November 1980.
- 24. General Services Administration, National Archives and Records Service, Office of the Federal Register, <u>Code of Federal Regulations 49</u>: <u>Transportation</u>, <u>Parts 100-179</u>. Government Printing Office, 1979.
- 25. Hawley, G. G. <u>The Condensed Chemical Dictionary</u>, Ninth Edition. Van Nostrand Reinhold Co., 1971.
- 26. Institute of Science and Public Affairs, The Florida State University. An Analysis of the Local, State, and Federal Authority and Responsibility for Hazardous Material Incidents in Leon County, Florida. June, 1980.
- 27. Leon County Emergency Operations Plan. Tallahassee, FL. 1981.
- 28. Ministry of the Solicitor General, Centre of Forensic Science. <u>Railroad</u> <u>Accident Report: Derailment of Canadian Pacific Railway Train No. 54 with</u> <u>Resultant Fire and Tank Car Ruptures at Mississauga, Ontario.</u> April 1, 1980.
- 29. National LP-Gas Association, LP-GAS Market Facts. 1978.
- National Transportation Safety Board. <u>Highway Accident Report: Accidental</u> <u>Mixing of Incompatible Chemicals, Followed by Multiple Fatalities, During a</u> <u>Bulk Delivery, Berwick, Maine</u>. April 2, 1971. NTSB Report No. NTSB-HAR-71-7. Adopted August 26, 1971.

- 31. National Transportation Safety Board. <u>Highway Accident Report: Liquefied</u> Oxygen Tank Explosion followed by Fires in Brooklyn, N.Y. May 30, 1970. Report No. NTSB-HAR-71-6.
- 32. National Transportation Safety Board. <u>Highway Accident Report: Multiple-</u> Vehicle Collision Followed by Propylene Cargo-Tank Explosion, New Jersey <u>Turnpike, Exit 8</u>. September 21, 1972. Report No. NTSB-HAR-73-4. Adopted October 17, 1973.
- 33. National Transportation Safety Board. <u>Highway Accident Report: Multiple-</u> <u>Vehicle Collisions under Fog Conditions, Followed by Fires, New Jersey</u> <u>Turnpike, North of Gate 2</u>. November 29, 1969. NTSB Report No. NTSB-HAR-71-3. Adopted January 20, 1971.
- 34. National Transportation Safety Board. <u>Highway Accident Report: Propane</u> Tractor Semitrailer Overturn and Fire, May, 1973.

- 35. National Transportation Safety Board. <u>Highway Accident Report: Surtigas</u>, S.A., Tank-Semitrailer Overturn, Explosion and Fire Near Eagle Pass, TX, <u>April 29, 1975</u>. Report No. NTSB-HAR-76-4.
- 36. National Transportation Safety Board. <u>Highway Accident Report: Truck-Automobile Collision Involving Spilled Methyl Bromide on U.S. 90 near Gretna, Florida.</u> August 8, 1971. NTSB Report No. NTSB-HAR-72-3. Adopted June 1, 1972.
- 37. National Transportation Safety Board. <u>Highway Accident Report: Transfer Co.</u>, of Texas, Tractor-Semitrailer (Tank) Collision with Bridge Column and Sudden Dispersal of Anhydrous Ammonia Cargo, 1-610 at S.W. Freeway, May 11, 1976. Report No. NTSB-HAR-77-1.
- 38. National Transportation Safety Board. <u>Hazardous Materials Accident Spill</u> <u>Maps. Hazardous Material Releases from Railroad Cars near Princeton</u>, <u>Kentucky. October 17, 1978. NTSB Report No. NTSB-HZM Map-80-3.</u>
- National Transportation Safety Board. <u>Hazardous Materials Spill Map.</u> <u>Inwood</u>, Indiana - November 8, 1979.
- 40. National Transportation Safety Board. <u>Railroad Accident Report: Burlington</u> <u>Northern, Inc. Monomethylamine Nitrate Explosion. Wenatchee,</u> <u>Washington.</u> August 6, 1974. NTSB Report No. NTSB-RAR-76-1. Adopted February 2, 1976.
- 41. National Transportation Safety Board. <u>Railroad Accident Report: Chicago and</u> <u>North Western Transportation Co., Freight Train Derailments and Collisions</u>, May 16,1972.
- 42. National Transportation Safety Board. <u>Railroad Accident Report: Chicago</u>, <u>Burlington</u>, and <u>Ouincy Railroad Company Train 64 and Train 824 Derailment and</u> <u>Collision with Tank Car Explosion. Crete</u>, Nebraska. February 18, 1969. NTSB Report No. NTSB-RAR-71-2. Adopted February 24, 1971.

- 43. National Transportation Safety Board. <u>Railroad Accident Report: Derailment</u> and Subsequent Burning of Delaware and Hudson Railway Freight Train at Oneonta, N.Y., February, 1974.
- 44. National Transportation Safety Board. Railroad Accident Report: Derailment of Atlanta and Saint Andrews Bay Railway Company Freight Train, Youngstown, Florida, February 26, 1978. Adopted: November 9, 1978. Report No. NTSB-RAR-78-7.
- 45. National Transportation Safety Board. Railroad Accident Report: Derailment of <u>Missouri Pacific Railroad Company's Train 94 at Houston, Texas, October 19,</u> <u>1971</u>. Report No. NTSB-RAR-72-6. Adopted: December 31, 1972.
- 46. National Transportation Safety Board. <u>Railroad Accident Report: Derailment of</u> <u>Toledo, Peoria, and Western Railroad Co. Train No. 20 with Resultant Fire</u> <u>and Tank Car Ruptures</u>. June 21, 1970.

- 47. National Transportation Safety Board. <u>Railroad Accident Report: Hazardous</u> <u>Materials Accident at the Southern Pacific Transportation Co., Englewood</u> <u>Yard, Houston, Texas</u>. September 21, 1974. Report No. NTSB-RAR-75-7. Adopted May 21, 1975.
- 48. National Transportation Safety Board. <u>Railroad Accident Report: Hazardous</u> <u>Materials Railroad Accident in the Alton and Southern Gateway Yard in East</u> <u>St. Louis, Illinois</u>. January 22, 1972. NTSB Report No. NTSB-RAR-73-1. Adopted January 31, 1973.
- 49. National Transportation Safety Board. Railroad Accident Report: Louisville and Nashville Railroad Co., Freight Train Derailment and Puncture of Hazardous Materials Tank Cars, Crestview, Florida. April 8, 1979. Report No. NTSB-RAR-79-11. Adopted September 13, 1979.
- 50. National Transportation Safety Board. <u>Railroad Accident Report: Louisville</u> <u>and Nashville Railroad Co., Freight Train Derailment and Puncture of</u> <u>Anhydrous Ammonia Tank Cars at Pensacola, Florida</u>, November 9, 1977. <u>Report No. NTSB-RAR-78-4</u>. Adopted July 20, 1978.
- 51. National Transportation Safety Board. <u>Railroad Accident Report: Pennsylvania</u> <u>Railroad Train PR-11A, Extra 2210 West and Train SW-6, Extra 2217 East</u> <u>Derailment and Collision, Dunreith, Indiana</u>. January 1, 1968. Adopted <u>December 18, 1968</u>.
- 52. National Transportation Safety Board. <u>Railroad Accident Report: St. Louis</u> <u>Southwestern Railway Company Freight Train Derailment and Rupture of Vinyl</u> <u>Chloride Tank Car. Lewisville, Arkansas.</u> March 29, 1978. NTSB Report No. <u>NTSB-RAR-78-8.</u> Adopted December 7, 1978.
- 53. National Transportation Safety Board. Railroad Accident Report: Southern Railway Company Train 154 Derailment with Fire and Explosion. Laurel, Mississippi. January 25, 1969. Adopted October 6, 1969.
- 54. Smith, Al Jr. Managing Hazardous Substances Accidents. McGraw-Hill, 1981.
- 55. SRI Internationl. <u>1981 Directory of Chemical Producers</u>. Menlo Park, CA. 1981.
- 56. State of Tennessee Hazardous Materials Training Institute. <u>Training</u> <u>Manuals A-C</u>.
- 57. United States Bureau of Census. <u>Census Transportation</u>, 1972, Volume III, <u>Commodity Transportation Survey</u>, Part 3., Area Statistics, South and West <u>Regions and U.S. Summary</u> Washington, D.C.: Government Printing Office, 1976.
- 58. Weiss, G. Hazardous Chemicals Data Book. Noyes Data Corp., 1980.

## APPENDIX H

#### GLOSSARY

<u>ABSORBENT</u> - (1) Any substance exhibiting the property of absorption. (e.g., substances capable of soaking up a particular HM(s))

<u>ACOUSTIC EMISSION</u> - Rapid release of energy through transient elastic waves, characteristic of structures under stress. (e.g., the characteristic sound waves produced by a structure under stress and which vary with change in stress or appearance of a crack).

<u>AMPULES</u> - Hermetically sealed small bulbous glass vessels used to hold chemicals/solutions.

<u>BERM</u> - A narrow shelf, path or ledge typically at the top or bottom of a slope. (e.g., along side a rail bed or highway).

<u>BILL OF LADING</u> - One type of shipping paper which lists goods shipped and other required identifying information.

<u>BLEVE</u> - The combination of a weakened structure and the buildup of intergal vapor pressure resulting in container rupture or instantaneous release and ignition of the vapor usually with violent effects. An acronym for boiling liquid expanding vapor explosion.

<u>CARBOY</u> - A glass, metal or plastic bottle or rectangular container of about 5 to 15 gallons capacity for liquids.

<u>CARGO TANK</u> - Any tank permanently attached to or forming a part of any motor vehicle or any bulk liquid or compressed gas packaging not permanently attached to any motor vehicle which by reason of its size, construction or attachment to a motor vehicle, is loaded or unloaded without being removed from the motor vehicle and is used for transporting the commodity(ies).

<u>CHEMTREC</u> - Abbreviation of Chemical Transportation Emergency Center. A division of the Manufacturers Chemical Association established as an emergency information source for transportaiton accidents involving hazardous materials (e.g., flammable, toxic or explosives).

<u>COMMODITY FLOW</u> - The quantity routing patterns for transportation of an article(s) of commerce (e.g., hazardous materials shipments and routes).

<u>COMMODITY INCOMPATIBILITY</u> - The situation whereby a chemical(s) is (are) capable of interacting with each other to create a hazard or unsafe condition and thus must be handled, packaged, stored and shipped with certain prescribed precautions.

<u>COMMUNICATIONS COORDINATION CENTER</u> - A central facility (mobile or fixed) which functions in the case to interlink and coordinate the different sources of information and communications at and away from the accident scene.

<u>COMPRESSED GAS</u> - Any material or mixture that, when enclosed in a container, has an absolute pressure exceeding 40 psi at  $70^{\circ}$ F or regardless of the pressure at  $70^{\circ}$ F, has an absolute pressure exceeding 140 psi. at  $130^{\circ}$ F; or any flammable material having a vapor pressure exceeding 40 psi absolute at  $100^{\circ}$ F.

<u>CONSIST</u> - Makeup or composition by classes, types, or grades and arrangement of rail cars in a train.

<u>CONTAINER STRUCTURAL INTEGRITY</u> - The existing condition of the container's structural components with respect to its original design and its capability to safely retain its contents as intended.

<u>CORROSIVE MATERIAL</u> - A solid that causes visible destruction or irreverable alterations in human skin tissue at the site of contact, or in the case of leakage from its packaging, a liquid that has a severe corrosion rate on steel.

<u>CYLINDER</u> - Pressure vessel designed for pressures higher than 40 psia and having a circular cross section. It does not include a portable tank, multi-unit tank car tank, cargo tank or tank car.

<u>**DENT</u>** - (1) A depression or hollow made by a blow or by pressure. (e.g., on a container surface)</u>

<u>DIKE</u> - (1) A bank usually of earth construction used to control or confine water or other fluid. (2) A barrier preventing passage, especially of something undesirable.

<u>DISPATCHER</u> - One who controls the movement of vehicles/persons. (e.g., trains, trucks, fire, police)

**EXPLOSIVE** - Any chemical coumpound, mixture or device, the primary or common purpose of which is to function by explosion. (i.e., with substantially instanteneous release of gas and heat).

**FLAMMABLE** - Capable of being easily ignited and of burning rapidly. See 49 CFR, Part 173 for precise types of flammable liquids and gases.

**FLAMMABLE GAS** - Any gas that will ignite easily and burn rapidly. See 49CFR, Part 173 for a more precise definition.

FLAMMABLE LIQUID - Any liquid that will ignite easily and burn rapidly. See 49CFR, Part 173 for precise definition of various types.

FLATCAR - A railroad freight car without permanent raised sides, ends, or covering.

<u>GAS CHROMATOGRAPHY-MASS SPECTROMETRY</u> - This is a combination of two chemical instrumental techniques. The gas chromatography involves seperation and identifying the components of a mixture by volatizing the sample into a carrier gas stream and passing it through a bed of special packing and comparing the times for thew various components to be released from the packing.

<u>GEL</u> - A colloid in which the disperse phase has combined with the continous phase to produce a viscous, jelly-like product.

GOUGE - A groove or cavity scooped out.

<u>GRAVITY FLOW</u> - The movement of material without using a driving force like a pump or pressure (i.e., used during product transfer).

<u>HACS</u> - Computerized portion of USCG Chemical Hazards Response Information System (CHRIS). Abbreviation for Hazards Assessment Computer System.

THERMITE GRENADE - A grenade using a mixture of aluminum powder and iron oxide that when ignited evolves a great deal of heat and is used in welding and in incendiary bombs.

**INFARED RADIOMETRY** - Absorption of radiation in the infared spectrum (0.78 - 300 microns). Can be used to assess temperature remotely

<u>IRRITATING MATERIAL</u> - Substances which give off dangerous or intensely irritating fumes when exposed to air or upon contact with fire.

LEAK, CONTINOUS - A steady continous loss of substance through an opening.

LEAK, INSTANTANEOUS - A sudden, abrupt loss of substance through an opening.

LONGITUDINAL AXIS OF CAR - The lengthwise axis of a car.

LOWER EXPLOSIVE LIMIT (LEL) - The lower limit for the range of concentration of a flammable gas or vapor (% by volume in air) in which explosion can occur upon ignition in a confined area.

<u>MASS SPECTROMETRY</u> - A method of chemical analysis in which ions are passed in a vacuum first through an accelerating electric field and then through a strong magnetic field. This has the effect of seperating the ions according to their mass, as they traverse the magnetic field at different velocities (electromagnetic separation).

NATIONAL MOTOR FREIGHT CLASSIFICATION NUMBER - The specific number assigned to commodities being transported over the road.

NATIONAL FESHONSE CENTER - The Coast Guard operated response center that provides telephonic assistance during emergencies and accidents.

NONDESTRUCTIVE MATERIAL - A material that cannot be broken apart or destroyed.

NONFLAMMABLE GAS - A gas that is not easily ignited and does not burn rapidly if ignited.

OFF-LOAD - To remove cargo from its container.

L L

<u>ON-SCENE COORDINATOR</u> - The authority at the scene of an accident who directs emergency handling and cleanup operations.

OXIDIZER - A substance that spontaneously releases oxygen at room temperature or upon heating (i.e., nitrogen tetroxide). Can react vigorously with organic and combustible materials.

PIT - A hollow or indentation on the surface of a substance.

<u>PLACARD</u> - Inverted, color-coded flat square placed on vessels transporting Lurardeus materials. Must be located on all four sides of the vessel, and can be used to aid material identification.

**POISON** A = A gas or liquid so toxic that an extremely low percentage of the  $||| = ||r||^{1/2}$ vapor formed by the liquid is dangerous to life.





<u>POISONOUS GAS</u> - A toxic or irritant gas or volatile liquid that is harmful to living tissues when applied in relatively small doses.

<u>POLYMERIZATION</u> - A chemical reaction, usually carried out with a catalyst, heat or light, and often under high pressure, in which a large number of relatively simple molecules combine to form a chain-like macro-molecule.

<u>PORTABLE TANK</u> - Any packaging (except a cylinder having a 1000-pound or less water capacity) over 110 U.S. gallons capacity and designed primarily to be loaded into or on or temporarily attached to a transport vehicle or ship, and equipped with skids, mounting, or accessories to facilitate handling of the tank by mechanical means.

<u>PRESSURE FLOW</u> - The steady movement of a material by applying a driving force using a pump or flow of gas or liquid.

<u>RADIOACTIVE MATERIAL</u> - A material which spontaneously emmits alpha or beta rays and sometimes also gamma rays by the disintegration of the nuclei of atoms.

<u>RAIL CAR TRUCK</u> - The structure supporting and attaching the wheels to the body of a rail car or tank car.

<u>**REMOTE SENSING</u>** - To detect a material property such as temperature or pressure from a distant location.</u>

RERAIL - To realign and put back in place a rail car that had been derailed.

<u>SELF CONTAINED BREATHING APPARATUS</u> - A breathing apparatus with air supply that keeps the individual completely independent of the surrounding atmosphere.

<u>SOLVENT</u> - A substance capable of dissolving another substance (solute) to form a uniformly dispersed mixture (solution) at the molecular or ionic size level. Solvents are either polar (high dielectric constant) or non-polar (low dielectric constant).

SOLUBILITY - The ability or tendency of one substance to blend uniformly with another, e.g., solid in liquid, liquid in liquid, gas in liquid, gas in gas.

SORBENT - A substance that takes up and holds by either adsorption or absorption.

<u>SYNTHESIS</u> - Creation of a substance which either duplicates a natural product or is a unique material not found in nature, by means of one or more chemical reactions, or (for . elements) by a nuclear change.

THRESHOLD LIMIT VALUES (TLV) - The upper values of a toxicant concentration to which an average healthy person may be repeatedly exposed to on day after day without suffering adverse effects.

TOXIC - Relating to, or caused by poison or toxin.

<u>UNIFORM CLASSIFICATION NUMBER</u> - The specific number assigned to commodities being transported by rail.

VAPOR DISPERSION - The movement of vapor clouds in air due to turbulence, gravity spreading and mixing.

<u>VAPOR SUPPRESSION</u> - The process of retaining vapors or preventing them from escaping from a liquid surface.

VENT AND BURN - To release a substance from its container and allow it to burn.

**VOLATILE** - A substance that will readily vaporize at a low temperature.

<u>WAYBILL</u> - A document prepared by the carrier of a shipment of goods that contains details of the shipment, route, and charges.